Northwest Atlantic Fisheries Organization (NAFO)



Meeting Proceedings of the General Council and Fisheries Commission for 2000

Printed and Distributed by: Northwest Atlantic Fisheries Organization P. O. Box 638, Dartmouth, Nova Scotia Canada B2Y 3Y9

• . · . .

Foreword

This is an annual publication of the Proceedings which contains the reports of all meetings of the General Council and Fisheries Commission including their subsidiary bodies through 2000. The objective of this publication is to provide the Contracting Parties with a detailed consolidated text of all discussions initiated during the year. The proceedings of the Scientific Council are published separately in an annual issue of *NAFO Scientific Council Reports*.

SECTION I contains the Report of the Joint Scientific Council and Fisheries Commission Working Group on Precautionary Approach, 29 February-2 March 2000, Brussels, Belgium.

SECTION II contains the Report of the Meeting on Shrimp Stocks in the Regulatory Area, 27-30 March 2000, Washington, D.C., USA.

SECTION III contains the Report of the Working Group on Allocation of Fishing Rights to Contracting Parties of NAFO, 28-30 March 2000, Washington, D.C., USA.

SECTION IV contains the Report of the Working Group on Dispute Settlement Procedures (DSP), 29-31 May 2000, Copenhagen, Denmark.

SECTION V contains the Report of the Standing Committee on International Control (STACTIC), 27-29 June 2000, Dartmouth, N. S., Canada.

SECTION VI contains the Report of the STACTIC Technical Working Group on Communications, 30 June 2000, Dartmouth, N. S., Canada.

SECTION VII contains the Report of the General Council including subsidiary bodies reports (STACFAD and STACFAC), 22nd Annual Meeting, 18-22 September 2000, Boston, Massachusetts, USA.

SECTION VIII contains the Report of the Fisheries Commission including subsidiary body (STACTIC), 22nd Annual Meeting, 18-22 September 2000, Boston, Massachusetts, USA.

• . . . · · ·

Contents

Foreword	• • • • • • • • • • • • • • • • • • • •		3
Мар			9
Structure of the	Northwest A	tlantic Fisheries Organization in 2000	10
SECTION I.		he Joint Scientific Council and Fisheries Commission	
Working Group on Precautionary Approach			13
	Annex 1.	List of Participants	27
	Annex 2.	Agenda	31
	Annex 3.	List of Papers Considered	32
	Annex 4.	Chairman's Summary of CWP Intersessional	<i></i>
		Meeting 2000	33
	Annex 5.	EU Working Paper	38
	Annex 6.		20
	Annex 0.	Next steps in the implementation of the PA	10
	A	- Cod in Divisions 3NO	43
	Annex 7.	Next steps in the implementation of the PA	
		- Yellowtail flounder in Divisions 3LNO	45
	Annex 8.	Next steps in the implementation of the PA	
		- American plaice in Divisions 3LNO	47
	Annex 9.	EU Summary - The Way Forward	49
SECTION II.	Report of t	he Meeting on Shrimps Stocks in the Regulatory Area	51
	Annex 1.	List of Participants	57
	Annex 2.	Opening Statement by the Representative of	
		the United States of America	62
	Annex 3.	Opening Statement by the Representative of Canada	63
. •	Annex 4.	Opening Statement by the Representative of Denmark	
		(in respect of the Faroe Islands and Greenland)	64
	Annex 5.	Opening Statement by the Representative of the European Union	65
	Annex 6.	Opening Statement by the Representative of Japan	
	Annex 0. Annex 7.		66
		Opening Statement by the Representative of Estonia	67
	Annex 8.	Agenda	68
	Annex 9.	Working Paper by Iceland	69
	Annex 10.		71
	Annex 11.	2	
		System for 3M Shrimp - Paper presented by Norway	72
	Annex 12.	Working Paper by the Chairman	73
		Working Paper by Denmark (in respect of the	
		Faroe Islands and Greenland)	78
	Annex 14.	Statement by the Delegate of Denmark (in respect of	, ,
		the Faroe Islands and Greenland)	79

SECTION III.		he Working Group on Allocation of Fishing Rights	01
	to Contrac	cting Parties of NAFO	81
	Annex 1.	List of Participants	89
	Annex 2.	Opening Statement by the Representative	0,
		of the European Union	94
	Annex 3.	Opening Statement by the Representative	
		of Canada	95
	Annex 4.	Agenda	96
	Annex 5.	Further interpretative notes by the Chair attempting to clarify discussion on Agenda point 4	97
	Annex 6.	Working Paper presented by Denmark (in respect of	97
	Annex 0.	Farce Islands and Greenland)	99
SECTION IV.	Report of th	• • • • • • • • • • • • • • • • • • •	
	Procedure	s (DSP)	103
	Annex 1.	List of Participants	109
	Annex 2.	Agenda	113
	Annex 3.	NAFO Dispute Settlement Procedures - A Possible	
		Way Forward (Paper by Chairman)	114
	Annex 4.	Working Paper by Denmark (in respect of Faroe Islands and Greenland)	116
	Annex 5.	Working Paper Presented by the European Union	118
	Annex 6.	Working Paper Presented by the European Union	120
	Annex 7.	Working Paper Presented by Canada	121
	Annex 8.	Working Paper Presented by Japan	122
	Annex 9.	Working Paper Presented by Canada	124
	Annex 10.		125
	Annex 11.		125
	Anney 12	Settlement of Disputes within NAFO - Consolidated	120
	7 mmox 12.	Text	128
SECTION V.	Report of the	he Standing Committee on International Control	
	(STACTI	C)	131
	Annex 1.	List of Participants	143
	Annex 2.	Agenda	147
	Annex 3.	Working Paper by Denmark (in respect of the	
		Faroe Islands and Greenland)	148
	Annex 4.	Proposal (by Norway) to amend the NAFO Conservation	
		and Enforcement Measures, Part VI.A.1(a) regarding	
		independent and impartial observers	149
	Annex 5.	Proposals (by Canada) to amend the NAFO Conservation	
		and Enforcement Measures regarding Protection of	
		Juvenile Groundfish	150
	Annex 6.	Additional Information (by Canada) - Depth Proposal	
		for Greenland halibut	153
	Annex 7.	Working Paper by the European Union	154
	Annex 8.	Working Paper by Japan	155
	Annex 9.	Statement from of the Representative of Canada	156

		Shrimp 3M Fishery Statistics, 1993-1995 Submission on shrimp catches and effort days (W.P.	158
	Aillex 11.	by Denmark (in respect of Faroe Islands & Greenland)	160
	Annex 12.	Compilation of Shrimp 3M Catches and Effort	
		Days for 1993-1999 (NAFO Secretariat)	163
		Statement from the Representative of Norway	178
	Annex 14.	Proposal (by EU) to amend the NAFO Conservation	
		and Enforcement Measures regarding Part VII-Port	
		Inspections'	179
anomiosium	D		
SECTION VI.		he STACTIC Technical Working Group on	100
	Commun	ications	193
	Annex 1.	List of Participants	200
	Annex 2.	Agenda	202
	Annex 3.	Presentation for an automated NAFO hail message	
		processing system as proposed by Software Kinetics	
		for use in providing a fully automated hail system	203
	Annex 4.	Draft amendment of the Conservation and Enforcement	•
		Measures	211
RECTION VII	D C.		
SECTION VII.		he General Council and its Subsidiary Bodies (STACFAD CFAC), 22nd Annual Meeting	015
	anu STAV	CFAC), 22lid Allitual Meeting	215
Part I.	Report of t	he Meeting of the General Council	217
	Annex 1.	List of Participants	223
	Annex 2.	Opening Statement by the President of NAFO -	
		E. Oltuski (Cuba)	236
	[·] Annex 3.	Speech of the Honorable Rolland A. Schmitten,	
		Deputy Assistant Secretary, International	
		Affairs, U. S. Dept. of Commerce	237
	Annex 4.	Opening Statement by the Representative of Canada	
		(P. Chamut)	239
	Annex 5.	Opening Statement by the Representative of the	
		European Union (E. Mastracchio)	241
	Annex 6.	Opening Statement by the Representative of the	
	. 7	United States of America (P. Kurkul)	242
	Annex 7.	Opening Statement by the Representative of France	2.42
	A O	on behalf of St. Pierre et Miquelon (G. Grignon)	243
	Annex 8. Annex 9.	Agenda NAFO Schedule of Meetings, 2001	244 246
		Press Release	240
		List of Decisions and Actions by the General Council	250
	· ••••••••••••••••••••••••••••••••••••	Let of Declorons and rectors by the General Couldinance	250
Part II.		he Standing Committee on Finance and Administration	
	(STACFA	AD)	251
	Annex 1.	List of Participants	258
	Annex 2.	Agenda	259
	Annex 3.	Travel and Meeting Costs Estimate for the NAFO	
		Secretariat	260

	Annex 4.	Schedule of Outstanding Contributions from	
		Bulgaria and Romania	262
	Annex 5.	Budget Estimate for 2001	263
	Annex 6.	Preliminary Calculation of Billing for 2001	264
	Annex 7.	Preliminary Budget Forecast for 2002	265
		,	200
Part III.	Report of th	e Standing Committee on Fishing Activities of Non-	
		ng Parties in the Regulatory Area (STACFAC)	266
	•		.
	Annex 1.	List of Participants	271
	Annex 2.		272
	Annex 3.	NAFO Letters	273
	-		
SECTION VIII.		ne Fisheries Commission and its Subsidiary Body	
	(STACTIC	C), 22nd Annual Meeting	277
Part I.	Report of th	ne Meeting of the Fisheries Commission	279
	A	List of Participants	202
	Annex 1.	List of Participants	293
	Annex 2.	rigendu	306
	Annex 3.	Amendment to the Conservation and Enforcement	
			308
	Annex 4.	Proposal to amend the NAFO Conservation and Enforcement	
		Measures with a view to introducing satellite based	
		vessel monitoring and related measures	309
	Annex 5.	Amendment to the NAFO Conservation and Enforcement	
		Measures regarding Shrimp in Division 3M	312
	Annex 6.	Working Group on 3M Shrimp Fishery Data,	
		27 March 2001, Copenhagen – Provisional Agenda	314
	Annex 7.	Special Fisheries Commission Meeting on Shrimp	
		Allocation and Management in the NAFO Regulatory Area,	
		28-29 March 2001, Copenhagen – Provisional Agenda	315
•	Annex 8.	Paper on Chartering	316
	Annex 9.	Formats for the Electronic Transmission of NAFO Hails	010
		from Contracting Parties to the NAFO Secretariat	318
	Anney 10	Proposal to amend the NAFO Conservation and Enforce-	510
	Annex 10.	ment Measures regarding Incidental Catch Limits	325
	Annov 11	Proposal to amend the NAFO Conservation and	
	Annex 11.		
		Enforcement Measures regarding Part VII - Port	226
	. 10	Inspections	320
	Annex 12.	Statement by the Representative of the European Union	226
		on 2J3KL Cod	336
	Annex 13.	Proposal to amend the Conservation and Enforcement	
		Measures, Part I.HOther Measures-Management	
		Measures for Shrimp in Div. 3LNO	337
	Annex 14.	Amendment to NAFO Conservation and Enforcement	
		Measures regarding 3L Cod	338
	Annex 15.	Amendment to NAFO Conservation and Enforcement	
		Measures regarding Witch Flounder in Div. 3L	339
	Annex 16.	Quota Table for 2001	340

	Annex 17.	Fisheries Commission's Request for Scientific Advice on Management in 2002 of Certain Stocks in Subareas 3 and 4, including supplementary questions on Division 3M Shrimp for 2001	341
	Annex 18.	List of Decisions and Actions by the Fisheries	
		Commission	347
Part II.	Report of th	e Standing Committee on International Control	
	(STACTIC	5)	348
	Annex 1.	Agenda	354
	Annex 2.	Report of STACTIC <i>Ad hoc</i> Working Group on Comparison of the NAFO Hail System with the	
		NEAFC Communication System	355
	Annex 3.	Proposal to amend the NAFO Conservation and Enforcement Measures regarding Incidental	
		Catch Limits	373
	Annex 4.	Proposal to amend the NAFO Conservation and Enforcement Measures regarding Part VII - Port	
		Inspections	374
	Annex 5.	Paper on Chartering	384
	Annex 6.	Proposal to amend the NAFO Conservation and	
		Enforcement Measures with a view to introducing	
		new rules concerning obligatory inspection presence	
		in the Regulatory Area	386



The Convention Area to which the Convention on Future Multilateral Cooperation in the Northwest Atlantic applies



Structure of the Northwest Atlantic Fisheries Organization (NAFO) in 2000 (as at 22nd Annual Meeting, September 2000)

Contracting Parties

Bulgaria, Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, European Union (EU), France (in respect of St. Pierre et Miquelon), Iceland, Japan, Republic of Korea, Latvia, Lithuania, Norway, Poland, Romania, Russia, Ukraine and United States of America (USA).

11

President

E. Oltuski (Cuba)

Constituent Bodies

General Council	Bulgaria, Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, EU, France (in respect of St. Pierre et Miquelon), Iceland, Japan, Korea, Latvia, Lithuania, Norway, Poland, Romania, Russia, Ukraine and USA.	Chairman – E. Oltuski (Cuba) Vice-Chairman – P. Chamut (Canada)
Scientific Council	Bulgaria, Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, EU, France (in respect of St. Pierre et Miquelon), Iceland, Japan, Korea, Latvia, Lithuania, Norway, Poland, Romania, Russia, Ukraine and USA.	Chairman – W. B. Brodie (Canada) Vice-Chairman – R. Mayo (USA)
Fisheries Commission	Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, EU, France (in respect of St. Pierre et Miquelon), Iceland, Japan, Korea, Latvia, Lithuania, Norway, Poland, Poland, Russia, Ukraine and USA. Standing Committees	Chairman – P. Gullestad (Norway) Vice-Chairman – D. Swanson (USA)
General Council	Standing Committee on Finance and Administration (STACFAD)	Chairman – G. F. Kingston (EU) Vice-Chairman –

J.-P. Plé (USA)

General Council (cont'd)	Standing Committee on Fishing Activity of Non-Contracting Parties in the Regulatory Area (STACFAC)	Chairman – JP. Plé (USA) Vice-Chairman – D. Silvestre (France in in respect of St. Pierre et Miquelon)
Scientific Council	Standing Committee on Fishery Science (STACFIS) Standing Committee on Research and Coordination (STACREC) Standing Committee on Publications (STACPUB)	Chairman – H. J. Rätz (EU) Chairman – R. Mayo (USA) Chairman – O.A. Jørgensen (Denmark- Greenland)
·	Standing Committee on Fisheries Environment (STACFEN)	Chairman – M. Stein (EU)
Fisheries Commission	Standing Committee on International Control (STACTIC)	<i>Chairman</i> – D. Bevan (Canada)

Secretariat

Executive Secretary Assistant Executive Secretary Administrative Assistant Senior Secretary Accounting Officer Desktop Publishing/Documents Clerk Statistical Officer/Conservation Measures Officer Graphic Arts/Printing Technician Graphic Arts/Printing Technician Word Processing Secretary Statistical Clerk Statistical Clerk L. I. Chepeł T. Amaratunga F. D. Keating B. J. Cruikshank S. M. Goodick F. E. Perry G. M. Moulton R. A. Myers B. T. Crawford D.C.A. Auby B. L. Marshall C. L. Kerr

Headquarters Location

2 Morris Drive, Dartmouth, Nova Scotia, Canada

SECTION I

(pages 13 to 50)

Report of the Joint Scientific Council and Fisheries Commission Working Group on Precautionary Approach 29 February - 2 March 2000 Brussels, Belgium

Report of the	Working Gi	oup	15
1.	Opening		15
2.		ent of Rapportcur	
3.	Adoption	of Agenda	15
4.	Harmoniz	ation of Concepts and Terminology	16
5.		alizing the Precautionary Approach into the Management	
		Three Model Stocks	20
6.		tation Plan for the Precautionary Approach to other	
		tocks	22
7.	Considera	tion of Changes or Additions to the Fisheries Commission's	
	Request	to the Scientific Council to Reflect the Precautionary Approach	23
8.	Considera	tion of Criteria for Reopening a Fishery in Light of the	
	Precautio	onary Approach	24
9.	Considera	tion of Additional Supportive Management Measures to	
	Compler	nent the Application of the Precautionary Approach	25
10.	Other Mat	ters	25
		of Report	
12.	Adjournm	ent	26
	Annex 1.	List of Participants	27
		Agenda	
	Annex 3.	List of Papers Considered	
	Annex 4.		
	Annex 5.	EU Working Paper	38
	Annex 6.	Next steps in the implementation of the PA	
		- Cod in Divisions 3NO	43
	Annex 7.	Next steps in the implementation of the PA	
		- Yellowtail flounder in Divisions 3LNO	45
	Annex 8.	Next steps in the implementation of the PA	
		- American plaice in Divisions 3LNO	
	Annex 9.	EU Summary - The Way Forward	49



Report of the Joint Scientific Council and Fisheries Commission Working Group on Precautionary Approach

(FC Doc. 00/2)

29 February - 2 March 2000 Brussels, Belgium

The Working Group was organized in accordance with the decision by the Fisheries Commission at the 21st Annual Meeting, 13-17 September 1999 (item 3.21 of the Fisheries Commission Report, FC Doc. 99/15).

1. Opening

The Joint Scientific Council and Fisheries Commission Working Group on Precautionary Approach was called to order by Co-Chairmen W. B. Brodie and J. Baird (Canada) at 1015 hr, 29 February 2000, at Albert Borschette Conference Centre, Brussels, Belgium. Representatives from Canada, Denmark (in respect of the Faroe Islands and Greenland), European Union, Iceland, Japan, Norway, Russian Federation and the United States of America and observers from NEAFC were present (Annex 1). The Chairman welcomed participants and expressed gratitude to the host Contracting Party (EU) for the invitation to host the meeting and for the excellent facilities.

The Co-Chairmen first outlined the history of the development of the Precautionary Approach (PA) at NAFO. In particular, the Scientific Council began discussions on the PA during its June 1997 Meeting. This was followed by the Scientific Council Workshop in March 1998 and the first Joint Fisheries Commission and Scientific Council Working Group Meeting in May 1998. Subsequent to the Scientific Council Meeting of 27 April – 1 May 1999, and the Joint Scientific Council and Fisheries Commission Working Group Meeting of 3-5 May 1999, the Terms of Reference and Agenda for this meeting of the Working Group were developed by the Fisheries Commission during its Annual Meeting in 13-17 September 1999. The Co-Chairmen highlighted the NAFO FC Doc. 99/13 on the Resolution to Guide Implementation of the Precautionary Approach within NAFO.

2. Appointment of Rapporteur

The Co-Chairmen proposed that T. Amaratunga, Assistant Executive Secretary, should act as the rapporteur for the general preparation of the report of this meeting, while individual rapporteurs will be appointed when necessary to address certain specific agenda items (e.g. Agenda items 4, 5, 6, 7, 8, 9 and 10).

3. Adoption of Agenda

In considering the agenda, the Chairman noted the Provisional Agenda circulated by the Executive Secretary on 31 December 1999 in accordance with Rule 4.2 of the Rules of Procedure.

The Working Group (WG on PA) noted that the CWP Intersessional Meeting held during 14-16 February in Copenhagen, Denmark, had considered inter-agency (NAFO, ICES, ICCAT and FAO) concepts and terminology of PA. The WG on PA agreed to review the Draft report of that meeting. The agenda was accordingly modified to include Item 4a for consideration of the CWP Working Group report, and <u>adopted</u> (see Annex 2). List of papers considered is at Annex 3.

4. Harmonization of Concepts and Terminology

a. Report of CWP Intersessional meeting – February 2000

The Draft report of the CWP Intersessional Meeting of 14-16 February 2000 was presented by Co-Chairman, W. B. Brodie (it was noted W. B. Brodie was also the Chairman of the CWP Intersessional meeting). The Chairman's summary of that report is given at Annex 4. The WG on PA noted that this was work produced by scientists from FAO, ICCAT, NAFO and ICES.

b. Discussion of an EU Paper

The EU representative presented a paper entitled "The Precautionary Approach in Fisheries – The issue of harmonization of concepts and terminology" (Annex 5). Following his presentation, the EU representative emphasized that the Scientific Council has been developing a framework for the implementation of the PA but that this framework has not yet been endorsed by the Fisheries Commission.

During discussions, some delegations disagreed with the paper's interpretations of existing international agreements and its corresponding conclusions relating to the activities in several international fisheries organizations and by Contracting Parties to implement the precautionary approach to fisheries management. It was also pointed out that the paper addresses matters other than the harmonization of concepts and terminology. There was considerable debate on the paper presented by the EU delegation. It was proposed by the EU to include the paper as an annex during the review of the WG on PA report. There was no agreement by the WG on PA on this paper, including whether or not to include it in the WG on PA Report as an Annex. Some delegations expressed the need for guidance from the Fisheries Commission on the inclusion of working documents in working group reports.

Some Contracting Parties also expressed diverging views with respect to the absence of consideration of F_{msy} as a limit in the approach suggested by the EU paper. The EU representative invited those delegations which disagreed to give their reasons. Further he stressed that existing international instruments offered no support for using F_{lim} as F_{msy} as a rule. In response, it was pointed out that, in the NAFO area target reference points of 2/3 F_{msy} and $F_{0.1}$ have been used and that, despite this, nearly 2 dozen groundfish have gone under moratorium or by-catch only fisheries. A view was expressed that promoting fishing mortality levels greater than or equal to F_{msy} in the context of the Northwest Atlantic was not consistent with conservation.

It was also pointed out that there are no compelling reasons to establish targets in a PA framework that are less conservative than the targets already agreed in recent management practices. The importance of setting targets was also pointed out by some Contracting Parties.

c. Contracting Parties' Experience with the Application of the Precautionary Approach

Canada

The Canadian delegation summarized Canada's activities in relation to the Precautionary Approach. Canada has been active for many years in implementation of precautionary fisheries management. The domestic Conservation Harvest Plan development process includes a number of precautionary measures. Canadian scientists and managers have been actively involved in the NAFO process, and scientists have been significantly involved in ICES through its development of the Precautionary Approach. The Precautionary Approach is embodied as an integral part of Oceans Act that came into force in January 1997. In addition, Canada ratified UNFA in August 1999. A Science Workshop was held in November 1999 during which Canadian scientists and managers explored application of the Precautionary Approach for 7 representative stocks that included finfish species, shellfish and marine mammals. The Fisheries Resource Conservation Council (FRCC), mandated to provide advice on Atlantic groundfish stocks, has prepared a discussion paper for domestic review and held a redfish workshop in January 2000 to explore, with industry, managers, and scientists, concepts of their discussion paper as they pertain to redfish. Canada has also been active for many years in conservation of Pacific Coho Salmon, culminating, in February 2000, with the adoption of a Wild Salmon Policy to conserve the resource that includes adherence to the Precautionary Approach. Canada has also been active in NASCO initiatives to adopt the Precautionary Approach for application to Atlantic salmon.

In summary, Canada strongly supports implementation of the Precautionary Approach as evidenced through: ongoing involvement in international fora dealing with Precautionary Approach, incorporation of Precautionary Approach into Oceans Act of 1997, ratification of UNFA in the summer of 1999 and ongoing activity on many domestic fronts.

USA

The U.S. delegation explained that the principal U.S. fisheries legislation mandated comanagement with regional fisheries management councils and that, with its most recent amendments (1996), required the setting of limit and threshold reference points, pre-agreed management actions according to timelines, and the possibility of setting target reference points in addition to management for optimum yield, which can be no greater than MSY. The impact of fisheries conservation and management measures on habitat and affected coastal communities must also be considered. The U.S. delegation provided a paper on the U.S. fisheries management experience.

European Union

The EU explained that long-term management arrangements based on a Precautionary Approach were being agreed upon and implemented for an increasing number of fish stocks in the Northeast Atlantic. These arrangements consist of predetermined biomass levels to define the critical level of stocks, pre-agreed fishing mortality rates which offer high probability of the stock not falling below the critical level and provision for specified safety margins which, if approached, will trigger remedial action. Such arrangements started with North Sea herring in 1997 and they now cover the following stocks:

- Norwegian spring spawning/Atlanto Scandian herring (involving the EU, Faroe Islands, Iceland, Norway and Russia)
- North East Atlantic mackerel (involving the EU, Faroe Islands and Norway)
- EU-Norway joint stocks in the North Sea:
- cod
- saithe
- haddock
- plaice
- stocks under the purview of the International Baltic Sea Fisheries Commission (IBSFC).
 Eastern cod stock and Western cod stock

Norway

Norway referred to the process of establishing a management plan for Norwegian spring spawning herring and underscored the following elements as important:

- simulation exercises to analyse the consequences of various exploitation rates on indicators as average yield, stability in yield and the risk of bringing the spawning stock below limit reference points.
- existence of a working group with both biologists and economists to evaluate the results of the simulation exercise
- decision made by the parties concerned.

Denmark (in respect of Faroe Islands and Greenland)

Denmark (in respect of Faroe Islands and Greenland) referred to the management practice with respect to the capelin stock off East Greenland, Iceland and Jan Mayen. Through many years the 3 Parties have used a limit of 400 000 tons as the minimum stock size required at the end of the fishing season for reproductive purposes.

Iceland

The history of the precautionary management strategy goes back to the early 1970s when the Icelandic summer spawning herring stock collapsed. At that time $F_{0.1}$ was introduced in the ICES/NAFO area. After a two year moratorium the Icelandic summer spawning herring has since then been managed according to $F_{0.1}$ concept. The SSB recovered in the 1970s from almost nothing to about 400-500 000 tons in the 1980s. At present the stock is at historical maximum of approximately 500 000 tons. The present TAC is 100 000 tons.

For capelin in the Iceland-Greenland-area a minimum target SSB of 400 000 tons was set in 1979. This management strategy of leaving 400 000 tons of mature capelin to spawn each season seems to work quite successfully and there has never been a reason to reconsider this target level of SSB.

Mainly due to overfishing the cod stock at Iceland declined from year to year until the midnineties. A risk analysis, originally three species model incorporating capelin, shrimp and economical aspects, was carried out in order to study different management strategies on the rebuilding of the stock. The model has been extended also to include marine mammals (whales). As a result from this modelling a harvest control rule was introduced in 1995 which restricts catches to 25% of the fishable stock (age groups 4+). The HCR has been enforced since then with excellent results, i.e. the fishable stock has almost doubled, the SSB has increased from 200 000 to about 500 000 tons and at the same time F has reduced by more than 50%.

In the early 1990s precautionary TACs were set for some groundfish species as dab, long rough dab, ling, blue ling and tusk according to the precautionary principle even though biological information in order to define the precautionary reference points was not available.

The saithe stock at Iceland, (also dealt with within ICES) is managed at present by using the PA reference point (F_{pa} and B_{pa}) for the first time in 1999. For plaice at Iceland (which is not dealt with within ICES) precautionary reference points were also implemented in 1999. This stock had shown a sharp decline in recent years and measurements to halt that decline failed. A TAC based on the PA reference points led to a decrease in the quota from 7 000 tons to 3 000 tons in one step. As plaice is also a by-catch in the other demersal fisheries such a

reduction in catches was technically hardly possible. The TAC was therefore revised to 4 000 tons.

The main aim of the fisheries management is to monitor the stocks and to keep a viable and sustainable fishery based on the precautionary principle. The goal can be achieved in different ways using different harvest control rules depending on the stock and fishery in view as can be seen in the examples given above.

Japan

Japan explained that the main method to manage its fisheries is fleet control system including reductions of fishing vessels for resource management. In addition to this, recently Japan has introduced the TAC system in its EEZ and manages its fisheries more cautiously.

Russia

Russian fishery management system is mainly based on Total Allowable Catches (TAC). Work on implementation of the Precautionary Approach (PA) into TAC assessments were started more than 3 years ago. Since then different options for the PA procedure of TAC estimation have been tested. As a result, a precautionary evaluation framework was designed which now is successfully used on a routine basis for 5 pollock stock units within Russian EEZ in the North Pacific. Besides, attempts are being conducted to apply the approach to some Pacific crab species and to several objects of Russian far-seas fishery in Atlantic.

Conclusion on Section 4

To this date, no formulations of the PA framework have been accepted by international fisheries organizations. However, several elements of the PA have been implemented by various management authorities (see item 4.c).

The WG on PA agreed that there are several broad similarities between the ICES and NAFO versions of the PA. The biomass limits (defined as B_{lim} in both frameworks), are virtually the same, although B_{lim} is also used in ICES as an indication of biomass below which recruitment is unknown. The biomass buffers (B_{pa} in ICES, B_{buf} in NAFO) generally correspond to a level of biomass at which there is a high probability of being above B_{lim} . However, the harvest control rules in the current formulations are different - the NAFO Scientific Council framework suggests no fishing below B_{buf} , whereas the ICES framework indicates a reduced fishing mortality below B_{pa} . The WG on PA concluded that determination of harvest control rules is the role of managers. In the NAFO context, it is the Fisheries Commission's responsibility to determine appropriate harvest strategies corresponding to reference biomass levels. The WG on PA preferred the B_{buf} term as opposed to B_{pa} .

The NAFO Scientific Council framework proposes that F_{lim} should be set no higher than F_{msy} , based on its interpretation of UNFSA. The ICES framework does not make specific reference to F_{msy} . The WG on PA did not reach agreement on which formulation was more appropriate. Differences of opinion may be related to experiences with fish stocks in the Northwest Atlantic as regards to their response to exploitation *vs* the Northeast Atlantic. Consequently seeking harmonization at this time may be premature.

A paper entitled "Considerations for the implementation of the Precautionary Approach into the Management Plans of Stocks Managed by the Northwest Atlantic Fisheries Organization (NAFO) - a discussion paper prepared by Canada" was presented by the Canadian delegation (PA WG WP 00/01). This paper was prepared to focus the discussions on the implementation of the precautionary approach, taking into account the elements of the resolution adopted by the Fisheries Commission. It outlined the progress made on the two model stocks used to first explore ways of implementing the precautionary approach, namely cod in Div. 3NO, and yellowtail flounder in Div. 3LNO. The document proposed additional steps for implementation of the precautionary approach for these stocks. For these stocks, the document provided a history of the precautionary approach and proposed practical steps to consider in its implementation under the headings of "harvest strategies and reference points", "conservation and management measures", and "research and monitoring". The following is a summary of information presented in the Canadian paper:

Cod in Divisions 3NO

The Div. 3NO cod stock has remained at a low level since the initial cessation of directed fishing in the early 1990s. Because current stock size is so low, the discussion necessarily focused on the strategy to reach the first benchmark to rebuilding, i.e. B_{lim}.

The NAFO Scientific Council framework for implementation of the precautionary approach identifies the need to "initiate precautionary monitoring" when the biomass is below B_{buf} . The paper proposed that any directed fishing below B_{lim} may only be allowed for the purpose of collecting information that would permit further evaluation of resource abundance. To safeguard against possible abuse, it was suggested that a protocol/guidelines be established respecting this activity.

It was noted that to this point in time, neither the Scientific Council nor the Fisheries Commission has focused much attention on eventual <u>targets</u> for stock rebuilding (SSB) or exploitation rates. It was also suggested that at current levels of SSB, the main objective of fisheries managers should be to minimize the by-catch of cod when fishermen are directing for other species. Some measures that could be considered to achieve this objective were outlined.

The Southeast Shoal area has been clearly identified as a nursery area for not only Div. 3NO cod, but also for yellowtail flounder. Information from research surveys also indicates that juveniles are found in other areas of the stock distribution. In order to afford pre-recruits of these stocks the best possible chance to survive and enter the fisheries and mature portion of the populations, the paper suggested that consideration be given to closures or other management measures in areas where juveniles are concentrated.

The paper identified the importance of having reliable information on catches taken as bycatch in other fisheries, as well as information on spawning times and locations, on juvenile nursery areas, on weight-at-length and maturity-at-length. Information on current spatial distribution of the stock compared to historical patterns may also be useful in indicating resource health and should be presented in the assessments.

Yellowtail Flounder in Divisions 3LNO

The Div. 3LNO yellowtail flounder stock appears to have rebuilt and its biomass seems to be within its expected productivity range. The re-opening of the Div. 3LNO yellowtail fishery in 1998 was based on a target relative exploitation rate of 6% which was believed to be conservative while allowing a commercial operation. The fishery has been conducted so as not to jeopardize the recovery of other stocks still under moratoria.

Based on general production analysis, the Scientific Council has tentatively identified F_{buf} and this level of fishing corresponds closely to the exploitation rate of 2/3 F_{msy} , a reference point used in the past for fisheries management. At this point in time, the Fisheries Commission has not focused much attention on eventual targets for stock rebuilding (SSB) or discussed whether the F_{buf} proposed by Scientific Council is an appropriate fishing mortality limit or target. This stock is considered to be in a data moderate situation and the paper suggested that scientists continue their work aimed at development of an age-structured model to estimate population size and, on that basis, recommend biological reference points as appropriate. In absence of progress in this area, the information from the production model should be further examined and the use of appropriate indices should be examined to determine the possible derivation of provisional biological reference points.

During 1998 and 1999, the fishery has been prosecuted with a suite of management measures aimed at protecting juvenile fish, minimizing the by-catch of American plaice, cod and witch flounder, and at allowing mature yellowtail flounder to spawn one more time. The paper suggested that such measures be continued to complement the PA. In particular, it suggested that it is important to conduct any fishery for yellowtail in a manner that will keep American plaice and cod by-catches at the lowest possible level. It noted that by-catches of Div. 3LNO yellowtail flounder have been increasing and suggested that a revision of conservation measures be undertaken so as to ensure that by-catch are truly incidental in nature.

The Southeast Shoal area has been clearly identified as a nursery area for Div. 3LNO yellowtail flounder as well as Div. 3NO cod. Information from research surveys also indicates that juveniles are found is other areas of the stock distribution. It was suggested that the Fisheries Commission should consider a resolution to close the Southeast Shoal area and other areas identified as having high concentrations of juveniles.

Prior to re-opening the yellowtail flounder fishery in Div. 3LNO, the scientific data collection programs were improved so as to obtain a better description of stock trends. In particular, joint Canadian industry-science surveys were introduced and undertaken seasonally. While the regular survey program was continued and was instrumental in the assessment of the stock, the additional information obtained from the industry surveys provided information on the expected performance of a commercial operation, on seasonal variations in the catch rates, and on probable levels of by-catch of other species. The information obtained during these surveys allowed the scientists to estimate stock abundance with some confidence, given the stability of the results from the various sources. The paper suggested that these research initiatives be continued and noted the June 1999 recommendation of the Scientific Council on the need "to restore the Council's ability to do age-structure analyses on this stock."

Discussion

It was noted that the proposed steps in the implementation of a PA did not include specific reference to harvest control rules. The need to address harvest control rules in an implementation plan was noted. The WG on PA agreed on the next steps in the implementation of the

Precautionary Approach for these two model stocks. These steps are outlined in Annexes 6 and 7 for Div. 3NO cod and Div. 3LNO yellowtail flounder, respectively.

The WG on PA did not consider the next steps for the implementation of the PA for shrimp in Div. 3M. It was noted that work by the Scientific Council related to a precautionary approach is ongoing and will be reviewed again in November 2000, prior to the 2001 fishing season.

6. Implementation Plan for the Precautionary Approach to Other NAFO Stocks

The template for an implementation plan developed for the model stocks was applied to one other stock managed by NAFO, namely American plaice in Div. 3LNO. The situation for this stock is similar to that of cod in Div. 3NO in the sense that the stock is at a very low level, much below the biomass limit reference point. The WG on PA agreed to the next steps in the implementation of the PA for Div. 3LNO American plaice as presented in Annex 8.

The implementation of the PA will, no doubt, take time but a detailed implementation plan including steps such as the ones agreed in Annexes 6, 7 and 8 for two of the model stocks, and for American plaice in Div. 3LNO, could help in channeling future efforts directed at the precautionary approach. It is suggested that similar detailed plans be developed for other stocks under the management of the NAFO Fisheries Commission.

For other stocks, the management objectives should be identified by the Fisheries Commission and should include, but not be limited to, the rebuilding and maintenance of stock biomass at a level that can support sustainable fisheries and produce stable yields.

As a general rule for all NAFO stocks, the Fisheries Commission should specify management strategies and ensure that data collection and analysis is carried out in support of the PA. If necessary, additional supportive management measures should be specified.

Management Strategy

The Fisheries Commission shall specify management objectives and strategies. Management actions include the selection of biomass and fishing mortality target reference points and setting corresponding limit and buffer reference points as calculated by the Scientific Council. Management strategies include specification of courses of action consistent with a Precautionary Approach Framework, specifically time horizons for stock rebuilding and fishing mortality adjustments to ensure stock recovery and/or avoid stock collapse. An evaluation of possible consequences of management actions shall include the specification of acceptable levels of risk.

Data Collection/Analysis

The Fisheries Commission, in consultation with the Scientific Council shall promote the collection and analysis of data to enhance the ability of the Scientific Council to evaluate the state of the resources. These shall include, but not be limited to the following:

- 1) conduct statistically sound, comprehensive research surveys,
- 2) obtain information on spawning times and locations as well as location of juvenile nursery areas,
- 3) collection of data on weight-at-length and maturity-at-length to be used to monitor SSB and for prediction of future trends,
- 4) develop information on the recent spatial distribution of the stocks with respect to historical distribution patterns,

5) stock assessment, modelling and forecasting using all appropriate data and up-to-date methods.

Supportive Management Measures/Good Practices

When the biomass of a stock is below B_{lim} , the main focus of the Fisheries Commission should be to minimize by-catch of adults and juveniles in fisheries directed at other species. As such, additional technical management measures may be specified, including but not limited to, the following:

- 1) Specification of technical conservation measures that permit only by-catch that is truly incidental in nature.
- 2) Closure of specific areas for specified time periods where by-catch has persisted, and where high concentrations of juveniles have been observed.

7. Consideration of Changes or Additions to the Fisheries Commission's Request to the Scientific Council to Reflect the Precautionary Approach

Proposals for modifications to the Fisheries Commission's Request for Advice to Scientific Council for 2001 were tabled by Canada and Norway. The WG on PA discussed the inclusion (or not) of references to various paragraphs, articles and annexes of the UN Fish Stock Agreement in the proposed modified request but there was no consensus reached. As such, the WG on PA decided not to draft revisions to the current Fisheries Commission's Request to Scientific Council for advice, i.e. the request agreed by Fisheries Commission in September 1999. Instead, it was agreed that the following items pertaining to advice under the PA would be submitted to Scientific Council for consideration.

It was agreed that the term 'Precautionary Approach Framework' would not be used as the Fisheries Commission has not yet formally adopted the PA Framework as proposed by the Scientific Council.

Additional Items for Consideration by the Scientific Council at its June 2000 Meeting include:

1. Many of the stocks in the NAFO Regulatory Area are well below any appreciable level of B_{lim} or B_{buf} . For these stocks, the most important task for the Scientific Council is to inform on how to rebuild the stocks. In this context and building on previous work of the Scientific Council in this area, the Scientific Council is requested to evaluate various scenarios corresponding to recovery plans with timeframes of 5 to 10 years, or longer as appropriate. This evaluation should provide the information necessary for the Fisheries Commission to consider the balance between risks and yield levels, including information on the consequences and risks of no action at all. Whenever possible, this evaluation should be cast in terms of risks analyses relating removals from various sources to B_{lim} (B_{buf}) and F_{lim} (F_{buf}).

References to "risk" and to "risk analyses" should refer to estimated probabilities of stock population parameters falling outside biological reference points.

- 2. Where reference points are proposed by the Scientific Council as indicators of biological risk, they should be accompanied by a description of the nature of the risk incurred if the reference point is crossed (e.g. short-term risk of recruitment overfishing, loss of long-term yield, etc.)
- 3. When a buffer reference point is proposed in order to maintain a low probability that a stock, measured to be at the buffer reference point may actually be at or beyond the limit reference point, the Scientific Council should explain the assumptions made about the uncertainty with

which the stock is measured, and also the level of 'low probability' that is used in the calculation.

- 4. Wherever possible, short and medium term consequences should be identified for various exploitation rates (including no fishing) in terms of yield, stability in yield from year to year, and the risk or probability of moving the stock beyond B_{lim} or B_{buf} . Whenever possible, this information should be cast in terms of risk assessments relating fishing mortality rates to the risks of falling below B_{lim} , the risks of stock collapse and recruitment overfishing, as well as the risks of growth overfishing and the consequences in terms of both short and long term yields.
- 5. When providing risk estimates, it is very important that the **time horizon** be clearly spelled out. By way of consequence, risks should be expressed in timeframes of 5, 10 and 15 years (or more), or in terms of other appropriate year ranges depending on stock specific dynamics. Furthermore, in order to provide the Fisheries Commission with the information necessary to consider the balance between risks and yield levels, each harvesting strategy or risk scenario should include, for the selected year ranges, the risks and yields associated with various harvesting options in relation to B_{lim} (B_{buf}) and B_{target} , and F_{lim} (F_{buf}) and F_{target} .

There was considerable debate on a paper presented by the EU delegation, entitled EU Summary-"A Way Forward" (Annex 9). This was proposed by EU as an explanatory memorandum, for Fisheries Commission consideration in future Fisheries Commission requests for advice. There was no agreement by the WG on PA on this paper, including whether or not to include it in the WG on PA report as an Annex. As was the case in Agenda item 4b), some delegations expressed the need for guidance from the Fisheries Commission on the inclusion of working documents in Working Group reports.

8. Consideration of Criteria for Reopening a Fishery in Light of the Precautionary Approach

Stocks under moratoria have been characterized by a very low spawning stock biomass and a reduced age-range. There is often a concern that the level of spawner biomass reached corresponds to a level where the chance of producing good year-classes is greatly reduced.

Once recovery has begun and spawner biomass has reached a level sufficient to allow consideration of reopening of the fishery, under a PA this reopening must be consistent with a strategy of continued stock rebuilding.

The discussion related to stocks under moratorium has necessarily focused on the strategy to reach the first benchmark to rebuilding, i.e. B_{lim} . In order to monitor the progress of stock rebuilding, milestones should be established so as to permit a review of the stock trajectory in relation to reference points within reasonable timeframes.

For the stock currently under moratorium, the other elements of a PA (i.e. other than B_{lim}), have not received detailed attention. Key considerations in the decision of re-opening include the determination of B_{lim} , the determination of the fishing mortality (F) at re-opening, the probability of continued growth in the stock, the trade-offs between yield/probability of growth in the stock and the risks that the stock could actually fall (again) below a pre-determined limit.

The other elements of a PA will need to be defined. Also, any reopening of commercial activity should only be contemplated under specific conditions. In particular, increased focus on additional conservation measures such as limitations on by-catch is required in order to afford the resource the best chance of recovery.

As such, additional technical management measures may be specified, including but not limited to, the following:

1. Protection of Spawners:

Management should incorporate controls to limit the catch during the main spawning periods in order to ensure the best possible spawning success. Information can be made available from scientists to guide managers in this regard. Scientists can also provide information regarding spawning areas for possible protection as well (see above).

An important conservation objective should be to allow development of a full age-range in the spawner population in order to promote the best possible stability in annual recruitment.

2. Protection of Pre-recruits (Area Closures):

Specific areas that have been clearly identified as significant nursery areas should be closed, as appropriate, for a specified time so as to minimize the mortality on small fish. In addition, other management measures to protect small fish should be considered.

3. Concerns with By-catch:

Fisheries for other species that might result in by-catch of the species under consideration must be conducted in such a manner so as to keep by-catch at the lowest possible level. This would necessitate careful review of possible management strategies including adequate monitoring.

4. Concerns with By-catch of Other Species:

Fisheries for the directed species that might result in bycatch of other species, especially those under moratorium, must be conducted in such a manner so as to keep bycatch at the lowest possible level. This would necessitate careful review of possible management strategies including adequate monitoring.

9. Consideration of Additional Supportive Management Measures to Complement the Application of the Precautionary Approach

The WG on PA noted a number of supportive management measures/good practices during discussion on the two model stocks (Div. 3NO cod, Div. 3LNO yellowtail flounder), as well as one additional stock (Div. 3LNO American plaice). The information on these three stocks is included in Annexes 6, 7 and 8.

10. Other Matters

The WG on PA considered some examples of supportive management measures as follows:

- The WG on PA noted that management of the NAFO stocks are based on single-species models. In the years to come, it will important to enhance our understanding of the ecosystem in order to base our management decision on models also taking into account of how fish stocks react to changes in the environment as well as the significance of stock interactions.
- The WG on PA noted that a primary cause of depleted fish stocks around the world is the existence of a too large fishing capacity relative to the fish resources. In order to achieve not only sustainable fish stocks, but sustainable fisheries, the Fisheries Commission should stimulate initiatives to curb overcapacity in the fishing fleet.

Some Contracting Parties considered the following measures as examples:

- TAC/Moratorium
- Limited Entry
- Vessel Replacement Restrictions
- Effort Control
- Conservation Harvesting Plans
- By-catch Protection Provisions
- Minimum Fish Size
- In-season Management
 - By-catch Protocols (In-season)
 - Small fish Protocols (In-season)
- Spawning Closures
- Juvenile Closures
- By-catch Closures
- Fishing Gear Restrictions Minimum Mesh
- Fishing Gear Restrictions Separator Grates
- Observers Canadian Zone
- Observers NRA % Coverage
- Dockside Monitoring % Coverage
- Vessel Monitoring Systems
- Air Patrols
- Ship Patrols
- On-board Inspections
- Basic Scientific Surveys
- Comprehensive Scientific Surveys

Some Contracting Parties considered these measures as example of already good management practices.

11. Adoption of Report

During the concluding session of the WG on PA on 2 March 2000, the draft report was reviewed and the report was adopted.

12. Adjournment

Noting the WG on PA work was brought to a successful completion, the Co-Chairmen, W. B. Brodie/J. Baird, thanked the participants, expressing hopes that the work done so far on the PA will continue to meet the Resolution on implementation of the PA outlined by the Fisheries Commission. Special thanks were extended to the NAFO Secretariat and the EU hosts for the arrangements and meeting facilities.

There being no further business, the Co-Chairmen adjourned the meeting at 1930 hrs.

CANADA

Representatives

J. Robichaud, Special Advisor ADM Fisheries, DFO, 200 Kent St., Station 13.029, Ottawa, Ontario K1A 0E6 Phone: +613-990-6794 - Fax: +613-954-1407

Advisers

C. Allen, Resource Management, DFO, 200 Kent St., Ottawa, Ontario K1A 0E6 Phone: +613-990-0105 - Fax: +613-990-7051 - E-mail: allenc@dfo-mpo.gc.ca D. B. Atkinson, Regional Director, Science, Oceans and Environment, DFO, P.O. Box 5667, St. John's, Newfoundland A1C 5X1 Phone: +709-772-2027 - Fax: +709-772-6100 - E-mail: atkinsonb@dfo-mpo.gc.ca J. W. Baird, Director, Resource Management, DFO, P.O. Box 5667, St. John's, Newfoundland A1C 5X1 Phone: +709-772-4543 - Fax: +709-772-2046 - E-mail: bairdj@dfo-mpo.gc.ca W. B. Brodie, Science Branch, DFO, P.O. Box 5667, St. John's, Newfoundland A1C 5X1 Phone: +709-772-3288 - Fax: +709-772-4188 - E-mail: brodieb@dfo-mpo.gc.ca D. Caron, Mission of Canada to the European Union, Avenue de Tervuren, 2, Brussels Belgium Phone: + 32 2 741 0688 Fax: +32 2 741 0629 B. Chapman, 1388 River Road, Manotick, Ontario Phone: +613-692-8249 - Fax: +613-692-8250 - E-mail: bchapman@sympatico.ca E. Dussault, Director, Atlantic Affairs Div., International Directorate, DFO, 200 Kent St, Ottawa K1A 0E6 Phone: +613-993-5316 - Fax: +613-993-5995 - E-mail: dussaulte@dfo-mpo.gc.ca G. Gregory, Fishery Product International Ltd., P O Box 550, Sation A, St. John's, Newfoundland A1C 5L1 Phone: +709-570-0427 - Fax: +709-570-0436 - Email: ggregory@fpil.com E. McCurdy, c/o FFAW/CAW, P O Box 10, St. John's, Newfoundland A1C 5H5 Phone: +709-576-7276 - Fax: +709-576-1962 - Email: D. Rivard, Fisheries Research Branch, DFO, 200 Kent St., Ottawa K1A 0E6 Phone: +613-990-0281 - Fax: +613-954-0807 - E-mail: rivardd@dfo-mpo.gc.ca R. Steinbock, Advisor, International Directorate, DFO, 200 Kent St., Ottawa K1A 0E6 Phone: +613-993-1836 - Fax: +613-993-5995 - E-mail: steinbob@dfo-mpo.gc.ca

DENMARK (in respect of Faroe Islands and Greenland)

Representatives

E. Lemche, Head of Representation, Greenland Home Rule, Denmark Office, Pilestræde 52, Box 2151, Copenhagen

Phone: +45 33 69 3400 - Fax: +45 33 69 3401- E-mail: Einar.Lemche@ghsdk.dk

A. Kristiansen, Ministry of Fisheries and Maritime Affairs, P. O. Box 347, FO-100 Torshavn, Faroe Islands Phone: +298 35 3030 - Fax: +298 32 0822 - E-mail: <u>andrask@fisk.fl.fo</u>

Advisers

E. Dahl, Faroese Representative in Brussels, Danish Embassy, Rue D'Arlon, 73, Brussels, Belgium Phone: +32 02 233 08 55 - Fax: +32 02 233 0966 - Email: <u>faroe.rep@shynet.be</u>
A. Nicolajsen, Fiskorannsoknarstovan, Noatun, Postboks 3051, FO-1100, Torshavn, Faroe Islands Phone: +298 31 50 92 - Fax: +298 31 8264 - Email: <u>arninic@frs.fo</u>
J. Persson, Greenland Home Rule, Dept. of Industry, P. O. Box 269, DK-3900 Nuuk, Greenland

Phone: +299 34 53 24 - Fax: +299 32 47 04 - E-mail: jpe@gh.gl

EUROPEAN UNION (EU)

Representative

F. Wieland, European Commission, Directorate Gen. for Fish., Rue de la Loi 200, B-1049 Brussels, Belgium Phone: +32 2 296 3205 - Fax: +32 3 299 4802 - Email: Freidrich.Wieland@cec.eu.int

Advisers

R. Åkesson, Ministry of Agriculture, 10333 Stockholm, Sweden

Phone: +46 08 405 1122 - Fax: +46 08 10 5061 - E-mail: rolf.akesson@agriculture.ministry.se V. Angot, European Commission, 200 Rue de la Loi, B-1049, Brussels, Belgium Phone: +32 2 296 6406 - Fax: +32 2 296 2338 - Email: Veronique.Angot@ cec.eu.int M. I. Aragon, Secretaria General de Pesca Maritima, Jose Ortega y Gasset, 57, 28006 Madrid, Spain Phone: +34 91 402 5000 - Fax: +34 91 309 3967 - E-mail: iaragonc@mapya.es E. De Cardenas, Institute Espanol de Oceanografia, Avenida de Brasil 31, 28020 Madrid, Spain Phone: +34 91-5974443 - Fax: +34 91 5974770 - E-mail: e.decardenas@md.ieo.es S. Ekwall, European Commission, J99 3/23, 200 Rue de la Loi, 1000 Brussels, Belgium Phone: + 32 02 299 6907 - Fax: - Email: steffan.ekwall@cec.eu.int H. Figueiredo, Director, Direction General Pescas Agricultura, Edificio Vasca da Gama, Rua General Gomes Araújo 1399-006, Lisbon, Portugal Phone: +351 21 391 4350 - Fax: +351 21 397 97 90/1 - Email: hfigueir@dg-pescas.pt F. Gauthiez, Ministère de l'Ágricuture et de la Pêche, Direction des Pêches Maritimes, 3 Place Fontenoy, 75700 Paris 07 SP, France Phone: +33(0) 1 49 55 8203 - Fax: +33 (0) 1 49 55 8200 - E-mail: francois.gauthiez@agriculture.gouv.fr O. Hagstrom, European Commission, DG XIV - Unit C-1, , 200 Rue de la Loi, B-1049, Brussels, Belgium Phone: +32 2 295 5612 - Fax: +32 2 296 6046 - E-mail: ulle.hagstroem@cec.eu.int J. Holbrook, EU Commission, DGXIV, J-11 99/1123, 200 Rue de la Loi, 1049, Brussels, Belgium Phone: + 32 2 296 0747 - Email: Joanna.Holbrook@cec.eu.int T. Kruse, Danish Representation to the EU, Rue D'Arlon, 73, 1040 Brussels, Belgium Phone: +32 02 233 0833 - Fax: +32 02 230 7242 - Email: thokrv@brubee.um.dk K. Patterson, EU Commission, DG Fish, Rue de la Loi 200, B-1040 Brussels, Belgium Phone: + 32 2 299 2179 - Fax: +32 2 295 5621 - Email: kenneth.patterson@cec.eu.int H. Pott, Bundesministerium für Ernahrung, Landwirtschaft und Forsten, Rochusstr. 1, D-53125 Bonn, Germany Phone+49 40 228 529 4124 - Fax: +49 40 228 529 4410 - Email: hermann.pott@bml.bund.de A. Y. Pinto, Edificio Vasco da Gama, DGPA, Alcantara -- Mar 1399-006, Lisbon, Portugal Phone: +35 21 391 427 - Fax: +35 21 397 9790 - Email: H.-J. Rätz, Institute of Sea Fisheries, Palmaille 9, D-22767 Hamburg, Germany Phone: +49 40 389 05169 - Fax: 49 40 389 05263 - Email: raetz.ish@bfa-fisch.de B. Steinarsson, European Commission, Rue de la Loi 200, B-1049 Brussels, Belgium Phone: + 32 2 299 1317 - Fax: +32 02 295 7862 - Email: bjorn.steinarsson@cec.eu.int L. E. Svensson, Council of the European Union, 4040 GH 26, Rue de la Loi, 175, B-1048 Brussels, Belgium Phone: +32 2 285 7853 - Fax: +32 2 285 69 10 - Email: lars-erik.svensson@consilium.eu.int

G. Taylor, Ministry of Agric., Fish. & Food, Nobel House, 17 Smith Square, London, England SW1P 3HX Phone: +44 71 238 6529 - Fax: +44 71 238 5721 - Email: G.taylor@fish.maff.gov.uk

A. Thomson, European Commission, Directorate General for Fisheries, J-II 99/3/27, 200 Rue de la Loi, 1049 Brussels, Belgium

Phone: +32 2 299 0180 - Fax: +32 2 299 4802 - Email: andrew.thomson@cec.eu.int

ICELAND

Representative

K. Arnason, Ministry of Fisheries, Skulagata 4 – 18, 150 Reykjavik Phone: +354-560-9670 - Fax: +354-562-1853 - E-mail: kolbeinn.arnason@sir.stjr.is

Advisers

S. A. Schopka, Marine Research Institute, Skulagata 4, P O Box 1390, Reykjavik
Phone: +354 552 0240 - Fax: +354 562 3790 - Email: sas@hafro.is
S. R. Palmason, Counsellor - Fisheries, Embassy of Iceland, 74 Rue de Treves, 1040 Brussels, Belgium
Phone: + 32 2 286 1700 - Fax: + 32 2 286 1770 - Email: snorri.runar.palmason@utn.stjr.is

JAPAN

Representative

K. Tanaka, Deputy Director, International Affairs Div., Fisheries Agency, 1-2-1 Kasumigaseki, Chiyoda-ku, Tokyo

Phone: +81 3 3591 1086 - Fax: +81 3 3502 0571 - E-mail: kengo_tanaka@nm.maff.go.jp

Adviser

H. Kobayashi, First Secretary, Japanese Mission to the EU, Square de Meenst, 6, 1000 Brussels, Belgium Phone: + 32 2 500 7756 - Fax: +32 2 513 3241 - Email: <u>hidetakakobayashi@mofa.go.jp</u>

NORWAY

Representative

P. Sandberg, Directorate of Fisheries, P. O. Box 185, 5804 Bergan Phone: +47 55 23 8000 - Fax: +47 55 23 8090 - E-mail: per.sandberg@fiskeridir.dep.telemax.no

Adviser

T. Lobach, Directorate of Fisheries, Box 185, 5804 Bergen Phone: +47 552 3800 - Fax: +47 552 38090 - E-mail: terje.lobach@fiskeridir.dep.telemax.no

Representative

RUSSIAN FEDERATION

V. Babayan, Russian Federal Research Institute of Fisheries & Oceanography (VNIRO), 17, V. Krasnoselskaya, Moscow, 107140 Phone: + 70 95 264 6983 - Fax: +70 95 264 9187

Advisers

Y. Efimov, Russian Federal Research Institute of Fisheries & Oceanography (VNIRO), 17, V. Krasnoselskaya, Moscow 117140
Phone: + 70 95 264-9289 - Fax: +70 95 264 9187 - Email:
A. Filin, PINRO, 6 Knipovich St., Murmansk 183763
Phone: +7 815 2 472962 - Fax: +47 789 10 518 - Email: matlad@pinro.murmansk.ru

UNITED STATES OF AMERICA (USA)

Representative

P. Kurkul, National Marine Fisheries, Northeast Region, 1 Blackburn Drive, Gloucester, MA 20910 Phone: +978-281-9331 – Fax: +978-281-9315 - E-mail: <u>pat.kurkul@noaa.gov</u>

Advisers

R. Mayo. National Marine Fisheries, 166 Water St., Woods Hole, MA 02543-1026 Phone: +508-495-2310 - Fax: +508-495-2393 - E-mail: ralph.mayo@noaa.gov F. Serchuk, National Marine Fisheries, 166 Water St., Woods Hole, MA 02543-1026
Phone: +508-495-2245 - Fax: +508-495-2258 - E-mail: Fred.serchuk@noaa.gov
D. Swanson, Chief, International Fisheries Div., NMFS, 1315 East-West Highway, Silver Spring, MD 20910
Phone: +301-713-2276 - Fax: +301-713-2313 - E-mail: Dean.Swanson@noaa.gov
S. Snyder, US Mission to the EU, 27 Blvd. Du regent, 1000 Brussels, Belgium
Phone: + 32 2 508 2670 - Fax: +32 2 511 2092 - Email: snyderss@state.gov

OBSERVERS

S. Engesaeter, Secretary, North-East Atlantic Fisheries Commission, 22 Berners Street, London W1P 4DY, UK

Phone: + 44 20 763 10016 - Fax: + 44 20 763 69225 - Email: <u>info@neafc.org</u> J. Neves, North-East Atlantic Fisheries Commission, 22 Berners Street, London W1P 4DY, UK

Phone: + 44 20 763 10016 - Fax: + 44 20 763 69225 - Email: joao@neafc.org

NAFO SECRETARIAT

T. Amaratunga, Assistant Executive Secretary

B. L. Marshall, Statistical Clerk

- 1. Opening (Co-Chairmen Bill Brodie and Jim Baird, Canada)
- 2. Appointment of rapporteur
- .3. Adoption of Agenda
- 4. Harmonization of concepts and terminology
 - a) Report of CWP Intersessional Meeting February 2000
 - b) Discussion of an EU paper
 - c) Contracting Parties' experience with the application of the Precautionary Approach
- 5. Operationalizing the Precautionary Approach into the Management Plans for Three Model Stocks
- 6. Implementation Plan for the Precautionary Approach to other NAFO Stocks
- 7. Consideration of changes or additions to the Fisheries Commission's Request to the Scientific Council to reflect the precautionary approach
- 8. Consideration of Criteria for reopening a fishery in light of the Precautionary Approach
- 9. Consideration of additional supportive management measures to complement the application of the Precautionary Approach
- 10. Other Matters
- 11. Adoption of report
- 12. Adjournment

Annex 3. List of Papers Considered

- 1. Resolution to Guide Implementation of the Precautionary Approach within NAFO. NAFO FC Doc. 99/13, Serial No. N4198, 1p.
- Considerations for the Implementation of the Precautionary Approach into the Management Plans of Stocks Managed by the Northwest Atlantic Fisheries Commission (NAFO) – a discussion paper prepared by Canada. PA W.G. Working Paper 00/1, 18 p.
- 3. The Precautionary Approach in Fisheries The issue of harmonization of concepts and terminology. EU Working Paper. PA W.G. Working Paper 00/2, 6 p.
- 4. Draft Report of the Working Group on Precautionary Approach Terminology. CWP Intersessional Meeting February 2000, 19 p. (CWP-19 in July 2001 will review this report before finalization).
- 5. Chairman's Summary. CWP Intersessional Meeting of the Working Group on Precautionary Approach Terminology, 12 p. (Co-Chairman W. B. Brodie's summary presented at this meeting see Annex 4.)
- The Precautionary Approach: A New Paradigm, or Business as Usual? V. R. Restrepo, P. M. Mace, and F. M. Serchuk. 1999. Feature Article 1, p.61-70. *In*: Our Living Oceans, Report on the Status of U.S. Living Marine Resources, 1999, NOAA Technical Memorandum NMFS-F/SPO-41, 301 p. (Submitted by the US Delegation, 6 p.)

Annex 4. Chairman's Summary, CWP Intersessional Meeting 2000 Meeting of the Working Group on Precautionary Approach Terminology [The complete report is submitted to CWP for finalization]

The CWP Intersessional was held during February 14-16, ICES HQ in Copenhagen. FAO, ICCAT, ICES and NAFO representatives attended the meeting.

TERMS OF REFERENCE FOR THE MEETING

- 1) Review the terminology and definitions of concepts in use by the different agencies.
- 2) Identify where concepts are identical and where these differ. Explore consequences of such differences in concepts to the reference points used for providing scientific advice within the Precautionary Approach.

PRESENTATION AND COMPARISON OF AGENCY PA FRAMEWORKS.

The CWP Intersessional noted NAFO and ICCAT both include science and management bodies, while ICES is strictly a scientific body.

Examination of PA work from other perspectives (EC, Canada, USA).

FAO

FAO presented a summary of main issues noted in the 5 years of PA implementation (1995-2000) eg. Marine Protected Areas Harvest Control Rules, role of science, operational management procedures, several others. The relevant papers were appended to the CWP Intersessional Draft Report.

ICCAT

The ICCAT presentation addressed the following:

- has not yet formalized an operational framework for implementing the PA.
- formed an ad hoc WG of the Standing Committee on Research and Statistics (SCRS) to examine the PA.
- proceeded along a slightly different track than either NAFO or ICES, noting that "Annex II of the Straddling Stocks Agreement states that F_{MSY} should be a minimum standard for a limit reference point. This is potentially in conflict with the objectives of the ICCAT Convention, which imply that F_{MSY} is the target."
- SCRS decided that it needs to conduct stock-specific evaluations using simulation methods.
- ICCAT has not yet.made a decision on what reference points would be treated as limits in providing PA advice.
- SCRS routinely provides estimates of stock status relative to MSY benchmarks for all stocks with quantitative assessments.

- SCRS provided working definitions of targets, limits, thresholds, and harvest control rules.

ICES

The ICES presentation addressed the following:

- In order for stocks and fisheries exploiting them to be within safe biological limits, there should be a high probability that :
 - 1) the spawning stock biomass is above the threshold where recruitment is impaired, and 2) the fishing mortality is below that which will drive the spawning stock to the biomass
 - threshold which must be avoided.
- To have a high probability to avoid the thresholds, ICES calculates a buffer that when applied to the limit reference points provide estimates of the precautionary reference points F_{pa} and $B_{pa}(_{pa})$ stands for precautionary approach).
- ICES proposed in 1998 and 1999 a number of "lim" and "pa" reference points as a provisional step to the implementation of a precautionary approach.
- F_{pa} and B_{pa} are thus the main devices in the ICES framework for providing advice. They are thresholds which constrain advice or trigger advice for implementation of management/recovery plans.
- If fishery management decisions lead to F_{pa} being exceeded, this would be regarded as overfishing and management would not be regarded as consistent with a precautionary approach.

NAFO

The NAFO presentation addressed the following:

- The PA framework was first defined within NAFO SC in 1997 characterized by limit, buffer, and target reference points for spawning stock biomass and fishing mortality.
- F_{lim} can be no higher than the fishing mortality rate which generates MSY. The target recovery level for biomass (B_{tr}) for overfished stocks is the total stock biomass which would produce MSY.
- B_{lim} is defined as the level of spawning biomass that the stock should not be allowed to fall below.
- Buffers (B_{buf} and F_{buf}) are defined for B_{lim} and F_{lim} to ensure that there is a high probability that the limit reference points are not reached.
- Within each of the biomass/fishing mortality zones defined by the reference points (collapsed, danger zone, recovery zone, recovered zone), specific courses of action are indicated.
- A full suite of reference points has not yet been developed for any NAFO stocks, but substantial progress has been made on some stocks, particularly those with age-based analytical assessments.

The CWP Intersessional review of comparisons contained the following:

COMPARISON OF TERMINOLOGY

- Terminology for limit reference points is consistent. ICES, NAFO and ICCAT use B_{lim} and F_{lim} terms to refer to biomass and fishing mortality limit reference points.
- Terminology for threshold reference points differs between agencies. ICES names these points F_{pa} and B_{pa}, NAFO names them F_{buf} and B_{buf}, ICCAT proposes to name them F_{thresh} and B_{thresh}.
- Target reference points:

Not presently proposed by ICES nor acknowledged in its precautionary framework. NAFO has a conceptual definition of targets for fishing mortality and biomass (F_{target} and B_{target}) but at present only proposes B_{target} reference points for rebuilding purposes. ICCAT notes that its Convention defines F_{msy} and B_{msy} as targets.

COMPARISON OF DEFINITIONS (LIMITS)

- For the biomass limit reference point, the operational definition is that it is a marker of the biomass below which low recruitment can be expected. However, in many cases ICES has also used this as a marker of the biomass below which recruitment is unknown. This alternative usage is not reflected in the nomenclature.
- For fishing mortality limit reference points, the operational definition varies:
 - ICES mostly uses F_{lim} to indicate a fishing mortality above which there is an unacceptable risk of the stock size declining below B_{lim} in some medium or long-term period. Hence it is a marker of the longer term risk of incurring recruitment overfishing.
 - In the NAFO framework $F_{\rm fim}$ is taken as corresponding to $F_{\rm msy}$, which means that it is used as a marker of decreasing stock stability and the loss of long-term yield.
 - ICCAT has yet to develop a position on this, but notes that UNFSA guidelines for a fishing mortality limit are in potential conflict with the ICCAT Convention which implies using F_{msy} as a target.

COMPARISON OF DEFINITIONS (THRESHOLDS)

- Definition of biomass threshold levels tends to be more consistent across agencies.

- Both ICES and NAFO use thresholds as markers of levels of probability, considered unacceptable, that a stock is measured (or forecast) to be at the threshold level, may actually be at or below the limit biomass, given some particular uncertainty assumptions.
- ICES also in some cases defines a threshold level as a marker of a region of unknown dependence of recruitment on stock size. This definition has been applied for some stocks with a history of only moderate exploitation.
- ICCAT's intended use of thresholds is as reference points that fall between limits and targets.

- Definition of F mortality threshold is less consistent.

- ICES has defined F_{pa} in four different ways, as marker of:
 - (a) an unacceptable probability that stock is fished at F_{lim} when it is measured to be F_{pa}, (ie for ICES, unacceptable long-term risk of recruitment overfishing)
 - (b) a high probability of growth overfishing in short term
 - (c) an unacceptable probability that SSB may fall below B_{pa} in medium term
 - (d) an unacceptable probability that SSB may fall below B_{lim} in medium term

- The NAFO definition is similar to (b) above. The ICCAT definition is still not developed.

CONCEPTS AND USAGE

Significant differences in operational definitions of reference points in the ICES, NAFO, and ICCAT areas were identified:

- Such differences have quite normally been driven by differences in the institutional framework in which these scientific bodies operate, and by the different dynamics of the stocks for which they provide advice (eg. many stocks in NAFO area at very low level).

One key difference is that the three organisations have made different interpretations of UNFSA.

NAFO: F_{msy} or a proxy should be adopted as the value for the limit reference point F_{lim} . **ICES:** does not incorporate F_{msy} in its PA framework. ICES considered that F_{msy} is an extremely difficult parameter to estimate reliably and was therefore reluctant to use this value in the provision of management advice.

ICCAT: UNFSA guidelines for a fishing mortality limit are in potential conflict with the ICCAT Convention which implies using F_{msy} as a target.

- Other technical differences in calculation of reference points exist between the NAFO and ICES frameworks.

HARVEST CONTROL RULES

NAFO, ICES and ICCAT all consider that it is the responsibility of the management agencies concerned to pre-agree conservation and management action in the event that they consider such pre-agreements to be necessary.

If a stock falls outside the "safe" or "target" area of its precautionary framework, action should be taken to :

- decrease fishing mortality below the threshold value
- take action to allow biomass to increase towards a rebuilding target.

NAFO has in many instances illustrated a linear reduction in fishing mortality in its precautionary framework. However, as presently most NAFO stocks are below B_{lim} , and such a linear reduction is not particularly germane to present conditions.

In the ICES area, many stocks are presently between limit and threshold reference points, and a diversity of approaches has been taken to proposing recovery plans. These are usually stock-specific and to a greater or lesser extent are evolved in dialogue with management agencies.
DATA MODERATE/POOR ENVIRONMENTS

About half the stocks assessed by ICES, more than 80% of the stocks assessed by the NAFO Scientific Council, and all stocks assessed by ICCAT are considered to be data moderate or data poor - age based assessments are unable to be successfully applied or indirect aging methods are used.

- In such cases alternative methods for assigning reference points are gradually being explored.
- For some stocks, ICES has introduced proxies to represent reference points using indices of stock size and other data sources. In ICES there is continued development of reference points.

In the NAFO SC, surplus production models (ASPIC) have been explored in some data moderate situations, whereas under data poor conditions, the "traffic light" approach has been evaluated. It is anticipated that these and other available methods will be examined in the context of all NAFO stocks in the near future.

ICCAT has a long tradition of using a wide variety of simple assessment methods and a suite
of proxies to reference points that are tailored to fit specific situations.

POSSIBILITIES FOR COMMON USAGE OF CONCEPTS AND TERMINOLOGY

Although specific interpretations of the UNFSA guidelines differed, the objectives of the three scientific agencies of ICES, ICCAT(SCRS) and NAFO (SC) share these common elements:

- Reference points should be chosen in such a way as to allow managers to operate a fishery to take sustainable yields close to the estimated long-term maximum. Reference points should generally lead to stock dynamics which satisfy these conditions, in order of priority:
 - a) Low probability of recruitment overfishing.
 - b) The choice of thresholds should be made so as to avoid a recruitment collapse or to minimize risk when approaching an area where the stock dynamics are poorly known.
- The pa reference points of ICES, the buf reference points of NAFO, and the threshold concept of ICCAT all refer to the same idea, ie. to provide a buffer or safety margin to ensure that here is a high probability that the limit reference points on biomass or fishing mortality will not be reached.
- There are a number of other initiatives on the PA underway in various organizations and national departments. Thus, even if it were possible, it may be premature to recommend a common approach to the PA. In many cases, work on the PA is very much in the exploratory stage.

Annex 5. EU Working Paper

THE PRECAUTIONARY APPROACH IN FISHERIES

The issue of harmonisation of concepts and terminology

1. Prologue

At the 1999 Annual Meeting of NAFO, the EU Delegation was requested to present a working paper on harmonisation of concepts and terminology of the Precautionary Approach. At that time, it was not clear that a CWP Inter-sessional Meeting 2000 would be held in Copenhagen from 14 to 16 February 2000 and bring together representatives from ICES, ICCAT, NAFO and FAO in order to review terminology and definitions of concepts of the Precautionary Approach. As the report of this meeting has not yet become available, the present paper can only be of a provisional nature.

2. The Precautionary Principle

Several international treaties relating to different subject matters such as marine pollution, climate change or biological diversity contain references to the Precautionary Principle. Definitions vary from instrument to instrument and writer to writer. A representative definition drawn from these treaties (e.g. the 1992 Convention on the Protection of the Marine Environment of the Baltic Sea Area) could summarised as follows:

"States shall take preventive measures in respect of action, which may have deleterious effects, even when there is no conclusive evidence of a causal relationship between inputs and their alleged effects."

This constitutes a departure from a former position which held that, if it cannot be convincingly demonstrated that some action will have deleterious effects, that action may be undertaken. The new principle brings with it a reversal of the burden of proof in that it stipulates that, if it cannot be convincingly demonstrated that some action will not have deleterious effects, that action should not be undertaken.

The formulation of the Precautionary Principle clearly gathered momentum at the UN Conference on the Environment and Development (UNCED). This lead to the 1992 Rio Declaration, Principle 15 of which provides that

"In order to protect the environment, the precautionary approach shall be widely applied by States according to their capabilities. Where there are threats of serious or irreversible damage, lack of full scientific certainty shall not be used as a reason for postponing cost-effective measures to prevent environmental degradation."

The preamble of the 1992 Convention on Biological Diversity draws upon this language as follows:

"Noting that it is vital to anticipate, prevent and attack the causes of significant reduction or loss of biological diversity at source,

Noting that where there is a threat of significant reduction or loss of biological diversity, lack. of full scientific certainty should not be used as a reason for postponing measures to avoid or minimise such a threat,"

3. The Precautionary Approach in the field of fisheries

3.1 The 1982 UN Convention on the Law of the Sea requires to rely on "the best scientific evidence available" when taking conservation and management measures (see Article 119 (1) (a) for high seas areas and Article 61 (2) for sea areas under national fisheries jurisdiction). This requirement is sometimes being misinterpreted in too strict if not perfectionist a sense as meaning that, in situations where scientific information offers no full certainty, no conservation measures could be taken. This misinterpretation ignores the inevitable imperfection of science and, therefore, can be seen as one of the motives for the emergence of the Precautionary Approach in the field of fisheries.

The shaping of the Precautionary Approach in the field of fisheries was very much influenced by UNCED and its follow-up processes. It has now been enshrined in both Article 6 of the 1995 UN Agreement on Straddling Fish Stocks and Highly Migratory Fish Stocks as well as Article 7.5 of the 1995 FAO Code of Conduct for Responsible Fisheries. Under these instruments, the Precautionary Approach is to be applied widely to conservation, management and exploitation of fisheries resources. It requires States to err on the side of caution "when information is uncertain, unreliable or inadequate. The absence of adequate scientific information shall not be used as a reason for postponing or failing to take conservation measures." Furthermore, States are required to determine both stock-specific precautionary reference points as well as the action to be taken when theses reference points are approached or exceeded.

This brings together two sets of rules which, at first sight, seem to be irreconcilable. On the one hand, there is acknowledgment that States will continue to work in a world of imperfect information. On the other hand, the technique of reference points requires risk management in a sophisticated decision-making process based on very detailed scientific risk analysis and other objective information. The latter shows that the Precautionary Approach is not only confined to cases where adequate scientific information is lacking.

3.2 The significance of Article 6 of the UN Agreement is that, for the first time in a multilateral fisheries treaty, it spells out the way in which the Precautionary Approach adopted at UNCED is to be applied. However, the UN Agreement *ratione materiae* only covers straddling fish stocks and highly migratory fish stocks. As treaty law, it will not be applicable to so-called 'discrete stocks', i.e. stocks which exclusively occur in high seas areas.

Another special feature of the UN Agreement is that it exemplifies in some detail the technique of precautionary reference points in the form of "Guidelines" set out in Annex II. Guidelines are indicative by their very nature. At the time of their drafting, they were intended to offer States Parties an example of how a system of precautionary reference points could work. They were, however, neither meant to be exhaustive nor were they intended to pre-empt future developments.

3.3 In actual fact, subsequent experience with the Precautionary Approach shows that recent developments have already started overtaking the system of reference points set out in Annex II. This system presupposes perfect knowledge of a given stock. On such a basis, the system would be limited to simply setting both a biomass related conservation (or limit) reference point which defines the critical level of the stock, below which the stock should never fall, and a pre-agreed (target) fishing mortality rate which offers a high probability of the stock not approaching or not falling below the defined critical level.

In view of uncertainties inherent to both the relevant scientific advice as well as the risk assessments needed when deciding upon management strategies, it was seen as a problem that one could never be sure about an entirely accurate selection of reference points. This has led to the concept "trigger points" to mark a security margin or a "buffer", whereby the distance between a conservation (limit) reference point and a "trigger point" is indicative of the risk which is considered to be acceptable in a given case. The more mechanistic system of Annex II does not provide for such security margins and, therefore, falls short of genuine risk management which, in order to judge and determine what is an "acceptable level of risk", presupposes meticulous assessment of potential consequences in terms of gains of lower risks set against losses in yield.

4. ICES/NAFO Frameworks for the implementation of the Precautionary Approach

4.1 Both ICES and the Scientific Council of NAFO have developed and, as this is a dynamic process, are in the course of refining Frameworks for the implementation of the Precautionary Approach. None of these Frameworks has yet been formally endorsed en block by competent management agencies.

4.2 Differences in nomenclature

ICES advises on <u>conservation limits</u> (limit reference points) which define the critical stock level, below which stock size should never fall, and <u>precautionary reference points</u> ("trigger points" or "buffers") which, if adhered to, offer a high probability of keeping the stock above the critical level and which, if approached or exceeded, should trigger remedial action to bring the stock within safe biological limits.

The former consist of **Blim** = absolute bottom line Spawning Stock Biomass (SSB) and **Flim** = abolute upper level of fishing mortality rate (F). The latter consist of **Bpa** = level of SSB higher than Blim and **Fpa** = level of F lower than Flim.

ICES regards Blim and Flim as incontrovertible values. However, the basis for some of the suggested values is debatable. The concept implies that if current F is estimated as greater than Flim or current SSB is estimated as less than Blim, the associated fisheries should be stopped.

NAFO uses three reference points for each SSB and F, namely Blim and Flim, Bbuf and Fbuf (buf = buffer) and Btr and Ftr (tr = target).

Blim and Flim seem to be consistent in both models. However, NAFO classifies Flim as equal to Fmsy (i.e. Flim can be no higher than the fishing mortality rate which generates maximum sustainable yield [msy]).

Bpa and Fpa should in principle correspond to NAFO's Bbuf and Fbuf. It has been suggested, however, that, if Bbuf and/or Fbuf were approached or transgressed, the associated fisheries should be stopped. If this were correct, the difference would not only be semantic. It would amount to using the same uncertainties twice and, thus, lead to defining absolute bottom line SSB at a higher level than really required. The establishment of Blim and Flim would then become a futile exercise.

In the ICES model, no attempt is made to define targets. If the aforementioned perception of Bbuf and Fbuf were the right one, NAFO's Btr and Ftr would in reality be "trigger points". Furthermore, NAFO's Btr is being used as the target recovery level for biomass for overfished stocks and defined as the total stock biomass which would produce maximum sustainable yield. This seems to conflict with the use of Flim as Fmsy as an absolute upper level of fishing mortality rate.

4.3 Differences in interpretation – Flim as Fmsy

Paragraph 7 of Annex II of the UN Agreement states that "The fishing mortality rate which generates maximum sustainable yield should be regarded as a minimum standard for limit reference points".

In the NAFO model, it is contended that this clause would make it a requirement to use Fmsy as the absolute upper level of fishing mortality rate. This would imply an automatism which would force to choose a much lower (target) fishing mortality rate in order to stay away from critical stock levels. This might unnecessarily restrict yields.

There is nothing in the UN Agreement which could support such a strict interpretation. By using the term "should", the clause itself is not constructed as a compulsory one. Furthermore, the clause is embedded in guidelines, the indicative nature of which has already been mentioned above (see point 3.2).

Such a strict interpretation would also conflict with the 1982 UN Convention on the Law of the Sea. Articles 119 (1) [for high seas areas] and 61 (3) [for sea areas under national fisheries jurisdiction] provide that conservation measures shall be designed "to maintain or restore populations of harvested species at levels which can produce the maximum sustainable yield". This implies a stock-oriented finality such that the fishing mortality rate which generates maximum sustainable yield is constructed as a management objective (i.e. a target) which should be achieved with a high probability on average. This excludes the use of Fmsy as a conservation (limit) reference point in the sense of the UN Agreement. Pursuant to the provisions of Article 4 of the UN Agreement, the relevant provisions of the 1982 UN Convention on the Law of the Sea prevail in cases of doubt or conflict.

It should also be noted that the use of Fmsy is extremely difficult to implement for most stocks because of great problems in computing reliable values of Fmsy. For this very reason, ICES' interpretation has been largely to ignore the clause in question.

4.4 Differences in interpretation – pre-agreed remedial action

ļ

Paragraph 4 of Annex II of the UN Agreement states inter alia that previously agreed reference points "shall be used to trigger pre-agreed conservation and management action". The NAFO model uses this clause to suggest that remedial action should consist of a linear decrease in fishing mortality in all cases where stock size falls below the predetermined level.

However, the clause cannot be invoked in support of this suggestion. As shown above (see point 3.3), the Annex II system of reference points only contemplates situations where stock size approaches or falls below the critical level. In such a case, "pre-agreed conservation and management action" in the sense of the said clause will consist of a closure of the associated fisheries.

In contrast to that, experience with "trigger points" or "buffers" has shown that any attempt of pre-determining remedial action in the event that stock size should approach or fall below the so defined safety margin would be too speculative in nature and, thus, fall short of the specific conditions prevailing at the time when remedial action becomes necessary. Indeed, if remedial action were to be pre-determined by a pre-agreed set of measures, the specificities encountered at the time when recourse to such action becomes necessary will almost certainly lead to divergence from the pre-agreed set of measures. In this sense, the pre-agreed set of measures might prejudice proper remedial action. It is clear, however, that the establishment of a precautionary "trigger point" or "buffer" carries with it an agreement of principle to take remedial action whever the relevant pre-determined value is approached or transgressed.

4.5 The need for harmonisation

Differences in terminology are normally indicative of differing concepts. As a general rule, terminology should be used in a harmonised fashion in all cases where there are no conceptual differences. Difference in terminology should be reserved to cases where different concepts so warrant.

Annex 6. Next steps in the implementation of the Precautionary Approach - Cod in Divisions 3NO

Objectives

The action plan for implementation of a Precautionary Approach should include the nine objectives discussed at the Joint Scientific Council/Fisheries Commission Working Group meeting in May 1999:

- 1. Restore and maintain stock at level that can support sustainable fisheries.
- 2. Rebuild SSB to a level that will increase the probability of good recruitment.
- 3. Keep directed fisheries closed in the short term.
- 4. Determine the spawning stock biomass at which the fishery will be re-opened.
- 5. Develop additional criteria to guide potential fishery re-openings.
- 6. Minimize the by-catch for cod in directed fisheries for other fisheries.
- 7. Identify and evaluate options for B_{lim} (60000 t SSB at high productivity level and 35000 t SSB at low productivity level). In doing so, use the following performance measures in the risk analysis:
 - The time (year) at which B_{lim} is reached at various probability levels
 - The yield potential at re-opening.
- 8. Evaluate risks of stocks being below B_{lim}.
- 9. Full review and analysis of 1) the stock recruitment data to determine the high and low productivity levels 2) options for B_{lim} and 3) the appropriate risk analysis.

Management Strategies

- As an initial management objective, Fisheries Commission should rebuild SSB to a level that will increase the probability of good recruitment and restore and maintain the stock at a level that can support sustainable fisheries.
- 2. Fisheries Commission should set a <u>provisional</u> limit SSB reference point of 60,000 t, and should determine harvest strategies and management measures in the context of this reference.
- 3. No directed commercial fishing should occur while SSB is below B_{lim}.
- 4. As there are indications of a possible shift to a lower productivity regime wherein B_{lim} may about 35,000 t, Fisheries Commission should request that Scientific Council should continue to monitor this resource and conduct further reviews of the biomass limit reference.
- 5. Fisheries Commission shall, as appropriate, review and revise these management measures and strategies based on any new advice provided by Scientific Council.

Data Collection/Analyses

- 1. A Contracting Party may submit a proposal to the Fisheries Commission for monitoring activity on 3NO cod to permit further evaluation of resource abundance. The Fisheries Commission, with the prior concurrence of the Coastal State on the proposed monitoring activity, shall seek the advice of Scientific Council with respect to ensuring appropriate data collection related to the proposed monitoring activity.
- It is important to continue to obtain information on spawning times and locations as well as on juvenile nursery areas.
- 3. Ongoing collections of weight-at-length and maturity-at-length data should continue and the data used in the context of the monitoring of SSB and prediction of future trends.
- Information on current spatial distribution of the stock compared to historical should be presented in the assessments.

44

ł

Supportive Management Measures/Good Practises

- 1. Below Blim, the main focus of Fisheries Commission should be to minimize the by-catch of cod, when fishers are directing for other species, and to minimize the catch of juveniles. Some measures that could be considered to achieve this objective are:
 - Review of current directed fisheries for the determination of specific cod bycatch problems so that remedies can be applied.
 - A revision of conservation and technical measures that only permit by-catch that is truly incidental in nature.
 - Closure of specific areas for specific periods of time identified as: a) areas where high levels of cod by-catch are persistent, b) nursery areas, and c) areas where high concentrations of juveniles are found.

Annex 7. Next steps in the implementation of the Precautionary Approach - Yellowtail flounder in Divisons 3LNO

Objectives

The action plan for implementation of a Precautionary Approach should include the eight objectives discussed at the Joint Scientific Council/Fisheries Commission Working Group meeting in May 1999:

- 1. Maintain harvest levels that will continue to rebuild and maintain the stock biomass above the rebuilt biomass level.
- 2. Continue with a comprehensive suite of management measures.
- 3. Ensure a conduct of the fishery in a manner that will not jeopardize recovery of other stocks in the area which are currently under moratorium, specifically 3NO cod and 3LNO American plaice.
- 4. Performance measures of interest to the managers could be expressed in terms of biomass and its trajectory and where it is with respect to the reference level and catch levels. With respect to catch, the performance measure was: cumulated yield, yield trajectories and trends (in
- particular, to identify declining trends).
- 5. It was noted that production models do not permit determination of all reference points. It should be ensured that data are available for scientists to move toward using age-structured modelling.
- 6. Despite these limitations, production modelling is a tool that could be used to start to evaluate real F limits and could be used to provide insight in what will happen if there are lower or higher fishing mortality levels.
- 7. There is a need to develop "target" biomass levels that could be higher than the biological limits so as to take into account management objectives including economic considerations.
- 8. Endorse the work of the Scientific Council in its attempts to develop a better understanding of the stock-recruit relationship.

Management Strategies

- 1. As a management objective, Fisheries Commission should maintain SSB at a level that will continue the probability of good recruitment and maintain the stock at a level that will support a sustainable fishery.
- 2. Given that the present estimate of F_{buf} is in the same range as the 2/3 F_{MSY} value used in past requests from Fisheries Commission, the value of 11% for exploitation rate could continue to be used by Fisheries Commission as a basis for establishing catch levels until such time as Scientific Council may recommend an alternative.
- 3. Fisheries Commission requests Scientific Council to give priority to work aimed at
- calculation of possible biological reference points as appropriate including age-based models and any other applicable stock evaluation methodologies.
- 4. Fisheries Commission shall, as appropriate, review and revise these management measures and strategies based on any new advice provided by Scientific Council.

Data Collection/Analyses

- 1. Scientific Council and Fisheries Commission should encourage continuation of multiple annual surveys in support of stock assessment.
- 2. Contracting Parties should ensure that appropriate data are collected and that scientists utilize stock evaluation techniques that allow for estimation of stock size and exploitation rates, risk assessment procedures, and a fuller evaluation of reference points.
- 3. Scientific Council continue efforts to develop a better understanding of the stock-recruit relationship.

- 46
- 4. Scientific Council and Contracting Parties continue to monitor expansion of the range into Div. 3L.
- 5. Scientific Council and Contracting Parties continue to monitor recruitment as well as trends in weight-at-age.
- 6. Scientific Council to review and update, as necessary, information on spawning locations and timing.
- 7. Scientific Council to provide updated information to the Fisheries Commission regarding the distribution of juvenile yellowtail flounder in relation to adult distribution.

Supportive Management Measures/Good Practises

- 1. Fisheries Commission should take steps to minimize the catch of juveniles, and ensure that the total catches of yellowtail flounder are in accordance with the target exploitation rate. Some measures that could be considered to achieve this objective are:
 - Review of current directed fisheries for the determination of specific yellowtail flounder by-catch problems so that remedies can be applied.
 - A revision of conservation and technical measures that only permit by-catch that is truly incidental in nature.
 - Closure of specific areas for specific periods of time identified as: a) nursery areas, and b) areas where high concentrations of juveniles are found.
- 2. Fisheries Commission to explore the utility of closure periods to protect spawners as well as the utility of closures of areas identified as spawning locations.

Annex 8. Next steps in the implementation of the Precautionary Approach - American plaice in Divisions 3LNO

Objectives

The action plan for implementation of a Precautionary Approach should include the following eight objectives that are similar to those developed during the 1999 meeting of the Working Group for the other 3 model stocks as follows:

- 1. Rebuild SSB to a level that will increase the probability of good recruitment.
- 2. Keep directed fisheries closed in the short term.
- 3. Determine the spawning stock biomass at which the fishery will be re-opened.
- 4. Develop additional criteria to guide potential fishery re-openings.
- 5. Minimize the by-catch for American plaice in directed fisheries for other fisheries.
- 6. Identify and evaluate options for B_{lim}. In doing so, use the following performance measures in the risk analysis:
 - The impacts of possible changes in natural mortality on est imates of B_{lim}
 - The time (year) at which B_{lim} is reached at various probability levels
 - The yield potential at re-opening.
- 7. Evaluate risks of stock being below Blim.
- Full review and analysis of 1) the stock recruitment data to determine the high and low productivity levels 2) the data as they pertain to possible changes in natural mortality 3) options for B_{lim} and 4) the appropriate risk analyses.

Management Strategies

- 1. As an initial management objective, Fisheries Commission should rebuild SSB to a level that will increase the probability of good recruitment and restore and maintain the stock at a level that can support sustainable fisheries.
- Fisheries Commission should adopt a conservation objective for 3LNO American plaice that ensures an ongoing full age range in the spawner population in order to promote the best possible stability in annual recruitment.
- 3. No directed commercial fishing should occur while SSB is below Blim.
- 4. Fisheries Commission should request Scientific Council to continue monitoring of resource and conduct a full review of reference points.
- 5. Fisheries Commission shall, as appropriate, review and revise these management measures and strategies based on any new advice provided by Scientific Council.

Data Collection/Analyses

- A Contracting Party may submit a proposal to the Fisheries Commission for monitoring activity on 3NO cod to permit further evaluation of resource abundance. The Fisheries Commission, with the prior concurrence of the Coastal State on the proposed monitoring activity, shall seek the advice of Scientific Council with respect to ensuring appropriate data collection related to the proposed monitoring activity.
- It is important to continue to obtain information on spawning times and locations as well as on juvenile nursery areas.
- Ongoing collections of weight-at-length and maturity-at-length data should continue and the data used in the context of the monitoring of SSB and prediction of future trends.
- Information on current spatial distribution of the stock compared to historical should be presented in the assessments.
- 5. Scientific Council should continue its investigations on the impact of possible changes in natural mortality, in particular with respect to the determination of reference points for

American plaice in 3LNO. The assessment framework (analytical or otherwise) should be investigated and defined in that context.

Supportive Management Measures/Good Practises

- 1. Below B_{lim}, the main focus of Fisheries Commission should be to minimize the by-catch of American plaice, when fishers are directing for other species, and to minimize the catch of juveniles. Some measures that could be considered to achieve this objective are:
 - Review of current directed fisheries for the determination of specific American plaice bycatch problems so that remedies can be applied.
 - A revision of conservation and technical measures that only permit by-catch that is truly incidental in nature.
 - Closure of specific areas for specific periods of time identified as: a) areas where high levels of American plaice by-catch are persistent, b) nursery areas, and c) areas where high concentrations of juveniles are found.

Annex 9. EU Summary

"The way forward"

Suggested wording for an "Explanatory Memorandum" which could be attached to the Fisheries Commission's future request for scientific advice

Stocks differ greatly in their inherent dynamics, in the amount of available information, and in the information content of the available data. The establishment of biological reference points, and the use of these points for management purposes, needs to be **highly stock specific**. Experience gained so far clearly shows that this is the most important prerequisite to obtain an acceptable result. With this in mind, the Precautionary Approach offers a suitable instrument to achieve the following goals (in order of implementation):

- 1. Ensure sustainability by maintaining a low risk of recruitment decline and stock collapse.
- 2. Where stocks are not overfished, threshold reference points (B_{pa} and B_{buf} , F_{pa} and F_{buf}) should be used in order to avoid entering an area of stock dynamics where either knowledge is poor or risk increases without any increase in yield.
- 3. Allow for sustainable fisheries with appropriate and stabilised yields in the long term.

In order to provide fisheries managers with the information needed to agree on management plans that fulfil these criteria, the Scientific Council should be requested to provide the following:

Risk assessment: whenever possible, estimates of the

- Risks of irreversible damage to the stock
- Risks of stock collapse and recruitment overfishing
- Risks in relation to long-term yield or growth overfishing

as associated with different fishing mortality rates.

When providing risk estimates, it is very important that the **time horizon** is clearly spelt out. By way of consequence, risks should be expressed in time frames of 5, 10 >15 years or other appropriate year ranges depending on stock specific dynamics. Furthermore, fisheries managers also need to consider the balance between risks and yields. For each alternative harvesting strategy or risk scenario, the corresponding yield should be presented over the same time period as related to the risk.

Many of the stocks in the NAFO Regulatory Area are well below an appreciable level of B_{lim} or B_{buf} . For these stocks, the most important task for Scientific Council is to inform on how to rebuild the stocks. The Scientific Council has made clear progress on some of the stocks. It must be emphasised, however, that it is of utmost importance for fisheries managers to obtain the aforementioned type of information. In this context, the importance of alternative recovery plans with time frames of 2-5 years or longer, as appropriate, and the corresponding risk/yield balances must also be stressed. One alternative scenario should always pertain to the consequences and risks of no action at all.

.

·

.

.

SECTION II

(pages 51 to 80)

Report of the Meeting on Shrimp Stocks in the Regulatory Area 27-30 March 2000 Washington, D.C., USA

Report of the N	seting	
1.	Dpening of the Meeting	
2.	Appointment of Rapporteur	
3.	Adoption of Agenda	
4.	Management systems for shrimp in the Regulatory Area	
5.	Report to the Fisheries Commission	
6.	Other matters	
7.	Adjournment of the Meeting	
	Annex 1. List of Participants	
	Annex 2. Opening Statement by the Representative of	
	the United States of America	
	Annex 3. Opening Statement by the Representative of Canada	
	Annex 4. Opening Statement by the Representative of Denmark	
	(in respect of the Faroe Islands and Greenland)	
	Annex 5. Opening Statement by the Representative of the	
	European Union 65	
	Annex 6. Opening Statement by the Representative of Japan	
·	Annex 7. Opening Statement by the Representative of Estonia 67	
	Annex 8. Agenda 68	
	Annex 9. Working Paper by Iceland 69	
	Annex 10. 3M Shrimp Catch Statistics 71	
	Annex 11. Possible Allocation Key for a TAC-based Management	
	System for 3M Shrimp - Paper presented by Norway 72	
	Annex 12. Working Paper by the Chairman 73	
	Annex 13. Working Paper by Denmark (in respect of the Faroe	
	Islands and Greenland) 78	
	Annex 14. Statement by the Delegate of Denmark (in respect of	
	the Faroe Islands and Greenland) 79	



Report of the Meeting on Shrimp Stocks in the Regulatory Area

(GC Doc. 00/3)

27-30 March 2000 Washington, D.C., USA

The Meeting was held in accordance with the decision taken by the General Council at the 21st Annual Meeting, September 1999 (GC Doc. 99/9, Part I, item 4.12).

1. Opening of the Meeting

The meeting was called to order by the Chair, Mr. H. Koster (EU), who welcomed delegates to the meeting. A list of participants is attached as Annex 1.

Several delegates made their brief opening statements. The delegates of USA, Canada, Denmark (in respect of Faroe Islands and Greenland), EU, Japan and Estonia provided their statements to the Rapporteur (Annexes 2-7).

4

2. Appointment of Rapporteur

P.E. Moran (USA) was elected as Rapporteur.

3. Adoption of Agenda

The agenda attached as Annex 8 was adopted.

4. Management systems for shrimp in the Regulatory Area

- 4.1 The Chair stated that the preceding opening comments seemed to indicate concern regarding the current effort allocation for 3M shrimp and its lack of success in controlling harvest to ensure levels of mortality below that advised by the Scientific Council. He noted general agreement among Parties that options should be examined regarding how to best achieve the goal of a 30,000 mt TAC and urged delegates to be open in their analyses. He thanked the delegates who had provided working papers on this subject in advance of the meeting and suggested that these papers be used, in conjunction with information from the Secretariat and the NAFO Conservation and Enforcement Measures, as the basis for initial discussions.
- 4.2 There followed a discussion on the current effort allocation system for 3M shrimp. Some Parties expressed the opinion that such a system of management could not succeed because it failed to take into account the ability of vessels to improve productivity and, thus, catch levels. It was noted that a TAC system provided concrete, scientifically based limits on catch that made such considerations unnecessary. Other delegates supported a continued use of the effort system, pointing out that it is premature to shift to TAC system, as any consideration on factors which caused the failure of current system including overfishing by Contracting Parties or one Party under the objection or "flag hopping" had not been conducted yet. The opinion was expressed that with proper regulation and reporting (e.g., through enhanced monitoring) the effort allocation system could be made effective.

Parties expressed a broad variety of opinions regarding possible future TAC-based allocation schemes for 3M shrimp. While it was generally agreed that any new TAC system should use as its basis elements of the current management scheme, there was a lack of consensus regarding which elements should be used and how they should be applied.

- 4.3 Particular concern was expressed regarding on how historical harvests (and opportunities for harvest) should be reflected in future TAC allocations, including the possible use of historic catch versus allocated fishing days. While there was broad support for the use of historical catch, there was no consensus on how such catches should be translated into TAC allocations. It was also pointed out that the current overall over harvest in the fishery would have to be taken into account in designing a new system and that any new scheme should not reward Parties that had undermined the efforts of the current effort scheme. Again, there was a lack of consensus regarding how such considerations should be reflected in a new scheme.
- 4.4 Delegates then entered into a discussion on the accuracy of the data table found in W.P. (Shrimp) 00/1 by Iceland (Annex 9). Iceland noted that this data was reflected in the paper based on information provided by the Secretariat. A large number of corrections and clarifications to this table were then provided by Parties to the Secretariat. The Executive Secretary stated that these figures were based on available data and that provided by Contracting Parties through hail reports. He also pointed out that the current effort scheme was based on the same data as provided by Parties for 1993 through August 1995. One delegate proposed that Parties submitting revised figures on catch, fishing days or number of vessels shall supplement such figures by stating catch per month (similar to Statlant 21A) and entry, exit and number of fishing days for each trip by the vessels flying the flag of the Contracting Party. After some consideration, it was generally agreed that Contracting Parties should provide data revisions to the Secretariat in time for the June 2000 STACTIC meeting. At this meeting, Parties would be expected to explain these revisions so that newly updated data could then be provided to the Fisheries Commission in time for the 2000 annual meeting. There was no consensus regarding acceptable sources for such data and how (if) they should be verified. However, the Secretariat agreed to make all raw data in its possession available to Parties.

<u>Note</u> (by the Secretariat): Following discussions at the STACTIC June meeting, the original and revised data on 3M shrimp catches were compiled in two Tables of Annex 10.

4.5 In addition, Parties expressed varying opinions regarding the use and appropriate length of a reference fishing period for determining future allocations. The Norwegian Delegation tabled its paper "Possible Allocation Key for a TAC-based Management System for 3M Shrimp" (Annex 11). Some Parties supported the use of relative catch levels at the time of initial allocation, while others proposed the use of a longer reference period. Such a longer period would take into account both the recent development of industries based on this fishery and the choices of Parties to refrain from fishing based on conservation concerns. Some Parties called the establishment of a date after which catches would not be considered when determining historical catch for future TAC allocations. However, there was no consensus on date.

It was pointed out that, regardless of the allocation system used, fishing opportunities should be maintained for all eligible Parties without a history in the fishery through the use of an "others" category. The need for (and amount of) such an allocation was not readily agreed. In addition, several Parties called for the establishment of a guaranteed minimum allocation for Parties with a history. One Party noted that Article XI (4) of the NAFO

Convention implies that the interests of coastal States should be taken into consideration for allocations on the Flemish Cap.

- After considerable discussion, the Chair noted that Parties appeared to be considering four 4.6 options regarding possible elements of a future TAC allocation scheme. These options were then summarized by the Chair in W.P (Shrimp) 00/8 and presented to the Parties for their consideration and comments. The Chair clarified that the data appearing in this paper were illustrative only and subject to revision. In addition, he noted that Parties should consider the four options presented as part of an on-going process. Following further discussion, this paper was reviewed based on the comments of Parties. The Chairman further advised that catch data and all calculations in the paper were still provisional and requested the delegations to provide their finalized data to the NAFO Secretariat. Such data would be incorporated in the Chairman's paper for further consideration. Note (by the Secretariat): All revised data from Annex 10 were incorporated in the Chairman's Paper. Although there was some support for the each of the options found in the revised version of the Chair's working paper, considerable disagreement remained on a variety of elements. Thus, there was no consensus that this paper could be adopted by the group and passed on to the Fisheries Commission for consideration at the 2000 NAFO Annual Meeting. It was only agreed that W.P. (Shrimp) 00/8 as would be revised by modification of catch data should remain a document of the Chair and be retained for use in guiding future work on the issue (Annex 12). The Chair urged that Parties reflect on the options outlined in the paper and be prepared to continue discussions at the 2000 annual meeting.
- 4.7 Regarding possible quota allocations for 3L shrimp, the delegate from Denmark (in respect of the Faroe Islands and Greenland) expressed great dissatisfaction with the current 3L allocation scheme and noted that his country has a track record in this fishery, having caught 1789 mt of 3L shrimp in 1993. This claim is supported by NAFO statistics. He also recognized the legitimate claim of Canada in this fishery based on its coastal State status. The delegate from Denmark then proposed that future allocations in this fishery be made with 2/3 of the TAC in the NRA allocated based on catch history and contribution to scientific data collection and the remaining 1/3 allocated into an "others" quota. This proposal, W.P. (Shrimp) 00/11, is attached as Annex 13.

There was little support among those present for the Danish proposal, although there was recognition that the current allocations of 67 mt did not provide for adequate fishing opportunities for Contracting Parties. It was pointed out that these measures were set to remain in place until the 2001 NAFO Annual Meeting. One Party suggested that it might be beneficial to link the 3L and 3M shrimp fisheries in an effort to provide greater opportunities for shrimp harvests, while others called for status quo until some experience and data could be accumulated in the fishery. It was noted that NAFO needed to determine both the distribution of the stock between the Canadian zone and the NRA as well as how allocations should take place in the NRA. After some discussion, two possible approaches were identified in addition to the Denmark proposal: 1) remain at status quo until an alternative allocation scheme can be agreed; and 2) place all available TAC in an "others" category and allow the fishery to develop. At this time the delegate from Denmark (in respect of the Faroe Islands and Greenland) made a statement (attached as Annex 14). It was agreed that all three of the proposed options should be presented to the Fisheries Commission for consideration at the 2000 NAFO Annual Meeting.

55

5. Report to the Fisheries Commission

It was agreed that the Chair's Working Paper (Annex 12) relating to the 3M shrimp fishery would be further revised as appropriate and used as the basis for continued discussion at the 2000 NAFO Annual Meeting. It was also agreed that advice would be sought from the Fisheries Commission on what future actions (if any) should be taken by the group with regard to 3M shrimp allocations.

With regard to 3L shrimp, it was agreed that all three options for future TAC management should be presented to the Fisheries Commission at the 2000 NAFO Annual Meeting for further discussion and advice on how to proceed.

6. Other matters

No other matters were considered.

7. Adjournment of the Meeting

The Chair adjourned the Meeting on Shrimp Stocks in the Regulatory Area on 30 March 00 at 13.30 hrs.

Annex 1. List of Participants

CANADA

Head of Delegation

E. Wiseman, Director-General, International Directorate, Dept. of Fisheries and Oceans, 200 Kent Street, Ottawa, Ontario K1A 0E6

Representative

E. Wiseman (see address above)

Advisers

C. J. Allen, Resource Management, Dept. of Fisheries and Oceans, 200 Kent St., Ottawa, Ontario K1A 0E6 J. Angel, Executive Director, Canadian Association of Prawn Producers, P. O. Box 1C1, Head of St. Margarets Bay, N.S. B0J 1R0

E. Dussault, Director, Atlantic Affairs Div., International Directorate, Dept. of Fisheries and Oceans, 200 Kent Street, Ottawa, Ontario K1A 0E6

D. Kulka, Northwest Atlantic Fisheries Centre, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

R. MacIntosh, Counsellor, Embassy of Canada, 501 Pennsylvania Ave., Washington, D.C., USA 20008

A. Saunders, Oceans, Environmental and Economic Law Div. (JLO), Dept. of Foreign Affairs and International Trade, 125 Sussex Drive, Ottawa, Ontario K1A 0G2

R. Steinbock, Atlantic Affairs Div., International Directorate, Dept of Fisheries and Oceans, 200 Kent Street, Stn. 1452, Ottawa, Ontario K1A 0E6

DENMARK (in respect of Faroes and Greenland)

Head of Delegation

E. Lemche, Head of Representation, Gronlands Hjemmestyre, Pilestraede 52, Box 2151, Copenhagen, Denmark

Alternate

A. Kristiansen, Ministry of Fisheries, Yviri við Strond 17, P. O. Box 87, FO-110 Torshavn, Faroe Islands

Representatives

E. Lemche (see address above)

A. Kristiansen (see address above)

Advisers

J. E. Hansen, Bondaheygur 9, FR-100 Torshavn, Faroe Islands

P. M. Pedersen, P. O. Box 310, 3900 Nuuk, Greenland

J. Persson, Greenland Home Rule, Dept. of Industry, Box 269, 3900 Nuuk, Greenland

J. H. Toftum, Ministry of Fisheries, P. O. Box 64, FO-100 Torshavn, Faroe Islands

ESTONIA

Head of Delegation

A. Soome, Officer, Fisheries Dept., Ministry of Environment, Kopli 76, 10416 Tallinn

Adviser

T. Saat, Director, Estonian Marine Institute, 18b Viljandi Road, 11216, Tallinn

EUROPEAN UNION (EU)

Head of Delegation

F. Wieland, European Commission, Directorate General for Fisheries, Rue de la Loi 200, B-1049 Brussels, Belgium

Representative

F. Wieland (see address above)

Advisers

H. Koster, European Commission, Directorate General for Fisheries, Rue Joseph II, 99, B-1049 Brussels, Belgium

S. Ekwall, Administrator, Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi/Wetstraat 200, B-1049 Brussels, Belgium

R. Rosenkranz, European Commission, Directorate General for Fisheries, Rue de la Loi 200, B-1049 Brussels, Belgium

R. Uher, Administrator, Council of the European Union, General Secretariat, Rue de la Loi 175, B-1048 Brussels, Belgium

M. H. Figueiredo, Directora de Servicos, Departamento de Relacoes Comunitarias Internacionais e de Cooperacao, Direccao Geral das Pescas e Aquicultura, Edificio Vasco da Gama, Rua General Gomes Araujo, 1399-006 Lisbon, Portugal

R. Akesson, Ministry of Agriculture, 10333 Stockholm, Sweden

S. Feldthaus, Ministry of Food, Agriculture and Fisheries, Holbergsgade 2, 1057 Copenhagen K, Denmark

H. Pott, Bundesministerium fur Ernahrung, Landwirtschaft und Forsten, Rochusstr. 1, D-53123 Bonn, Germany S. Segura, Conseiller des Affaires Etrangeres, Direction des Affaires Juridiques, Ministere des Affaires

Etrangeres, 37 Ouai d'Orsay, 75700 Paris, France

J. Turenne, Chargé de Mission, Ministere de l'Agriculture et de la Peche, 3 Place de Fontenoy, 75007 Paris, France

I. Ybanez, Sub-Director General de Organismos Multilaterales de Pesca, Ministerio de Agricultura, Pesca y Alimentacion, Direccion General de Recursos Pesqueros, Jose Ortega y Gasset, 57, 28006 Madrid, Spain L. M. Esteruelas, Counselor for Agriculture, Embassy of Spain, 2375 Pennsylvania Ave., N.W., Washington,

L. M. Esterueias, Counselor for Agriculture, Embassy of Spain, 2375 Pennsylvania Ave., N. W., Washington, D.C. 20037

G. Taylor, Ministry of Agriculture, Fisheries & Food, Nobel House, 17 Smith Square, London SW1P 3JR

FRANCE (in respect of St. Pierre and Miquelon)

Head of Delegation

D. Silvestre, Sccretariat General de la Mer, 16 Boulevard Raspail, 75007 Paris

Representative

D. Silvestre (address above)

ICELAND

Head of Delegation

T. Asgeirsson, Director of Fisheries, Ingolfsstræti 1, 150 Reykjavik

Representative

T. Asgeirsson (see address above)

Advisers

S. Asmundsson, Ministry of Fisheries, Skulagata 4, 150 Reykjavik

K. Arnason, Ministry of Fisheries, Skulagata 4, 150 Reykjavik

E. Gudnason, Ministry for Foreign Affairs, Raudararstigur 25, 150 Reykjavik

G. Hannesson, Directorate of Fisheries, Ingolfsstræti, 150 Reykjavik

JAPAN

Head of Delegation

T. Iwata, International Affairs Div., Fisheries Agency, Government of Japan, 1-2-1 Kasumigaseki, Chiyoda-ku, Tokyo

Alternate

N. Yagi, Embassy of Japan, 2520 Massachusetts Ave., N.W., Washington, D.C. 20008

Advisers

N. Takagi, Executive Secretary, Japan Deep Sea Trawlers Association, Ogawacho-Yasuda Bldg., 6 Kanda-Ogawacho, 3-Chome, Chiyoda-ku, Tokyo 101-0052

LATVIA

Head of Delegation

N. Riekstins, Director, National Board of Fisheries of the Ministry of Agriculture, 2 Republikas laukums, Rigá LV-1010

Representative

N. Riekstins (see address above)

Adviser

U. Rinkis, National Board of Fisheries of the Ministry of Agriculture, 2 Republikas laukums, Riga LV-1010

LITHUANIA

Head of Delegation

A. Raudonius, Vice-Minister, Ministry of Agriculture, 19 Gedimino pr., Vilnius 2600

Alternate

A. Rusakevicius, Chief Specialist, Fisheries, Dept. of the Ministry of Agriculture, 19 Gedimino pr., Vilnius 2600

Representatives

A. Raudonius (see address above)A. Rusakevicius (see address above)

Adviser

R. Bogdevicius, Deputy Director of Fish Resources Dept. of the Ministry of Environment, A. Juozapavichiaus St. 9, Vilnius 2600

NORWAY

Head of Delegation

T. Lobach, Directorate of Fisheries, P. O. Box 185, N-5804 Bergen

Representative

T. Lobach (see address above)

Advisers

W. Barstad, Norwegian Fishing Vessel Owners Association, P.O. Box 67, Aalesund

H. P. Johansen, Fisheries Counselor, Royal Norwegian Embassy, 2720 34th Street, N.W., Washington, D.C. 20008

P. Sandberg, Directorate of Fisheries, P. O. Box 185, N-5804 Bergen

POLAND

Head of Delegation

L. Dybiec, Director, Fisheries Dept., Ministry of Agriculture and Rural Development Wspólna 30, 00-930 Warsaw

Representative

L. Dybiec (see address above)

Advisers

D. Kravtsova, Parkowa 13/17, M123, Warsaw

M. Kucharski, Embassy of the Republic of Poland, 443 Daly Avenue, Ottawa, Ontario K1N 6H3 B. Szemioth, Parkowa 13/17, M123, Warsaw

RUSSIA

Head of Delegation

M. G. Botvinko, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031

Representative

M. G. Botvinko (see address above)

Advisers

V. M. Kolesnikov, Deputy of Head of Resource Department, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow

I. Mikhno, Embassy of the Russian Federation, 1609 Decatur Street N.W., Washington, D.C. 20011

A. Okhanov, Representative of the Russian Federation in Canada on Fisheries, 47 Oceanview Dr., Bedford, Nova Scotia, Canada B4A 4C4

V. N. Shibanov, PINRO, 6 Knipovich St., Murmansk 183763

V. Solodovnik, VNIRO, 17 V. Krasnoselskaya Stret, Moscow 107140

E. A. Yakunin, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031

60

UNITED STATES OF AMERICA

Head of Delegation

A. Rosenberg, Deputy Assistant Administrator, National Marine Fisheries Service, 1315 East-West Highway, Silver Spring, MD 20910

Representative

A. Rosenberg (see address above)

Advisers

D. Warner-Kramer, International Affairs Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

P. Kurkul, Regional Administrator, Northeast Region, National Marine Fisheries, 1 Blackburn Drive, Gloucester, MA

P. Moran, International Fisheries Div., F/SF4, National Marine Fisheries Service, U.S. Dept. of Commerce, 1315 East-West Hwy., Silver Spring, MD 20910

J.-P. Plé, Senior Atlantic Affairs Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

D. E. Swanson, Chief, International Fisherics Div., F/SF4, National Marine Fisheries Service, U.S. Dept. of Commerce, 1315 East-West Highway, Silver Spring, MD 20910

SECRETARIAT

L. I. Chepel, Executive Secretary

B. J. Cruikshank, Senior Secretary

SECRETARIAT ASSISTANCE

P. Fenwick, Office of Marine Conservation (Rm 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

C. Smith, Office of Marine Conservation (Rm 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

Annex 2. Opening Statement by the Representative of the United States of America (USA)

Mr. Chairman,

I extend a warm welcome to you and all participants to the United States and Washington, D.C. We are happy to see you again and to host this meeting.

As many of you know, I work for the National Oceanic and Atmospheric Administration, which is also the parent organization of the National Weather Service. For those of you who arrived over the weekend and experienced some of the finest weather Washington has to offer, I arranged for those favorable conditions. I have additionally requested that the weather over the course of the week match the progress made here in this room. I have hope for sunny, bright days.

We welcome the pending discussions of shrimp management and the NAFO allocation practice. There should be many ways in which our primarily theoretical discussions of allocation approaches can be advanced by considering the practical cases of 3L and 3M shrimp management and alternatives to them. Conversely, our consideration of shrimp management should further inform our more general allocation discussions.

We are prepared to work with you, Mr. Chairman, and all delegations to carry out the terms of reference of these two meetings. I wish everyone two successful meetings and a pleasant stay in Washington.

Annex 3. Opening Statement by the Representative of Canada

Mr. Chairman, distinguished Representatives, it is a pleasure for Canada to participate at this meeting on shrimp management. We would like to thank the U.S. Government for hosting this meeting and providing the meeting facilities. We would also like to thank the NAFO Secretariat for providing the usual high level of logistical support.

This meeting on shrimp management is timely. For a number of years the Scientific Council has recommended that shrimp catches on the Flemish Cap should not exceed 30,000t; at its meeting in November 1999, it recommended that 3M shrimp catches in 2001 should not exceed 30,000t. It appears that this advice was significantly exceeded last year as 1999 catches of 3M shrimp were over 41,000t - based on the provisional catch reports submitted to NAFO.

Canada would like to thank Iceland for its paper and its proposal for a TAC and quota management regime. As noted in the paper, there are flaws with the current effort limitation scheme. These include the absence of a catch limit, the lack of control on advances in fishing efficiency and the potential for a fishery that can produce a significantly higher level of catch than to date.

Canada is open to any management solution that will ensure that an effective, conservation-based management regime is in place for 3M shrimp for 2001.

Mr. Chairman, Canada looks forward to discussing practical solutions to ensure the conservation of the Flemish Cap shrimp stock.

Annex 4. Opening Statement by the Representative of Denmark (in respect of the Faroe Islands and Greenland)

In order to prohibit an olympic fishery for 3M shrimp NAFO decided to implement an effort limitation system from 1996. For many Parties this was a new approach of managing fishery. For the Faroe Islands, however, this was a well-known system. Since 1984 Faroese shrimp trawlers have fished under such regulation system in the area with Svalbard in the North East Atlantic. Furthermore, the Faroese Parliament in 1994 decided to switch from a quota system for the demersal species in Faroese waters to an effort system. This step was taken due to problems getting the quota system to work properly.

After the implementation of the effort system for vessels fishing for 3M shrimp, some Contracting Parties have questioned this system. They have claimed that due to improvement in fishing technique and equipment the fishing will pass far beyond 30,000 metric tonnes per year. Our delegation does not regard it is of any use to try to prove whether this prophecy is right or wrong. However, we can agree that the catches have increased slightly in the years 1997 to 1999.

The statistics for catches and fishing days given in the attachment to NAFO document GF/00-164 clearly demonstrate that the problem is not the effort limitation system. Based on this information we have made some calculations concerning how the fishing would have been if all Contracting Parties had implemented the effort system. Furthermore, we have made calculations about the overfishing by some Parties who actually did adopt the effort limitation system.

The results of these calculations are very interesting. They show overfishing by especially 3 Parties, varying from 20% to 330% in the years 1996-1999. This overfishing amount from 6% to 72% of the total catches. If the total catches are adjusted for this overfishing, the catches in 1996-1998 would have been below 30,000 tonnes each year.

In other words, we can state that there is no proof for, that the effort limitation system has failed. On the contrary the problem discovered so far is that a number of Contracting Parties have failed to accept and implement the decisions made by NAFO. Furthermore they have fished much more than they have been entitled to.

Having said this we also would like to inform, that even Denmark (in respect of the Faroe Islands and Greenland) has not decided to leave the effort limitation system and adopt a quota system, we are fully prepared to participate in a constructive and creative approach in the discussions about a possible quota allocation system for shrimps in the NAFO Regulatory Area.

Annex 5. Opening Statement by the Representative of the European Union

Thank you Mr. Chairman,

I would first of all like to thank the Government of the United States for hosting this meeting in Washington, D.C., which is extremely pleasant to visit at this time of year with cherry blossoms and nice Spring weather.

Concerning the issues ahead, I would very much like to echo the opening remarks of other Contracting Parties that this is indeed an important exercise. We must most of all look at the system established for 1996 and try to evaluate advantages and disadvantages of both the current system as well as a possible total allowable catch (TAC) and quota system. We must also bear in mind that this is a new fishery since 1995.

I have also some sympathy for what has already been said by Norway. Contrary to the Icelandic suggestion, we believe that both the issues of a TAC and its allocation should be addressed at the same time.

Finally, I would like to stress that we are not meeting in a working group but, as expressed at last year's annual meeting, rather in an exploratory dialogue. Nevertheless, I am looking forward to today's discussions and I hope they will be constructive.

Thank you.

Annex 6. Opening Statement by the Representative of Japan

Thank you Mr. Chairman,

Our delegation also extends our special thanks to the Government of the United States for hosting the meeting.

The basic Japanese position on this fishery is to seek a sustainable use of resources through proper management mechanisms. We respect the NAFO regulations on shrimp in Division 3M.

Japan has allocation of shrimp in Divisions 3M and 3L, but has voluntarily refrained from exercising its rights with regard to these fisheries. It did not operate shrimp fisheries in these areas until last year. This is because Japan was concerned about the possibility of adverse effects of these shrimp fisheries on other fish stocks through by-catch.

However, from the year 2000, Japan is planning to exercise its shrimp fishing rights in 3M and 3L. We think that the by-catch concern regarding demersal fish would be alleviated by using sorting grates.

We hope the outcome of this meeting is successful to the proper management of these shrimp stocks and our delegation is willing to contribute to the discussion.

Thank you.

Annex 7. Opening Statement by the Representative of Estonia

It is always difficult to make rapid changes. The Estonian position is that the possibilities of the effort regulation system are not exhausted, and we suggest to continue the effort regulation of the 3M shrimp fishery using fishing days. To ensure stability and reduce the risk of overfishing, allocation of fishing days to Contracting Parties should take into account the actual number of fishing days used during the previous year.

Estonia is not against introducing the TAC system in the future. However, to achieve this, a transition period is needed before TAC regulation is applied. During the transition period, the state of the stock and the catches should be monitored and the TAC allocation system worked out.

Thank you.

- 1. Opening by Chairman (H. Koster-EU)
- 2. Adoption of Agenda
- 3. Appointment of Rapporteur
- 4. Management system(s) for shrimps in the Regulatory Area
 - Current management system for 3M shrimp
 - Possible TAC-based quota allocation systems for 3M shrimp
 - Possible quota allocation systems for 3L shrimp
- 5. Report to the Fisheries Commission
- 6. Other matters
- 7. Adjournment of the Meeting

Annex 9. Working Paper by Iceland

(W.P. (Shrimp) 00/1)

In 1995, NAFO Contracting Parties agreed to initial management measures for 3M shrimp. Despite the management measures, catches have increased substantially, to an unsustainable level. In the year the measures were agreed upon, 1995, the catches were 28,235 mt but were over 42,000 mt in 1999 according to provisional statistics. This number will almost certainly become even higher when more accurate information becomes available. It is clear that these catches are not sustainable as they are significantly above the scientific recommendation of 30,000 mt. In addition, catches are likely to increase even further this year. In the light of the fact that less than 58% of the allocated fishing days were used in 1999 it is clear that this management system allows for a total catch of over 73,000 mt, based on all fishing days being used with catch per fishing day staying at the 1999 level.

In order to conserve the stock and ensure that the fishery is sustainable in the future it is necessary to change the current management as it is clearly not working as intended. Limiting the number of days used in the fishery has not been enough to keep catches at a sustainable level. The management must limit the actual catches of 3M shrimp. It is therefore necessary to set a TAC which will then be allocated to NAFO Contracting Parties. This would result in the management of 3M shrimp being in line with other NAFO management measures, including the 3L shrimp management measures agreed upon at NAFO's annual meeting last year. It would also bring the management in line with what is the norm in international fisheries management.

As in other cases where a TAC has been decided upon, the main criterion which should be looked at in deciding the national allocations is the relative catches of individual Contracting Parties. This is the case since the rights of coastal states do not apply to 3M shrimp.

Other criteria, such as dependence, should also be considered in deciding the allocation.

Iceland proposes that the NAFO Contracting Parties agree at this meeting on two separate issues regarding the management of 3M shrimp:

- 1. In order to ensure the conservation of the stock and the sustainability of the fishery it is necessary to set a TAC and national allocations thereof.
- 2. In deciding the national allocations, the main criterion to be looked at should be the relative catches of individual Parties.

It is further proposed that new management measures for 3M shrimp, based on a TAC and national allocations thereof, be agreed on at NAFO's annual meeting in September 2000.

Statistics of Shrimp Fishery in the NAFO Regulatory Area

Notes:

- Shrimp effort limitation scheme was introduced from 1996.
 - Days used for 1993-96 taken from STATLANT 21B.
 - Days used for 1997-99 taken from hails.
- Catches for 1993-1998 taken from STATLANT 21A & B.
- Catches for 1999 (also 1998 Faroes) taken from provisional monthly catches. -i ci co 4i vi

Contracting Party Canada		1993		1994		1995		1996			1997			1998			1999	
Canada	Used	Catch	7	Catch	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	Catch
	202	3191	333	1042	319	968		311	806	443	156	784	443	82	435	456	62	385
			•	•	,	1		'		100	•	1	100	,	•	6	ŝ	119
Den Faroes		7076	•	4998	•	5993	-	'	8685	1606	1241	7387	1607	1271	7741	1606	1111	9119
DenGreenland	•	3788	•	2275		2400		•	1107	515		104	515	108	865	515	56	576
Estonia	•	'	•	1051	•	2380	1852	666	1973	1217	692	3239	1217	916	5694	1667	1645	10846
European Union.	139	754	97	432	44	487	508	•	198	457	83 83	593	457	105	1553	457	268	1265
France (SP)	•	•			•	,	N/N		1	100	23	•	N/A	•	F	8		
Iceland	279	2195	638	2355	1842	7481	N/A	5256	20680	N/A	1362	7197	N/A	968	6572	N/A	1312	7643
Japan		•	•	,	,	T	N/A		,	N/A			N/A		•	10	, , ,	
Korea	•	•	•	1		•	N/A	•	•	N/A	•	•	N/A	•	•	100	'	
Lalvia	•	•	190	324	545	679	421	504	1253	400	369	997	400	313	1191	416	598	2765
Lithuania	1		453	863	638	086	638	918	1585	579	611	1785	579	866	3107	579	209	3370
Norway	1354	7075	2130	8625	2113	9534	2206	1482	5805	1985	334	1831	1985	214	1339	1985	428	2976
Poland	•	·	•		'	•	N/A	•	-	N/A	100	,	400	40	148	100	104	207
Russia	76	54	41	350	1533	3327	N/A	2458	444	2600	807	1090	2600	•	,	2100	417	1126
USA				-	•	•]	N/A	,	•	N/A	•		100	•	,	100		
Total	2355	24133	3882	22315	7034	34229	8527	11922	46638	10002	5757	25007	10403	4883	28645	10381	6760	40897
ł		000																
Contraction Deter	-	1950		1884		1995		1996			1997			1998		-	11 666	
contracting rarry	nsea	Catch	2	Catch	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	Catch
Canada	201	3191	333	1042	319	968	•	311	906	443	156	784	443	82	435	456	62	385
		•			•					<u>6</u>	'.	,	100	•		100	33	119
Den Cranes	1324	/333	C8/1	6791	1093	5993	•	**	8688	1606	1250	7410	1606	1292	9368	1606	1051	9199
Dell-Greenland Ectoria	2/6	08/2	482	2722	265	2316			1098	515	31	105	515	113	862	515	65	537
Estoria Currente Haine	149	297 297	609 5	1051	2153	2379	-	990	1898	1217	1254	3240	1217	1454	5533	1667	1651	10834
European Union Eraneo (SD)	951	154	76	432	44	487		•	198	457	83	593	457	105	1553	457	268	1265
		• 1		1			8			100	22	•	100		•	6	'	
iceland	5/2	2195	638	2355	1842	7481	N/A	5256	20682	N/A	1327	6473	N/A	980	6580	N/A	1222	9286
Japan	•	<u>,</u>	1	,	•	, "	<u>6</u>	•		100	•	•	100	`		10	•	
Korea	•	,	•		•	, _ _	100	•		10	•	,	100	`	•	100	'	
Lalvia		,	190	324	649	679			1253	490	439	266	490	402	1191	490	438	3080
Lithuania .		•	453	863	838	980		918	1585	579	611	1785	579	866	3107	579	620	3371
Norway	1403	7074	2206	8625	2162	9391	2206	1549	5648	1985	329	1886	1985	211	1339	1985	394	2975
roland	1	'		'	•			•	'	<u>10</u>	100	817	100	40	148	100	104	859
Hussia	22	24	4	350	1533	3327		2458	4444	2600	807	1090	2600	•	_;	2100	417	1126
	,			•	•		3	•	-	3	•	•	ŝ			ç		
												•	~	i	ĺ	3		•

Annex 10. 3M Shrimp Catch Statistics

Annex 11. Possible Allocation Key for a TAC-based Management System for 3M Shrimp - Paper presented by Norway (W.P. (Shrimp) 00/4)

The allocation key for fishing days for the current effort-regulation system of 3M shrimp is based on the reference period 1993, 1994, and first 8 months of 1995. By applying the same reference period when establishing an allocation key for a TAC-based management system, the shares, and the quotas, for the various Parties will be as illustrated in the table below. In the table a TAC of 30,000 tonnes has been used.

Contracting Parties with no track record in the reference period could be entitled to fish under an others-quota of approx. 3% of the TAC (1,000 t).

Contracting Parties (FC Members)	1993 Catch	1994 Catch	1995 first 8 months Catch 1)	Sum of catch	Share of Catch quota	Quota According to 30000 t 29,000
Canada	3,191	1,042	645	4,878	7.0%	2,042
Cuba	0	0	0	0	0.0%	2,012
Denmark:		Ĭ) v	Ŭ	0.070	Ŭ
Faroes	7,076	4,998	3,995	16,069	23.2%	6,727
Greenland	3,788	2,275	1,600	7,663	11.1%	3,208
Estonia	0	1,051	1,587	2,638	3.8%	1,104
EU	754	432	325	1,511	2.2%	632
France (SPM)	0	0	0	0	0.0%	0
Iceland	2,195	2,355	4,987	9,537	13.8%	3,993
Japan	0	0	0	0	0.0%	. 0
Korea	0	0	0	0	0.0%	0
Latvia	0	324	453	777	1.1%	325
Lithuania	0	863	653	1,516	2.2%	635
Norway	7,075	8,625	6,356	22,056	31.8%	9,234
Poland	0	0	0	0	0.0%	
Russia	54	350	2,218	2,622	3.8%	1,098
Ukraine	0	0	0	0	0.0%	0
USA	0	0	0	0	0.0%	0
TOTAL	24,133	22,315	22,819	69,267	100.0%	29,000

<u>Shrimps in 3M</u>

1) The catch figure for each Contracting Party for the first 8 months of 1995 is found as 8/12 of the total catch in 1995 respectively.
Annex 12. Working Paper by the Chairman

(W.P. (Shrimp) 00/8, Revision 3)⁻

DRAFT (all data to be scrutinized)

Identification of some options for the purpose of guiding the process initiated by the Fisheries Commission at its 21st Annual Meeting in September 1999

Noting the advice provided by the Scientific Council on 3M shrimp (catches should not exceed 30,000 tonnes in 2000 and 2001);

Noting that the catches of 3M shrimp exceeded in 1996 and 1999 30,000 tonnes and are likely to exceed this level in 2000;

A reinforcement of the current management measures needs, therefore, to be considered by the Fisheries Commission.

The options for doing so are the setting of a catch limit e.g. in the form of a total allowable catch (30,000 tons or less) or a maximum number of fishing days (less than 4762 days which corresponds to a reduction of the current number of days allocated to Contracting Parties by approximately 60%)¹.

In the event that a catch limit is set in the form of a total allowable catch, the following options are identified as a basis for allocation of quota to Contracting Parties.

Acknowledging that options presented in this Working Paper do not reflect considerations pursuant to Article XI (4) of the NAFO Convention nor possible other relevant criteria. The options are identified in no order of priority.

⁻ The current total number of fishing days allocated is 11,704 days

⁻ The total number of days used is 6670 days

⁻ The total catch in 1999 is 42,554 tonnes

⁻ The average catch per day can therefore be calculated at 6.3 tonnes per day

⁻ The maximum number of fishing days compatible with the scientific advice can be calculated by dividing 30,000 tonnes by 6.3 tonnes.

OPTION A

This option takes as point of departure the allocation of fishing days under the current management scheme which includes the following elements:

- a). Limitation of the number of vessels fishing for shrimp to the number that have participated in the 3M shrimp fishery from 1 January 1993 to 31 August 1995.
- b) Limitation to the maximum number of fishing days observed for their vessels in one of the years 1993, 1994 or 1995 (until 31 August 1995).
- c) For Contracting Parties with a track record in the period from 1 January 1993 to 31 August 1995 a level of 400 days is permitted.
- d) For Contracting Parties with no track record in this period a level of 100 days with one vessel is permitted.

A basis for quota allocation can be derived as follows:

- 1) allocation will be based on the highest catch in one of the years 1993, 1994 or 1995 (until August 1995)
- 2) or alternatively
- For Contracting Parties with a track record in the period 1 January 1993 to 31 August 1995 the catch figure will be at least 1600 (400 x average catch per day (mt?))
- For Contracting Parties with no track record in the period 1 January 1993 to 31 August 1995, the basis will be at least 400 (100 x average catch per day (mt?))

Contracting	Highest	Minimum		
Party	Catch	Level	Basis	%
Canada	3191	-	3191	7.38
Cuba	-	400	400	0.93
Denmark:				
Faroes	8545	-	8545	19.76
Greenland	3780	-	3780	8.74
Estonia	2379	-	2379	5.50
European Union	754	1600	1600	3.70
France (SPM)	-	400	400	0.93
Iceland	5422 ¹⁾	-	5422	12.54
Japan	-	400	400	0.93
Korea	-	400	400	0.93
Latvia	679	1600	1600	3.70
Lithuania	980	1600	1600	3.70
Norway	9391	-	9391	21.72
Poland	-	400	400	0.93
Russia .	3327	-	3327	7.70
USA	-	400	400	0.93
TOTAL			43235	100%

Basis for allocation (1993, 1994, 1995/1 Jan-31 Aug)

1) corrected on the basis of average catch rate per day for period 1 January - 31 August 1995

OPTION B

This option takes as point of departure the period of application of the current management scheme for 3M shrimp.

As allocation basis, it will be taken the catches in each of the years from 1996-1999 subject to certain corrections of the figures.

Two sub-options are identified:

- B_1 Catches for the period 1996-1999 with the adjusting of the catch figures of the Contracting Parties which are inconsistent with the fishing pattern (e.g. the catches of Contracting Parties which exceeded their allocated fishing days those catches were adjusted to the allocated fishing effort).
- B_2 Catches for the period 1997-1998 with elimination of the years 1996 and 1999 with "extreme" catches.

It should be further clarified that in this table Contracting Parties with no "track record" allocated with a "constant-nominal" 400 mt through the whole period, which most probably should not change principal proportional values of the whole mathematical estimates and basic "shares" but in full fairness, reflect a presence and interest of all Contracting Parties as stakeholders of this resource.

Contracting (96-99	Sub-Opt	ion B 1	97-98	Sub-Opt	ion B 2
Party	1996	1997	1998	1999	Catch	%-1	%-2	Catch	%-1	%-2
Canada	908	784	435	385	2512	1.9%	1.6%	1219	2.1%	2.1%
Cuba	400	400	400	119	1319	1.0%	0.9%	800	1.4%	1.3%
Denmark:						0.0%	0.0%		0.0%	0.0%
Faroes	8688	7410	9368	9199	34665	25,7%	22,7%	16778	28.4%	28.3%
Greenland	1098	105	862	537	2602	1.9%	1.7%	967	1.6%	1.6%
Estonia	1898	3240	5533	10834	21505	16.0%	14.1%	8773	14.8%	14.8%
European Union	198	593	1553	· 1265	3609	2.7%	2.4%	2146	3.6%	3.6%
France (SPM)	400	400	400	400	1600	1.2%	1.0%	800	1.4%	1.3%
Iceland-1	5205	6293	6580	6938	25016	18.6%		12873	21.8%	
Iceland-2	20682	6473	6580	9286	43021		28.2%	13053		22.0%
Japan	400	400	400	400	1600	1.2%	1.0%	800	1.4%	1.3%
Korea	400	400	400	400	1600	1.2%	1.0%	800	1.4%	1.3%
Latvia	1253	997	1191	3080	6521	4.8%	4.3%	2188	3.7%	3.7%
Lithuania	1585	1785	3107	3371	9848	7.3%	6.4%	4892	8.3%	8.3%
Norway	5648	1886	1339	2975	11848	8.8%	7.8%	3225	5.5%	5.4%
Poland	400	817	148	859	2224	1.7%	1.5%	965	1.6%	1.6%
Russia	4444	1090	-	1126	6660	4.9%	4.4%	1090	1.8%	1.8%
USA	400	400	400	400	1600	1.2%	1.0%	800	1.4%	1.3%
Total-1	33325	27000	32116	42288	134729	100.0%		59116	100.0%	
Total-2	48802	27180	32116	44636	152734		100.0%	59296		100.0%

NOTES:

a) Iceland 1- data adjusted for reference

number of fishing days i.e. 1323 days calculated on the basis of the average catch per day

Iceland 2 - actual catch data as reported by Iceland

- b) %-1 this is a ratio from Total 1
 - %-2 this is a ratio from Total 2
- c) All catch data should be verified by Contracting Parties and reported back to the NAFO Secretariat.

d) The data notified by Poland for 1997 are also included in the Icelandic figures.

A further adjustment may be considered for the maximum number of vessels fishing for shrimp which shall not exceed the number that participated in the reference period (total number of named vessels during the reference period).

OPTION C

This option takes as point of departure catch history.

As allocation basis will be taken the catches in each of the years from 1993-1999.

Two sub-options are presented:

- C_1 the sum of the catches for the whole observation period, 1993-1999. In future probable scenario, if decided, the relative share of each Contracting Party would be applied on 90% of the TAC, and the remaining 10% would be set aside as OTHERS quota.
- C2 the sum of the catches for a short reference period (1997,1999). As in C₁, the relative share of each Contracting Party would be applied on 90% of the TAC and, remaining 10% would be set aside as OTHERS quota.

Contracting		-						93-99	Sub-Op	tion C 1	97-99	Sub-Opti	on C 2
Party	1993	1994	1995	1996	1997	1998	1999	Catch	%-1	%-2	Catch	%-1	%-2
Canada	3191	1042	968	908	784	435	385	7713	3.7%	3.4%	1604	1.7%	1.6%
Cuba	-	-	-	-	-	-	119	119	0.1%	0.1%	119	0.1%	0.1%
Denmark:													
Faroes	7333	6791	5993	8688	7410	9368	9199	54782	26.2%	24.1%	25977	27.1%	26.4%
Greenland	3780	2272	2316	1098	105	862	537	10970	5.2%	4.8%	1504	1.6%	1.5%
Estonia	268	1051	2379	1898	3240	5533	10834	25203	12.0%	11.1%	19607	20.5%	19.9%
European Union	754	432	487	198	593	1553	1265	5282	2.5%	2.3%	3411	3.6%	3.5%
France (SPM)		-	-	-	-	-	-	0	0.0%	0.0%	0	0.0%	0.0%
Iceland-1	2195	2355	7481	5205	6293	6580	6938	37047	17.7%		19811	20.7%	
Iceland-2	2195	2355	7481	20682	6473	6580	9286	55052		24.2%	22339		22.7%
Japan	-	-	-	-	-	-	-	0	0.0%	0.0%	0	0.0%	0.0%
Korea		-	-	-	-	-	-	0	0.0%	0.0%	0	0.0%	0.0%
Latvia		324	679	1253	997	1191	3080	7524	3.6%	3.3%	5268	5.5%	5.4%
Lithuania		863	980	1585	1785	3107	3371	11691	5.6%	5.1%	8263	8.6%	8.4%
Norway	7074	8625	9391	5648	1886	1339	2975	36938	17.6%	16.2%	6200	6.5%	6.3%
Poland	-	-	-		817	148	859	1824	0.9%	0.8%	1824	1.9%	1.9%
Russia	54	350	3327	4444	1090	-	1126	10391	5.0%	4.6%	2216	2.3%	2.3%
USA	-	-	-	-	-	-	-	0	0.0%	0.0%	0	0.0%	0.0%
Total-1	24649	24105	34001	30925	25000	30116	40688	209484	100.0%		95804	100.0%	
Total-2	24649	24105	34001	46402	25180	30116	43036	227489	l	100.0%	98332	ļ	100.0%

NOTES:

a) Iceland 1 - data adjusted for reference

number of fishing days i.e. 1323 days calculated on the basis of the average catch per day

Iceland 2 - actual catch data as reported by Iceland

- b) %-1 this is a ratio from Total 1
 %-2 this is a ratio from Total 2
- All catch data should be verified by Contracting Parties and reported back to the NAFO Secretariat.
- d) The data notified by Poland for 1997 are also included in the Icelandic figures.

OPTION D

This option takes as point of departure the conversion of allocated fishing days in an allocation basis.

Contracting Party	Allocated fishing days	%
Canada	456	3.9
Cuba	100	0.9
Denmark:		
Faroes	1606	13.8
Greenland	515	4.4
Estonia .	1667	14.3
European Union	457	3.9
France (SPM)	100	0.9
Iceland	1191 ¹⁾	10.2
Japan	100	0.9
Korea	100	0.9
Latvia	490	4.2
Lithuania	579	5.0
Norway	1985	17.0
Poland	100	0.9
Russia	2100	17.9
USA	100	0.9
TOTAL	11646	100%

¹⁾ corresponding to allocated fishing days reference level minus 10%

% Annex raw data.

Annex 13. Proposal by Denmark (in respect of the Faroe Island and Greenland) (W.P. (Shrimp) 00/11)

Allocation of 3L Shrimps

Taking into account the criteria for quota allocation discussed at the meeting of the Quota Allocation Working Group in March 2000, which most Parties can agree upon – that is fishing track records and contribution to scientific data collection, Denmark, in respect of Faroe Islands and Greenland propose, that:

- 1. 2/3 of the quota in NRA shall be allocated according to catch statistics and contribution to scientific data collection,
- 2. and taking into account the large number of Parties entitled to participate in utilization of the "others" quota, that 1/3 of the quota in NRA is allocated as "others" quota.

Annex 14. Statement by the Delegate of Denmark (in respect of the Faroe Islands and Greenland)

On several occasions Denmark (in respect of Faroe Islands & Greenland-DFG) has flagged its view on the 3L shrimp fishery. Based on research fishery and exploratory fishery we have argued for a reopening for a commercial fishery for shrimp in 3L. To the very last end the Fisheries Commission at the last annual meeting adopted management measures which allow Contracting Parties to undertake commercial fishery in 2000 and 2001. The solution was that each Contracting Party is allocated a quota of 67 tonnes in the Regulatory Area.

However, DFG made it clear at the Fisheries Commission meeting, that this is not a satisfactory solution for DFG. Faroe Islands and Canada have track record for shrimp in 3L. And it is at least our definite view that this track record should be taken into account in the allocation of the available quota.

Therefore we only accepted the equal sharing as a preliminary solution. We have been looking forward for this process to come up with a recommendation to Fisheries Commission which takes into account the interests of those Contracting Parties with a track record as well other relevant criteria such as data collection and scientific surveys.

In this regard I would like to point to the fact that the Faroe Islands have contributed to data collection and scientific research with regard to this stock. In 1994 and from 1996-1999 the Faroe Islands conducted a row of 9 surveys in Div. 3L in order to provide NAFO with data on the shrimp in this area and the potential opportunities for commercial fishery.

The reopening of the 3L shrimp fishery was mainly based on information from this work.

Based on the track record and the contribution to data collection and scientific surveys DFG during the first session of this meeting proposed that two-thirds of the quota for the Regulatory Area be allocated to Contracting Parties with a fishing track record in the area and one-third be set aside as an others quota.

Unfortunately, Contracting Parties do not show any substantial support for this proposal.

Our delegation has listened carefully to the opinions expressed by other Parties regarding the allocation of the 3L shrimp quota. We have noted a general view by a number of Parties, that track record for one year is not considered as enough for allocation purposes. Some CP (USA) indicated 3 years to be more appropriate and referred to Working Paper 00/2 for the W.G. on Allocation Fishing Rights. This is the same time period as was used as basis for the allocation of the 3M shrimp fishery.

However, it has to be borne in mind, that not only the Faroe Islands had the opportunity to fish in 3L in 1993. Vessels from other Contracting Parties could as well have participated in this fishery, but they did not use the opportunity. The result is that DFG has to suffer from the lack of interest by vessels from other Contracting Parties to participate in the 3L shrimp fishery prior to the closing of it effective from 1994.

Taking the fishing track record as indication of "real interests" the DFG was the only Contracting Party showing a "real interest" in this fishery.

DFG has presented its proposal for a future allocation of the 3L shrimp for the Regulatory Area. The proposal is based on criteria we have been discussing during the meeting of the Quota Allocation W.G. and to which most Contracting Parties can agree upon - fishing track record and contribution to data collection and scientific research.

Although DFG seems to stand alone in this topic I can assure all Contracting Parties that DFG will not accept that the track record from 1993 and the contribution to data collection and scientific research be set aside in the allocation of the quota for 3L shrimp.

At relevant up-coming meetings of NAFO, DFG will revert to this issue.

SECTION III

(pages 81 to 102)

Report of the Working Group on Allocation of Fishing Rights to Contracting Parties of NAFO 28-30 March 2000 Washington, D.C., USA

Report of the	Working Group	83
1.	- F 0 0	83
2.	Adoption of Agenda	84
3.	Appointment of Rapporteur	84
4.	Development of a broad strategy of allocation of future fishing	
	opportunities for stocks not currently allocated	84
5.		
	fishing opportunities in connection with the stocks under TACs	86
6.	Allocation of the bloc quota	
7.	Report to the Fisheries Commission	
8.	Other matters	
9.	Adjournment	
	Annex 1. List of Participants	89
	Annex 2. Opening Statement by the Representative of	
	the European Union	
	Annex 3. Opening Statement by the Representative of Canada	95
	Annex 4. Agenda	96
	Annex 5. Further interpretative notes by the Chair attempting	
	to clarify discussion on Agenda point 4	97
	Annex 6. Working Paper presented by Denmark (in respect of Faroe	
	Islands and Greenland)	99

.

•

Report of the Working Group on Allocation of Fishing Rights to Contracting Parties of NAFO

(GC Doc. 00/2)

28-30 March 2000 Washington, D.C., USA

The Working Group on Allocation of Fishing Rights to Contracting Parties of NAFO met in accordance with the decision taken by the General Council at the 21st Annual Meeting, September 1999 (GC Doc. 99/9, Part I, item 4.12).

1. Opening of the Meeting

The meeting was called to order by the Chairman, Mr. H. Koster (EU), who welcomed delegates and made some comments regarding organizational aspects of the meeting. A list of participants is attached (Annex 1).

The delegations of the EU, USA, Canada and Japan made brief opening statements.

The Representative of the EU stated that this meeting was part of an important on-going process and that all relevant elements must be considered in this process. He noted that these elements included questions dealing with equity and balance (among others) and that the real issues associated with quotas and utilization must be addressed. The EU Representative expressed concern that the stability of the organizations should not be negatively effected and urged the Working Group to be realistic in its examination of the available alternatives. The EU Statement was provided to the Rapporteur (Annex 2).

The Representative of the United States pointed out that NAFO had already seen some instances in which there was a clear need for procedures relating to allocation and noted that the work of Denmark (in respect of the Faroe Islands and Greenland) and others should provide a strong basis for continued progress. The US Representative supported the EU statement that equity and stability are key points to bear in mind during the up-coming discussions. Finally, he expressed the US hope that this meeting would result in concrete recommendations to the Fisheries Commission regarding the NAFO allocation process.

The Representative of Canada stated that the issues faced by NAFO with regard to allocation are challenging. He noted that the Working Group thus far has explored some broad international legal issues and stated that the NAFO Convention is the legal basis for allocations within the Organization. After briefly reviewing the progress of the Working Group thus far, he called on Parties to be sensitive to issues relating to stability and conservation in its efforts to achieve consensus on this complex topic. The Canadian Statement was provided to the Rapporteur (Annex 3).

The Representative of Japan noted that his government values the work accomplished thus far by the Working Group. He expressed his hope that the Working Group might contribute to sustainable fisheries and stability within NAFO. He also clarified that the Japanese position on this issue (as outlined in Working Group Working Paper $99/4^*$) remains unchanged. He called for a positive review of unutilized and underutilized allocations within NAFO.

^{*} Note: During this meeting, the Working Group referred a number of working papers from its proceedings, 1999 (April, Halifax).

2. Adoption of Agenda

The agenda was adopted with revisions (Annex 4). It was agreed that the Representatives of Estonia, Latvia, Lithuania, and Russia would meet privately with the Chair to discuss the issue of the bloc quota and that the results of this meeting would be reported to the Working Group at the appropriate time during this meeting.

3. Appointment of Rapporteur

Mr. P.E. Moran (United States) was elected as Rapporteur.

4. Development of a broad strategy of allocation of future fishing opportunities for stocks not currently allocated

4.1 The Chair noted that several working papers regarding allocation had been submitted for the consideration of the Working Group in 1999 and he suggested that these papers could provide the basis for discussions over the next few days. In advance of the meeting, two working papers were distributed. The first paper (Allocation Fishing Rights W.G. W.P. 00/1) provided further interpretive notes by the Chair on the progress of the Working Group. This paper was based on the Chair's notes from the 13-15 April 1999 Working Group meeting (W.P. 99/8 Revised) and sought to further clarify the issues before the Working Group. The second paper distributed in advance of the meeting (Allocation Fishing Rights W.G. W.P. 00/2) was a redistribution of the 1999 working paper by Denmark (in respect of the Faroe Islands and Greenland). Both of these working papers are attached as Annexes 5 and 6, respectively.

The Chair then requested that the Working Group examine the qualifying and allocation criteria outlined in W.P. 99/8 Revised and comment on the current "shopping lists" as found in this paper. The goal of this examination was to further clarify and update W.G. W.P. 00/1.

4.2 Initial discussion on this topic focused on the sources and nature of both types of criteria. While it was generally agreed that Article XI of the NAFO Convention provided the primary basis for both qualification and allocation criteria within the Organization, some Parties also expressed support for the consideration of relevant provisions of the 1995 UN Agreement on Straddling Fish Stocks and Highly Migratory Fish Stocks to guide NAFO's allocation process. A view was expressed that the UN Agreement was not addressing the issue of allocation criteria in the strict sense. There was general agreement among those present that any lists of qualifying and allocation criteria should not be prioritized in any way or considered exhaustive. It was also agreed that qualification should not be considered the right for an allocation.

Regarding qualifying criteria, it was generally agreed that Contracting Parties wishing to be eligible for allocations should be in "good standing" and "interested" (as found W.P. 99/8 Revised), although there was some question as to how such standing should be established. It was also agreed that references in the Working Paper to "Other Contracting Parties" and "Future new members" were not applicable and that they should be dropped from the list. After a brief discussion regarding how the status of "good standing" might be established, it was agreed that text should be inserted to indicate that Contracting Parties who are members of the Fisheries Commission and may exercise the right to vote (based on NAFO rules) would be considered eligible for allocations.

4.3 The Working Group then examined qualifying criteria relating to "interest". Discussion touched on each of the qualifying criteria listed in W.P. 99/8 Revised under "Interested

Contracting Parties" and there was general support for the inclusion of these items in an updated list. Some debate followed regarding the issue of Contracting Parties whose economies are overwhelmingly dependent on fisheries. A number of those present spoke in favor of the inclusion of a criterion relating to overwhelming economic dependence, although there was some concern how such dependence might be substantiated. The view was expressed that, even if it were possible to quantify dependency, it remained doubtful whether this was a suitable criterion in a situation where all Contracting Parties were in principle entitled to be treated on an equal footing. In addition, one Party suggested that special geographic considerations should be taken into account. It was also clarified that Contracting Parties who are members of the Fisheries Commission and may exercise the right to vote must only fulfill one of the various criteria relating to "interest" in order to be considered eligible for allocations.

- 4.4 Regarding allocation criteria, the Chair noted that the items on this list would be used to determine the amount of allocations to eligible.Contracting Parties. It was also agreed that there should be no attempt to weight these criteria at this point. There was general support for the view that allocation criteria should reflect the principle of equity. Although there was general acceptance of the allocation criteria listed in W.P. 98/8 Revised, discussion touched on each of the items in the paper. One Party suggested that all of the qualifying criteria should also be included in the list of allocation criteria, although it was also recognized that too many allocation criteria could complicate the allocation process.
- 4.5 There was general support for the inclusion of an allocation criterion relating to reference fishing patterns during a representative reference period. It was pointed out that such a criterion is, comparatively, easier to quantify. However, concern was expressed that Parties should not be awarded for reference patterns established in a way that undermined NAFO conservation and management. It was noted that, although the allocation criteria did not currently include a compliance element, reference patterns should be chosen that were representative of generally responsible fishing practices. It was agreed that some flexibility would be necessary with regard to this element.

Some concern was also expressed regarding the W.P. 99/8 Revised allocation criterion dealing with Coastal State considerations. In particular, some Parties questioned the inclusion of a consideration relating to "zonal attachment" in criteria designed to provide allocations in the Regulatory Area. After some debate on the issue, it was agreed that the principle of zonal attachment would be addressed by the Fisheries Commission (based on Scientific Council advice) when it determined what proportion of a relevant stock in the NAFO Convention Area would be allocated to the Regulatory Area for eligible and interested Contracting Parties.

4.6 Regarding the creation of an "Others" category containing a lump sum allocation, much of the debate on this issue took place during discussion of the next agenda item. This issue was also discussed in the concurrent Meeting on Shrimp Stocks in the Regulatory Area.

A number of other considerations were discussed with regard to allocation criteria. There was strong support for the inclusion of a reference to Article XI(4) of the NAFO Convention, which relates to allocation within the Organization. It was also generally agreed that considerations from the qualifying criteria relating to contributions to research and data collection and overwhelming dependence on fisheries should be included in the allocation criteria as well. One Party suggested that other contributions to NAFO should also be considered. In addition, arguments were made by some Parties that there should be a specific reference to the needs of small coastal communities.

4.7 There was also some discussion relating to the possibility of setting aside fishing opportunities for vessels of non-Contracting Parties that have demonstrated a high degree of cooperation with NAFO. It was generally agreed that, given the basic qualifying criterion of Contracting Party status, such opportunities could not be considered to be a formal part of the allocation procedure. Instead, it was suggested that such opportunities could be considered by the Fisheries Commission on an ad hoc basis. Parties stressed the need for some type of written agreement (e.g., a protocol) demonstrating a commitment between the non-Contracting Party and NAFO if such an allocation were to be considered. It was pointed out that such a system is currently under consideration by the International Commission for the Conservation of Atlantic Tunas. Although reaction to this idea was somewhat mixed, it was agreed that it should be considered and included in the revised Chair's paper.

5. Exploration of possible margins to accommodate requests for fishing opportunities in connection with stocks under TAC

5.1 Some Parties stated that the current status of most stocks within NAFO made consideration of reallocations particularly difficult. Others generally supported the idea that repeated underutilization of allocations should result in reallocation, although concern was expressed that there might be valid reasons that such underutilization might take place. For example, it was noted that Parties might opt not to harvest an allocation on the basis of conservation, economic, or domestic concerns and that reallocation under such circumstances would be unfair. It was suggested that a time period might be considered in order to firmly establish a pattern of underutilization and that some minimum percentage could be identified below which an allocation might be considered underutilized. However, it was also pointed out that it might not be desirable to obligate Contracting Parties to fully utilize allocated quota and that such a requirement could lead to false catch reporting.

It was also suggested that if all NAFO allocations were reviewed on a regular basis, reallocations would not be necessary. One Party noted that when fisheries are active, the transfer procedure takes care of reallocation as appropriate. The Chair noted that constant reviews of allocations could threaten stability within the Organizations, but agreed that a reasonable review process in conjunction with the use of transfers (in the short term) could be useful.

5.2 Regarding possible allocations of or to the "Others" quotas, there was general agreement that an "Others" quota is desirable, but concern was expressed regarding how changes to the amount of such allocations would effect country-specific allocations. Parties again noted that the current status of NAFO stocks made such discussions difficult. One Party stated that fishing from "Others" quotas was difficult due to practical issues relating to planning and preparation. There was some support for the idea that NAFO might regulate allocations within the "Others" category to ensure a minimum level of allocation available to all eligible Parties.

With regard to the acceptable level of "Others" quotas, some Parties called for a standardized amount for all fisheries. Other representatives expressed the opinion that flexibility was necessary and that the proportions of this quota should be dealt with on a fishery-by-fishery basis. There was some support for the establishment of a range of TAC percentages (e.g., 2%-15%) representing benchmarks within which "Others" quotas might be set on a fishery-by-fishery basis. It was pointed out that this quota should be high enough to allow a Contracting Party with no allocation to participate. Some representatives cautioned against the establishment of "mini quotas" which would not allow for a viable fishery. It was also suggested that this quota should allow such a Party to build a fishing history and, possibly, establish an eventual permanent quota allocation.

Discussion also touched on the issue of who should have access to such an allocation. Some Parties suggested that it might be beneficial to allow access by holders of country-specific quotas to fish in the "Others" category at some point in the fishing season. It was noted that a system could be put into place in which Parties signified an intent to fish within an "Others" quota. If no interest was expressed by a certain deadline, Contracting Parties with country-specific quotas would be allowed to fish this quota.

5.3 The Chair summarized the issues relating to reallocation and the use of an "Others" quota, noting that there was no consensus that fishing opportunities for those without country-specific allocations should come from already allocated fish. Thus, such a system could currently only be recommended for new fisheries on a case-by-case basis. He stated that as stocks increase biologically the Fisheries Commission must decide if it is possible to look at increases to "Others" quotas. The Chair urged the Working Group to consider the needs of all Contracting Parties with regard to fishing opportunities in NAFO.

6. Allocation of the bloc quota

The Representative of Latvia presented the result of an informal meeting between those NAFO Parties that share the bloc quota (Estonia, Latvia, Lithuania, and Russia). The Chairman of the Working Group, H. Koster, was present at that meeting. It was tentatively agreed that all involved Parties would meet before the 2000 NAFO Annual Meeting with the goal reporting to the Commission presenting a proposal for allocations of the current bloc quotas. The Parties further considered that three issues should be reviewed during the up-coming meeting: 1) all relevant stocks to be discussed for further allocation of the block quota; 2) the appropriate reference period to be used in determining block quota percentage shares; and 3) principles to be used for determining percentage share allocations.

7. Report to the Fisheries Commission

One representative noted the difficult nature of the tasks facing the Working Group and suggested that the Fisheries Commission should consider a timetable with benchmarks for addressing the allocation issues facing the organization. It also was suggested that the issue of allocation of new stocks should be a high priority in such considerations. After a brief discussion, it was agreed that the Working Group should recommend that the Fisheries Commission reflect in view of the work done thus far by the Working Group and suggest when it might be appropriate to meet again.

8. Other matters

The Representative from the EU presented information and expressed grave concern regarding the practice of "flag hopping". He clarified that preliminary information indicated that vessels of one Contracting Party appeared to be seeking permission from their government to arrange for double flagging with another Contracting Party. This permission was then being used in conjunction with chartering or other similar arrangements to allow these vessels to operate (apparently at their convenience) under two flags. He stated that such a practice endangers the NAFO quota system by weakening the link between NAFO quota beneficiaries and harvesting vessels. This raises the question if NAFO is an organization of fishing States or an organization of quota sellers. The EU representative pointed out that vessels that engage in flag hopping could be considered Stateless and, thus, should be subject to the new rules adopted by NAFO regarding Stateless vessels.

There was general agreement that the practice of flag hopping could have a negative effect on the NAFO allocation system and many Parties called for an examination of the current NAFO rules regarding bareboat charters. The Chair noted that Contracting Parties are required under the

NAFO Conservation and Enforcement Measures to notify NAFO of all bareboat charters. It was greed that this issue should be discussed at greater length during the 2000 NAFO Annual Meeting in Boston, USA, in September.

9. Adjournment

The Chair adjourned the meeting at 11:55 am on 30 March 2000.

Annex 1. List of Participants

CANADA

Head of Delegation

E. Wiseman, Director-General, International Directorate, Dept. of Fisheries and Oceans, 200 Kent Street, Ottawa, Ontario K1A 0E6

Representative

E. Wiseman (see address above)

Advisers

C. J. Allen, Resource Management, Dept. of Fisheries and Oceans, 200 Kent St., Ottawa, Ontario K1A 0E6 J. Angel, Executive Director, Canadian Association of Prawn Producers, P. O. Box 1C1, Head of St. Margarets Bay, N.S. B0J 1R0

E. Dussault, Director, Atlantic Affairs Div., International Directorate, Dept. of Fisheries and Oceans, 200 Kent Street, Ottawa, Ontario K1A 0E6

D. Kulka, Northwest Atlantic Fisheries Centre, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

R. MacIntosh, Counsellor, Embassy of Canada, 501 Pennsylvania Ave., Washington, D.C., USA 20008

A. Saunders, Oceans, Environmental and Economic Law Div. (JLO), Dept. of Foreign Affairs and International Trade, 125 Sussex Drive, Ottawa, Ontario K1A 0G2

R. Steinbock, Atlantic Affairs Div., International Directorate, Dept of Fisheries and Oceans, 200 Kent Street, Stn. 1452, Ottawa, Ontario K1A 0E6

DENMARK (in respect of Faroes and Greenland)

Head of Delegation

E. Lemche, Head of Representation, Gronlands Hjemmestyre, Pilestraede 52, Box 2151, Copenhagen, Denmark

Alternate

A. Kristiansen, Ministry of Fisheries, Yviri við Strond 17, P. O. Box 87, FO-110 Torshavn, Faroe Islands

Representatives

E. Lemche (see address above)

A. Kristiansen (see address above)

Advisers

J. E. Hansen, Bondaheygur 9, FR-100 Torshavn, Faroe Islands

P. M. Pedersen, P. O. Box 310, 3900 Nuuk, Greenland

J. Persson, Greenland Home Rule, Dept. of Industry, Box 269, 3900 Nuuk, Greenland

J. H. Toftum, Ministry of Fisheries, P. O. Box 64, FO-100 Torshavn, Faroe Islands

ESTONIA

Head of Delegation

A. Soome, Officer, Fisheries Dept., Ministry of Environment, Kopli 76, 10416 Tallinn

Adviser

T. Saat, Director, Estonian Marine Institute, 18b Viljandi Road, 11216, Tallinn

EUROPEAN UNION (EU)

Head of Delegation

F. Wieland, European Commission, Directorate General for Fisheries, Ruc de la Loi 200, B-1049 Brussels, Belgium

Representative

F. Wieland (see address above)

Advisers

H. Koster, European Commission, Directorate General for Fisheries, Rue Joseph II, 99, B-1049 Brussels, Belgium

S. Ekwall, Administrator, Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi/Wetstraat 200, B-1049 Brussels, Belgium

R. Rosenkranz, European Commission, Directorate General for Fisherics, Rue de la Loi 200, B-1049 Brussels, Belgium

R. Uher, Administrator, Council of the European Union, General Secretariat, Rue de la Loi 175, B-1048 Brussels, Belgium

M. H. Figueiredo, Directora de Servicos, Departamento de Relacoes Comunitarias Internacionais e de Cooperacao, Direccao Geral das Pescas e Aquicultura, Edificio Vasco da Gama, Rua General Gomes Araujo, 1399-006 Lisbon, Portugal

R. Akesson, Ministry of Agriculture, 10333 Stockholm, Sweden

S. Feldthaus, Ministry of Food, Agriculture and Fisheries, Holbergsgade 2, 1057 Copenhagen K, Denmark

H. Pott, Bundesministerium fur Ernahrung, Landwirtschaft und Forsten, Rochusstr. 1, D-53123 Bonn, Germany

S. Segura, Conseiller des Affaires Etrangeres, Direction des Affaires Juridiques, Ministere des Affaires

Etrangeres, 37 Quai d'Orsay, 75700 Paris, France

J. Turenne, Chargé de Mission, Ministere de l'Agriculture et de la Peche, 3 Place de Fontenoy, 75007 Paris, France

I. Ybanez, SubDirector General de Organismos Multilaterales de Pesca, Ministerio de Agricultura, Pesca y Alimentacion, Direccion General de Recursos Pesqueros, Jose Ortega y Gasset, 57, 28006 Madrid, Spain

L. M. Esteruelas, Counselor for Agriculture, Embassy of Spain, 2375 Pennsylvania Ave., N.W., Washington, D.C. 20037

G. Taylor, Ministry of Agriculture, Fisheries & Food, Nobel House, 17 Smith Square, London SW1P 3JR

FRANCE (in respect of St. Pierre and Miquelon)

Head of Delegation

D. Silvestre, Secretariat General de la Mer, 16 Boulevard Raspail, 75007 Paris

Representative

D. Silvestre (address above)

ICELAND

Head of Delegation

T. Asgeirsson, Director of Fisheries, Ingolfsstræti 1, 150 Reykjavik

91

Representative

T. Asgeirsson (see address above)

Advisers

S. Asmundsson, Ministry of Fisheries, Skulagata 4, 150 Reykjavik

K. Arnason, Ministry of Fisheries, Skulagata 4, 150 Reykjavik

E. Gudnason, Ministry for Foreign Affairs, Raudararstigur 25, 150 Reykjavik

G. Hannesson, Directorate of Fisheries, Ingolfsstræti, 150 Reykjavik

JAPAN

Head of Delegation

T. Iwata, International Affairs Div., Fisheries Agency, Government of Japan, 1-2-1 Kasumigaseki, Chiyoda-ku, Tokyo

Alternate

N. Yagi, Embassy of Japan, 2520 Massachusetts Ave., N.W., Washington, D.C. 20008

Advisers

N. Takagi, Executive Secretary, Japan Deep Sea Trawlers Association, Ogawacho-Yasuda Bldg., 6 Kanda-Ogawacho, 3-Chome, Chiyoda-ku, Tokyo 101-0052

LATVIA

Head of Delegation

N. Riekstins, Director, National Board of Fisheries of the Ministry of Agriculture, 2 Republikas laukums, Riga LV-1010

Representative

N. Riekstins (see address above)

Adviser

U. Rinkis, National Board of Fisherics of the Ministry of Agriculture, 2 Republikas laukums, Riga LV-1010

LITHUANIA

Head of Delegation

A. Raudonius, Vice-Minister, Ministry of Agriculture, 19 Gedimino pr., Vilnius 2600

Alternate

A. Rusakevicius, Chief Specialist, Fisheries, Dept. of the Ministry of Agriculture, 19 Gedimino pr., Vilnius 2600

Representatives

- A. Raudonius (see address above)
- A. Rusakevicius (see address above)

Adviser

R. Bogdevicius, Deputy Director of Fish Resources Dept. of the Ministry of Environment, A. Juozapavichiaus St. 9, Vilnius 2600

NORWAY

Head of Delegation

T. Lobach, Directorate of Fisheries, P. O. Box 185, N-5804 Bergen

Representative

T. Lobach (see address above)

Advisers

W. Barstad, Norwegian Fishing Vessel Owners Association, P.O. Box 67, Aalesund

H. P. Johansen, Fisheries Counselor, Royal Norwegian Embassy, 2720 34th Street, N.W., Washington, D.C. 20008

P. Sandberg, Directorate of Fisheries, P. O. Box 185, N-5804 Bergen

POLAND

Head of Delegation

L. Dybiec, Director, Fisheries Dept., Ministry of Agriculture and Rural Development Wspólna 30, 00-930 Warsaw

Representative

L. Dybiec (see address above)

Advisers

D. Kravtsova, Parkowa 13/17, M123, Warsaw M. Kucharski, Embassy of the Republic of Poland, 443 Daly Avenue, Ottawa, Ontario K1N 6H3 B. Szemioth, Parkowa 13/17, M123, Warsaw

RUSSIA

Head of Delegation

M. G. Botvinko, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031

Representative

M. G. Botvinko (see address above)

Advisers

V. M. Kolesnikov, Deputy of Head of Resource Department, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow

I. Mikhno, Embassy of the Russian Federation. 1609 Decatur Street N.W., Washington, D.C. 20011 A. Okhanov, Representative of the Russian Federation in Canada on Fisheries, 47 Oceanview Dr., Bedford, Nova Scotia, Canada B4A 4C4

V. N. Shibanov, PINRO, 6 Knipovich St., Murmansk 183763

V. Solodovnik, VNIRO, 17 V. Krasnoselskaya Stret, Moscow 107140

E. A. Yakunin, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031

UNITED STATES OF AMERICA

Head of Delegation

A. Rosenberg, Deputy Assistant Administrator, National Marine Fisheries Service, 1315 East-West Highway, Silver Spring, MD 20910

Representative

A. Rosenberg (see address above)

Advisers

D. Warner-Kramer, International Affairs Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

P. Kurkul, Regional Administrator, Northeast Region, National Marine Fisheries, 1 Blackburn Drive, Gloucester, MA

P. Moran, International Fisheries Div., F/SF4, National Marine Fisheries Service, U.S. Dept. of Commerce, 1315 East-West Hwy., Silver Spring, MD 20910

J.-P. Plé, Senior Atlantic Affairs Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

D. E. Swanson, Chief, International Fisheries Div., F/SF4, National Marine Fisheries Service, U.S. Dept. of Commerce, 1315 East-West Highway, Silver Spring, MD 20910

SECRETARIAT

L. I. Chepel, Executive Secretary

B. J. Cruikshank, Senior Secretary

SECRETARIAT ASSISTANCE

P. Fenwick, Office of Marine Conservation (Rm 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

C. Smith, Office of Marine Conservation (Rm 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

Annex 2. Opening Statement by the Representative of the European Union

The European Community would first like to thank the United States for their hospitality in hosting this important meeting in Washington, D.C. We would also like to commend you, Mr. Chairman, for the skill with which you have guided us through the earlier sessions of this Working Group.

We see this meeting as yet another step in an extremely important process. The topics at issue are as challenging as they are complex. All relevant elements must be carefully examined. It is somehow in the nature of things that the more topics are touched upon, the more questions come up. These questions pertain to substantive issues such as of equity but also to factual elements, e.g. the real reasons for quota under-utilization.

The European Community is prepared to discuss with an open mind possible allocation criteria for stocks not currently allocated and any other topic of principle which might be relevant in the given context. We should, however, recall the elements of balance and stability enshrined in the established allocation practice and we, therefore, share the concerns expressed by others that requests put forth in the course of this process might have implications for the stability of NAFO. We trust that participants will give due regard to these concerns and that a constructive dialogue will help to address all the questions at issue in a realistic manner.

This process may be difficult and may take some time to conclude with solutions which are agreeable to all Contracting Parties. We should not be discouraged by those difficulties as we proceed in a process, the ultimate aim of which is to achieve lasting and sustainable results.

The delegation of the European Community is looking forward to working with you, Mr. Chairman, and with all our partners from the other Contracting Parties to meet the challenges ahead.

Annex 3. Opening Statement by the Representative of Canada

Mr. Chairman, distinguished Representatives, it is a pleasure for Canada to participate in the Working Group on Allocation of Fishing Rights. We would like to again thank the U.S. Government for hosting this meeting and the NAFO Secretariat for providing the usual high level of logistical support.

There is no doubt that the issues before us are challenging. The March 1998 and the April 1999 meetings of the Working Group have highlighted the complexity and sensitivity of these issues. We have explored some of the broad principles of international fisheries law - all delegations have concurred that a variety of established or emerging international law as well as recent international declarations gave guidance on participatory rights within organizations such as NAFO. We have also agreed that the NAFO Convention is the legal framework within which quota allocations must be decided.

During 1999 some of these issues were advanced. Based on the recommendations of the April 1999 meeting of the Working Group, General Council at the 1999 Annual Meeting endorsed the resolution to guide the expectations of future new members with regard to fishing opportunities in the NAFO Regulatory Area. It was important not to raise expectations of potential new members on the fishing opportunities in the NAFO area. Also based on the recommendations of the Working Group, NAFO adopted rules on a pilot basis during the year 2000 for non-flag state vessel charters as well as for notification procedures for "bare-boat" charters. We will need to assess these rules in the near future.

Canada would like to thank the Chairman of the Working Group for his deft handling of the meeting last year and we look forward to his guidance at this meeting. Last year he developed an inclusive "shopping list" of criteria for Contracting Parties to qualify for fishing rights and secondly considerations for the allocation of fishing rights. The Working Group also advanced a number of ideas for possible further consideration on possible margins for allocation in regard to stocks currently under TAC.

Developing a consensus on these questions raises several questions. Is there a set of universal allocation criteria or will each situation require its own criteria? What relative weight should be assigned to the various allocation criteria? The agenda item on possible margins in the current quota table to accommodating requests for fishing opportunities will be complex and sensitive. There are various proposals for reallocation of existing quotas based on some concept of "use it or lose it". These proposals raise substantive issues of equity as well as questions as to the real reason for quota underutilization. These questions need to be looked at carefully.

Canada continues to share the concerns expressed by others that these discussions could have the potential to adversely affect the conservation of the stocks and the stability of the Organization. Based on our discussions to date and the progress made last year, I am confident that the Parties will continue to be sensitive to these concerns and find ways to develop solutions through open, constructive dialogue.

Mr. Chairman, Canada looks forward to examining these questions and making further progress on these issues.

Annex 4. Agenda

- 1. Opening by Chairman (H. Koster-EU)
- 2. Adoption of Agenda
- 3. Appointment of Rapporteur
- Development of a broad strategy of allocation of future fishing opportunities for stocks not currently Allocated (see Annex 11 of NAFO/GC Doc. 99/4, "Interpretive notes by the Chair attempting to clorify discussions on Accord 2," and Accord 2, of NAFO/GC D. 2014

to clarify discussions on Agenda points 6 and 7," and Annex 2 of NAFO/GC Doc. 99/4, "Terms of Reference")

- Qualifying criteria
- Allocation criteria
- Exploration of possible margins to accommodate requests for fishing opportunities in connection with the stocks under TAC (see Annexes 2 and 11 of NAFO/GC Doc. 99/4)
 - Re-utilization, re-allocation
 - Allocation of or to the "Others" quota
- 6. Allocation of block quota
- 7. Report to the Fisheries Commission
- 8. Other Matters
- 9. Adjournment

Annex 5. Further interpretative notes by the Chair attempting to clarify discussion on Agenda point 4 (W. P. 00/1 by the Chairman)

Agenda point 4:

Development of a broad strategy of allocation of future fishing opportunities for stocks not currently allocated (see Annex 11 of NAFO/GC Doc. 99/4" interpretative notes by the chair attempting to clarify discussions on Agenda points 6 and 7", and Annex 2 of NAFO/GC Doc. 99/4, "Terms of Reference").

- Qualifying criteria
- Allocation criteria

When allocating fishing opportunities, the Fisheries Commission will proceed in accordance with the following points:

- A. The Commission will identify the Contracting Parties which are eligible for and interested in the allocation of the relevant fishing opportunities. Contracting Parties who are members of the Fisheries Commission and may exercise the right to vote, will be considered eligible for allocation. The Fisheries Commission will consider Contracting Parties which fulfill one or more of the following criteria as interested in the allocation:
 - Where appropriate (straddling stocks) the relevant coastal state.
 - Contracting Parties whose vessels have traditionally fished the relevant resources.
 - Contracting Parties who have undertaken extensive efforts to ensure the conservation of such stocks in particular by providing surveillance and inspection of international fisheries under the international scheme of joint enforcement.
 - Contracting Parties who have undertaken significant substantial contribution to research and data collection for the relevant resources.
 - Contracting Parties whose economy is overwhelmingly dependent on fisheries.
 - Contracting Parties hosting small coastal communities which are dependent mainly on fishing for the stocks regulated by NAFO.
- B. The Commission will determine, in taking into account any relevant information or advice provided to it by the Scientific Council, the fishable stock(s) or, where appropriate, the portion of the fishable stock(s) in the Regulatory Area to be allocated to Contracting Parties who are eligible and interested in the allocation.
- C. The Commission may take into account the following criteria for the determination of the size of the fishing opportunities to be allocated to Contracting Parties who are eligible and interested in the allocation.

Allocation Criteria

- Reference fishing pattern converted in the relative share of the Contracting Parties concerned.
- The setting aside of a lump sum as others quota intended for Contracting Parties who have no record of fishing on the stock concerned.
- Fixing a minimum size for quota to be allocated to Contracting Parties
- Considerations

- pursuant to Article XI (4) of the NAFO Convention

- relating to the contribution to research and data collection
- relating to the needs of small coastal communities
- relating to the dependency on fisheries
- D. The criteria listed under points A and C are indicative, apply simultaneously and do not represent an order of importance or priority.
- E. Notwithstanding points A and C, the Fisheries Commission may set aside and regulate certain fishing opportunities available to vessels of parties which are not a Contracting Party to the NAFO Convention, who have signed a protocol on the integral acceptance of the NAFO Conservation and Enforcement Measures, enabling such Party to cooperate with NAFO.

Annex 6. Working Paper by Denmark (in respect of Faroe Islands and Greenland) (W.P. 00/2)

The Working Paper is inspired by the U.S.A. paper "Proposal by the U.S.A. for a Northwest Atlantic Fisheries Organization Policy on Allocation of Quotas" (Working Group W.P. 98/6).

The attachment to this Working Paper is divided into 3 sections: "Questions to be Addressed", "Suggested Solutions" and "Remarks". We consider that such a division will facilitate discussion. Delegations might agree with the "Questions to be Addressed" while not agreeing with the "Suggested Solutions".

Denmark (in respect of Faroe Islands and Greenland) considers that any quota mechanism should contain an element of stability. However, stability should not be interpreted as unchangeability. Hence, the proposal in the Attachment should not have a duration of more than 5-10 years.

We also are aware of the risk that any new mechanism for setting TAC's or quotas might be applied so strictly that the role of the Fisheries Commission would evaporate. Therefore the proposal in the Attachment should be regarded as "Principal Guidelines", from which exemptions can be made if concrete circumstances so warrant.

The proposal in the Attachment seeks to reflect – to a reasonable extent – existing NAFO principles, whilst also taking into account appropriate changes caused by developments since the existing quota sharing system was taken over by NAFO twenty years ago.

(Attachment to W.G. Working Paper 00/2)

QUESTIONS TO BE ADDRESSED	SUGGESTED SOLUTIONS	REMARKS
A: ALLOCATION BETWEEN PRESENT NAFO MEMBERS		
1. Should the receiver of allocations be "in good standing"?	Yes	Same as U.S.A. proposal.
 Should preference be given to Coastal States? 	Yes, with respect to straddling stocks.	The weight of the preference should vary from stock to stock, and reflect the different zonal attachments.
3 Should allocation only take place when the allocation has a certain minimum size?	Yes, but the minimum size may vary from stock to stock.	An allocation should, as a minimum, enable the Contracting Party to conduct an economically meaningful fishery (e.g. one full hold for one vessel).
		Minimum sizes should be fixed for each stock, and may vary from year to year.
	If an amount to be allocated under 4a-d falls short of the minimum size, the amount should be transferred to the "others" quota.	
4a. Current fisheries with TAC's.		In 1999 only four NAFO TAC's are bigger than 0: 3M redfish, 3LNO yellowtail, 3LMNO Greenland halibut and 3+4 squid.
	TAC's up to traditional levels: should be allocated corresponding to NAFO's present allocation key.	"Traditional levels" = highest TAC's within the period [1980-1990].

REMARKS		The suggested solution should not be considered as a proposal to establish a TAC/quota system for 3M shrimps.		The suggested solution should not apply if the fishery has been closed for less than [5] years or more than [15] years.
SUGGESTED SOLUTIONS	Increases of TAC's beyond traditional levels should be split in three parts: one should be allocated corresponding to NAFO's present allocation key; one should be allocated to present Contracting Parties whose economies are overwhelmingly dependent on the exploitation of iving marine resources; one should be transferred to the "others" quota.	 [70] percent of the TAC should be allocated proportional to catches in a relevant period. [20] percent of the TAC should be allocated proportional to the present allocation of fishing days. [10] percent should be set aside for the "others" quota. 	[90] percent of the TAC should be allocated proportional to fisheries in the latest [3] years. [10] percent of the TAC should be set aside for the "others" quota.	Upon reopening: TAC's up to traditional levels: [90] percent of the TAC should be allocated to previous quota receivers, corresponding to the latest NAFO allocation key before the fishery was closed. [10] percent should be set aside for the "others" quota. Increases of TAC's beyond traditional levels should be allocated as in 4a.
QUESTIONS TO BE ADDRESSED		4b. Fisheries previously regulated by effort limitation.	4c. New fisheries	4d. Closed fisheries

QUESTIONS TO BE ADDRESSED	SUGGESTED SOLUTIONS	REMARKS
5. Should transfers be allowed?	Yes, on the same conditions as today.	
 Should underutilization of guotas imply consequences for future allocations? 	Yes. Repeated underutilization of a Contracting Party's quota should imply that [part of] the unutilized quota is transferred to the "others" quota.	Transfers should be regarded as under- utilization by the original quota receiver.
B: PARTICULARS WITH RESPECT TO FUTURE NEW MEMBERS.		This proposal takes no stand on the question of eventual future conditions
 From which TACs may future new members receive allocations? 	TAC's mentioned under A 4c, <u>not</u> TAC's mentioned under A 4a, b and d.	for NAFOs approval of applications from new members ("real interest").
2. How much may future new members fish?	Future new members participate on equal footing with present Contracting Parties in allocations mentioned in A 4c.	
	In addition, future new members may fish under the "others" quotas mentioned in A 4a-d.	

SECTION IV

(pages 103 to 130)

Report of the Working Group on Dispute Settlement Procedures (DSP) 29-31 May 2000 Copenhagen, Denmark

Report of the Working Group	
1. Opening by the Chairman	
2. Appointment of Rapporteur	
3. Adoption of Agenda	
4. Examination of the desirability and, as appropriate, the deve	lopment
of procedures for the settlement of disputes between NAFO	
Contracting Parties a) by implementing in a NAFO context	
Agreement and UNCLOS dispute settlement procedures, ar	
including additional measures if needed	
5. Report to the General Council	
6. Other matters	108
7. Adjournment	108
Annex 1. List of Participants	109
Annex 2. Agenda	
Annex 3. NAFO Dispute Settlement Procedures – A Possib	le
Way Forward (Paper by Chairman)	114
Annex 4. Working Paper by Denmark (in respect of Faroe	
Islands and Greenland)	
Annex 5. Working Paper Presented by the European Union.	
Annex 6. Working Paper Presented by the European Union.	
Annex 7. Working Paper Presented by Canada	
Annex 8. Working Paper Presented by Japan	
Annex 9. Working Paper Presented by Canada	
Annex 10. Working Paper Presented by Canada	
Annex 11. Settlement of Disputes within NAFO –	
Compilation of Proposals	126
Annex 12. Settlement of Disputes within NAFO –	
Consolidated Text	

.

-1 i i

Report of the Working Group on Dispute Settlement Procedures (DSP)

(GC Doc. 00/4)

29-31 May 2000 Copenhagen, Denmark

The Working Group met in accordance with the decision taken by the General Council at the 21st Annual Meeting, September 1999 (item 4.8 of the General Council Report, GC Doc. 99/9).

1. Opening by the Chairman

The meeting was opened by the Chairman, Mr. Stein Owe (Norway) at 10.15 on 29 May 2000. He welcomed all delegates and thanked the delegation of Denmark (in respect of the Faroe Islands and Greenland) for hosting the meeting. The following Contracting Parties were represented at the meeting: Canada, Denmark (in respect of Faroe Islands and Greenland), Estonia, European Union, France (in respect of St. Pierre and Miquelon), Iceland, Japan, Latvia, Lithuania, Norway and the United States of America (Annex 1).

In his welcoming remarks, the Chairman anticipated a constructive debate and expressed a wish to finalise the work of the Working Group as soon as possible.

2. Appointment of Rapporteur

Mr. Staffan Ekwall (EU) was appointed Rapporteur.

3. Adoption of Agenda

The Provisional Agenda was adopted (Annex 2).

4. Examination of the desirability and, as appropriate, the development of procedures for the settlement of disputes between NAFO Contracting Parties a) by implementing in a NAFO context the 1995 UN Agreement and UNCLOS dispute settlement procedures, and b) by including additional measures if needed.

- 4.1 The Chairman presented DSP W.G. W.P. 00/1 (Annex 3), which was meant as an attempt to move the discussion forward. He explained that the best way to make progress and to move closer to a quick solution would in his opinion be to abandon the idea of an *ad hoc* Panel Procedure discussed at earlier Working Group meetings. This procedure, in the Chairman's opinion, has created many questions of its own, and considerable work to establish procedural rules would be needed. One objective of a NAFO DSP would be to settle a dispute expeditiously. However, it is not likely that the Panel Procedure would speed up the process since the losing Party, especially in case of serious disputes, will probably not accept the outcome of such proceedings. It would, therefore, be better to stick with the procedures stipulated in the UN Convention on the Law of the Sea (UNCLOS) and the 1995 UN Agreement on straddling stocks with some modifications, if necessary.
- 4.2 There was a wide-ranging debate on this paper. Some delegations welcomed the paper and expressed a wish to have simple dispute settlement rules or guidelines and to avoid a complex and time consuming negotiation exercise on a completely new DSP applicable only in NAFO. Other delegations appreciated the initiative by the Chairman but stressed

that the task of the Working Group was to establish a NAFO DSP in the proper sense in light of recent international developments. Other Regional Fishing Organisations are presently discussing this issue, and in the view of some delegations it would be odd if NAFO remains the only Regional Fishing Organisation without a special DSP. In this context reference was made to the recent "Blue Fin Tuna case", which to some delegations shows the importance of having a NAFO DSP.

Many delegations emphasised a need to have 1) a speedy procedure 2) a binding procedure and 3) a mandatory obligation for an objecting Contracting Party to indicate its post-objection behaviour. Some delegations underlined that a NAFO DSP based on the relevant provisions of UNCLOS and the 1995 UN Agreement could, and should, apply to all Contracting Parties, whether or not they have ratified UNCLOS and the 1995 UN Agreement or only one of these instruments. A view was expressed that the provisions of the 1995 UN Agreement should also apply to disputes on discrete stocks, and that both the declaration of intentions following an objection or notice not to be bound as well as the actual post-objection behavior could be the subject of DSP. Some delegations either had misgivings about an extended application of the 1995 UN Agreement or queried how this could be achieved in practice. The delegate of Japan felt that most disputes will derive from scientific and political decisions and that such disputes can not be resolved by judicial arguments.

- 4.3 The delegation of Denmark (in respect of Faroe Islands and Greenland) introduced Working Paper DSP W.G. W.P. 00/3 (Annex 4) and explained that the paper was an attempt to show some goals that could be achieved by a DSP. Most delegations considered this paper as contributing considerably to a clarification of the questions at hand and thanked the Danish delegation for its effort.
- 4.4 Following the discussions, the Chairman concluded that he did not have sufficient support for his idea of a simplified scheme.

4.5 EU Revised Paper

The Chairman therefore suggested to take as a point of departure for the further discussions the revised EU paper presented at the last Working Group meeting in Bergen, Norway, 1999 (Annex 5).

The EU delegation presented DSP W.G. W.P. 00/2 (Annex 6) as an attempt to clarify the different procedural options available under the procedures laid down in the aforementioned revised EU paper. The main idea was to provide for a voluntary *ad hoc* Panel which would offer a much more swift and cost effective process for disputes over conservation measures and which would help to resolve these very disputes within NAFO. In the event that a dispute should not be resolved at this stage and one of the parties to the dispute should have recourse to the general binding procedures, the recommendation of the panel should nevertheless apply as a provisional measure pending the definitive and binding settlement of the dispute. The parties to the dispute would remain the 'masters of the game' at this stage as well because it would be in their hands to either agree otherwise or request the competent court or tribunal to prescribe other provisional measures. Many delegations welcomed the Working Paper as a clarification of the procedures proposed by the EU.

i) On the first point in the revised EU Paper, many delegations felt that a Contracting Party must fulfil three obligations when making an objection or notice of intention not to be bound by a measure, namely 1) state the reasons for the objection or notice, 2) state its intentions following the objection or notice and 3) give a description of the possible autonomous measures to be taken. Some delegations underlined that this description should include relevant control and enforcement measures.

One delegation stated that it should be possible to initiate a DSP as well on the basis of the stated intentions as on the actual post-objection behaviour. A new text (DSP W.G. W.P. 00/4 - Annex 7) was presented to capture these possibilities. Some delegations expressed concern that a possibility to initiate a DSP already on the basis of the declaration of intentions might lead to a limitation of the right to object. It was emphasised that an objection itself cannot form the matter of a dispute. It was added that the objecting Party might not always be able to give the requested information at such an early stage.

- ii) On the second point in the revised EU Paper, some delegations pointed out that some Contracting Parties have not ratified UNCLOS and/or the 1995 UN Agreement. A reference to these instruments might be seen as implying an indirect acceptance by these Parties of the relevant instruments. Other delegations pointed out that the intention was never to make a Contracting Party bound to international instruments outside the NAFO context. The intention was only to 'import' the procedures laid down in these instruments and, thus, take advantage of a legal technique which has been used frequently and which would make already existing rules available for the purposes of settling disputes within NAFO. A view was also expressed that there should be a level playing field: all Contracting Parties should be bound by the same rules and those rules should reflect the most modern standards (i.e. UNCLOS and the 1995 UN Agreement) and apply equally to NAFO straddling and discrete stocks. Annex 9 was based on that approach.
- iii) On the third point in the revised EU paper, most delegations preferred to focus the discussion on the concept of having an *ad hoc* Panel and, if this would be the case, adopt procedural rules at a later stage. Some delegations stressed that the Panel Procedure must in any case be voluntary. Parties should not be constrained from resorting directly to the binding procedures. Other delegations stressed the need to have an established NAFO dispute settlement mechanism that the Parties to the dispute are encouraged to use. This would in their view speed up the process, since the Parties to the dispute otherwise must first agree on all procedural matters. One delegation asked for a mechanism that would avoid repeated disputes on exactly the same issue and inquired, if such a mechanism was not possible to foresee, who should pay the cost for such repetitious exercises.
- iv) On the fourth point in the revised EU Paper, the EU delegation explained that the Parties to the dispute, when agreeing to choose the Panel Procedure, also agree to apply the recommendation from the Panel provisionally. The main aim was to bridge the gap of time between the delivery of the recommendation and the final binding settlement of the dispute. Some delegations supported this approach. Other delegations, however, disagreed with a provisional application of the panel recommendation. They stressed that the Parties to the dispute must have the right to choose, at the time when the recommendation is given, if it should apply provisionally or not.

One delegation indicated that the Parties to a dispute should take such a decision already when they agree on a Panel Procedure. As a compromise, this delegation

suggested that two different Panel Procedures could be established, one where the Parties to the dispute agree to the provisional application of the recommendation when they agree on a panel procedure, and another where no such automatic application will occur.

- v) On the fifth point in the revised EU paper, the Latvian and the Japanese delegations once again expressed their concerns that a reference to UNCLOS and the 1995 UN Agreement could constitute an indirect "ratification" by those Contracting Parties who have not ratified these instruments. Those delegations requested their reservations on this subject be recorded in the minutes of the Meeting. The Lithuanian and Estonian delegations shared the view of Latvia and Japan. More neutral wordings were proposed by some delegations.
- 4.6 The Japanese delegation presented DSP W.G. W.P. 00/8 (Annex 8) containing an alternative DSP specially designed for NAFO. The main idea was to have a compulsory but non-binding *ad hoc* Panel Procedure with a final resolving of the dispute by the Fisheries Commission if the Parties to the dispute do not accept the *ad hoc* Panel recommendation. This proposal was welcomed for giving new ideas. However, the proposal was not discussed in detail since it was already clear from the earlier debate that the Working Group could not agree on a binding DSP in this form.
- 4.7 As an attempt to summarise the outcome of the first round of discussions, the Chairman presented DSP W.G. W.P. 00/09 (Annex 11). This paper was later revised following the discussions in the Working Group. DSP W.G. W.P. 00/10-Revised (Annex 12) which contains text and alternatives in brackets, reflects the current level of agreement and views expressed to-date in the Working Group.

5. Report to the General Council

Following the extensive discussion which took place during the meeting, the Working Group agreed to submit the Consolidated Text (Annex 12) to the General Council together with its report.

The Group discussed the possibility to meet again on Monday, 18 September 2000 in connection with the Annual Meeting.

6. Other matters

No other matters were discussed.

7. Adjournment

The meeting was adjourned at 17.30 on 31 May 2000.
Annex 1. List of Participants

CANADA

Head of Delegation

H. Strauss, Director, Oceans, Environmental and Economic Law Div., Dept. of Foreign Affairs and International Trade Canada, 125 Sussex Drive, Ottawa, Ontario K1A 0G2 Phone: +613 992 2104 - Fax: +613 992 6483 - E-mail: <u>howard.strauss@dfait-maeci.gc.ca</u>

Advisers

N. Bouffard, Senior Advisor, Legal Issues, International Affairs Directorate, Dept. of Fisheries and Oceans, 200 Kent St., 13th Floor, Stn. 13159, Ottawa, Ontario K1A 0E6 Phone: +613 993 1860 - Fax: +613 993 5995 - E-mail: <u>bouffardn@dfo-mpo.gc.ca</u>

DENMARK (IN RESPECT OF FAROE ISLANDS & GREENLAND)

Head of Delegation

E. Lemche, Head of Representation, Gronlands Hjemmestyre, Pilestraede 52, Box 2151, DK-1016 Copenhagen K, Denmark Phone: +45 33 69 34 35 - Fax: +45 33 69 34 01 - E-mail: <u>Einar.Lemche@ghsdk.dk</u>

Advisers

O. H. Larsen, Minister Counsellor, Royal Danish Ministry of Foreign Affairs, Legal Service, 2 Asiatisk Plads, DK-1448, Copenhagen K, Denmark

Phone: +45 33 92 00 00 - Fax: +45 31 54 05 33 - E-mail:

L. Lind, Head of Section, Royal Danish Ministry of Foreign Affairs, 2 Asiatisk Plads, DK-1448, Copenhagen K, Denmark

Phone: +45 33 92 00 00 - Fax: +45 32 54 05 33 - E-mail: lilind@um.dk

J. Persson, Head of Section, Greenland Home Rule, Department of Industry, Box 269, 3900 Nuuk, Greenland

Phone: +299 34 53 24 - Fax: +299 32 47 04 - E-mail: jpe@gh.gl

H. Weihe, Ministry of Fisheries, P. O. Box 64, FR-110 Torshavn, Faroe Islands Phone: +298 353030 - Fax: +298 353035 - E-mail: hedinw@fisk.fl.fo

ESTONIA

Head of Delegation

K. Mühlbaum, Ministry of the Environment, Fisheries Department, Marja 4d, 10617 Tallinn Phone: +372 6566720 - Fax: +372 6567599 - E-mail: <u>kristiina@klab.envir.ee</u>

Adviser

A. Soome, Ministry of the Environment, Fisheries Department, Marja 4d, 10617 Tallinn Phone: +372 6112987 - Fax: +372 6567599 - E-mail: ains@klab.envir.ee

EUROPEAN UNION (EU)

Head of Delegation

F. Wieland, Deputy Head of Unit, International Fisheries Organizations and Fisheries Agreements; Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi 200, B-1049 Brussels, Belgium Phone: +32 2 296 3205 - Fax: +32 2 299 4802 - E-mail: Friedrich.Wieland@cec.eu.int

Alternate

T. van Rijn, Conseiller juridique, European Commission, Legal Service, Avenue des Nerviens 85, B-1040 Brussels, Belgium

Phone: +32 2 295 1818 - Fax: +32 2 295 2485 - E-mail: Thomas.van-Rijn@cec.eu.int

Advisers

S. Ekwall, Administrator, Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi/Wetstraat 200, B-1049 Brussels, Belgium Phone: +32 2 299 6907 - Fax: +32 2 299 4802 - E-mail: <u>Staffan.Ekwall@cec.eu.int</u>

M. H. Figueiredo, Directora de Servicos, Dept. de Relacoes Comunitarias, Internacionais e de Cooperacao, Direccao Geral das Pescas e Aquicultura, Edificio Vasco da Gama, Alcantara Mar, 1350 Lisbon, Portugal

Phone: +351 21 3914350 - Fax: +351 21 3979790 - E-mail: hfigueir@dg-pescas.pt

S. Salvador, Chefe da Divisao de Relacoes Internacionais, Direccao Geral das Pescas e Aquicultura, Edificio Vasco da Gama, Alcantara Mar, 1399-006 Lisbon, Portugal

Phone: +351 21 3914352 - Fax: +351 21 3979790 - E-mail: susanas@dg-pescas.pt

J. Neisse, Administrator, Council of the European Union, DG I, Rue de la Loi 175, B-1048 Brussels, Belgium

Phone: +32 2 285 7097 - Fax: +32 2 285 7822 - E-mail: jurgen_neisse@consilium.eu.int

S. Feldthaus, Head of Section, Ministry of Food, Agriculture and Fisheries, Holbergsgade 2, 1057 Copenhagen K, Denmark

Phone: +45 33 92 35 60 - Fax: +45 33 11 82 71 - Internet: sfe@fvm.dk

B. Prince, Ministere de l'Agriculture et de la Peche, Direction des Peches Maritimes, 3 Place de Fontenoy, 75007 Paris, France

Phone: +45 33 92 35 60 - Fax: +45 33 11 82 71 - E-mail: berengere.prince@gouv.fr

I. Ybanez, Subdirector General de Organismos Multilaterales de Pesca, Secretaria General de Pesca Maritima, Jose Ortega y Gasset, 57, 28006 Madrid, Spain

Phone: +34 91 402 74 04 - Fax: +34 91 309 3967 - E-mail: iybanezr@mapya.es

FRANCE (IN RESPECT OF ST. PIERRE ET MIQUELON)

Head of Delegation

S. Ségura, Conseiller des Affaires Etrangères, Direction des Affaires Juridiques, Ministère des Affaires Etrangères, 37 Quai d'Orsay, 75700 Paris, France Phone: +01 43 17 53 28 - Fax: +01 43 17 43 59 - E-mail: <u>serge.segura@diplomatie.fr</u>

ICELAND

Head of Delegation

K. Arnason, Head of Division, Ministry of Fisheries, Skulagata 4, 150 Reykjavik Phone: +354 560 9670 - Fax: +354 562 1853 - E-mail: <u>kolbeinn.arnason@sjr.stjr.is</u>

JAPAN

Head of Delegation

Y. Arima, Assistant Director, Legal Affairs Div., Treaties Bureau, Ministry of Foreign Affairs, 2-2-1 Kasumigaseki, Chiyoda-ku, Tokyo 100 Phone: +81 33581 2047 - Fax: +81 33580 5074 - E-mail: <u>vutaka.arima@mofa.go.jp</u>

LATVIA

Head of Delegation

N. Riekstins, Director, National Board of Fisheries, Ministry of Agriculture, 2, Republikas laukums, LV-1010 Riga Phone: +371 732 3877 - Fax: +371 733 4892 - E-mail: <u>fish@com.latnet.lv</u>

Alternate

R. Derkacs, Head of International Agreements and Legal Div., National Board of Fisheries, 2, Republikas laukums, LV-1010 Riga Phone: +371 732 3877 - Fax: +371 733 4892 - E-mail: fish@com.latnet.lv

LITHUANIA

Head of Delegation

A. Raudonius, Vice-Minister of Agriculture, Ministry of Agriculture, 19 Gedimino pr., 2600 Vilnius Phone: +3702 391306 - Fax: +3702 391308

Alternate

A. Rusakevicius, Chief Specialist-International Relations, Fisheries Dept., Ministry of Agriculture, 19 Gedimino pr., 2600 Vilnius Phone: +3702 391183 - Fax: +3702 391176 - E-mail: <u>algirdasr@zum.lt</u>

Adviser

G. Babcionis, Senior State Inspector, International Relations Div. of Fish Resources Dept. of the Ministry of Environment, A. Juozapavichiaus St. 9, Vilnius 2600 Phone: +3702 723631 - Fax: +3702 723273 - E-mail: <u>Genadijus@nt.gamta.lt</u>

NORWAY

Head of Delegation

T. Lobach, Directorate of Fisheries, Strandgaten 229, P. O. Box 185, N-5802 Bergen Phone: +47 55 23 80 00 - Fax: +47 55 23 80 90 - E-mail: terje.lobach@fiskeridir.dep.telemax.no

Alternate

J. P. Groenhof, Executive Officer, Royal Ministry of Foreign Affairs, P. O. Box 8114 Dep., N-0032 Oslo Phone: +47 22 24 36 00 - Fax: +47 22 24 95 80

Adviser

S. Owe, Director General, Ministry of Fisheries, P. O. Box 8118 Dep., N-0032 Oslo Phone: +47 22 24 64 71 - Fax: +47 22 24 26 67 - E-mail: <u>stein.owe@fid.dep.telemax.no</u>

UNITED STATES OF AMERICA

Head of Delegation

G. S. Martin, Attorney, Office of the General Counsel, Northeast Region, National Oceanic and Atmospheric Administration, U.S. Dept. of Commerce, 1 Blackburn Dr., Gloucester, MA 01930 Phone: +1 978 281 9242 - Fax: +1 978 281 9389 - E-mail: <u>Gene.S.Martin@noaa.gov</u>

NAFO SECRETARIAT

L. I. Chepel, Executive Secretary B. Cruikshank, Senior Secretary

Annex 2. Agenda

- 1. Opening by Chairman, S. Owe (Norway)
- 2. Appointment of Rapporteur
- 3. Adoption of Agenda
- 4. Examination of the desirability and, as appropriate, the development of procedures for the settlement of disputes between NAFO Contracting Parties:
 - a) by implementing in a NAFO context the 1995 UN Agreement and UNCLOS dispute settlement procedures, and
 - b) by including additional measures if needed.
- 5. Report to the General Council
- 6. Other Matters
- 7. Adjournment

Annex 3. NAFO Dispute Settlement Procedures - A Possible Way Forward

(Paper presented by the Chairman - DSP W.G. W.P. 00/1)

In our deliberations in the DSP Working Group we seem to have run into considerable difficulties in reaching consensus and even in moving ahead with our work. I have tried to think of ways to break this situation and hopefully conclude our endeavours in the near future. In this paper I would like to present some suggestions in this regard to the Working Group.

The main idea would be to simplify our scheme, even if this means settling for something that would not be regarded as ideal by all (and may be not by any) of the Contracting Parties. As I see it, the simplification is probably necessary to avoid some complicated issues that as such generate disagreement and to arrive at a speedy finalisation of our work.

A certain simplification also seems to be supported by the changes that were made to our terms of reference at the last annual meeting. The relevant parts now read:

"Examine the desirability and, as appropriate, the development of procedures for the settlement of disputes between NAFO Contracting Parties

- by implementing in a NAFO context the 1995 UN Agreement and UNCLOS dispute settlement procedures, and
- by including additional measures if needed;"

My understanding of the changes that were made is that we are to focus to a lager degree on the thoroughly developed system for dispute settlement which is found in the UN Agreement and UNCLOS.

We have in our discussions identified four main reasons for establishing separate NAFO DSP:

- 1. The UN Agreement, and thus its provisions on dispute settlement, has not yet entered into force.
- 2. Even when it enters into force it is not certain that all NAFO Contracting Parties will become parties to the agreement and thus bound by its provisions. The same applies to UNCLOS.
- 3. A NAFO DSP gives the possibility to include disputes regarding discrete stocks, which is also found in the NAFO Regulatory Area.
- 4. Finally, it makes it possible to create more expeditious procedures.

Many Contracting Parties have regarded this last element as rather important. Much effort has gone into designing rules that could lead to settlement of disputes as quickly as possible within the given fishing season. However, this is an area where we have encountered considerable problems as well. The dispute settlement process should be compulsory and lead to binding

results. It does not seem feasible to have such dispute settlement by a specially designed body such as the ad hoc panel we have discussed. This being the case, we have entered into rather complex deliberations on the possible role and use of such a panel, including the option of keeping it out of a concrete dispute all together. It has been argued that since it is not likely that both parties to a dispute will accept the recommendation of the panel, this procedure is actually just delaying the final settlement of the dispute that will have to be brought before the bodies in the UN Agreement/UNCLOS system anyway.

It would greatly simplify our work if we leave the idea of establishing a specific body like the ad hoc panel in a NAFO scheme. The procedure itself will be much more straightforward and many of the questions we have dwelt on in our last meetings will loose their relevance. Furthermore, we would not have to deal with the issues contained in the annex on establishment of an ad hoc panel or the rules of procedure for panel proceedings.

It might be said that we would loose the opportunity of having the disputes settled expeditiously and within the fishing season. However, as has been mentioned it does not seem likely that the parties to a dispute both (or all) will accept the recommendation of the panel. One or more of the parties will probably exercise the right to take the dispute to binding settlement. At least in cases that are regarded as important this presumption seems reasonable. The reality might thus be that such disputes would not in any case be solved expeditiously. In less important cases the attitude might be different. But if the parties so wish, they would still have at their disposal the general possibility to try non-binding solution of the dispute through any peaceful means of their own choice.

Another aspect of leaving the idea of a specific NAFO body would be that there would be no recommendation by such a body that can be used as an immediate provisional measure if the dispute is pursued. However, as has been pointed out both the UN Agreement and UNCLOS contain provisions on provisional measures. It may take some time to have these measures established. Nevertheless, provisional measures may take effect earlier in the handling of a dispute than if an ad hoc panel process is the first stage. This is due to difficulties in finding anything that could be suitable and acceptable as provisional measures during a panel process. In general it does not seem possible to avoid any risk of a dispute leading to some damage to NAFO stocks.

We should be able to establish rules that comprise the other main reasons for having separate NAFO DSP. The goal of having DSP that is binding on all NAFO Contracting Parties whether or not they are also parties to the UN Agreement or UNCLOS seems to favour aiming for incorporation of these new rules in the NAFO Convention. As part of the amendments we should include an obligation to give reasons for an objection or notice of intention not to be bound by a management measure in force, as well as information on the relevant Commission member's intentions following the objection or such a notice. This information should include a description of the conservation and management measures that are planned or already taken. It seems to be agreement that such a provision would be important, not at least to assess whether there is reason to initiate DSP.

As a point of departure it seems natural that NAFO DSP cover all possible disputes within the organisation. Thus disputes concerning discrete stocks will be included. (What rules shall be applicable to such disputes is a different question.)

Annex 4. Working Paper Presented by Denmark (in respect of Faroe Islands and Greenland) (DSP W.G. W.P. 00/3)

In the attached Scheme we try to illustrate how possible goals could be achieved through different dispute settlement measures.

Other Means	yes	Maybe	Maybe	Maybe	Maybe
FULL NAFO DSP	, yes	yes	yes	yes	yes
NAFO DSP only referring to UNCLOS & UN Agreement DSP	yes	S	0		
NAI only to U UN DSF	У¢	yes	ou	оп	yes
UN Agreement DSP	Only when ratified	оц	ou		yes
UN AB DSP	Only w			•	
	ıtified				
DSP	Only when ratified	yes	ou	ou	yes
		s and	ettled g season)	ile who NAFO	r nally)
	All NAFO Parties may participate	Straddling stocks and discrete stocks	Speed (dispute settled vithin the fishing season)	Decision by people who are familiar with NAFO	Decision binding (at least provisionally)
	All N may	Strad	Spee	Deci: are f	Dcci: (at le

Annex 5. Working Paper Presented by the European Union (DSP W.G. W.P. 99/4)

[obligation to cooperate]

1. Contracting Parties shall cooperate in order to prevent disputes.

In particular, any Contracting Party may invite a Commission Member that has objected to a proposal of the Commission or has given notice of its intention not to be bound by a measure of the Commission to state the reasons for its objection or its notice of intention, as well as to describe the conservation and management measures it has taken or intends to take for the fishery resource in question.

[Ist sentence from Chairman's paper; voluntary declaration of intent added]

[binding dispute settlement procedure]

2. Without prejudice to para. 3 a Contracting Party may refer any dispute concerning the interpretation or application of the Convention to DSP.

Such procedures shall be governed <u>mutatis mutandis</u> by the provisions relating to the settlement of disputes set out in Part XV of UNCLOS or, where the dispute concerns one or more straddling stocks, by the provisions set out in Part VIII of the UN Agreement.

The relevant parts of UNCLOS and the UN Agreement shall apply whether or not the Parties to the dispute are also State Parties to these instruments.

[rephrase of No. 2 of Chairman's paper to make text simpler.]

[ad hoc panel procedure]

3. Where the dispute concerns the interpretation or application of a proposal adopted by the Commission pursuant to Article XI or matters related thereto Parties to the dispute shall within x days after the notification of the dispute to the Executive Secretary proceed to an exchange of views regarding its settlement through an *ad hoc* panel procedure. When the Parties do not agree to such a procedure or to any other peaceful means to resolve the dispute, the dispute shall be referred, if one of the Parties concerned so requests, to a binding DSP as provided in para. 2.

Where a dispute has been submitted to the *ad hoc* panel procedure, the panel constituted as provided in Annex ... to this Convention shall at the earliest possible opportunity confer with the Parties concerned and shall endeavour to resolve the dispute expeditiously. Within x weeks after being constituted the panel shall present a report to the Parties concerned. The report shall as far as possible include any recommendations which the panel considers appropriate to resolve the dispute.

Where a dispute has not been resolved through agreement between the Parties following an *ad hoc* panel procedure it shall be referred, if one of the Parties concerned so requests, to a binding DSP as provided in para. 2.

[text of yesterday's paper slightly modified to take into account comments from delegations]

[provisional application during and after ad hoc panel procedure]

4. Where the Parties to a dispute have agreed to submit the dispute to the *ad hoc* panel procedure, they may agree at the same time to apply provisionally the relevant proposal adopted by the Commission until the report of the panel or the dispute is resolved, whichever occurs first.

Pending the settlement of disputes according to para. 2 the Parties to the dispute shall apply provisionally any recommendation made by a panel where the Parties had agreed an *ad hoc* panel procedure. That provisional application shall cease when the Parties agree on arrangements of equivalent effect, when a court or tribunal to which the dispute has been submitted in accordance with para 2 has taken a provisional or definitive decision or, in any case, at the end of the calendar year in which the report of the panel has been presented.

[text of the Chairman's paper adapted to the new subpara. 3]

[law to be applied by court, tribunal or panel]

A court, tribunal or panel to which any dispute has been submitted under this Article shall apply the relevant provisions of this Convention, of the 1982 UN Convention, of the 1995 UN Agreement, as well as generally accepted standards for the conservation and management of living marine resources and other rules of international law not incompatible with the 1982 UN Convention and the 1995 UN Agreement, with a view to ensuring the conservation of the fish stocks concerned.

[same text as the Chairman's paper]

5.

Annex 6. Working Paper Presented by the European Union (DSP W.G. W.P. 00/2)



Annex 7. Working Paper Presented by Canada (DSP W.G. W.P. 00/4)

An objection according to paragraph 1 and a notice of objection not to be bound by a measure according to paragraph 3 shall be accompanied by a declaration setting out the autonomous conservation and management measures (including control and enforcement measures) to be established and the rationale for the objection and the autonomous measure. The declaration and post-objection behaviour may be challenged through dispute settlement procedures.

Annex 8. Working Paper Presented by Japan

(DSP W.G. W.P. 00/8)

- 1. Contracting Parties should cooperate in order to prevent disputes.
- 2. Where the dispute concerns the interpretation or application of a proposal adopted by the Fisheries Commission pursuant to Article XI or matters related thereto, any Party may request the other Parties concerned in the dispute to have consultations to resolve the dispute with a written notice. Parties to the dispute should, with X days from the receipt of the notice, proceed to an exchange of views with a view to resolving the dispute as soon as possible.
- 3. If the dispute is not resolved within X weeks after the written notice mentioned in paragraph 2 is given, any Party to the dispute may request the dispute to be submitted to an *ad hoc* panel of experts. The *ad hoc* panel is established by the General Council in accordance with Article IV paragraph 6, and constituted in accordance with the Annex to this proposal.
- 4. The Panel should at the earliest possible opportunity confer with the Parties concerned and should endeavour to resolve the dispute expeditiously. The Panel should issue recommendations for resolving the dispute as necessary. The Parties should cooperate with the members of the Panel and should endeavour to resolve the dispute as faithfully as possible.
- 5. If the dispute is not resolved with the involvement of the Panel after X weeks from the request referred to in paragraph 3, any Party to the dispute may request the Panel to submit the recommendation to the Fisheries Commission. The Fisheries Commission may consider such recommendations as proposals prescribed in Article XI paragraph 2 and adopt them. Article XII applies to the adoption of such recommendations by the Fisheries Commission.

<u>Annex</u>

(to proposal by Japan)

- 1. The Executive Secretary will prepare a list of experts on fisheries matters whose competence in scientific or technical aspects of fisheries matters is established and generally recognized and who enjoy the highest reputation for fairness and integrity. Each Party may nominate, at any time, two experts for this list and the persons so nominated will constitute this list of experts.
- 2. The *ad hoc* panel of experts will be constituted from three experts which should be chosen from the list of experts prepared by the Executive Secretary referred to in paragraph 1.
- 3. Each Party to the dispute should choose one expert to be a member of the panel. The third member should be appointed jointly by the Parties to the dispute. If the Parties to the dispute cannot agree on the third expert, any Party to the dispute may request the Chairman of the General Council to make the appointment from the list of experts referred to in paragraph 1.

Annex 9. Working Paper Presented by Canada (DSP W.G. W.P. 00/6)

2. A Contracting Party may refer any dispute concerning the interpretation or application of the Convention to DSP.

The Contracting Parties agree to apply the 1995 UN Agreement provisionally both to straddling stocks and discrete stocks that occur in the NAFO Regulatory Area, whether or not the Contracting Parties are party to the Agreement.

Annex 10. Working Paper Presented by Canada (DSP W.G. W.P. 00/7)

3. Where the dispute concerns the interpretation or application of a proposal adopted by the Commission pursuant to Article XI or matters related thereto, a party to the dispute may invite the other party to submit the dispute to a panel. The panel shall confer with the States concerned and shall endeavour to resolve the dispute expeditiously without recourse to binding procedures for the settlement of disputes.

Annex 11. Settlement of Disputes within NAFO

COMPILATION OF PROPOSALS (DSP W.G. W.P. 00/9)

New Paragraph 4 of Article XII (If NAFO Dispute Settlement Procedures are not incorporated as amendments to the NAFO Convention this provision may possibly be adopted in another form.)

An objection according to paragraph 1 and a notice of intention not to be bound by a measure according to paragraph 3, shall be accompanied by a statement of the relevant Commission member's reasons for the objection or notice of intention as well as a declaration of its intentions following the objection or such notice, including a description of any conservation and management measures[, including control and enforcement measures,] it has taken or intends to take. [The declaration and post-objection behaviour may be challenged through dispute settlement procedures.]

(New) Article...

1. Contracting Parties shall cooperate in order to prevent disputes.

2. If any dispute arises between two or more Contracting Parties concerning the interpretation or application of this Convention, those Contracting Parties shall consult among themselves with a view to resolving the dispute, or to having the dispute resolved by negotiation, inquiry, mediation, conciliation, *ad hoc* panel procedures, arbitration, judicial settlement or other peaceful means of their own choice.

[3. A Contracting Party may refer any dispute concerning the interpretation or application of the Convention to DSP.

The Contracting Parties agree to apply the 1995 UN Agreement provisionally both to straddling stocks and discrete stocks that occur in the NAFO Regulatory Area, whether or not the Contracting Parties are party to the Agreement.]

[3. Where a dispute concerns the interpretation or application of a proposal adopted by the Commission pursuant to Article XI or matters related thereto parties to the dispute shall within x days after the notification of the dispute to the Executive Secretary proceed to an exchange of views regarding its settlement through *ad hoc* panel procedures.

Where a dispute has been submitted to *ad hoc* panel procedures, the panel constituted in accordance with provisions adopted by the General Council shall at the earliest possible opportunity confer with the Contracting Parties concerned and shall endeavour to resolve the dispute expeditiously. Within x weeks after being constituted the panel shall present a report to the Contracting Parties concerned and through the Executive Secretary to the other Contracting Parties. The report shall as far as possible include any recommendations which the panel considers appropriate to resolve the dispute.

Where a dispute has not been resolved through agreement between the Contracting Parties following the recommendations of the *ad hoc* panel it may be referred, on request of one of the Contracting Parties, to a binding DSP as provided in para. 5.]

[4. Where a dispute concerns the interpretation or application of a proposal adopted by the Commission pursuant to Article XI or matters related thereto, a party to the dispute may invite the

other party to submit the dispute to a panel. The panel shall confer with the States concerned and shall endeavour to resolve the dispute expeditiously without recourse to binding procedures for the settlement of disputes.]

4(or 5). Where the parties to a dispute have agreed to submit the dispute to *ad hoc* panel procedures, they [may agree at the same time to] [shall] apply provisionally the relevant proposal adopted by the Commission until the report of the panel is presented or the dispute is resolved, whichever occurs first.

[Pending the settlement of a dispute according to para. 5 the parties to the dispute shall, if one of these Contracting Parties so desire, apply provisionally any recommendation made by a panel where the Contracting Parties had agreed an *ad hoc* panel procedure.] [The parties to a dispute may agree to apply provisionally any recommendation made by a panel pending the settlement of the dispute according to para 5.] That provisional application shall cease when the Contracting Parties agree on arrangements of equivalent effect, when a court or tribunal to which the dispute has been submitted in accordance with para 5 has taken a provisional or definitive decision or, in any case, at the date of expiration, if applicable, of the propsal of the Fisheries Commission.

[5. If the Contracting Parties do not agree to any other peaceful means to resolve a dispute, or no settlement has been reached by recourse to these means, the dispute shall be referred, if one of the Contracting Parties concerned so requests, to binding dispute settlement procedures. Such procedures concerning the interpretation and application of this Convention shall be governed <u>mutatis mutandis</u> by the provisions relating to the settlement of disputes set out in Part XV of United Nations Convention on the Law of the Sea of 10 December 1982 (1982 UN Convention) or[, where the dispute concerns one or more straddling stocks,] by the provisions set out in Part VIII of the United Nations Agreement for the Implementation of the Provisions of the United Nations Convention on the Law of the Sea of 10 December 1982 relating to the Conservation and Management of Straddling Fish Stocks and Highly Migratory Fish Stocks of 4 August 1995 (1995 UN Agreement)[, whether or not the parties to the dispute are also State parties to these instuments].]

[6. A court, tribunal or panel to which any dispute has been submitted under this Article shall apply the relevant provisions of this Convention, of the instruments referred to in para. 2, as well as generally accepted standards for the conservation and management of living marine resources and other rules of international law not incompatible with the said instruments, with a view to ensuring the conservation [and optimum utilization] of the fish stocks concerned.]

Annex 12. Settlement of Disputes within NAFO

CONSOLIDATED TEXT (DSP W.G. W.P. 00/10-Revised)

New Paragraph 4 of Article XII (If NAFO Dispute Settlement Procedures are not incorporated as amendments to the NAFO Convention this provision may possibly be adopted in another form.)

On request of any Contracting Party, a Member of the Fisheries Commission, which has presented an objection to a proposal in accordance with Article XII (1) or given notice of its intention not to be bound by a measure in accordance with Article XII (3), shall within [...] days give a statement of the reasons for its objection or notice and a declaration of its intentions following the objection or notice, including a description of any measures it intends to take or has already taken for the conservation and management [, including control and enforcement measures,] of the fish stock or stocks concerned. [The declaration and post-objection behaviour may be challenged through dispute settlement procedures.]

(New) Article...

1. Contracting Parties shall cooperate in order to prevent disputes.

2. If any dispute arises between two or more Contracting Parties concerning the interpretation or application of this Convention, those Contracting Parties shall consult among themselves with a view to resolving the dispute, or to having the dispute resolved by negotiation, inquiry, mediation, conciliation, *ad hoc* panel procedures, arbitration, judicial settlement or other peaceful means of their own choice.

3. Where a dispute concerns the interpretation or application of a proposal adopted by the Fisheries Commission pursuant to Article XI or matters related thereto, the parties to the dispute may submit the dispute to an *ad hoc* panel constituted in accordance with procedures adopted by the General Council. The Contracting Parties that so agree shall within [...] days of the notification of the dispute to the Executive Secretary proceed to an exchange of views concerning the constitution of the panel and the resolution of the dispute through the panel.

Where a dispute has been submitted to *ad hoc* panel procedures, the panel constituted in accordance with provisions adopted by the General Council shall at the earliest possible opportunity confer with the Contracting Parties concerned and shall endeavour to resolve the dispute expeditiously. Within x weeks after being constituted the panel shall present a report to the Contracting Parties concerned and through the Executive Secretary to the other Contracting Parties. The report shall as far as possible include any recommendations which the panel considers appropriate to resolve the dispute.

Where a dispute has not been resolved through agreement between the Contracting Parties following the recommendations of the *ad hoc* panel it may be referred, on request of one of the Contracting Parties, to a binding DSP as provided in para. 5.

4. Where the parties to a dispute have agreed to submit the dispute to *ad hoc* panel procedures, they may agree at the same time to apply provisionally the relevant proposal adopted by the Commission until the report of the panel is presented or the dispute is resolved, whichever occurs first.

[Pending the settlement of a dispute according to para. 5 the parties to the dispute shall, if one of these Contracting Parties so desire, apply provisionally any recommendation made by a panel where the Contracting Parties had agreed an *ad hoc* panel procedure.] *or* [The parties to a dispute may agree to apply provisionally any recommendation made by a panel pending the settlement of the dispute according to para 5.] That provisional application shall cease when the Contracting Parties agree on arrangements of equivalent effect, when a court or tribunal to which the dispute has been submitted in accordance with para 5 has taken a provisional or definitive decision or, in any case, at the date of expiration, if applicable, of the proposal of the Fisheries Commission.

[5. If the Contracting Parties do not agree to any other peaceful means to resolve a dispute, or no settlement has been reached by recourse to these means, the dispute shall be referred, if one of the Contracting Parties concerned so requests, to binding dispute settlement procedures. Such procedures concerning the interpretation and application of this Convention shall be governed <u>mutatis mutandis</u> by the provisions relating to the settlement of disputes set out in Part XV of United Nations Convention on the Law of the Sea of 10 December 1982 (1982 UN Convention) or[, where the dispute concerns one or more straddling stocks,] by the provisions set out in Part VIII of the United Nations Agreement for the Implementation of the Provisions of the United Nations Convention on the Law of the Sea of 10 December 1982 relating to the Conservation and Management of Straddling Fish Stocks and Highly Migratory Fish Stocks of 4 August 1995 (1995 UN Agreement)[, whether or not the parties to the dispute are also State parties to these instuments].]

[6. A court, tribunal or panel to which any dispute has been submitted under this Article shall apply the relevant provisions of this Convention, of the instruments referred to in para. 5, as well as generally accepted standards for the conservation and management of living marine resources and other rules of international law not incompatible with the said instruments, with a view to ensuring the conservation [and optimum utilization] of the fish stocks concerned.]

OR (instead of 5 and 6)

[A Contracting Party may refer any dispute concerning the interpretation or application of the Convention to DSP.

The Contracting Parties agree to apply the 1995 UN Agreement provisionally both to straddling stocks and discrete stocks that occur in the NAFO Regulatory Area, whether or not the Contracting Parties are party to the Agreement.]

. v · · · ·

.

SECTION V

(pages 131 to 192)

Report of the Standing Committee on International Control (STACTIC) 27-29 June 2000 Dartmouth, N.S., Canada

Report of the N	Meeting		
1.	Opening of the Meeting		
2.	Appointment of Rapporteur		
3.	Adoption of the Agenda		
4.	Program for Observers and Satellite Tracking		
5.	Possible Amendments to Conservation and Enforcement		
	Measures Regarding Juvenile Fish		
6.	Other Matters		
7.	Adoption of the Report142		
8.	Adjournment		
	Annex 1. List of Participants		
	Annex 2. Agenda147		
	Annex 3. Working Paper by Denmark (in respect of the Faroe Islands and Greenland)		
	Annex 4. Proposal (by Norway) to amend the NAFO Conservation and Enforcement Measures, Part VI.A.1(a) regarding		
	independent and impartial observers		
	Annex 6. Additional Information (by Canada) - Depth Proposal for Greenland halibut		
	Annex 7. Working Paper by the European Union154		
	Annex 8. Working Paper by Japan155		
	Annex 9. Statement from the Representative of Canada156		
	Annex 10. Shrimp 3M Fishery Statistics, 1993-1995158		
	Annex 11. Submission on shrimp catches and effort days (W.P. by Denmark (in respect of the Faroe Islands and Greenland)		
	Annex 12. Compilation of Shrimp 3M Catches and Effort Days for 1993-1999 (NAFO Secretariat)		
	Annex 13. Statement from the Representative of Norway		
	Annex 15. Statement from the Representative of Norway		
	and Enforcement Measures regarding Part VII-Port		
	Inspections179		

. •

Report of the Standing Committee on International Control

(FC Doc. 00/4)

27-29 June 2000 Dartmouth, N.S., Canada

At the 1999 Annual Meeting of the Fisheries Commission, STACTIC's recommendation was accepted that an inter-sessional meeting of the Committee should take place to begin work on the scientific requirements for the observer program, the existing program and the observer manual. Furthermore, an examination was required to ensure that observers are independent and impartial.

The Fisheries Commission also requested STACTIC to review management options to reduce catches of juvenile fish with a view to incorporating measures into the NAFO Conservation and Enforcement Measures.

Contracting Parties also considered it useful to begin discussions on a number of other issues, in particular on the follow up to the March joint working group on the Precautionary Approach, and on the issues of charters and "flag hopping". Furthermore, the meeting on shrimp stocks held in Washington D.C. in March 2000 requested that STACTIC examine possible new information on shrimp fishing activity in the NAFO Regulatory Area, in order that newly updated data could be provided to the Fisheries Commission before the 2000 Annual Meeting. Other items for discussion are covered in the report below.

1. Opening of the Meeting

The Chairman, Mr. David Bevan (Canada), opened the meeting at 10.10 on 27 June 2000. Representatives from the following Contracting Parties were present: Canada, Cuba, Denmark (in respect of Faroe Islands and Greenland), Estonia, the European Union, Iceland, Japan, Norway, Russian Federation and the United States. A list of participants is given at Annex 1.

2. Appointment of Rapporteur

Mr. Andrew Thomson (European Union) was appointed rapporteur.

3. Adoption of the Agenda

Following some protracted discussion between the Contracting Parties, it was agreed to adopt the agenda as amended (Annex 2).

The representative from the European Union initially felt that it would be relevant to discuss all issues concerning the Program for Observers and Satellite Tracking under the same agenda item. However, it was pointed out that at its meeting in September 1999, the Fisheries Commission had not given STACTIC a mandate to discuss the review and possible revision of the Program. The three sub-points under point 4 had in fact been carried over from the September 1999 STACTIC meeting. It was therefore agreed that the heading of this item should be amended so that the discussion under point 4 could reflect the full contents of the said Program. However, discussion under point 6 e) would remain separate.

4. Program for Observers and Satellite Tracking

a. Scientific requirements

The representative of Denmark (in respect of Faroe Islands and Greenland) introduced their suggestion for an amendment to the existing Program (Annex 3). From their experience and from research carried out, it appeared that the actual amounts of by-catch and discards were much higher than the estimates, which were usually made on a visual basis. He suggested that it would be necessary and compulsory to collect by-catches in boxes or containers (say 20kg capacity) in order to allow for a proper assessment of the quantities involved. He particularly noted the potential dangers in respect of a possible quota of shrimp in area 3M.

Support for the suggestion by Denmark (in respect of Faroe Islands and Greenland) came from the representative of the United States, as he felt it would help to alleviate ambiguities and improve the stock assessment. The representative of Japan also supported the proposal, as did the representative of the Russian Federation, although the Canadian representative supported the proposal in principal but felt that further review of the practical implications is required. The representative of Iceland went along with this approach.

The representative of the European Union was not convinced by the Danish paper of the actual value of the suggestion. He felt that it was necessary to have further detailed examination of the underlying problem and the implications of the proposed measures, given that they would involve changes to the processing lines onboard the ships. The representatives of both Canada and Iceland understood this latter concern.

The Chairman asked delegations to gather the needed information on the potential impacts of the Danish suggestion to facilitate a return to this issue at the Annual Meeting in September 2000 and examine possible improvements to data gathering. The representative of Canada suggested that Denmark (in respect of Faroe Islands and Greenland) return at the time of the Annual Meeting with a firm proposal for amendment to the Conservation and Enforcement Measures.

Dave Kulka (Canada) made a presentation of a Scientific Council proposal for a harmonised NAFO Observer Data System (NAFO SCS Doc. 00/23). An ad hoc working group of NAFO Scientists had worked inter-sessionally and prepared a series of four draft collection forms and associated documentation designed to capture the basic information required for assessing removals from stocks in the Regulatory Area and presented to STACTIC in September 1999. STACTIC in turn requested that the Scientific Council produce a data description for these forms.

The Scientific Council Observer Working Group reviewed the progress of this work in June 2000. At this time, two separate initiatives were reported, namely a Canadian initiative for a database, which has been capturing observer data since 1998, and a European Union form set, which was a catch-tracking system designed by the European Union NAFO inspectors. There was a high degree of overlap in the European Union system with the one formulated by the Scientific Council working group. However, there were also additional elements in the European Union system not required by NAFO. In essence, the only item not in the European Union system was the length frequency catch data retrieval.

The representative of the European Union noted that observer coverage in its current version made it impossible to place scientific observers on board vessels. Furthermore, he noted that it was necessary to distinguish the idea of using the information already gathered by the control observers for scientific purposes from the idea of requiring observers to carry out additional scientific work. The latter should be done without putting undue additional burdens on the observers. Furthermore, the future of the whole Program was still in question. He also stressed that it was necessary to highlight those tasks of the observers, which could be of specific use to the scientists.

The representative from Denmark (in respect of Faroe Islands and Greenland) was also concerned at giving observers too many tasks. He noted that in Greenland, it would be necessary to have two observers on board to carry out the duties adequately.

The Canadian representative, supported by Mr. Kulka, also noted that in Canada, observers had been carrying out scientific tasks along with control functions since the late 1970s. Furthermore, with 100% observer coverage, control observers would only be required to take two or three samples per week occupying six to nine hours of their time. This could easily be achieved with adequate efficiency. The Japanese representative was able to support this proposal.

In view of the overall discussion, the Parties agreed that it was the element of length-frequency catch data retrieval, which should be considered as the only additional scientific element for the observers. Evaluation of this point should also take place in full co-ordination with the general evaluation requested of the Contracting Parties under item 4 (c) below.

b. Amendments to existing Program

The representative of Norway introduced a proposal to amend Part VI.A.1 (a) of the Conservation and Enforcement Measures with regard to independent and impartial observers (Annex 4). He explained that his proposal was to ensure that anyone working as an observer had that sole responsibility. The Russian representative was able to concur with this approach. The representative of Japan queried whether an observer could work for the company owning the fishing vessel.

The feeling of the representative of the European Union was that the Norwegian approach was incomplete. He questioned whether there really was a problem. If so, what was it? He also pointed out that it might be necessary to clarify what was independent and impartial, as well as to define what was a crewmember.

The Parties recognised that there was a need to ensure that observers were able to perform the duties, which had been established for them, in an independent and impartial manner. After considerable further deliberation, the Parties agreed that a new amendment proposed by the Chairman could replace that proposed by Norway and would be inserted at the end of point A.1 (a) of the existing Program for Observers and Satellite Tracking. The amendment would read as follows:

"Observers are not to perform duties, other than those described in Sections 3, 4 and 5 below."

It was agreed that it would be helpful if Contracting Parties could demonstrate at the Annual Meeting how they themselves ensure impartiality and independence for their own observers. The representative of Denmark (in respect of Faroe Islands and Greenland) pointed out that this exercise had already been carried out in 1998 (Ref. to STACTIC Working Paper 98/12). It was agreed, therefore, that all Contracting Parties would provide the next Annual Meeting with updated information on this matter.

c. Observer Manual

The representative of Canada reminded Parties that at the September 1999 STACTIC meeting, it was agreed that there was a need to develop a consistent approach with regard to the duties of observers in NAFO. In order to help expand the discussion in STACTIC, they provided the heads of each delegation with a copy of the existing manual used by Canadian observers in the NAFO Regulatory Area. It was felt that this could provide a useful guideline for the eventual development of a NAFO-specific observer manual. The Canadian manual, whilst in need of updating, was developed in 1996 as a reference for observers and not as a training tool and covers all the duties required of an observer. Using the basis of an existing manual was thought to be easier than starting from scratch.

It was pointed out by the representative of Denmark (in respect of Faroe Islands and Greenland) that whilst the Canadian manual was comprehensive, we were seeking a checklist which allowed our observers to operate appropriately.

It was noted that this was a good but ambitious document consisting of three parts, namely training, tasks for observers and working methodology. The representatives of the European Union suggested that discussion should focus on the latter. In line with that, he presented a "NAFO Observer Manual" as proposed by the EU (STACTIC Working Paper 00/10) suggesting a working methodology, which would ensure enhanced transparency. The other aspects covered in the Canadian document were not felt to be relevant in this context. The paper consisted of two parts. Part I covered the tasks to be performed by the observers, Part II of the proposed NAFO Observer Report Form. The United States representative noted that Part I would be very useful, whilst there were similarities of Part II to document SCS 00/23 from the Scientific Council.

The Parties took full account of the paper presented from the Scientific Council meeting of June 2000 (NAFO SCS Doc. 00/23 as referred to under item 4(a) above). They noted that the information contained in the EU proposal encompassed the information set out in the Scientific Council document. The representative of the European Union explained that the codes used in the European Union paper were the standard ISO and FAO international codes, with the primary methodology taken from the North Atlantic format. This enabled the Contracting Parties to avoid being locked into a single system. The representative of the United States was able to endorse document SCS 00/23 meeting the scientific requirements of the observer manual. The representative of Japan supported the use of document SCS 00/23 as an observer manual.

However after some protracted discussion, it was concluded that Contracting Parties should examine and evaluate both the paper from the European Union and document SCS 00/23 prior to the Annual Meeting. This would enable a finalised discussion to take place at the Annual Meeting.

5. Possible Amendments to Conservation and Enforcement Measures Regarding Juvenile Fish

The representative of Canada introduced two proposals to amend the existing Conservation and Enforcement Measures in respect of juvenile fish (Annex 5). He also referred to an information note (Annex 6) which went into further detail on the issue of Greenland halibut. The Chairman noted that no other delegation had a proposal at this stage. In particular the Canadian representative noted that at the Fisheries Commission meeting of September 1999, STACTIC had been directed as follows:

"In light of the advice of the Scientific Council, STACTIC shall review all management options by which catches of juvenile fish can be reduced taking into account the various NAFO fisheries and elaborate and recommend feasible measures to be incorporated in the NAFO Conservation and Enforcement Measures."

The measures proposed by Canada were:

- 1. Increase in the mesh size from 130mm to 145mm for all principal groundfish in the Regulatory Area (with redfish and capelin being excluded).
- 2. Restriction on the directed fishing for Greenland halibut in Divisions 3LNO to be prohibited at depths of less than 400 metres. The 400-metre contour would be delineated by a number of fixed co-ordinates to be determined.

The Canadian representative explained that the measures currently in operation in the Regulatory Area were inadequate for the protection of the juvenile fish. This was hindering the rebuilding of the groundfish stocks. The Canadian mesh size was already 145mm and sometimes 155mm irrespective of the fishing grounds.

With respect to the Greenland halibut, adequate protection must be given to the juveniles. With a depth restriction of 400 metres, great benefit could be accorded to the stock. It was suggested that the 400-metre depth was only an example and perhaps the restriction may need to be at a lower depth. In particular, it was noted that the current Greenland halibut fishery is a juvenile-based fishery. With a depth restriction, far less of the juvenile part of the stock would be targeted since the juveniles do not swim at the greater depths.

The representative of the European Union questioned the reasoning behind the retention of the mesh size for redfish and for restricting the proposed depth restriction measure to Divisions 3LNO.

The Canadian representative explained that while the depth restriction was aimed at protecting juvenile Greenland halibut, reductions in by-catch of other groundfish, including yellowtail flounder and American plaice could also be realised. This, he believed, was an added benefit to such a depth restriction. For redfish, it was not felt appropriate to increase the mesh size; some have even expressed the view in the past that it could be reduced. The omission of area 3M was an oversight on the part of Canada.

The representative of the United States gave full support to the Canadian proposal, although he acknowledged that there could be difficulties in enforcement for the depth restriction measure pending final geographic co-ordinates of such a depth restriction.

The Japanese representative was not at all convinced of the need to take measures to protect the juvenile groundfish using an increased mesh size, or of the need to impose depth restrictions for Greenland halibut. He did, however, acknowledge that excessive incidental by-catch of juveniles was undesirable. The Russian representative concurred with this view.

Once again, the representative of Canada explained the background to the Canadian proposals and in particular, the fact that the Scientific Council had brought the attention of the Fisheries Commission to their concern about the need for the Parties to take measures to reduce catches of juvenile Greenland halibut. It was felt that we could not return to the Fisheries Commission without a suitable result. The Precautionary Approach indicates that when in doubt, managers should err on the side of caution. It appeared, from the point of view of the representative of Norway, that there was little to back the demand for an increased mesh size to 145mm, which appeared to do little to protect the juveniles. However, they could go along with the proposal based on the fact that the coastal State has a mesh size of 145mm. He noted that in any case, Norway employed sorting grids. Regarding the depth restriction, Norway was positive to closures to protect juvenile fish, but more evidence was required to support the proposed measure.

The representative of Canada explained the depth surveys, which had been carried out from 1995 to 1999 and which clearly demonstrated the potential positive effect of depth restrictions for the juveniles. For example, Greenland halibut juveniles generally prefer to remain in waters shallower than 500 metres. He also explained for the benefit of Japan that while the mesh size required for avoiding juveniles would in fact be 205mm, the 145 mm mesh size proposed was a compromise to minimise the impact on commercial fishing while reducing juvenile catches. The Japanese representative considered that this would make any commercial fishery very difficult.

In conclusion, the representative of the European Union noted that the mesh size had been discussed on numerous occasions but that no new arguments had been put forward. Any new measures should be appropriate and suitable. With respect to the depth restrictions, the European Union was of an open mind. The matter should be examined carefully and the Scientific Council should make an assessment and report back accordingly. Acknowledging that something needed to be done, the representative of the United States agreed with the need for such an assessment. The representative from Canada, whilst continuing to be frustrated at the lack of real progress, presented a paper as the basis of a request to the Scientific Council on possible depth restrictions in the Greenland halibut fishery. In order to seek advice from the Scientific Council on the costs and benefits of various closure options and fishing mortality rates, the European Union representative formulated a more detailed request to the Scientific Council (Annex 7). The Japanese representative did, however, note that any restrictions additional to those already in place should still enable there to be commercial fisheries. Existing restrictions were considered by Japan to be already sufficient to protect and increase the Greenland halibut stock. The Japanese representative formulated a request to the Scientific Council (Annex 8).

In order to reflect the urgency of the need for scientific information on the Greenland halibut fishery, it was agreed to reformulate the requests of the European Union and Japan into a single request concentrating on Greenland halibut. The request to the Scientific Council will read as follows:

"The Scientific Council is requested to evaluate:

- "1. Whether the current measures, with minimum size, mesh size and requiring vessels to move from areas where high percentages of undersized fish (less than 30cm in length) are caught, allow for the continued rebuilding of the stock in the presence of the current fishery.
- "2. The bio-mass of Greenland halibut available to the commercial fishery over the whole distribution area of this species, in depth strata of 0 99 metres, 100 199 metres, 200 299 metres, 300 399 metres, 400 599 metres, 600 799 metres and 800 1,000 metres.

"Separate values should be provided for:

- "a. Fish above and below the length of 50% maturity.
- "b. Fish above and below the current minimum landing size."

138

Other elements in the European Union proposal will be retained for discussion at a later date.

The Canadian representative read a statement, which is attached to this report (Annex 9). He was particularly insistent on the relationship of NAFO to the United Nations Fish Stocks Agreement of 1995 and the consistency of NAFO to the coastal States. The Parties agreed that there would be further discussion of this matter at the Annual Meeting in September 2000 following a reply from the Scientific Council.

6. Other Matters

a. Review of submissions on shrimp catches and effort days

The meeting on shrimp stocks held in Washington D.C. in March 2000 requested that STACTIC examine possible new information on shrimp fishing activity in the NAFO Regulatory Area. This would allow for any newly updated data to be provided to the Fisheries Commission before the 2000 Annual Meeting.

The Executive Secretary introduced a paper on the allocations of days, used days and catches as discussed at the Washington D.C. meeting and as revised for the STACTIC meeting (Annex 10). Any data received since the shrimp meeting had been incorporated. However, it was noted that the data contained in this paper was still open to modification.

The Norwegian representative introduced a working paper (STACTIC Working Paper 00/1), which referred to the meeting in Washington D.C. In particular, he referred to Working Paper (Shrimp) 00/12, which specified the level of detail to be presented by Contracting Parties. It was felt that the current Norwegian working paper enhanced the transparency of Norway's shrimp fishery in area 3M. Furthermore, they would like to see other Contracting Parties providing similar details in their submissions to NAFO.

The representative of Denmark (in respect of Faroe Islands and Greenland) introduced a paper covering the revision of data from Greenland on shrimp (Annex 11). In his submission, he agreed with the Norwegian approach, in particular, as this would help the ongoing discussion in the meeting on shrimp and improve the transparency. Furthermore, Denmark (in respect of Faroe Islands and Greenland) cautioned the use of data from the STATLANT reports as data in these reports may have been statistically processed by other authorities outside the fisheries management. Data in the STATLANT reports is based on information from fishing logbooks which reflects the actual fishing days and not the fishing days as calculated according to the entry-and exit-hail reports.

The Canadian representative was able to support the Norwegian approach, but had some doubts on where the data should actually be revised. He also felt that it would be necessary for any changes submitted to be clearly explained. Whilst the United States was able to agree with Canada, there was general agreement by all Parties on the need for clear explanation. The Japanese representative noted the doubts raised as a result of the uncertain data.

The representative of the European Union questioned whether it was wise to use figures as far back as 1993. The measure for shrimp was established in 1995. Subsequently, figures had been constantly changing and as is normal for fisheries, would continue to change. Prior to 1995, the' fishery had been entirely unregulated with consequences and uncertainty for any figures from that time. Questioned by Norway about the high number of days used by the European Union for the reference period, the representative of the European Union felt that the emphasis being laid upon this issue by Norway was entirely due to their own high catches in the earlier years.

The representative of Estonia explained, that his Country had difficulties in being able to provide suitable statistics for the earlier years in question.

The Chairman referred to the compilation of shrimp catches in area 3M prepared by the Executive Secretary (Annex 12). This was the best available data and was to be read in conjunction with Annex 10 (Working Paper 00/2). It was therefore suggested that this data be forwarded to the Fisheries Commission.

The Norwegian representative still insisted on getting further clarification from other Contracting Parties at this stage from both Iceland and the Russian Federation, in particular for the period 1993 to 1995. He noted the enormous difference in levels of detail contained in the compilation. Enhanced transparency was essential for the discussion at the Annual Meeting. The representative of the European Union felt that we were drowning in data and that there was still enormous uncertainty, suggesting that there should be some form of cut off date and that explanations should only be necessary from those Contracting Parties with revised figures. The representative of the European Union also expressed misgivings about an increased use of STACTIC to address topics other than issues of international control. The Canadian representative suggested that it should be for the Fisheries Commission to establish any cut off date.

In conclusion, the Chairman suggested that the data, being the best available, be forwarded to the Fisheries Commission as soon as possible and in any case, no later than 3 July. In so doing, the different quality of information available would be noted, particularly for the period from 1993 to 1995. The Fisheries Commission should also consider a cut off date for the input of data.

The representative of Norway requested that a statement be attached to this report (Annex 13).

The Japanese delegation suggested that, due to the uncertainty in the data and the ongoing changes, the original data be used.

b. Possible follow-up to the Working Group on the Precautionary Approach

The Chairman referred to the report of the Joint Scientific Council and Fisheries Commission Working Group on the Precautionary Approach held in Brussels from 29 February to 2 March 2000 (FC Doc. 00/2). In particular, he noted that STACTIC needs to examine the report and decide on what steps should be taken next. The report is as yet not adopted by the Fisheries Commission and will be examined by them at the meeting in September 2000.

The Canadian representative noted that the next steps were already set out for three stocks (cod 3NO, yellowtail flounder 3LNO and American plaice in 3LNO) in Annexes 6 to 8 of the report. Their motive for adding this point to the agenda was to deal with supportive management measures and good practices for the three stocks in question and hence, to discuss how to deal with these points. It follows on from the Canadian proposal at the 1999 Annual Meeting for a revision of part I.A.5 of the Conservation and Enforcement Measures.

The representative of the European Union felt that at this stage, it was necessary to get further guidance from the Fisheries Commission and that STACTIC should not be addressing questions of a general nature.

The Chairman noted that the proposal had endeavoured to pre-empt the discussion at the forthcoming Annual Meeting and acknowledged the need at this stage to have further guidance from the Fisheries Commission.

c. Charters / "Flag hopping"

The Canadian representative noted that at the last Annual Meeting, new rules on **chartering** had been adopted under Part I.B of the Conservation and Enforcement Measures. This had led to a

140

pilot project on chartering for 2000 and resulted in a charter between Poland and the Russian Federation. Clarification of this project was requested. Did it comply with the Conservation and Enforcement Measures? Were catch statistics available from the charter? The Executive Secretary indicated that information on this charter had been received from the authorities of both Contracting Parties. The question now arose from the Canadian side as to whether the charter itself had been properly notified to the other Contracting Parties. Both Canada and the European Union had doubts as to whether the Fisheries Commission had given approval in the prescribed manner. The Executive Secretary believed that in his interpretation of the rules, the charter had been properly authorised under Article XI (2) of the Convention. The Parties agreed that the issue of the pilot project should be raised for discussion in the Fisheries Commission at the Annual Meeting in September 2000. It was agreed that Canada would prepare a proposal to the Fisheries Commission to this effect. The representative of the European Union recalled that the currently applicable measures were limited in time to 2000 only. The representative of Japan also noted that his country could only accept chartering if it was in full compliance with the full conservation and enforcement measures.

On the separate subject of **flag hopping**, the representative of the European Union wanted to flag this issue, which, he felt, needs to be addressed in detail at a later stage. The European Union wanted to restate its concerns about the practice of vessel owners from one Contracting Party seeking double registry agreements with other Contracting Parties. It was noted that double-flag vessels are flagless and that this was of concern to both the European Union and Iceland. Material was still being compiled on the magnitude of this problem. The question arises as to whether NAFO wants to be an organisation of fishing States or become an organisation of quota buyers and sellers. This issue will need to be discussed again at the next meeting of the Fisheries Commission in September 2000. There was general support from other Contracting Parties, in particular Canada, Denmark (in respect of Faroe Islands and Greenland), Japan and Iceland. In particular, the Japanese representative noted his country's firm opposition to re-flagging as a means to avoid enforcement in regional fisheries organisations.

d. Possible harmonisation of port inspection reports

The representative of the European Union introduced a paper (Annex 14), which would lead to possible harmonisation of port inspection reports by the Contracting Parties under Part VII of the Conservation and Enforcement Measures. He explained the existing disparities in terms of delay experienced by the European Union, the increased practice of vessels landing in ports of other Contracting Parties and thus the difficulties in obtaining port inspection reports in good time. Harmonised port inspection would ensure a better exchange of information as well as improved data flow. It is felt that port inspection under Part VII of the Conservation and Enforcement Measures is one of the pillars of the existing scheme and an important source of information. The proposal of the European Union utilises the North Atlantic format and furthermore, will allow for any subsequent computerisation of data if so required.

It was agreed by the Parties, in particular Denmark (in respect of Faroe Islands and Greenland) and Canada, that this was a good starting point for discussion. The representative of Denmark (in respect of Faroe Islands and Greenland) suggested that there should be greater consistency and harmony between the systems operating on both sides of the Atlantic with regard to the North Atlantic format. The Parties agreed that they would review this proposal in greater depth before the Annual Meeting in September 2000. A two-stage approach would be taken which would examine the manual report and also the relevant codes. It was agreed that the Contracting Parties would prepare for these discussions.

e. Preparation of the review and, as appropriate, the revision of the "Program for Observers and Satellite Tracking"

The representative of the European Union referred to Part VI of the Conservation and Enforcement Measures (Program for Observers and Satellite Tracking). He noted that it was agreed in 1998 that the provisions of the Program are subject to review during 2000 and, as appropriate, revision. If there is a lack of agreement on what to do with this Program, the measures will terminate on 31 December 2000. The measures originally formed part of a package negotiated in 1995. The last evaluation of them was carried out in 1998, but only on the observer component. Satellite tracking is to be on a 100% basis by 1 January 2001 and thereafter, the appropriateness of 100% observer coverage will be questioned. Subsequently, there will be a need to see how the two components of the Program can be properly balanced. At this stage, it is important to flag this issue. The representative of the United States disagreed and indicated that if no changes were necessary to the Program, it should be retained as it is.

Both the representatives of Iceland and Japan agreed with the European Union on the importance of this issue. The representative of Iceland stated that he did not consider 100% observer coverage necessary. However, the representatives of both Canada and the United States did not agree on the interpretation that the measures would drop if there were no agreement of the result of a review. They felt the need to seek further guidance from his authorities and from the Fisheries Commission in September 2000 before proceeding any further. The representative of Denmark (in respect of Faroe Islands and Greenland) felt that it was too early to review the Program as there was still too little experience of Contracting Parties with satellite tracking.

f. New developments / possible overhaul of the Conservation and Enforcement Measures

The representative of the European Union explained that in the opinion of his delegation, it was necessary for all Contracting Parties to be aware that there may need to be a complete overhaul of the Conservation and Enforcement Measures. These measures had evolved over a number of years and clearly needed to be consolidated. Furthermore, there were newer and more recent developments in international fisheries, such as the 1995 UN Agreement on Straddling Fish Stocks and the FAO Compliance Agreement, which should be examined with a view to reviewing the NAFO measures.

The European Union would suggest at the 2000 Annual Meeting that a working group be established to assist NAFO in this respect. A similar exercise was being carried out in other regional fisheries organisations such as NEAFC in the Northeast Atlantic. It was inappropriate to await the entry into force of or adherence to the UN Agreement. NAFO needs to prepare already considering the practical effects of the current changes. Furthermore, NAFO will need to address the issue of the relationship between the special NAFO control rules and the general enforcement provisions of the UN Agreement. The aim of all this would be to strengthen NAFO rules and keep NAFO at the forefront of developments.

The Parties recognised the enormous task ahead of NAFO and agreed to address this issue at the Annual Meeting.

7. Adoption of the Report

The report was adopted by STACTIC on 29 June 2000.

8. Adjournment

The meeting adjourned at 15.05 on 29 June 2000.

Annex 1. List of Participants

CANADA

Head of Delegation

C. J. Allen, Chief, Groundfish, Pelagics and Foreign Fisheries, Resource Management, Dept. of Fisheries and Oceans, Ottawa, Ontario K1A 0E6

Phone: +613 990 0105 - Fax: +613 990 7051 - E-mail: allenc@dfo-mpo.gc.ca

Advisers

D. Bevan, Director General, Resource Management, Dept. of Fisheries and Oceans, 200 Kent Street, Ottawa, Ontario K1A 0E6

Phone: +613 990 6794 - Fax: +613 954 1407 - E-mail: bevand@dfo-mpo.gc.ca

T. Blanchard, Chief, NAFO Unit, Fisheries Management, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 0928 - Fax: +709 772 5983 - E-mail: blanchardt@dfo-mpo.gc.ca

D. W. Kulka, Division Manager, Groundfish Div., Science Br., Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland AIC 5X1

Phone: +709 772 2064 - Fax: +709 772 4188 - E-mail: kulkad@dfo-mpo.gc.ca

L. Penney, Fishery Officer, Fisheries Management, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 3630 - Fax: +709 772 5983 - E-mail: penneyl@dfo-mpo.gc.ca

P. Steele, Program Manager-Enforcement, Conservation and Protection Directorate, Dept. of Fisheries and Oceans, 200 Kent St., 13th Floor, Ottawa, Ontario K1A 0E6

Phone: +613 990 0109 - Fax +613 941 2718 - E-mail: steelep@dfo-mpo.gc.ca

R. Steinbock, Advisor, Atlantic Affairs Div., International Directorate, Dept. of Fisheries and Oceans, 200 Kent St., Ottawa, Ontario K1A 0E6

Phone: +613 993 1836 - Fax: +613 993 5995 - E-mail: steinbob@dfo-mpo.gc.ca

CUBA

Head of Delegation

L. Albelo Leon, Cuban Fishing Fleet Representative, 1881 Brunswick St., Ph-B, Halifax, Nova Scotia, Canada B3J 3L8

Phone: +902 425 5773 - Fax: +902 423 8871 - E-mail: Pfphfx@aol.com

Adviser

E. Valdes, Centro de Investigaciones Pesqueras, 5ta. Ave y 248, Barlovento, Sta Fe, Playa C. Habana 19100 Phone: +537 298055 – Fax: +537 339168 – E-mail: <u>cubacip@ceniai.inf.cu</u>

DENMARK (IN RESPECT OF THE FAROE ISLANDS & GREENLAND)

Head of Delegation

M. T. Nedergaard, Fiskerilicensinspektor, Head of Unit, Gronlands Fiskerilicenskontrol, Postbox 501, DK-3900 Nuuk, Greenland

Phone: +299 345377 - Fax +299 323235 - E-mail: GFLK@gh.gl

Advisers

L. Geråe, Head of Development and Planning, Gronlands Fiskerilicenskontrol, Postbox 501, DK-3900 Nuuk, Greenland

Phone: +299 345373 - Fax +299 323235 - E-mail: <u>GFLK@gh.gl</u>

M. Kruse, Vaktar-og Bjargingartaenastan, P. O. Box 347, FR-110 Torshavn, Faroe Islands Phone: +298 11065 - Fax: +298 13981 - E-mail: mk@vb.fo

144

J. H. Toftum, Ministry of Fisheries, P. O. Box 64, FO-100 Torshavn, Faroe Islands Phone: +298 353030 - Fax: +298 353035 - E-mail: jenst@fisk.fl.fo

ESTONIA

Head of Delegation

L. Vaarja, Ministry of the Environment, Fisheries Department, Marja Str. 4d, 10617 Tallinn Phone: +372 6112987 Fax: +372 6567599

Adviser

A. Soome, Ministry of the Environment, Fisheries Department, Marja Str. 4d, 10617 Tallinn Phone: +372 6112987 Fax: +372 6567599 E-mail: <u>ains@klab,envir.ee</u>

EUROPEAN UNION

Head of Delegation

F. Wieland, Deputy Head of Unit, International Fisheries Organizations and Fisheries Agreements; Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi Wetstraat 200, B-1049 Brussels, Belgium

Phone: +32 2 296 3205 - Fax: +32 2 299 4802 -E-mail: Friedrich, Wieland@cec.eu.int

Alternate

A. Thomson, Principal Assistant, International Fisheries Organizations and Fisheries Agreements; Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi/Wetstraat 200, B-1049 Brussels, Belgium Phone: 132.2.200.0180. East: 132.2.200.4802. E. mail: Andrew Thomson @eco.ex.int.

Phone: +32 2 299 0180 - Fax: +32 2 299 4802 - E-mail: Andrew, Thomson@cec.eu.int

Advisers

K. Patterson, European Commission, Directorate General for Fisheries, Rue de la Loi 200, B-1049 Brussels, Belgium

Phone: + 32 2 299 2179 - Fax: +32 2 295 5621 - Email: kenneth.patterson@cec.eu.int

B. O'Shea, European Commission, Directorate General for Fisheries, 200 Rue Joseph II 99, Rm 1/27, B-1049, Brussels, Belgium

Phone: +32 2 296 6748 - Fax: +32 2 296 2338 - Email: Brendan.O'Shea@DG14.be

- S. Salvador, Chefe da Divisao de Relacoes Internacionais, Direccao Geral das Pescas e Aquicultura, Edificio Vasco da Gama, Cais da Alcantara Mar, 1399-006 Lisbon, Portugal
- Phone: +351 21 3914352 Fax: +351 21 3979790 E-mail: susanas@dg-pescas.pt
- G. F. Kingston, Senior Adviser, Delegation of the European Commission, 45 O'Connor Street, Suite 1900, Ottawa, Ontario, Canada KIP 1A4

Phone: +613 238 6464 - Fax: +613 238 5191 - E-mail: fred.kingston@eudelcan.org

T. Heaton, Director, DG BIII-Fisheries, Council of the European Union, Rue de la Loi 175, B-1048 Brussels, Belgium

Phone: +32 2 285 6486 - Fax: +32 2 285 8261 - E-mail: Trevor, Heaton@consilium.eu.int

- S. Feldthaus, Head of Section, Ministry of Food, Agriculture and Fisheries, Holbergsgade 2, 1057 Copenhagen K, Denmark
- Phone: +45 33 92 35 60 Fax: +45 33 11 82 71 Internet: sfe@fvm.dk
- Y. Auffret, Ministere de l'Agriculture et de la Peche, Direction des Peches Maritimes, 3 Place de Fontenoy, 75007 Paris, France -

Phone: +33 1 49558245 - Fax: +33 1 49558200 - E-mail: yves.auffret@agriculture.gouv.fr

H. Pott, Bundesministerium fur Ernahrung, Landwirtschaft und Forsten, Rochusstr. 1, D-53125 Bonn, Germany Phone: +49 228 529 4124 - Fax: +49 228 529 4410 - Email: <u>hermann.pott@bml.bund.de</u>
I. Ybanez, Subdirector General de Organismos Multilaterales de Pesca, Secretaria General de Pesca Maritima, Jose Ortega y Gasset, 57, 28006 Madrid, Spain

Phone: +34 91 402 74 04 - Fax: +34 91 309 3967 - E-mail: iybanezr@mapya.es

J. Del Hierro, Subdireccion General de Inspeccion Pesquera, Secretaria General de Pesca Maritima, c/Castellama 112, Madrid, Spain

Phone: +34 91 3471847 - Fax: +34 91 3471512

ICELAND

Head of Delegation

S. Asmundsson, Legal Advisor, Ministry of Fisheries, Skulagata 4, IS-150 Reykjavik Phone: +354 560 9670 - Fax: +354 562 1853 - E-mail: stefas@hafro.is

Advisers

G. Hannesson, Directorate of Fisheries, Ingilfsstraeti 1, 101 Reykjavik Phone: +354 569 7900 - Fax: +354 569 7991 - E-mail: greta@hafro.is

JAPAN

Head of Delegation

 Y. Sakamoto, Deputy Director, Far Seas Fisheries Div., Resources Management Dept., Fisheries Agency, Government of Japan, 1-2-1Kasumigaseki, Chiyoda-ku, Tokyo 100-8907
 Phone: +81 33 591 6582 - Fax: +81 33 591 5824

Advisers

Y. Kashio, Japan Fisheries Association, Suite 1209 Duke Tower, 5251 Duke St. Tower, Halifax, N.S., Canada B3J 1P3

Phone: +902 423 7975 - Fax: +902 425 0537 - E-mail: jfa-hfx@ns.sympatico.ca

N. Takagi, Director Executive Secretary, Japan Deep Sea Trawlers Association, Ogawacho-Yasuda Bldg.,

6 Kanda-Ogawacho, 3-Chome, Chiyoda-ku, Tokyo 101-0052

Phone: +81 33 291 8508 - Fax: + 81 33 233 3267 - E-mail: nittoro@mx3.mesh.ne.jp

NORWAY

Head of Delegation

S.-A. Johnsen, Directorate of Fisheries, P. O. Box 185, 5804 Bergen Phone: +47 55 238000 - Fax: +47 55 238090

Adviser

P. Oma, Directorate of Fisheries, P. O. Box 185, 5804 Bergen Phone: +47 55 238000 - Fax: +47 55 238090

RUSSIA

Head of Delegation

 A. Okhanov, Representative of the Russian Federation in Canada on Fisheries, 47 Oceanview Dr., Bedford, Nova Scotia, Canada B4A 4C4
 Phone: +902 832 9225 - Fax: +902 832 9608

UNITED STATES OF AMERICA

Head of Delegation

D. T. Mathers, Lieutenant Commander, Coast Guard Liaison Officer, Office of Marine Conservation (Rm 5806), U.S. Department of State, 2201 C Street NW, Washington, D.C. 20520 Phone: +202 647 3177 - Fax: +202 736 7350 - E-mail: <u>dmathers@comdt.uscg.mil</u>

Advisers

J. Anderson, Fisheries Management Specialist, Northeast Region, National Marine Fisheries Service, U.S. Department of Commerce, 1 Blackburn Dr., Gloucester, MA 01930

Phone: +978 281 9226 - Fax: 978-281-9135 - E-mail: Jennifer.Anderson@noaa.gov

K. Rodrigues, Fishery Policy Analyst, Northeast Region, National Marine Fisheries Service, U.S.

Department of Commerce, 1 Blackburn Dr., Gloucester, MA 01930

Phone: +978 281 9324 - Fax: +978 281 9135 - E-mail: Kathi.Rodrigues@noaa.gov

NAFO SECRETARIAT

L. I. Chepel, Executive Secretary

F. D. Keating, Administrative Assistant

G. M. Moulton, Statistical /Conservation Measures Officer

B. Cruikshank, Senior Secretary

F. E. Perry, Desktop Publishing/Documents Clerk

Annex 2. Agenda

- 1. Opening by the Chairman (D. Bevan Canada)
- 2. Appointment of Rapporteur
- 3. Adoption of Agenda
- 4. Program for Observers and Satellite Tracking
 - (a) scientific requirements
 - (b) amendments to existing program
 - (c) observer manual
- 5. Possible amendments to Conservation and Enforcement Measures regarding juvenile fish

6. Other matters

- a) Review of Submissions on shrimp catches and effort days
- b) Possible follow-up to the Working Group on the Precautionary Approach
- c) Charters: "Flag hopping"
- d) Possible harmonization of port inspection reports
- e) Preparation of the review and, as appropriate, the revision of the "Program for Observers and Satellite Tracking"
- f) New developments/possible overhaul of the Conservation and Enforcement Measures
- 7. Adjournment

Annex 3. Working Paper by Denmark (in respect of Faroe Islands and Greenland) (STACTIC Working Paper 00/5)

During the discussion of the scientific requirements for the observer program in September 1999 the accuracy of the by-catch estimations and discards were questioned.

As quantities of by-catches and discards normally are based on a visual estimation made by the masters of the fishing vessels and the observers, Greenland biologists and the Greenland observers carried out a number of tests in order to evaluate the accuracy of by-catch estimations on board shrimp trawlers.

The results of the research, carried out in Greenland waters is displayed in the graphs below.

The estimate is based on a visual judgement of the catch in the codend and when it is emptied into the bin as well as during the processing/sorting of the catch.

The difference is striking, bearing in mind that the estimates are made by experienced observers.

In order to improve the quality of the by-catch- and discard data Denmark (in respect of Greenland and Faroe Islands) suggests that it becomes compulsory to collect by-catches in boxes or containers in order to make a proper estimate before any quantity is discarded.



Annex 4. Proposal (by Norway) to amend the NAFO Conservation and Enforcement Measures, Part VI.A.1(a) regarding independent and impartial observers (STACTIC Working Paper 00/7)

At the STACTIC Meeting during the NAFO Annual Meeting in September 1999, it was agreed that it was needed to look at an amendment to the Conservation and Enforcement Measures, Part VI.A.1(a), to ensure that observers are independent and impartial.

We propose the following amendment:

These Observers are not to perform other duties e.g. working as crew members onboard the fishing vessel.

Annex 5. Proposals (by Canada) to amend the NAFO Conservation and Enforcement Measures Regarding Protection of Juvenile Groundfish (STACTIC Working Paper 00/3)

General Background

At the September 1999 annual NAFO meeting, the Fisheries Commission directed that "In light of the advice of the Scientific Council, STACTIC shall review all management options by which catches of juvenile fish can be reduced taking into account the various NAFO fisheries and elaborate and recommend feasible measures to be incorporated in the NAFO Conservation and Enforcement Measures."

The Fisheries Commission made this statement in the context of discussions surrounding the setting of a TAC for 2+3KLMNO Greenland halibut. The subsequent TAC set by the Fisheries Commission was considerably higher than Canada and some other Contracting Parties had favoured, particularly in light of the continuing concern expressed by the Scientific Council over excessive catches of juvenile Greenland halibut.

The Scientific Council has, on a number of occasions, expressed similar concern regarding catches of juveniles in other groundfish stocks as well. The Scientific Council has also raised concerns regarding the need to keep bycatches of stocks, particularly those subject to NAFO moratoria, to the lowest possible level and reducing and controlling the amount of discards in the Regulatory Area.

The February 29-March 2, 2000 report of the Joint Scientific Council and Fisheries Commission Working Group on Precautionary Approach proposes 'next steps' in the implementation of the Precautionary Approach for the three stocks being considered on a pilot basis (3NO cod, 3LNO American plaice and 3LNO yellowtail). In all cases, under the 'Supportive Management Measures/Good Practices' section, the Working Group recommends that the Fisheries Commission take steps to minimize the catch of juveniles. While the Working Group's overall report has not yet been adopted by the Fisheries Commission, it would seem to be only common sense that measures, or good practices, be adopted to protect juveniles.

Adequate measures must be put in place to preserve young, immature fish, giving them a chance to develop and survive in sufficient numbers to spawning age so as to allow stocks to recover. Secondly, discarding of undersized fish at sea must be reduced. The inadequate measures currently in place have hindered the rebuilding of a number of NAFO-managed groundfish stocks. As in other areas of the world the size of fish being taken is too small.

(1) <u>Increase in Mesh Size</u>

Background

The current mesh size for all groundfish in the Regulatory Area is 130 mm. Canada began increasing its minimum mesh size a number of years ago from this level, in consultation with fish managers, scientists and fishermen, because of concerns with the capture of too many juvenile fish.

The minimum mesh size for Canadian fishermen fishing NAFO-managed stocks in both Sub-Areas 2+3 (except redfish and skate) is 145 mm both inside Canadian waters and within the NAFO Regulatory Area and many believe that this is still too small to adequately protect juveniles. This mesh size was increased a number of years ago as a precautionary measure to enable some greater escapement of small fish without preempting the economics of a trawler fishery. In the context of 75-81 % of the 2+3KLMNO Greenland halibut biomass, for instance, being distributed within coastal state waters but 74 % of the total allocation and 80 % of the catch taking place in the NRA, it would be appropriate for NAFO to adopt the same minimum mesh size as the coastal state. Any benefit that might accrue to the resource as a result of this conservation measure by the coastal state will be effectively undermined if the minimum mesh size stays at 130 mm in the NRA.

Proposal #1

Proposed Amendment to Part V, Schedule IV of the Conservation and Enforcement Measures .

Authorized Mesh Size of Nets

	Species	Mesh Size
a)	All principal groundfish, flatfishes and other groundfish and other fish with the exception of capelin <u>and redfish</u> as listed in Part V, Schedule II, Attachment II.	145 mm
b)	redfish	130 mm

Existing (b) and (c) be re-lettered (c) and (d).

(2) Depth Restriction for Greenland halibut

Background

Continued rebuilding of the Greenland halibut resource will depend on the ability of recruiting juvenile fish to reach spawning age. The probability of good recruitment will also be enhanced through the establishment of a rebuilt and stable spawning stock biomass. However, virtually 100% of the fishing mortality in the NAFO Regulatory Area, and much of the fishing mortality in coastal state waters, consists of juvenile fish. Unlike other groundfish fisheries in the NRA, where fishing mortality cuts across a broader age structure consisting primarily of adult fish, the Greenland Halibut fishery is essentially a 'recruitment fishery'.

Previously, the Scientific Council noted that recovery of 2+3KLMNO Greenland Halibut has commenced for the fishable population (>35 cm) which currently was about 40% of levels of the late 1970s through early 1980s. The population of the female spawning stock biomass (>60 cm) remains at or near record lows (less than 10% of historic levels). In its June 2000 meeting, the Scientific Council noted that the high exploitation of immature fish and the low abundance of sexually mature fish (>60 cm) is indicative of a situation of significant biological risk, although this risk cannot be quantified at present. The Council again recommended that measures be considered to reduce, as much as possible, the exploitation of juvenile Greenland halibut in all fisheries.

The Council, in its June 2000 report also notes that it is concerned that increased catches of Greenland halibut will result in increased catches of other species, some of which are currently under moratorium. They strongly recommend that the Fisheries Commission take steps to ensure

that any bycatches of other species during the Greenland halibut fishery are true and unavoidable bycatches.

While the fishable biomass appears to be recovering, the same cannot be said for the female spawning biomass (i.e. >60 cm) which remains at or near record low levels. The initial recovery trends of this stock is primarily a result of the emergence of several good year classes. Its continued recovery and future viability will depend in part on the rebuilding of a broad age structure within the spawning stock biomass.

The precautionary approach, and simple common sense, suggests that greater caution is required when managing a recruitment or juvenile-based fishery. If the reality of the commercial trawler fishery results in a greater mortality on juveniles than would otherwise be the case, then specific measures should be undertaken to mitigate any associated impact on the long-term health on the resource, particularly when viewed in the context of a re-building objective. It is not prudent management to rely on recent high recruitment trends from a low spawning stock biomass.

It is also important to note that a natural separation between juvenile and older Greenland halibut appears to follow the 500-fathom contour, as younger halibut prefer depths less than 500 fathoms.

Significant quantities of cod, yellowtail, and American plaice have been caught as by-catch in the NRA. There are higher relative abundance of these species and of juvenile fish (including Greenland halibut) in shallower waters. While permitted under the current by-catch regime, it is apparent that these fish are not being caught as a true incidental catch, at least during the directed Greenland halibut fishery, as the distribution of this fishable biomass occurs in deeper waters. It would be effective and feasible for directed Greenland halibut fisheries to be restricted from geographic coordinates that involve depths less than 400 meters (or perhaps even deeper).

There is virtually no overlap in the 'commercial-size' distribution of Greenland halibut and yellowtail. Similarly, overlap in distribution of Greenland halibut and American plaice/cod generally occurs at depths greater than 200 meters for all sizes and greater than 400-750 meters for commercially fished sizes. Based on this information, it would be effective and feasible for directed Greenland halibut fisheries to be restricted from geographic coordinates that involved depths less than 400-750 meters. Such a restriction would be effective in minimizing by-catch of cod, yellowtail and American plaice, in mitigating the catch of witch, and in mitigating the catch of 'pre-recruit' Greenland halibut. Such a restriction would be enforceable, yet would not place undue hardship on the economic viability of the directed Greenland halibut fishery conducted by the trawler fleet.

Proposal #2

Proposed Amendment to Part I, Management of the Conservation and Enforcement Measures

Addition of new section L as follows:

- L. Other Measures Management Measures for Greenland halibut in Divisions 3LNO
 - 1. Directing for Greenland halibut in Divisions 3LNO will be prohibited in waters of depths less than 400 meters.
 - 2. For the purpose of paragraph (1), the 400 meter contour will be delineated by the following coordinates:

Annex 6. Additional Information (by Canada) - Depth Proposal for Greenland halibut

(STACTIC Working Paper 00/3, Addendum)

A total of 1803 successful Campelen sets were examined from fall surveys in 3LNO from 1995-99. The following table shows the percentage of catch numbers, by depth zone, for Greenland halibut, yellowtail, American plaice, cod, witch, and skate. It is important to note that while representative in a general sense, these percentage figures are overstated in relation to the depth distribution of the respective species that would be available to commercial gear. To illustrate, the percentage of fishable biomass of Greenland halibut (>35 cm) that are at depths less than 400 meters would be significantly lower than the 50.5 % that relates to the small mesh Campelen trawl. It is also important to note that a natural separation between juvenile and older Greenland halibut appears to follow the 500 meters contour; as younger halibut prefer depths less than 500 meters.

Depth	Gr. Halibut	Yellowtail	A. Plaice	Cod	Witch	T. Skate
<100 m	2.1 %	99.9 %	36.2 %	53.1 %	20.8 %	67.5 %
<200 m	5.8 %	100 %	74.7 %	73.8 %	39.6 %	73.8 %
<400 m	50.5 %	100 %	89.9 %	98.2 %	51.5 %	95.4 %
<750 m	78.7 %	100 %	96.7 %	100 %	88.9 %	99.7 %
<1000 m	91.4 %	100 %	99.9 %	100 %	98.9 %	99.9 %

There is virtually no overlap in the 'commercial-size' distribution of Greenland halibut and yellowtail. Similarly, overlap in distribution of Greenland halibut and American plaice/cod generally occurs at depths greater than 200 meters for all sizes and greater than 400-750 meters for commercially fished sizes. Based on this information, it would be effective and feasible for directed Greenland halibut fisheries to be restricted from geographic coordinates that involved depths less than 400-750 meters. Such a restriction would be effective in minimizing by-catch of cod, yellowtail and American plaice, in mitigating the catch of witch, and in mitigating the catch of 'pre-recruit' Greenland halibut. Such a restriction would be enforceable, yet would not place undue hardship on the economic viability of the directed Greenland halibut fishery conducted by the trawler fleet.

Annex 7. Working Paper by European Union (STACTIC W.P. 00/11)

Draft of Request to Scientific Council on Greenland Halibut Depth-Distribution and Protection of Juveniles

Scientific Council is requested to evaluate:

1. The fishable biomass of the main commercial species of fish in depth strata of 0-99m, 100-199m, 200-299m, 300-399m.

For all species, separate values should be provided for

- a. Fish above and below the length of 50% maturity.
- b. Fish above and below the current minimum landing size.
- 2. The likely future medium-term development for Greenland Halibut, Yellowtail Flounder, cod in 3NO and as many other stocks as possible, under the following assumed constraints:
 - a. Closure of targeted Greenland Halibut fishery in depths less than 100, 200, 300, or 400 metres, and redirection of effort so removed onto the remaining depth strata according to recent fishing practices. These cases should be compared with evaluation of current fishing practices.
 - b. Subject to the above, likely future medium-term consequences (5 to 10years) for the yield, spawning biomass, exploitable biomass and recruitment, stating the relevant biological assumptions.
 - c. The scenarios should be explored for a range of fishing effort assumptions corresponding to :
 - i) Maintaining overall fishing effort at the same levels as estimated in the last year for, which good information is available.
 - ii) Increase or decreases of +/- 30% in fishing effort from this value.
 - iii) Additional scenarios as considered appropriate by Scientific Council

In the above scenarios, Scientific Council should evaluate whether these fishing strategies provide adequate long-term protection to juvenile fish to allow maintenance of the spawning biomass at an appropriate level.

Annex 8. Working Paper by Japan (STACTIC W.P. 00/12)

Draft of Request to Scientific Council to evaluate Greenland Halibut

Whether the current restriction is enough to protect Juveniles

- 1. Do the current measures with minimum size, mesh size and requiring vessels to move from areas where high percentages of juveniles are caught, allow for the continued rebuilding of the stock in the presence of the current fishery?
- 2. How much catch of juvenile fish will result in risks to the stock rebuilding?
- 3. If the fishing mortality is largely concentrated on adult fish what is the potential impact on spawning stock biomass?
- 4. Is a mesh size requirement sufficient to achieve the same conservation goals as a combination of minimum depth and small fish size restrictions?

Annex 9. Statement from the Representative of Canada

Agenda Item 5 - Possible amendments to Conservation and Enforcement Measures regarding juvenile fish

Mr. Chairman,

Canada is getting a little frustrated at lack of any progress on this issue. As I said this morning, the Fisheries Commission gave STACTIC, what we thought, were very clear instructions – I'll read them again:

"In light of the advice of the Scientific Council, STACTIC shall review all management options by which catches of juvenile fish can be reduced taking into account the various NAFO fisheries and elaborate and recommend feasible measures to be incorporated in the NAFO Conservation and Enforcement Measures."

We do not understand what is unclear about this sentence. It makes no mention as to whether anything should be appropriate or not. (I'm referring here to our earlier discussion on possible revisions to the Program for Observers and Satellite Tracking, if appropriate.) It clearly states that STACTIC should be recommending measures or amendments to existing measures to reduce catches of juvenile fish. It is talking about all fish stocks – not just Greenland halibut.

Once again, I would like to remind delegates why we got these instructions – they were linked to the agreement on a TAC for Greenland halibut for 2000. They came out of the Heads of Delegation meeting. Canada, and others, finally accepted a higher TAC for Greenland halibut but only if STACTIC was instructed to come up with measures to protect juveniles.

So – what ideas have we come up with? Canada has made 2 proposals, neither of which appear to be acceptable to the majority of participants here. But no one else has come up with any other proposals.

A number of statements were made this morning by delegations that had difficulty with accepting our proposals – yet they have not offered any alternatives.

Some have questioned whether or not the Scientific Council has presented any views to back up our proposals. This has always been the excuse in STACTIC for not moving forward on unfavourable proposals. I can understand why some may wish to query the Scientific Council on our proposal for depth restrictions – this is an issue that has never before been contemplated by STACTIC or NAFO. But on mesh size – STACTIC has had plenty of discussions on increasing mesh sizes before – this is not a new concept.

Whatever happened to the concepts embodied in UNFA. Now, we know that not all Contracting Parties around this table have ratified UNFA, but surely to goodness fisheries management around the world has at least bought into the idea embodied in Article 6 of UNFA that "states shall be more cautious when information is uncertain, unreliable or inadequate. The absence of adequate scientific information shall not be used as a reason for postponing or failing to take conservation and management measures."

I would just like to remind delegates that Canada's interpretation of the NAFO Convention is that NAFO is supposed to be consistent with the coastal states when it comes to managing straddling stocks – not the other way around.

Canada has put in place a whole suite of management measures that are much more restrictive than what is in place within the NRA. Just like within the NRA, no-one measure by itself will necessarily make a difference – but taken as a whole, yes they can make a difference.

In Canada we reacted a number of years ago to continuing concern about catches of juvenile groundfish. One of the measures we adopted was to increase mesh size. We also implemented what we call a small fish protocol. We have explained these measures and all of our other measures to STACTIC before and to other NAFO Working Groups.

I for one, do not want us to go back to the Fisheries Commission saying that we discussed a couple of ideas but need more input from the Scientific Council before we act.

Annex 10. Shrimp 3M Fishery Statistics, 1993-1999 (STACTIC Working Paper 00/2)

- Allocated/used days and catches (data as discussed at the Washington Meeting, March 2000) -Table 1
- Revised catches and allocated/used days (as received at the Secretariat by June 26, 2000) -Table 2

	-	1233	-	1994	-	1995	-	1996		<i></i>	1997		7	1998		31	1999	
Contracting Party	Used	Catch	Used	Catch	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	Catch
Canada	507	3191	333	1042	319	968	445	311	908	443	156	784	443	82	435	456	64	385
Cuba	,	•	•	ŀ	•	,	100	•		100	•	•	100	•		<u>10</u>	33	119
DenFaroes	•	7076	1	4998	•	5993	1785	•	8685	1606	1241	7387	1607	1271	7741	1606	1111	9119
DenGreenland		3788	•	2275	,	2400	572	•	1107	515	•	104	515	108	865	515	8	576
Estonia	'	•	•	1051	•	2380	1852	993	1973	1217	692	3239	1217	916	5694	1667	1645	10846
European Union	139	754	67	432	44	487	508		198	457	3	593	457	105	1553	457	268	1265
France (SP)	•	'	•	1		,	N/A	'	•	100	ನ	•	N/A	•	,	<u>6</u>	,	
Iceland	279	2195	638	2355	1842	7481	N/A	5256	20680	N/A	1362	7197	N/A	9 6 8	6572	N/A	1312	7643
Japan -	,	•	•	,	•	1	N/A		1	N/A	•		N/A	•	1	100	,	,
Korea	,	'	•	,	•	•	N/A	•	'	N/A	•	•	N/A	•	•	100	,	,
Lalvia		•	190	324	545	679	421	504	1253	400	369	697	400	313	1191	416	598	2765
Lithuania	•		453	863	638	980	638	918	1585	579	611	1785	579	866	3107	579	602	3370
Norway	1354	7075	2130	8625	2113	9534	2206	1482	5805	1985	334	1831	1985	214	1339	1985	428	2976
Poland	•	•	1	,	4	•	N/A	1	,	N/A	100		400	4	148	100	104	707
Russia	76	54	41	350	1533	3327	N/A	2458	444	2600	807	1090	2600	•	4	2100	417	1126
USA		'	·	-	•	·	N/A	•	•	N/A	•	•	100	•	•	00 1	'	,
Total	2355	24133	3882	22315	7034	34229	8527	11922	46638	10002	5757	25007	10403	4883	28645	10381	6760	40897
	-	1993		1994	-	1995	L '	1996			1997			1998		Ē	1999	
Contracting Party	Used	Catch	Used	Catch	Used	Catch	Alloc.	Used	Catch	Alloc.	Used	- Catch	Alloc.	Used	Catch	Alloc.	Used	Catch
Canada	507	3191	333	1042	319	968	492	311	806	443	156	784	443	82	435	456	6/	385
Cuba		,	1	•	•	1	100	•	· ·	<u>1</u> 0	'		100	•	,	100	ŝ	119
DenFaroes	1324	7333	1785	6791	1093	5993	1785	1831	8688	1606	1250	7410	1606	1292	9368	1606	1051	9199
DenGreenland	572	3780	482	2272	265	2316	572	202	1098	515	31	105	515	113	862	515	65	537
Estonia	149	268	609	1051	2153	2379	1852	066	1898	1217	1254	3240	1217	1454	5533	1667	1651	10834
European Union	139	754	97	432	44	487	508	'	198	457	63	593	457	105	1553	457	268	1265
France (SP)	•	1	1		4	k	100	'	•	100	22	- <u>-</u>	100	•	•	<u>8</u>		1
lceland	279	2195	638	2355	1842	7481	N/A	5256	20682	N/A	1327	6473	N/A	980	6580	N/A	1222	9286
Japan		•		•	•	'	001	•	,	100	1	· ·	100	'	•	100	•	1
Korea	'	•	'	•	٠	1	100	1.	1	10		,	100	•	•	100		
Latvia		•	190	324	649	679	544	504	1253	490	439	266	490	402	1191	490	438	3080
Lithuania	•	'	453	863	638	980	638	918	1585	579	611	1785	579	866	3107	. 579	620	3371
Norway	1403	7074	2206	8625	2162	9391	2206	1549	5648	1985	329	1886	1985	211	1339	1985	394	2975
Poland	•	•	•	'	•	•	100	,	'	100	100	817	100	40	148	100	104	859
Russia	76	5	41	350	1533	3327	N/A	2458	4444	2600	807	1090	2600	•	•	2100	417	1126
USA		'				ʻ	100		1	100		1	100			100	,	'
Totai	4449	24649	6834	24105	10698	34001	9197	14019	46402	10492	6389	25180	10492	5545	30116	10455	6242	43036

,

,

159

.

Annex 11. Submission on shrimp catches and effort days - Working Paper by Denmark (in respect of Faroe Islands & Greenland) (STACTIC Working Paper 00/4, Rev. - submitted by Greenland)

With regards to the STACTIC agenda p. 6a and with reference to the Working Group meeting on Shrimp in 3M in Washington, D.C., 27 March 2000 it was agreed that Contracting Parties should provide data revisions to the Secretariat in time for the June 2000 STACTIC meeting.

Greenland hereby forwards information on vessels, catches and effort days for the period 1993-1999.

Entry and Exit dates are according to the hail reports of the vessels and catches are accumulated catches based on logbook entries and landing documentation.

Furthermore a specification on shrimp catches by year and months is also attached.

					Green	and			÷ .		
1993	<u> </u>	T	Trip1	_		Trip2			Trip3		
/essel Name	RC	In		Days	In	<u></u>	Days	In	10	Days	Total Da
linniamiut	aiv	4Jun93	16.11-93	43			0			0	
Jesper Belinda	am	28 May 90	13.1n90		16.1.0.93	26.11.93	41	7.4 n.03	15 Aug 93	g	
Tasemiut	avai	31-May-90	4.01-90		7-11-90	20.11-32	14	17033	Joruga	a	
	OWI	26Jun-93	4580-93	71	7-Sep-90						
villiit	OWM	30-Aug-93	4900	6	8-Sep-90	301-90	- 28				
Tunuik	OK	29-May-90			24Jun98					- 0	
Tasilag	OHD	31-May-93	1-Aug-96		2400130	/00/50		-			
Cipcopaq	ONK	8Jun93	9.04				— d			- 0	
Gipuquaq Betty Belinda			7-11-90	30						0	
Nanco Trawi	OXT	1-Jun 90	22-11-92				- 0				
	01ZL	7-Jun 93	7-11-50	31	10.11-93	1-Aup-93	23				
Anso Mølgård Kaassassuk	014	8Jn93	16.11-90	39	10.01-50	1-AUGO	0				
Total	Uzna	ourise	10-00-50	437			126		-	0	
10a 1994			Third	4.3/		Tile 0			T :- 0	5	5
Vessel Name	RC	In l	Tript Out l	0.2.5	ln l	Trip2	Dec		Trip3	D00	Total
	akv	29-May-94		Days		<u> </u>	Days		<u>Ot</u>	Days	Total Da
Tirmiamut			9-11 -94 	42 43							
Tasemiut Dalor Diagona II	OVAL	23-May-94		_	<u> </u>		q		I		
		7-Jul-94		88			0				
Regina C	OMBZ	26Jn94	8,11-94	13			0			0	
Tasiilag Dattu Daliarata	OHO	30-May-94	14.11-94	46			0			C	
Betty Belinda	OAL	29Jin94	20-01-94	2			0			0	2
Anso Mølgård	0/21.	7-Apr-94		39			48	7-11-94	13-Aug-94	38	
Nuk	ON THE	1-May-94	2Jun94	33	6Jun94	19,11-94	44			0	
Kaassassuk	040	12Jn94	14JU-94	33	· · ·		0			0	
Total				364			90			38	4
1995		<u> </u>	Trip1			Trip2			Trip3		~
Vessel Name	RC		<u> </u>	Days	In	<u>0</u>	Days	In	<u></u>	Days	Total Da
Kiliutaq	OVGG		23Jn95	33	27Jin95	4A <u>.</u> 995	39				
Tæemiut	ovan -	30-May-95	211-95	34	ł		0	·		q	
Tasiilaq	OHD	23Jn95	20,11-96	- 28			q			Q	2
BettyBelinda	ONT	25Jn95	<u>30-Jrn96</u>	6	ł		q			0	
Nencq Trawl	OVXT	14Jn95	27-JU-95	44			0			0	4
Nuk	OZDH	15-May-96	22-J.n95	39	26Jin96	6Aug96	42			0	_
Total			<u> </u>	184			81				2
1996											
			_Trip1			Trip2			Trip3		
Vessel Name	RC	In	<u>Ó</u>	Days	In	Cut	Days	In		Days	
Tasiilaq	OHO	27-May-96	0.1 4.11-96	39	in .		d	<u>In</u>		0	
Tasiilaq Nancq Trawl	OHO OYXT	27-May-96 8Jun-96	Out 4Jul-96 17Jul-96	39 40	in 		d	<u>In</u>			
Tasiilaq Nanoq Trawl Regina C	onho oxt onez	27-May-96 8Jun-96 18Jun-96	Out 4JUI-96 17JUI-96 20JUI-96	39 40 33	In 		0	In		0	
Tasiilaq Nanoq Trawl Regina C Nodine C	OHD OXT OXZ OYZ	27-May-96 8Jun-96 18Jun-96 17Jun-96	Qi 4JU-96 17JU-96 20JU-96 23JU-96	39 40 33 37	in		0 0 0 0	In		0	
Tasiilaq Nanoq Trawl Regina C Nocline C Kaassassuk		27-May 96 8-Jin 96 18-Jin 96 17-Jin 96 9-May 96	Qi 4JU96 17JU96 20JU96 23JU96 2JU96	39 40 33 37 25	in		0 0 0 0	In		0	
Tasiilaq Nanoq Traw Regina C Nodine C Kaassassuk Rolar Reaja	OHD OXT OXZ OYZ	27-May-96 8Jun-96 18Jun-96 17Jun-96	Qi 4JU96 17JU96 20JU96 23JU96 2JU96	39 40 33 37 28 28	in			In			
Tasiilaq Nanoq Traw Regina C Nodine C Kaassassuk Rolar Raaja Total	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	27-May 96 8-Jin 96 18-Jin 96 17-Jin 96 9-May 96	Qi 4JU96 17JU96 20JU96 23JU96 2JU96	39 40 33 37 25	In	L CL	0 0 0 0	In		000	
Tasiilaq Nanoq Trawl Regina C Nodine C Kaassassuk Polar Reaja Total 1997	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	27-May 96 8-Jin 96 18-Jin 96 17-Jin 96 9-May 96	Qi 4JU96 17JU96 20JU96 23JU96 2JU96	37 37 37 38 37 37 37 37 37 37 37 37 37 37 37 37 37	1n		0 0 0 0 0 0	In 			
Tasiilaq Naroq Trawl Regina C Nocline C Kaassasu k Rolar Rasija Total 1997 Vessel Name		274/by96 8Jun96 18Jun96 17Jun96 9Mby96 3Sep96	Ot 4.1198 17.1198 20.1198 23.1198 23.1198 2.1.098 30.Spp98 Thip1 Ot	39 40 33 37 25 28 28 202 08 ₉ 6	in In In	L CL	0 0 0 0 0 0	In	0,1		
Tasilaq Nancq Traw Regina C Nocline C Kaassassik Rdar Paaja Total 1997 Vessel Name Tasilaq		274/by96 8Jun96 18Jun96 17Jun96 9Mby96 3Sep96 	Q1 4,4,49 17,449 23,449 23,449 23,449 24,149 30,58 24,149 30,59 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50,	99 40 33 37 25 28 20 20 20 20 20 20		Cut		-	Ot Trip3		Total Da
Tasiilaq Nanoq Trawl Regina C Nocline C Kaassasuk Polar Reaja Total 1997 Vessel Nano Tasiilaq Nanoq Trawl		274/by96 8Jun96 18Jun96 17Jun96 9Mby96 3Sep96	Q1 4,4,49 17,449 23,449 23,449 23,449 24,149 30,58 24,149 30,59 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50 50,50,	99 40 33 37 28 28 28 28 28 20 20 20 11	In	Cut		-	Ot Trip3		Total Da
Tasiilaq Nanoq Trawl Regina C Nocline C Kaassasuk Polar Reaja Total 1997 Vessel Nano Tasiilaq Nanoq Trawl		274/by96 8Jun96 18Jun96 17Jun96 9Mby96 3Sep96 	Q1 4,1149 17,1149 23,1149 23,1149 23,1149 23,1149 30,5p,96 Thip 1 Q1 5,1n49 23,1149 23,1149	99 40 33 37 25 28 20 20 20 20 20 20	In	Cu Trip2 Cu	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	-	Ot Trip3 Ot		Total Day
Tasiilaq Nanoq Trawl Regina C Nodine C Keassesuk Rolar Regia Total 1997 Vessel Name Tasiilaq Nanoq Trawl Total 1998	0140 0121 0122 0122 0122 0122 0122 0122	274/by96 8Jun96 18Jun96 17Jun96 9Mby96 3Sep96 	Q1 4,4,49 17,449 23,449 23,449 24,149 30,58 24,149 30,58 24,149 30,58 24,149 30,58 24,149 30,58 24,149 30,58 24,149 30,58 24,149 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,58 30,59 30,50 30,50 30,50 30,50 30,50 30,50 30,50 30,50 30,50 30,50 30,50 30,	99 40 33 37 28 28 28 28 28 20 20 20 11	In	Cut		-	Ot Trip3		Z Z Z Total Da
Tasilaq Nanoq Trawl Regina C Nodine C Keassessuk Rolar Raeja Total 1997 Vessel Name Tasilaq Nenoq Trawl Total 1998 Vessel Name	0140 0121 0122 0122 0122 0122 0122 0122	27-May96 8-Un96 18-Un96 9-May96 3-Sep96 17-Un96 9-May96 17-Vay97 17-Way97 13-U1-97	Qt 4,1148 17,1148 20,1	99 40 33 37 28 28 28 28 28 20 20 20 11	In	Cu Trip2 Cu		-	Ot Trip3 Ot		Total Day
Tasiilaq Nanoq Trawl Regina C Nodine C Keassesuk Rolar Regia Total 1997 Vessel Name Tasiilaq Nanoq Trawl Total 1998	0750 0752 0752 0752 0752 0752 0752 0752	27-May96 8-Un96 18-Un96 17-Un96 9-May96 3-Sep96 	Qt 4,1148 17,1148 20,1	39 40 33 37 28 28 202 08 5 202 11 11 31		Ct Trip2 Ct Trip2 Trip2			Ot Trip3 Ot Trip3		Total Da
Tasilaq Nanoq Trawl Regina C Nodine C Keassessuk Rolar Raeja Total 1997 Vessel Name Tasilaq Nenoq Trawl Total 1998 Vessel Name	07±0 07±7 07±2 07±2 07±2 07±0 07±0 07±1 07±1 07±1 07±1 07±1 07±1	27-May96 8-Un96 18-Un96 9-May96 3-Sep96 17-Un96 9-May96 17-Vay97 17-Way97 13-U1-97	Qt 4,1148 17,1148 20,1	39 40 33 37 25 202 08 5 202 202 202 202 202 202 202 202 202 2	ln In	Trip2 Cu Trip2 Cu Trip2 Cu	C C C C C C C C C C C C C C C C C C C		Ot Trip3 Ot Trip3		Total Da
Tasilaq Nanoq Trawl Regina C Nodine C Keassessuk Rolar Ragia Total 1997 Vessel Name Tasilaq Nenoq Trawl Total 1998 Vessel Name Rolar Amaroq	07±0 07XT 078Z 07CZ 07CZ 07CZ 07CZ 07CZ 07CZ 07CZ 07C	27-May96 8-Un96 18-Un96 9-May96 3-Sep96 17-Un96 9-May96 17-Unay97 17-Way97 17-Way97 13-U-97 1-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0	Qt 4,1148 17,1148 20,1	39 40 33 37 28 20 20 20 20 20 20 20 20 20 20 20 20 20	In In 29JLn98	Trip2 Cu Trip2 Cu Trip2 Cu	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		Ot Trip3 Ot Trip3		Total Da
Tasilaq Nanoq Trawl Regina C Nodine C Kaassessuk Rolar Paaja Total 1997 Vessel Name Tasilaq Nanoq Trawl Total 1998 Vessel Name Rolar Amaroq Regina C	0140 07XT 078Z 07CZ 07CZ 02KQ 02KQ 02KQ 07CZ 07CZ 07CZ 07CZ 07CZ 07CZ 07CZ 07CZ	27-May96 8-Un96 18-Un96 9-May96 3-Sep96 17-Un96 9-May96 17-Unay97 17-Way97 17-Way97 13-U-97 1-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0	Qt 4,1148 17,1148 20,1	39 40 33 37 28 28 28 28 28 28 28 28 28 20 20 20 20 20 20 20 20 20 20 20 20 20	In In 29JLn98	Cu Trip2 Cu Trip2 Cu 2Ag96	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		Ot Trip3 Ot Trip3 Ot		Total Da
Tasilaq Nanoq Traw Regina C Vodine C Kaassassuk Rotar Rasila Total 1997 Vessel Name Tasillaq Nanoq Traw Nanoq Traw Nanoq Traw Nanoq Traw Rasilaq Nanoq Traw Rasilaq Nanoq Traw Rasilaq	0140 07XT 078Z 07CZ 07CZ 02KQ 02KQ 02KQ 07CZ 07CZ 07CZ 07CZ 07CZ 07CZ 07CZ 07CZ	27-May96 8-Un96 18-Un96 9-May96 3-Sep96 17-Un96 9-May96 17-Unay97 17-Way97 17-Way97 13-U-97 1-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0	Qt 4,1148 17,1148 20,1148 20,1148 20,1148 20,1148 20,1148 20,1148 30,59p 20,1148 7,1148 20,1148 20,1148 31,1148	39 40 33 37 28 28 28 28 28 28 28 28 28 20 20 20 20 20 20 20 20 20 20 20 20 20	In In 29JLn98	Trip2 Cu Trip2 Cu Trip2 Cu	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		Ot Trip3 Ot Trip3		Total Da

Greenland -	Summary	1993-1	999
-------------	---------	--------	-----

Year	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec	TOTAL
1002					47.00	1050.00			1 40 01		<u> </u>	<u> </u>	
1993					47.85	1859.02	1460.54	242.03	160.81	9.75	<u> </u>		
1994		<u> </u>		80.39	375.71	854.36	689.49	165.68	106.37	 	ļ		
1995	 			<u> </u>	279.07	933.04	1003.72	100.17			_	<u> </u>	
1996				<u> </u>	191.29	466.85	392.86	47					
1997`					44.25	14.75	46				1		
1998					133.89	262.60	448.77	16.74					
1999				<u> </u>	115.66	231.32	190.02	-					

Annex 12. Compilation of Shrimp 3M Catches and Effort Days for 1993-1999 (STACTIC Working Paper 00/8 - NAFO Secretariat)

<u>NOTE</u>: This is confidential information from Contracting Parties and not for public release.

Submissions as received from Contracting Parties up to June 27, 2000 indicating revised catches and efforts days for the shrimp fishery in 3M.

Denmark (Faroe Islands)

Year	No. Vessels*	Fishing Days	Catch, tonnes
1993	9 ·	1.324	7.333
1994	10	1.785	6.791
1/1-31/8 1995	7	705	4.228
1995	7	1.093	5.993
1996	10	1.831	8.688
1997	6	1.250	7.410
1998	7	1.292	9.368
1999	6	1.051	9.199

3M Shrimp Catch and Effort, 1993-1999

* The number of different vessels 1/1-1993 to 31/8-1995 was 11.

3L shrimp catch, 1993-1999

Year	Catch, tonnes ¹⁾
1993	1.789
1994	356
1995	
1996	79
1997	485
1998	515
1999	700

¹⁾Catches in 1994 and following years are in connection with research fishery.

Denmark (Greenland)

3M Shrimp Ca	tch and Effort,	1993-1999
--------------	-----------------	-----------

			3M Shrimp	Catch/Effort	1993-1999						
1993			Trip1	1		Trip 2			E chT		
Vessel Name	R/C	tn	Out	Davs	In	Out	Days	tn	Out	Days	Total Days
Timmiarmiut	OUKV	4-Jun-93	16-Jul-93	43						0	43
Jesper Belinda	0000	28-May-93	13-Jun-93	17	16-Jun-93	26-Jul-93	41	7-Aug-93	15-Aug-93	9	67
Tasarmiut	owau	31-May-93	4-Jul-93	35	7 Jul 93	20-Jul-93	14			0	49
Polar Princess II	OWTI	26-Jun-93	4-Sep-93	71	7-Sep-93	14-Sep-93				0	79
Killin	OWVM	30-Aug-93	4-Sep-93	6	8-Sep-93	3-Oct-93	26				32
Tunnulik	OYCK	29-May-93	15-Jun-93	18	24-Jun-93	7-Ju⊢93	14			0	32
Tasiilag	OYHO	31-May-93	1-Aug-93	63	24 001 00		0			0	63
Qipoqqaq	ОҮКК	8-Jun-93	9-Ju⊦93	32		+					32
Betty Belinda	OYRT	8-Jun-93	7-Jul-93	30			0			0	32
Nanog Trawl	OYXT	1-Jun-93	22 Jul-93	52							
Anso Mølgård	OYZL	7-Jun-93	7-Jul-93	31	10 64 00	1 1	23			0	52
					10-Jul-93	1-Aug-93				0	54
Kaassassuk	OZKO	8-Jun-93	1 <u>6-Jul-93</u>	39			0			0	39
Total				437			126			9	572
1994	210		tqh7			Trip 2	ł		Trip 3		
Vessel Name	R/C	In	Out	Days	<u>In</u>	Out	Days	ln	Out	Days	Total Days
Timmiarmiut	OUKV	29-May-94	9-Jul-94	42			0			00	42
Tasermiut	OWOU	23-May-94	4-Jul-94	43			0			0	43
Polar Princess II	OWTI	7-Jul-94	27-Sep-94	83			0			Ö	83
Regina C	OYBZ	26-Jun-94	8-Jul-94	13			0		1	Ö	13
Tasiilaq	ОҮНО	30-May-94	14-Jul-94			ľ	0		_	0	46
Betty Belinda	OYRT	29-Jun-94	20-Jul-94	22			0			0	22
Anso Mølgård	OYZL	7-Apr-94	15-May-94	39	19-May-94	3-Jul-94	46	7-Jul-94	13-Aug-94	38	123
Nuuk	OZDH	1-May-94	2-Jun-94	33	6-Jun-94	19-Jul-94	44			0	77
Kaassassuk	OZKO	12-Jun-94	14-Jul-94	33			0			0	33
Total				354			90		- î	38	482
1995			Trio1			Tria 2			Trio 3		
Vessel Name	RVC	In	Out	Days	In	Out T	Days	In	Out	Days	Total Days
Kiliutao	OWGG	22-May-95	23-Jun-95	33	27-Jun-95	4-Aug-95	39			0	72
Tasermiut	OWOU	30-May-95	2-Jul-95	34			0			ő	34
Tasiilag	ОҮНО	23-Jun-95	20-Jul-95	28			0			Ő	28
Betty Belinda	OYRT	25-Jun-95	30-Jun-95	6			Ő			0	6
Nanog Trawl	ΟΥΧΤ	14-Jun-95	27-Jul-95	44			0			0	44
Nuuk	OZDH	15-May-95	22-Jun-95	39	26-Jun-95	6-Aug-95	42			0	81
					20-301-331	07/00-301			-		
				184			011			0	
Total				184			81			0	265
1996			Tdp1			Trip 2			Trip 3		
1996 Vessel Name	R/C	In	Out	Days		Trip 2 Out	Days	In	Trip 3 Out	Days	Total Days
1996 Vessel Name Tasiilaq	R/C OYHO	In 27-May-96	Out 4-Jul-96	Days 39	In		Daγs 0	 In		Days_0	Total Days 39
1996 Vessel Name Tasiilaq Nanoq Trawl	R/C OYHO OYXT	In 27-May-96 8-Jun-96	Out 4-Jul-96 17-Jul-96	Days 39 40	In		Daγs 0 0	 In		Days_0	Total Days 39 40
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C	R/C OYHO OYXT OYBZ	In 27-May-96 8-Jun-96 18-Jun-96	Out 4-Jul-96 17-Jul-96 20-Jul-96	Days 39 40 33	<u>In</u>		Daγs 0 0 0	ln		Days_ 0 0	Total Days 39 40 33
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicoline C	R/C OYHO OYXT OYBZ OYCZ	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96	Days 39 40 33 37			Daγs 0 0 0 0	In		Days0 0 0 0	Total Days 39 40 33 37
1996 Vessel Name Tasiilag Nanog Trawl Regina C Nicoline C Kaassassuk	R/C OYHO OYXT OYBZ OYCZ OZKQ	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 9-May-96	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 2-Jun-96	Daγs 39 40 33 37 25			Daγs 0 0 0 0 0 0	In		Days_0 0 0 0	Total Days 39 40 33 37 25
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicotine C Kaassassuk Polar Raaja	R/C OYHO OYXT OYBZ OYCZ	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 2-Jun-96	Daγs 39 40 33 37 25 28			Daγs 0 0 0 0 0 0 0 0	in		Days_0 0 0 0 0	Total Days 39 40 33 37 25 28
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicotine C Kaassassuk Polar Raaja Total	R/C OYHO OYXT OYBZ OYCZ OZKQ	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 9-May-96	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 2-Jun-96 30-Sep-96	Daγs 39 40 33 37 25		Out	Daγs 0 0 0 0 0 0	in	Out	Days_0 0 0 0	Total Days 39 40 33 37 25
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicoline C Kaassassuk Polar Rasia Total 1997	R/C OYHO OYXT OYBZ OYCZ OZKQ OUPJ	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 9-May-96 3-Sep-96	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 2-Jun-96 30-Sep-96 Trip1	Days 39 40 33 37 25 28 202		Out	Days 0 0 0 0 0 0 0 0		Out	Days0 0 0 0 0 0 0 0	Total Days 39 40 33 37 25 28 202
1996 Vessel Name Tasiilag Nanog Trawl Regina C Nicoline C Kaassassuk Polar Raaja Total 1997 Vessel Name	R/C OYHO OYXT OYBZ OYCZ OZKQ OUPJ R/C	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 9-May-96 3-Sep-96	Out 4-Jul-96 17-Jul-96 20-Jul-96 2-Jul-96 2-Jun-96 30-Sep-96 Trlp1 Out	Days 39 40 33 37 25 28 202 Days		Out	Daγs 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ln 	Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 39 40 33 37 25 28 202 202 Total Days
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicotine C Kaassassuk Polar Raaia Total 1997 Vessel Name Tasiilaq	R/C OYHO OYXT OYBZ OYCZ OZKQ OUPJ R/C OYHO	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 9-May-96 3-Sep-96 In 17-May-97	Out 4-Jul-96 17-Jul-96 20-Jul-96 2-Jun-96 30-Sep-96 Trlp1 Out 5-Jun-97	Days 39 40 33 37 25 28 202 Days 20		Out	Daγs 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 39 40 33 37 25 28 202 202 Total Days 202 202 202 202 202 202 202 202 202 20
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicotine C Kaassassuk Polar Raaia Total 1997 Vessel Name Tasiilaq Nanoq Trawl	R/C OYHO OYXT OYBZ OYCZ OZKQ OUPJ R/C	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 9-May-96 3-Sep-96	Out 4-Jul-96 17-Jul-96 20-Jul-96 2-Jul-96 2-Jun-96 30-Sep-96 Trlp1 Out	Days 39 40 33 37 25 28 202 Days 202 Days 20 11		Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 39 40 33 37 25 28 202 Total Days 200 11
1996 Vessel Name Tasiilaq Nanoq Trawl Reginą C Nicoline C Kaassassuk Polar Raaja Total 1997 Vessel Name Tasiilaq Nanoq Trawl Total	R/C OYHO OYXT OYBZ OYCZ OZKQ OZKQ OUPJ R/C OYHO OYHO OYXT	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 9-May-96 3-Sep-96 In 17-May-97	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 2-Jun-96 30-Sep-96 Trlp1 Out 5-Jun-97 23-Jul-97	Days 39 40 33 37 25 28 202 Days 20		Out Trip 2 Out	Daγs 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		Out - Trtp 3 Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 39 40 33 37 25 28 202 202 Total Days 202 202 202 202 202 202 202 202 202 20
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicoline C Kaassassuk Polar Raaja Total 1997 Vessel Name Tasiilaq Nanoq Trawl Total 1998	R/C OYHO OYXT OYBZ OYCZ OZKQ OZKQ OZKQ OZKQ OZKQ OZKQ OYHO OYXT	In 27-May-96 8-Jun-96 17-Jun-96 17-Jun-96 3-Sep-96 3-Sep-96 In 17-May-97 13-Jul-97	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 30-Sep-96 Trip1 Out 5-Jun-97 23-Jul-97 723-Jul-97	Days 39 40 33 37 25 28 202 Days 202 11 31		Out Trip 2 Out Trip 2	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	in	Out Trtp 3 Out Trtp 3	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 39 40 33 37 25 28 202 Total Days 20 11 31
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicoline C Kaassassuk Polar Raaja Total 1997 Vessel Name Tasiilaq Nanoq Trawl Total	R/C OYHO OYXT OYBZ OYCZ OZKQ OVCZ OZKQ OYHO OYHO OYHO OYXT R/C	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 3-Sep-96 3-Sep-96 In 17-May-97 13-Jul-97	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 23-Jul-96 30-Sep-96 Trlp1 Out 5-Jun-97 23-Jul-97 7-10-17 7-10-17 0ut	Days 39 40 333 37 25 28 202 Days 20 11 31 31 Days		Out Trip 2 Out Trip 2 Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		Out - Trtp 3 Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 39 40 33 37 25 26 202 Total Days 70 11 31 70 40 70 40 40 40 40 40 40 40 40 40 40 40 40 40
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicoline C Kaassassuk Polar Raaja Total 1997 Vessel Name Tasiilaq Nanoq Trawl Total 1998	R/C OYHO OYXT OYKZ OYCZ OZKQ OUPJ R/C OYHO OYHO OYYT R/C OZMA	In 27-May-96 8-Jun-96 17-Jun-96 17-Jun-96 3-Sep-96 3-Sep-96 In 17-May-97 13-Jul-97	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 2-Jun-96 30-Sep-96 Trlp1 Out 5-Jun-97 23-Jul-97 Trlp1 Trlp1 Out 25-Jun-98	Days 39 40 33 37 25 28 202 Days 202 11 31		Out Trip 2 Out Trip 2	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	in	Out Trtp 3 Out Trtp 3	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 39 400 33 37 22 202 202 Total Days Total Days Total Days 76
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicotine C Kaassassuk Polar Raaia Total 1997 Vessel Name Tasiilaq Nanoq Trawl Total 1998 Vessel Name	R/C OYHO OYXT OYBZ OYCZ OZKQ OVCZ OZKQ OYHO OYHO OYHO OYXT R/C	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 3-Sep-96 3-Sep-96 In 17-May-97 13-Jul-97	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 2-Jun-96 30-Sep-96 Trip1 Out 5-Jun-97 23-Jul-97 Trip1 Out 25-Jun-98	Days 39 40 333 37 25 28 202 Days 20 11 31 31 Days		Out Trip 2 Out Trip 2 Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	in	Out Trtp 3 Out Trtp 3	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 36 4(4 33 33 22 20 20 20 20 20 20 20 20 20 20 20 20
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicotine C Kaassassuk Polar Asi 1997 Vessel Name Total 1998 Nanoq Trawl Total 1998 Vessel Name Polar Amaroq	R/C OYHO OYXT OYKZ OYCZ OZKQ OUPJ R/C OYHO OYHO OYYT R/C OZMA	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 3-Sep-96 3-Sep-96 3-Sep-96 17-May-97 17-May-97 13-Jul-97 1-16-May-98	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 2-Jun-96 30-Sep-96 Trlp1 Out 5-Jun-97 23-Jul-97 Trlp1 Trlp1 Out 25-Jun-98	Days 39 400 333 37 25 28 202 202 202 202 111 31 31 Days 41		Out Trip 2 Out Trip 2 Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	in	Out Trtp 3 Out Trtp 3	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 36 4 (3 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicoline C Kaassassuk Polar Rasia Total 1997 Vessel Name Tasiilaq Nanoq Trawl Total 1998 Vessel Name Polar Amaroq Regina C	R/C OYHO OYXT OYBZ OYCZ OZKQ OUPJ R/C OYHO OYXT R/C OZMA OYBZ	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 3-Sep-96 3-Sep-96 3-Sep-96 17-May-97 17-May-97 13-Jul-97 1-16-May-98	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 2-Jun-96 30-Sep-96 Trlp1 Out 5-Jun-97 23-Jul-97 Trlp1 Trlp1 Out 25-Jun-98	Days 39 39 202 202 Days 200 11 31 Days 203 200 31 31 31 31 31 37		Out Trip 2 Out Trip 2 Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	in	Out Trip 3 Out Trip 3 Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 36 4(4 3) 22 20 Total Days 7(11 3) Total Days 7(3) 3
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicoline C Kaassassuk Polar Raaia 1997 Vessel Name Tasiilaq Nanoq Trawl Total 1998 Vessel Name Polar Amaroq Regina C Total	R/C OYHO OYXT OYBZ OYCZ OZKQ OUPJ R/C OYHO OYXT R/C OZMA OYBZ	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 3-Sep-96 3-Sep-96 3-Sep-96 17-May-97 17-May-97 13-Jul-97 1-16-May-98	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 23-Jul-96 30-Sep-96 30-Sep-96 30-Sep-97 23-Jul-97 23-Jul-97 23-Jul-97 7rlp1 Out 25-Jun-98 31-Jul-98 31-Jul-98	Days 39 39 202 202 Days 200 11 31 Days 203 200 31 31 31 31 31 37		Out Trip 2 Out Trip 2. Out 2-Aug-98	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	in	Out Trtp 3 Out Trtp 3	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 36 44 33 33 22 28 20 20 20 20 20 20 20 20 20 20 20 20 20
1996 Vessel Name Tasiilaq Nanoq Trawl Regina C Nicotine C Kaassassuk Polar Raaia Total 1997 Vessel Name Tasiilaq Nanoq Trawl Total 1998 Vessel Name Polar Amaroq Regina C Total 1999	R/C OYHO OYXT OYRZ OYRZ OYCZ OZKQ OUPJ R/C OYHO OYHT R/C OZMA OYBZ	In 27-May-96 8-Jun-96 18-Jun-96 17-Jun-96 3-Sep-96 3-Sep-96 17-May-97 13-Jul-97 13-Jul-97 13-Jul-97 16-May-98 25-Jun-98	Out 4-Jul-96 17-Jul-96 20-Jul-96 23-Jul-96 23-Jul-96 2-Jun-97 30-Sep-96 Trlp1 Out 5-Jun-97 23-Jul-97 Trlp1 Out 25-Jun-98 31-Jul-98 31-Jul-98 Trlp1	Days 39 400 333 37 25 28 202 202 202 202 11 31 31 203 5 200 11 31 31 37 78	In	Out Trip 2 Out Trip 2 Out 2-Aug-98 Trip 2	Days C C C C C C C C C C C C C C C C C C C	in In	Out Trip 3 Out Trip 3 Out Trip 3 Out	Days 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Total Days 36 4(4) 33: 22: 22: 22: 22: 22: 21: Total Days Total Days 77: 77: 71: Total Days

Greenland - Summary 1993-1999

Year	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec	TOTAL
1993				<u></u>	47.85	1859.02	1460.54	242.03	160.81	9.75			
1994	-			80.39	375.71	854.36	689.49	165.68	106.37		1		
1995					279.07	933.04	1003.72	100.17					
1996					191.29	466.85	392.86	47					
1997					44.25	14.75	46						
1998					133.89	262.60	448.77	16.74					
1999					115.66	231.32	190.02						

<u>Estonia</u>

3M Shrimp Catch and Effort, 1993-1999

·	1993			1994			1995			19	996	
Days	No. of		Days	No. of	0.1	Days	No. of	Catch	Days	Days	No. of	
Used	Vessels	Catch	Used	Vessels	Catch	Used	Vessels		Allocated	Used	Vessels	Catch
149	_1	268	609	4	1051	2153	9	2379	1852	990	5	1898
		•				1	Up to 31 Aug	ust				
						Days	No. of					
						Used	Vessels	Catch				
						1852	9	1654				

	19	997			199	8			199	9	
Days	Days	No. of		Days	Days	No. of		Days	Days	No, of	
 Allocated	Used	Vessels	Catch	Allocated	Used	Vessels	Catch	Allocated	Used	Vessels	Catch
1217	1254	6	3240	1217	1454	7	5533	1667	1651	9	10834

<u>Iceland</u>

3M Shrimp Catch and Effort, 1993-1999

June July August Splember October November December 0 119296 0 <th></th>	
July August Sptember October 119296 0 0 0 0 119296 0 0 0 0 0 119296 0 0 0 0 0 0 138569 0 0 0 0 0 0 0 17111 155277 159653 0 0 0 0 0 152812 161223 0 140996 248116 0	
July August Sptember October 119296 0 0 0 0 119296 0 0 0 0 0 119296 0 0 0 0 0 0 138569 0 0 0 0 0 0 0 17111 155277 159653 0 0 0 0 0 152812 161223 0 140996 248116 0	
July August July August 119296 0 138569 0 17111 155277 123812 161223 123812 161223 128098 457496 788098 457496 788098 457496 83354 0 83354 0 245771 137945 0 92270 0 78605 81904 87079 0 78605 0 74605 0 74605 0 745145 0 74517 0 74517 0 74517 0 74517 0 74517 0 168233 0 168233 0 168823 0 168233 0 163823	
July August July August 119296 0 138569 0 171111 155277 122812 161223 152812 161223 152812 161223 205310 14096 788098 457496 83254 0 83354 0 245771 137945 0 92270 0 92270 0 78605 81904 87079 0 78605 0 78605 0 16823 0 16823 0 16823 0 16823 0 16823 0 16823 0 16823 0 16823 0 16823	<u> </u> -
┡╼╄╼╄╾╄╌╂╶╎╴┠╶┧╴╄╍┾╍╄╌┞╼┦╌╂╼┨╴	
May 000000000000000000000000000000000000	
March March 0	
February 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
January 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2355091
No. No. 2506 659 669 669 669 669 669 669 669 669 66	
NS REVE EN SIN SIN SIN SIN SIN SIN SIN SIN SIN SI	
Vessels name. Skutull Jón Finnsson Riakon Pétur Jónsson Suma Total catch 1993: Total catch 1993: Total catch 1993: Total catch 1993: Total catch 1993: Total catch 1993: Total catch 1993: Amarres Skutul Jón Finnsson Bunna Hafrafell Baldur Andvari Pétur Jónsson Pétur Jónsson	Total catch 1994:
Year Reg. No. 1993 1742 1993 1383 5993 1742 1993 1383 5051 5993 1507 11 1993 1807 11 1993 1809 11 1993 2061 5994 1128 / 7 1994 1128 / 7 1994 12306 1994 22061 59 1994 22061 50 1994 22016 1994 22016 1994 22016 1994 22016 10994 22018 / 700000000000000000000000000000000000	
Year 1993 1993 1993 1994 1994 1994 1994	

46216	0	S6768	0	0	ð	0	0	0	0	0	0	0	0	0	84505	0	103691	72524	83125	56887	503716	
50857	c	12499	0	0	0	0	c	0	0	0	0	0	56852	0	111550	c	96196	134301	0	0	462255	
48048	0	0	118275	0	0	0	0	0	0	0	0	90£6L	209626	0	137208	0	218953	105156	0	0	916572	
59314	0	0	0	0	142486	0	63733	84773	0	0	0	0	93440	13655	148459	0	0	104419	0	0	710279	
59862	73665	0	77204	37752	119743	c	58819	84608	90420	c	104369	106020	82496	177093	696861	194667	136310	144530	0	0	1746527	
74008	87458	0	111320	0	174928	138724	0	0.	210975	100880	212424	0	103646	0	0	0	0	160001	0	0	1374364	
21188	0	0	0	0	129817	130017	0	0	252501	92393	0	94378	100559	0	0	0	0	0	0	0	870853	
59629	0	¢	79541	0	185939	0	0	0	0	¢	0	25000	102082	0	205830	0	0	0	0	0	658021	
64110	0	0	0	0	0	0	0	0	0	0	0	0	0	0	174336	0	0	0	0	0	238446	
0	0	0	0	0	c	0	0	0	0	c	0	0	0	0	0	0	0	0	0	0	0	
0	Ð	0	0	0	0	0	0	0	c	0	0	0	0	0	0	0	0	0	0	0	0	
0	0	0	0	0	0	0	0	0	0	0	0	c	0	0	0	0	0	0	0	0	0	7481033
70	180	8	45	15	373	172	30	2	67	383	117	222	100	69	96	7	317	50	204	101		
SI	S	SU	IS	DII	RE	SI	OF	EA	IS	SU	NK	IS	VE	RE	NS	EA	FΑ	SU	ΒA	ΒΛ		
Arnames	Skutull	Klettur	Guðmundur Péturs	Nökkvi	Helga II	Jöfur	Sigurfari	Bliki	Sunna	Brimir	Blængur	Hafrafell	Andvari	Pétur Jónsson	Otto Wathne	Svalbakur	Dalborg	Klara Sveinsdóttir	Erik	Kan	Total	Total catch 1995:
1128	1383	1626	1753	1768	1903	1905	1916	1942	2061	2155	2197	2204	2211	2216	2218	2220	2237	2244	2258	2259		
1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995	1995		

۰

۰.

··· · - · - - -

167

.

0	18382	0	0	0	•	0	-	•	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	165025	0	80770.	83889	55860	58983	0	0	0	9836	572745	٦
Н	-		_					_				-				_	-		_	_						-			-				_	_			_		-	_
0	148616	0	0	0	0	0	0	0	0	0	c	0	0	•	0	0	•	0	31789	71240	0	0	0	0	68826	122847	c	0	0	0	59367	88818	79551	54467	0	0	•	46091	771612	
62335	124520	0	0	0	51685	0	143284	0	0	0	. 0	0	86967	•	48957	56899	54434	0	103153	144754	0	0	28880	0	0	197409	0	0 -	113671	0	71961	92544	87389	146802	159581	106414	61470	77251	2020360	
48956	0	0	57801	0	55839	0	66914	0	0	0	- 0	0	85980	c	65221	110819	0	0	1.32456	0	52469	0	30227	0	100258	229133	0	0	0	0	0	0	83229	0	98525	0	0	69181	1287008	
92636	136989	54769	57128	0	57105	0	113448	0	0	0	0	0	72802	67940	67937	79039	71202	0	141395	0	65734	0	24377	0	99555	0	0	0	144713	117576	106033	130369	0	90243	0	110712	73000	0	1974702	
103431	119821	43656	53112	0	136601	83165	96985	158405	0	0	97.194	80094	52864	96794	78678	71876	155154	49936	118420	189414	76869	148177	0	66408	95254	235950	205784	0	99461	205264	142226	154815	153604	108672	104735	107832	0	0	3690651	
141476	194832	112744	120820	0	107336	82835	117313	122965	0	142622	88207	59913	85910	0	83767	99504	67318	80096	0	170724	100887	167546	23141	113451	120253	315065	195823	0	229875	249181	149496	121583	177185	132043	94886	103193	0	0	4187902	
0	0	100296	48014	95030	12162	105015	98825	0	0	117619	0	0	0	0	75771	86110	61742	89196	c	164749	18662	0	58418	100235	87471	249445	193180	0	0	0	105666	0	112071	102225	0	138926	0	0	2349156	
0	0	0	0	139192	0	105122	0	86898	0	155709	c	0	0	c	86344	113683	c	92966	0	170305	c	c	0	0	87308	308725	250357	0	0	271858	0	141350	0	0	0	0	0	0	2009817	
0	0	0	0	0	0	0	• 0	0	0	96073	0	0	0	0	95824	0	0	108203	0	204017	0	0	40000	0	0	0	0	0	0	0	0	176468	169503	161330	0	0	0	0	1051418	
0	0	0	0	0	0	0	0	0	31229	0	0	0	0	0	0	0	0	0	0	201510	0	0	0	0	0	0	c	150353	0	0	90670	0	6179	102992	0	0	0	0	676533	
0	0	.0	0	0	0	0	0	0	0	.0	0	0	0	0	0	0	0	0	0	90328	0	0	0	0	0	0	¢	0	0	0	0	0	0	0	0	0	0	0	90328	20682232
307	302	150	6	710	4	310	0	360	8	70	4	9	45	201	ŝ	172	g	12	410	67	383	117	222	72	1001	\$	69	ß	19	1	317	50	204	101	[_	~	316	4		
EA	SI	SI			S	EA	Hđ.	Hđ	SU	ST	ÅR	ÁRI	<u> </u>	HS	Ē	S	G	EA		SI		Ж	S	ÓР	Ë		<u> </u>	SZ		EA		SU	BA		þH	E	E	RE		
Sólbakur	Svalbarði	Siglfirðingur	bórunn Havsteen	Margrét	I leiðrin	IIjalteyrin	Kolbeinsey	Stakfell	Kleitur	Hólmadrangur	l lersir	Gissur	Guðmundur Péturs	Hamra-Svanur	Nökkvi	Jöfur	Sigurfari	Bliki	ßessi	Sunna	Brimir	Blængur	Hafrafell	Hvannaberg	Andvari ·	Guðbjörg	Pétur Jónsson	Otto Wathne	Snæfell	Svalbakur	Dalborg	Klara Sveinsdottir	Erik	Kan	Július Havsteen	Helga Björg	Arnarborg	Andenes	Total	Total catch 1996:
1276	1352	1407	1462	1484	1506	1514	1576	1609	1626	1634	1742	1752	1753	1757	1768	1905	1916	1942	Γ	2061		2197	2204	2206	t	Ì	T	2218			<u> </u>	1	1	1	1	_	1-	2271	1-	
9661	9661	1996	1996	9661	9661	1996	9661	1996	9661	1996	1996	9661	1996	1996	9661	9661	1996	1996	9661	9661	1996	9661	9661	1996	9661	1996	9661	9661	1996	1996	1996	9661	1996	1996	1996	1996	1996	9661		

Regn.no.	Vessels name	ln	Out	Days	Port of unloading	Catch(kg)	Total Catch	Catch pr. d
2288	Pétur Jónss. RE-69	20-May	18-Jun	30	Argentia	201,570		6.71
2288	Pétur Jónss, RE-69	23-Jun	26-Jul	34	Hafnartjörður	313,770		9,22
				64		\$15,340	515,340	8,05
1352	Svalbarði SI-302	27-Jul	24-May	28	Harbour Grace	114,100		4,075
1352	Svalbaroi SI-302	1-Jun	28-Jun	28	Argentia	123,789		4,421
1352	Svalbarði SI-302	6-Jul	10-Aug	36	Harbour Grace	193,037		5,362
1352	Svalbarði SI-302	19-Aug	14-Sep	30	Argentia	146,051	<u> </u>	4,868
1352	Svalbarði 51-302	21-Scp	19-Oct	29	Harbour Grace	138,634		4,780
1352	Svalbarði SI-302	24-Oct	10-Nov	18	Harbour Grace	66,470		3,693
1352	Svalbarði SI-302	17-Nov	14-Dec	28	Siglufjörður	101,421		3,622
				197		883,502	883,502	4,485
2258	Erik BA-101	l 2-Jan	27-Jan	16	Argentia	0		·
2258	Erik BA-101	30-Jan	22-Feb	27	Argentia	125,498		4,64
		ļ		43		125,498	125,498	2,91
2013	Bessi IS-410	18-Jun	22-Jul	35	Argentia	185,761	+	5,307
2013	Bessi IS-410	27-Jul	26-Aug	31	Argentia	149,041		4,808
2013	Bessi IS-410	2-Sep	30-Sep	29	İsafjörður	155,624	400.42(5,366
		+		95		490,426	490,426	5,162
2061	Sunna SI-67	28-Apr	29-May	32	Argentia	174,792		5,462 7,403
2061	Sunna SI-67	5-Jun	Z-Jul	28	Argentia	207,270	- <u>}</u> }	
2061	Sunna SI-67	9-Jul	4-Aug	27 87	Siglufjörður	173,806	555,868	6,437
		10.1.1	20 1			149,110	555,000	4,518
1383	Skutull IS-180	. 19-Jul	20-Aug	33	Isafjörður	149,110	149,110	4,518
	0. () () (740	P. Maria	11 1	35	Hostowe Canan	160,906	147,110	4,597
2218	Snæfell SH-740	8-May	11-Jun 15-Jul	31	Harbour Grace Harbour Grace	186,410		6,013
2218	Snæfell SH-740	15-Jun		34		181,355		5,334
2218	Snæfell SH-740	21-Jul	23-Aug 15-Oct	34	Harbour Grace Harbour Grace	80,940		2,188+
2218	Snæfell SH-740	9-Sep	21-Nov.	32 -	Ólafsvík	337,857		10,558
2218	Snæfell SH-740	20-Oct	21-1104.	169	UIGISYIN	947,468	947,468	5,606
2286	Bliki EA-12	28-May	15-Jun	24		0	211,100	
2286	Bliki EA-12	20-Jun	28-Jun	9	Argentia	86,400		
2286	Bliki EA-12	4-Jul	5-Aug	33	Argentia	161,300		
2286	Bliki EA-12	.' Aug	14-Sep	35	Dalvik	155,600		
2280	UNKLATI	<u> </u>		101		403,300	403,300	3,993
2197	Blængur NK-117	8-Jun	12-Jul	35	Argentia	201,668	· · · · · · · · · · · · · · · · · · ·	5,762
2197	Blængur NK-117	18-Jul	19-Aug	33	Neskaupsstaður	183,719		5,567
	5.00 B			68		385,387	385,387	5,667
1628	Slettanes IS-808	15-Jul	31-Jul	17	Ť	0		0
1628	Slettanes IS-808	7-Aug	24-Aug	18	İsafjörður	153,425		8,524
				35		153,425	153,425	4,384
1216	Húsvíkingur ÞH-1	22-Aug	22-Sep	32	Argentia	123,143		3,848
1216	Húsvikingur ÞH-1	28-Sep	25-Oct	28	Akureyri	296,260		10,581
				60		419,403	419,403	6,990
2206	Hvannaberg OF-72	28-Apr	5-Jun	39	Olafsfjörður	123,919	122.222	3,177
		+ <u> </u>	10.17	39 20	Arunchin	103,058	123,919	3,177
2211 2211	Andvari VE-100 Andvari VE-100	21-Apr 17-May	10-May 8-Jun	20	Argentia Argentia	103,058	- <u> </u>	4,436
2211 2211	Andvari VE-100 Andvari VE-100	15-Jun	5-Jul	23	Argentia	113,261	1	5,393
2211	Andvari VE-100	12-Jul	1-Aug	21	Argentia	116,514		5,548
2211	Andvari VE-100	9 Aug	29-Aug	21	Argentia	115,227		5,487
2211	Andvari VE-100	5-Sep	26-Sep	22	Argentia	101,186	}	4,599
2211	Andvari VE-100	2-001	24-Oct	23 151	Argentia	99,575 750,838	750,838	4,329
7360	Von De 101	15-Jan	27-Jan	0	. 0	0	100,020	0
2259	Kan BA-101	30-Jan	13-Feb	0	0	0	+	0.
2259	Kan BA-101	15-Feb	25-Feb	38	Argentia	81,440	+	2,143
2259	Kan BA-101 Kan BA-101	20-Apr	23-760 28-May	38	Argentia	113,000	<u>†</u> ───┤	2,897
2259	Kan BA-101 Kan BA-101	3-Jun	25-Jun	23	0	0	1	0
2259	Kan BA-101	28-Jun	12-Jul	15	Harbour Grace	100,705	1 1	6,714
2259	Kan BA-101	29-Jul	1-Sep	35	Harbour Grace	132,100		3,774
2259	Kan BA-101	17-Scp	7-Oct	21	0	0	1	0
2259	Kan BA-101	9-Oct	Z2-Oct	14	Argentia	142,500	1 1	10,179
	Nan OA-TOT	+		185		569,745	569,745	3,080

;

1

į

,

Rcgn.no.	Vessels name	In	Out	Days	Port of unloading	Catch(kg)	Total Catch	Catch pr. day
2288	Pétur Jónss, RE-69	II-May	6-Jun	27	Argentia	306,431	Total Calcin	11,349
2288	Pétur Jónss, RE-69	Jun	8-Jul	28	Argentia	377,177		3,471
2288	Pétur Jónss, RE-69	12-Jul	8-Aug	28	Argentia	267,714		9,56
2288	Pétur Jónss, RE-69	13-Aug	7-Sep	23	Argentia	235,159		9,045
2288	Pétur Jónss, RE-69	12-500	16-Oct	35	Argentia	217,771		6,222
	Tetal Jons, ICC-177	12-340	hrou	144	Argenna	1,404,252	1,404,252	9,752
1352	Svalbarði S1-302	19-Fcb	16-Mar	26	Harbour Grace	177,216	1,-10-1,232	6,816
1352	Svalbarði \$1-302	23-Mar	20-Apr	29	Harbour Grace	221,771		7,647
1352	Svalbarði SI-302	25-Apr	25-May	31	Harbour Grace	224,748	†	7,250
1352	Svalbarði SI-302	31-May	13-Jun	14	Harbour Grace	102,139		7,296
1352	Svalbarði SI-302	22-Jun	19-Jul	28	Harbour Grace	231,208		8,257
1352	Svalbarði SI-302	26-Jul	24-Aug	30	Harbour Grace	179,951		5,998
1352	Svalbarði SI-302	30-Aug	1-Sep	3	Harbour Grace	0	<u> </u>	0
1352	Svalbarði SI-302	7-Sep	5-Oct	29	Harbour Grace	155,451		5,360
				190	Autoral Chart	1,292,484	1,292,434	6.803
2190	Evborg EA-59	16-Mav	8-Jun	24	Argentia	89,483	.,	3,728
2190	Eyborg EA-59	18-Jun	12-Jul	25	Argentia	100,821		4,033
2190	Eyborg EA-59	18-Jul	25-Jul	8	St. Jhons	0		
2190	Eyborg EA-59	28-Jul	18-Aug	22	Akaureyri	134,913		6,132
	<u> </u>			79		325,217	325,217	4,117
2216	Húsvíkingur ÞH-I	12-May	13-Jun	33	Argentia	364,165		11,035
2216	Húsvíkingur ÞH-I	20-Jun	19-Jul	30	Bay Roberts	386,463		12,882
2216	Húsvíkingur ÞH-1	24-Jul	26-Aug	34	Hafnarfiðrður,	303,566		8.928
	2			97		1,054,194	1,054,194	10,868
2061	Sunna SI-67	7-Sep	5-Oct	29	Argentia	188,157		6,488
2061	Sunna SI-67	10-Oci	16-Nov	38	Siglufjörður	255,290		6,718
				67		443,447	443,447	6,619
1609	Stakfell PH-360	22-May	24 Jun	34	lsafjörður	181,033		5,325
				34		181.033	181,033	5,325
2218	Snæfell SH-740	7-Sep	11-Oct	35	Harbour Grace	174,939		4,998
2218	Snæfell SH-740	18-Oct	17-Nov	31	Harbour Grace	95,964		3,096
2218	Snæfell SH-740	21-Nov	15-Dec	25	Reykjavík.	189,102		7,564
				91		460,005	460,005	5,055
2242	Oni İS	7-Sep	6-Oct	30	Argentia	0	· · ·	
2242	Oπi İS	10-Oct	8-Nov	30	Argentia	209,402		6,980
2242	Orri İS	14-Nov	16-Dec	33	Ísafjörður	298,858		9,056
				93		508,260	508,260	5,465
2279	Lómur HF-177	25-May	24-Jun	24	Harbour Grace	143,786		5,991
2279	Lómur HF-177	l-Jul	28-Jul	28	Hafnarfjörður.	147,766		5,277
			•	52		291,552	291,552	5,607
2212	Guobjörg IS-46	9-Sep	29-Sep	21	Argentia	49.950		2,379
2212	Guðbjörg IS-46	4-Oct	26-Oct	23	Akureyri	187,790		8,165
				44		237,740	237,740	5,403
2286	Bliki EA-12	25-Jun	22-Jul	28	Harbour Grace	137,700		4,918
2286	Bliki EA-12	27-Jul	23-Aug	28	Bay Roberts	124,200		4,436
2286	Bliki EA-12	31-Aug	2-Oci	33	Dalvík	119,500		3.621
				89		381,400	381,400	4,285
			Effort days	980		Toltal Catch:	6.579.584	6.7/4

Regn.no.	Vessels name	In	Out	Days	Port of unloading	Catch(kg)	Total Catch	Catch pr. d
2288	Pétur Jonss, RE-69	16. febr.	16, mars.	29	Bay Roberts	272,678		• 9,40
2288	Pétur Jónss. RE-69	20, mars,	20. april.	32	Bay Roberts	364,633		11,39
2288	Pétur Jónss. RE-69	24, april.	25. mai	32	Bay Roberts	315,597		9,86
2288	Pétur Jónss. RE-69	29. mai.	29. Júní.	32	Bay Roberts	331,580	1	10,36
2288	Pétur Jónss, RE-69	3. Júli	3. Ágúst.	32	Bay Roberts	318,953		9,96
2288	Pétur Jónss. RE-69	7. ágúst.	7. Sept.	32	Bay Roberts	306,585		9,58
2288	Pétur Jónss. RE-69	11. Sept.	12. okt.	32	Bay Roberts	289,213		9,03
2288	Pétur Jónss. RE-69	16.okt,	16.nóv.	32	Bay Roberts	225,865	-	7,05
2288	Pétur Jónss. RE-69	20. nóv.	16. des.	27	Hafnarfjörður	285,663		
2200		1		280	Í Í	2,710,767	2,710,767	9,68
1768	Nökkvi HU-15	2, mars.	22. mars.	21	Argentia	81,367		3,87
1768	Nökkvi HU-15	28. mars.	11. april.	15	Argentia	81,253		5,41
1768	Nőkkvi HU-15	17. april.	4. mai.	18	Argentia	82,144		4,56
1768	Nõkkvi HU-15	11. mai.	28.mai	18	Blönchuos	80,479		
1700	NUMBER 10-15			72	(325,243	325,243	4.51
2286	Bliki EA-12	7. mars.	30. mars.	24	Bay Roberts	154,500		6,43
2286	Bliki EA-12	4. april.	26, april.	23	Bay Roberts	136,500		5,93
2286	Bliki EA-12	Z. mai.	30, mai.	29	Bay Roberts	144,500	1.	4,98
2286	Bliki EA-12 Bliki EA-12	4. júni.	1, Júlí.	28	Dalvik.	167,400		5,97
2280	DIN LA-12	jum,		104	Durna	602,900	602,900	5,75
1352	Svalbarði SI-302	5, apríl.	4. mai.	30	Harbour Grace	210,529		7,018
1352	Svalbarði SI-302	9. mai.	7. júni.	30	Bay Roberts	238,716		7,95
1352	Svalbarði SI-302	15, Júni,	12. Júlí.	31	Siglufjörður.	244,125		7,87
1352	34404101 514502	19. 9411	12. 741.	91	Digitaljototal.	693,370	693,370	7.61
2190	Eyborg EA-59	. 21. april.	19. mai.	29	Argentia	134,470		4,63
2190	Eyborg EA-59	27. mai.	22. Júni.	27	Argentia	103,063	1	3,81
2190	Eyborg EA-59	28. Júní.	22. júlí.	25	Dalvik.	104,908		
2190	Eyoorg EA-39	20. 700	22. juli.	81	Darrix	342,441	342,441	4,228
1634	Hólmadrangur ST-70	20. april.	20. mai.	31	Hólmavík	127,193		4,103
1634	Hólmadrangur ST-70	15, Júní,	15. júlí.	31	Hólmavík	168,776		1,105
1634	Holmaurangur 51-70	15. Juni	15. jun.	62	TIOUNAVIK	295,969	295,969	4,774
20(1	Sunna SI-67	25. april.	17. mai	23	Argentia	207,211		9,009
2061 2061	Sunna SI-67	22. april. 22. mai,	31, Mai.	10	Ekki landað.	207,211		7,007
2061	Sunna SI-67	2.Júni.	20. júní.	21	Argentia	238,285	1	11,347
2061	Sunna SI-67	24. Júni	21 júlí.	28	Argentia	247,689		8,846
2061	Sunna SI-67	24. Júlí.	17.sept.	23	Argentia	195,028	ł. 1	8,479
2061	Sunna SI-67	20. run. 22 ágúst.	28-Aug	7	Ekki landað.	175,626		0,477
2061	Sunna SI-67	31 ágúst.	17 sept.	18	Bay Roberts **	198,602		7,944
2061	Sunna \$1-67	22. Sept.	19. okt.	28	Bay Roberts **	251,286		8,975
2061	Sunna SI-67	22. Stat. 24. okt.	23, nov.	31	Siglufjörður **	273,956		8,837
2001	30110 31-07	24. 084	25, 101	189	olginijoron.	1,612,057	1,612,057	8,529
1383	Skutull ÍS-180	13. nóv.	13. des.	31	Hafnarfjörður.	151,886	1,010,057	01241
1383	3Kutun 13-160	15. 104.	15. des.	31	nanarijotoat.	151,886	151,886	
3340	Unine DF 40	4, mai.	1	29	Bay Roberts	279,176	,000	9,627
2249 2249	Heiga RE-49.		1. júní. 4. júli.	30	Bay Roberts	327,973		10,932
	Helga RE-49.	5. júní. 8. Júlí.	 9. Ágúst. 	33	Bay Roberts	331,654	<u> </u>	10,932
2249	Helga RE-49.			33	Bay Roberts	298,574	1	9,631
2249	Helga RE-49.	13. ágúst.	12 sept. 19. okt.			298,574 295,665		11038
2249	Helga RE-49.	16 sept.	19. OKt.	123	Reykjavik.	1,533,042	1,533,042	
	0		c : / - /		999039	1,000,042	1,042	
2242	OrrilS	22, mai	5 júní.	15	??????	121.027	1 1	7,043
2242	Oπi ÍS	9. juni.	10. júli	32	Argentia Roy Doberts	331,027		7,043
2242	Orri İS	16. Juli	9. ágúst.	25	Bay Roberts		1	6,434
2242	Orti ÍS	13. ágúst.	7. Sept.	26	Ísafjarðar.	167,289	693,055	0,434
		<u> </u>		98		693,055	CCU,EX0	
2332	Askur AR	24. mai	7. júní.	15		has	1 1	
2332	Askur AR	12. júní.	4. júlí.	23	Bay Roberts	196,238	<u> </u>	5,164
2332	Askur ÁR	<u>9. júlí.</u>	30 júlí.	22	Reykjavík.	128,539	324,777	<u>5,843</u> 5,413
t				60)	324,777	1 474 TTT	5415

<u>Latvia</u>

3M Shrimp Catch and Effort 1993-1999

	1993	1994	1995/	1996	1997	1998	1999
			8 months				
Number of vessels	-	2	4	4	4	2	3
Fishing days allocated*	-	-	-	544	490	490	490
Fishing days used	-	190	649/544	504	439	402	438
Catches of shrimp (mt)	-	324	679/605	1253	997	1191	3080

NOTE: Concerning the way Latvia accounted fishing days and how they were shown in the Statlant 21B form, we have concluded, that during 1993-1995 the number of days was previously fixed only for the days spent directly for fishing, but not for the total number off days on the fishing ground. In subsequent years 1996-1999 al the days spent in shrimp fishery were counted in a different way, taking into account the total number of the days which vessels were represented in the NAFO area. Furthermore, it should be mentioned, that the NAFO Conservation and Enforcement Measures did not lay down the principles or rules for the accounting of fishing days as in hail reports.

On that background we have made a correction for the year 1995 taking as a basis the days of entry and exit from the fishing area. Accordingly it is necessary to update the number of fishing days allocated for Latvia from 1996 to 2000.

Lithuania

3M Shrimp Catch and Effort 1993-1999

Year:	1994	1995	1996	1997	1998	1999
Catch, MT	863	980	1585	1785	3107	3371
Used days	453	638	918	611	866	620

NOTE: The data as presented to the NAFO Secretariat in Statlant 21A and B forms.

<u>Norway</u>

3M Shrimp Catch and Effort, 1993-1999

Year							Mo	onth					Total
	January	February	March	April	May	June	July	August	September	October	November	December	1
1993			41	30	384	1,695	1,026	1,669	187	829	1,213		7,074
1994			1,072	443	169	134	2,138	2,174	597	1,009	339	550	8,62
1995		1	145	140	217	1,413	2,031	1,886	2,482	372	426	277	9,391
1996					141	171	779	771	760	559	474	1,993	5,648
1997	. 0			172.6	392	156.4	217.4	456.2		256	130.5	104.8	1,886
1998						280		622.2	194.9	242.1			1,339
1999					737.8	616.8	249.7	388	4.2	324.4	198.2	455.7	2,975
Total	0	1	1,258	785	2,041	4,466	6,441	7,966	4,226	3,592	2,781	3,380	36,937

1993		٦	Frip 1		٦ ٦	rip 2		٦	Trip 3		٦	rip 4		-	Trip 5		-	Trip 6		
Vesseiname	Radiosign	IN	OUT	Days	IN	ουτ	Days	N	OUT	Days	IN	ουτ	Days	IN	OUT	Days	iN	ουτ	Days	Total days
Arctic	LHIY	11-Jun	18 Jul	36	11-Aug	4-Sep	25	8-Sep	8-Sep	1										64
Bjørgvin Senior	јхск	17-Sep	28-Oci	42																42
Gisund	LHOL	30-May	22-Jun	24																24
ingar iversen	TXXT	18-Jun	11-Aug	55	23-Aug	19-Oct	58	1-Nov	22-Dec	52										165
John Longva	LGSO	8-Sep	4-Oct	27	7-Oct	27-Oct	21	13-Nov	13-Nov	1										49
Kap Farvel	LCKT	9-Jun	6-Jul	28	24-Jul	31-Aug	39	13-Sep	13-Sep	t										68
Lyshaug	LMEM	24-May	16-Jun	24																24
Ocean Trawler	LNBR	11-Jun	9-Aug	60																60
Ole Nordgård		27 Jun	31 Jul	35	11-Aug	17-Sap	38												[73
Olympic Prawn	LMJF	13-Jun	4 Jul	22	8-Jul	21-Jul	14	23-Jul	7 Aug	16	15-Sep	3-Nov	50							102
Polar Prawns	LDVP	9-Sep	29-Oct	51																51
Polarfangst	LGPZ	3-Nov	6-Dec	34																. 34
Remøy	JWYW	2-Jun	4-Jul	33	19-Jul	14-Sep	58	30-Sep	5-Dec	67										158
Remøytrål	JXOK	14-Jun	14-Jul	31	28-Jul	1-Sep	36	13-Sep	13-Sep	1					1			1		68
Rossvik	LNJV	24 May	8-Jun	16												[1	16
Stältind I	LKON	17-Jul	31-Aug	46	24-Sep	10-Oct	17	14-Oct	10-Nov	28										91
Stältor	LARD	23-May	11-Jun	20	23-Jun	23-Jul	31													51
Sytteljord	LNYG	13-Jul	13-Aug	- 32	3-Sep	10-Oct	38								i				1	70
Tromsbas	LEMA	20-Jun	24-Jul	35																35
Valderey	JWVC	22-Jul	5-Aug	15	10-Aug	31-Aug	22													37
Vikatrál	JXLV	11-Nov	10-Dec	30															[30
Volstad Viking	LAIR	14-Jun	24-Jul	41	5-Aug	23-Sep	50									Ι				91
Total				739			447			167			50			0			1	1403

1994		Т	rip 1	1	T	rip 2	T	T	rip 3	T	T	tip 4		T	rip 5		T	rip 6		ļ
		IN		Days	- N	OUT	Days	IN		Days	IN	OUT	Days	١N	<u> </u>	Days	IN	OUT	Days	Total days
/esselname	Radiosign						43													97
		28-Jan	22-Mar		26-May	7.Jul	23	-+								-1				66
	JXCK	11-Jun	23-Jul	43	29-Jul	20-Aug				-+						-				85
Gisund	LHQL	25-May	6-Jul	43	11-Jul	21-Aug	42				. 6	15.00	45			1				197
Hekktind	LAVJ	19 Mar	14-May	57	21-May	5-Jul	46	11-Jul	28-Aug	49	1 Sep	15-Oct			22-Oct		16-Dec	26-Dec	- 11	218
ngar iversen	JXXI	5-Jan	16-Mar	71	20-Mar	3-Apr	15	10-May	15-Jun	37	25-Ju ^l	10-04	78	17-Oct	22-001		10-000	20.060		134
John Longva	LGSO	5-Jan	26-Feb	53	2 Jun	24 Jul	53	30-Jul	26-Aug	28			-						┝─┨	86
Kap Farvel	LCKT	11-Jan	20-Feb	41	12 - Jun	26-Jul	45													58
Nyhorizont	LGAT	13-Jun	15-Jun	3	18 Jun	6-Jul		16-Jul	9-Aug	25	14-Aug	24-Aug	- 11						┝╼┛	
Ocean Trawler	LNBR	26-May	30-Jun	36	26 Jul	3-Oct	70						4						┝╼╉	106
Ole Nordgård		28-Jan	25-Mar	57	19-May	20-Jun	33	6-Aug	23-Aug	18									┝─┨	108
Olympic Prawn	LMJF	<u>11-</u> Jan	15-Ma	64	6-Jun	8-Aug	64	9-Sep	29-Oct	51		L	ļ						╞──┨	179
Polar Prawns	LOVP	1-Mar	4-May	65	27-May	17-Ju	52	7-Aug	19-Sep	44			·							161
Remey	JWYW	3-Jun	- 23-Ju	51	19-Sep	26-Oc	38						<u> </u>		L					89
Remøytrål	JXOK	18-May	3.Ju	47	7-Ju	18-Aug	43		Γ				1_							90
Statind)	LKON	19-Mar		60	22-May	10-Ju	T	17-Jul	28-Aug	43	1-Sep	11-00	41	L					\square	194
Státor	LARD	5-May	1-Ju		6-Jun	20-Ju													1	70
Tromsbas	LEMR	6-Jun		1					1			[Г							4
Tromstand	JXDH	27-JU		<u> </u>	2-Sep	5-00	34		\neg			Γ				<u> </u>				66
		12-Jar	6-Ma		25-May	19-JL		22-Ju	6-Sep	47		T	1			T				15
Volstad Viking	LAIN	12-041	i omini		Lonna	,0.00	771		ب م	342		†—	175			e	;		11	220
Total	<u></u> _	<u>!</u>	<u> </u>	901				<u>ا</u>	<u></u>	1		<u> </u>		<u> </u>			<u> </u>		1	
1995 Vesselname	Radiosign	IN	Trip 1 out	Days	IN	Trip 2 [ουτ	Days	IN	<u>Trip 3</u> [_{ουτ}	Days	N	Trip 4	Days	IN	Trip 5	Days		Trip 6	Days	Total day
	LLOW	2-Au				1 001	1	<u> </u>	1			1	1		1	1	<u> </u>	1	1	3
Andenesfisk 1	LHIY	12-Ma				14-Au	34		+-	1		†—	1						Г	6
Arctic		13-Ju				1	≝		+			t	+						T	•
Bjørgvin Senior	JXCK	14.00	0.00																	5
Gisund		20 40	e 1 10		Gallur	18.1	43		┼──			╞──	+			┢	<u> </u>	<u> </u>	+	
1		20-Ap			t		+	10.1		<u>ت</u> ه ا	24-Au	1 9-Se	0 1	ļ		F			┼──	8
Hekktind	LAVJ	6-Ap	1 21-Ma	n 43 y 44	25-Ma	y 6-Ji	4				24-Au	·*	-+	<u> </u>	13-Au		0 18-Au	8-54	22	8
Ingar iversen	LAVJ JXXJ	6-Ap 1-Ja	u 21-Ma n 9-Ja	n 43 y 44 n 4	25-Ma 9 11-Ja	y 6-Ja 1 11-Ja	ul 43	23-Fe	- <u> </u>		24-Au	·*	-+	<u> </u>	13-Au	9 6	0 <u>18</u> -Au	9 8-Se	p 22	8 14
Ingar iversen John Longva	LAVJ JXXJ LGSO	6-Ар 1-Ја 26-Ма	v 21-Ma n 9-Ja y 25-Ju	n 43 y 44 n 9	25-Ma 9 11-Ja	y 6-Ja 1 11-Ja	ul 43	23-Fe				·*	-+	<u> </u>	13-Au		0 18-Au	8-56	22	8 14 14 6
Ingar iversen John Longva Kap Farvel	LAVJ JXXJ LGSO LCKT	8-Ap 1-Ja 26-Ma 18-Ma	r 21-Ma n 9-Ja y 25-Ju y 1-J	n 43 y 44 n 3 1 4	25-Ma 11-Ja 28-Ju	y 6-Ji 1 11-Ja 1 26-Ji	ul 43	1 23-Fe	6 17-Ma	1 23		·*	-+	<u> </u>	13-Au	9 6	0 <u>18</u> -Au	8-Se	22	8 14 14 6 4
Ingar iversen John Longva Kap Farvel Myrefisk II	LAVJ JXXJ LGSO LCKT LGBZ	6-Ap 1-Ja 26-Ma 18-Ma 15-Ma	v 21-Ma n 9-Ja y 25-Ju y 1-J y 27-Ju	n 45 n 9 n 3 1 40	25-Ma 11-Ja 28-Jun 1 1-Ju	y 6-Ji n 11-Ja n 26-Ji	ul 43	1 23-Fe	6 17-Ma	1 23		·*	-+	<u> </u>	13-Au	2 6 2 6	0 18-Au	8-56	p 22	8 14 14 6 4 10
Ingar iversen John Longva Kap Farvel Myrefisk II Ocean Trawler	LAVJ JXXJ LGSO LCKT LGBZ LNBR	8-Ар <u>1-Ja</u> 26-Ма 18-Ма 15-Ма 26-Ма	r 21-Ma n 9-Ja y 25-Ju y 1-J y 27-Ju ry 2-Ar	n 43 n 9 1 40 n 3 1 40 n 40 n 40	25-Ma 11-Ja 28-Ju 1 1 1 -Ju	y 6-Ja 1 11-Ja 1 26-Ja 1 12-Au	ul 43	1 23-Fe	6 17-Ma	1 23 20 20	14-Ma	y 12-Jt	in 3) 15-Ju	13-Au	9 6	0 18-Au	8-Se	p 22	8 14 14 6 4 10 6
Ingar iversen John Longva Kap Farvel Myrefisk II Ocean Trawler Odd Erik	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX	8-Ap 1-Ja 26-Ma 18-Ma 15-Ma 26-Ma 21-Ju	r 21-Ma n 9-Ja y 25-Ju y 1-J y 27-Ju y 2-Au n 18-J	n 43 y 44 n 9 n 3 ul 40 n 44 ul 2	25-Ma 11-Ja 28-Ju 4 1-Ju 5 5 8 23-Ju	y 6-Ja 1 11-Ja 1 26-Ja 1 12-Au	ul 43	1 23-Fe	6 17-Ma	1 23 20 20		y 12-Jt	in 3) 15-Ju	13-Au		0 18-Au	8-Se	p 22	8 14 14 6 4 10 6 10 10
Ingar iversen John Longva Kap Farvel Myrefsk II Ocean Trawler Odd Erik Ole Nordgard	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNOA	6-Ap 1-Ja 26-Ma 18-Ma 15-Ma 26-Ma 26-Ma 21-Ju 29-Ma	r 21-Ma n 9-Ja y 25-Ju y 1-J y 27-Ju y 2-Au n 18-J ny 12-J	n 43 y 44 n 3 ul 44 n 4 ul 2 ul 4	25-Ma 11-Ja 28-Jui 4 1-Ji 8 23-Ju 5	6-Ja 1 11-Ja 26-Ja 1 12-Au 1 22-Au	19 43 19 43	23 Fe	6 17-Ma	1 23 20 20	14-Ma	y 12-Jt	in 3) 15-Ju	13-Au	2 6	0 18-Au	8-5e	22 	8 14 6 4 10 6 10 10
Ingar iversen John Longva Kap Farvet Myretick II Ocean Trawler Odd Erik Ole Nordgard Olympic Prawn	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNOA LMJF	6-Ap 1-Ja 26-Ma 18-Ma 26-Ma 26-Ma 21-Ju 29-Ma 21-Ju 29-Ma 7-Ap	r 21-Ma n 9-Ja y 25-Ju y 1-J y 27-Ju y 27-Ju y 27-Ju n 18-J n 18-J n 18-J n 18-J	n 43 n 3 n 3 n 4 n 4 n 4 n 4 u 2 u 2 u 4 m 6	25-Ma 11-Ja 28-Ju 29-Ju 1-Ja 29-Ju 5 5 1 24-Ju	y 6-Ja 1 11-Ja 1 26-Ja 1 12-Au 1 22-Au 1 22-Au	19 43 19 43 19 43	23 Fe	6 17-Ma	1 23 20 20	14-Ma	y 12-Jt	in 3) 15-Ju	13-Au	9 6	0 18-Au	8-56	p 22	8 14 14 14 6 4 4 10 6 6 10 10 10 10 10 10 10
Ingar iversen John Longva Kap Farvel Myrefsk II Ocean Trawler Odd Erik Ole Nordgard	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNOA LMJF JWOP	6-Ap 1-Ja 26-Ma 18-Ma 26-Ma 26-Ma 26-Ma 21-Ju 29-Ma 21-Ju 29-Ma 7-Ap 4-J	r 21-Ma n 9-Ja y 25-Ju y 27-Ju y 27-Ju y 27-Ju y 27-Ju n 18-J n 18-J n 18-J n 18-J n 12-Ju n 12-Au	n 43 y 44 n 3 n 3 ul 45 ul 45 ul 2 ul 4 m 6 49 4	25-Ma 11-Ja 28-Jun 28-Jun 28-Jun 28-Jun 1-Jl 28-Jun	y 6-Ji 111-Ja 26-Ji 122-Au	19 43 19 43 19 43 19 43 19 43 19 4	23-Fe	6 17-Ma	1 23 20 20	14-Ma	y 12-Jt	in 3) 15-Ju	13-Au	2 6		8-5e	p 22	8 14 14 14 16 4 10 6 12 12 12 14
Ingar iversen John Longva Kap Farvet Myretick II Ocean Trawler Odd Erik Ole Nordgard Olympic Prawn	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNOA LMJF	6-Ap 1-Ja 26-Ma 18-Ma 15-Ma 26-Ma 26-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 20-Ap	r 21-Ma n 9-Ja y 25-Ju y 1-J y 27-Ju y 27-Ju y 2-Au n 18-J ny 12-J or 6-Ju ul 12-Au un 10-M	n 43 n 3 n 3 n 3 ul 43 n 4 ul 2 ul 4 ul 2 ul 4 an 6 ig 4 ar 4	25-Ma 11-Ja 28-Ju 28-Ju 3 4 1-Ju 28-Ju 5 5 1 24-Ju 0 17-Au 4 4-Ju	y 6-Ji n 11-Ja 1 26-Ji 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au	1 43 n 25 1 25	1 23-Fe	9 4·Se	r 23	14-Ma	y 12-Jt	in 3) 15-Ju	13-Au		0 18-Ац	8-Se		8 14 14 14 14 14 14 14 14 14 16 16 15 15 15 16 16 16 16 16 16 16 16 16 16 16 16 16
Ingar kreisen John Longva Kap Farvel Myretisk II Ocean Trawler Odd Erk Ole Nordgard Olympic Prawn Orion	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNOA LMJF JWOP	6-Ap 1-Ja 26-Ma 18-Ma 26-Ma 26-Ma 26-Ma 21-Ju 29-Ma 21-Ju 29-Ma 7-Ap 4-J	r 21-Ma n 9-Ja y 25-Ju y 1-J y 27-Ju y -Ju y 27-Ju y 2	n 43 n 3 n 3 n 4 n 4 n 4 n 4 n 4 u 2 u 4 n 6 n 6 n 6 n 4 n 6 n 3 n 4 n 4 n 4 n 4 n 4 n 4 n 4 n 4 n 4 n 4	25-Ma 11-Ja 28-Jun 28-Jun 1 -Ju 28-Jun 5 23-Jun 5 1 -Ju 1 -Ju 28-Jun 1 -Ju 1 -Ju 28-Jun 1 -Ju 28-Jun 1 -Ju 28-Jun 1 -Ju 28-Jun 5 - 1 -Ju 28-Jun 5 - 1 -Ju 28-Jun 5 - 1 -Ju 28-Jun 5 - 1 -Ju 28-Jun 5 - 1 -Ju 28-Jun 5 - 1 -Ju 28-Jun 5 - 1 -Ju 28-Jun 5 - - - - - - - - - - - - -	y 6-Ja 1 11-Ja 26-Ja 1 12-Au 1 12-Au 1 12-Au 1 22-Au 1 22-Au 1 22-Au 1 12-Au 1 12-A	1 43 1 43 1 25 1 25 1 25 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2	1 23-Fe	b 17-Ma	1 23 20 20 41 43	14-Ma	y 12-Jt	in 3) 15-Ju	13-Au		0 18-Au	8-Se		
Ingar kreisen John Longva Kap Farvel Myretick II Ocean Trawler Odd Erk Ole Nordgard Olympic Prawn Orion Remey	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNOA LMJF JWOP JWYW	6-Ap 1-Ja 26-Ma 18-Ma 18-Ma 15-Ma 26-Ma 21-Ju 29-Ma 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 20-Ju 29-Ma 20-Ju 20-J	r 21 Mz n 9-Jz y 25-Ju y 27-Ju y 27-Ju y 27-Ju y 27-Ju n 18-J ny 12-J y 6-Ju ul 12-Au un 10-M xb 4-F	n 45 n 3 n 3 n 3 n 4 n 4 n 4 n 4 n 4 n 4 n 4 n 4 n 4 n 4	25-Ma 11-Ja 28-Jun 28-Jun 5 5 5 1 24-Ju 0 17-Au 0 17-Au 1 9-Fe 4 10-J	y 6-Ja 1 11-Ja 26-Ja 1 12-Au 1 12-Au 1 22-Au 1 22-Au 1 22-Au 1 12-Au 1 12-A	19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 19 4 10 1 10 10 1 10	I 23-Fe 1 23-Fe 1 29-Au 29-Au 5 5 7 23-Mi 9 19-Au	6 17-Ма 9 4-Se 9 10-Ол ау 2-J 19 Ал	1 23 20 20 1 43	14-Ma	y 12-Jt	in 3) 15-Ju	13-Au		0 18-Au	8-56		88 144 144 144 144 144 144 144 144 144 1
Ingari Versen Jehn Longva Kap Farvel Myretek II Ocean Trawler Odd Erk Ole Nordgard Olympic Prawn Orion Romey Romeytrat	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNOA LMJF JWOP JWYW JXOK	6-Ap 1-Ja 26-Ma 18-Ma 15-Ma 26-Ma 26-Ma 21-Ju 29-Ma 29-M	r 21 Mz n 9-Jz y 25-Ju y 27-Ju y 27-Ju y 27-Ju y 27-Ju n 18-J ny 12-J y 6-Ju ul 12-Au un 10-M xb 4-F	n 45 n 3 n 3 g 6 g 6 g 6 g 6 g 6 g 6 g 6 g 6 g 6 g 6	25-Ma 11-Ja 28-Ju 28-Ju 28-Ju 4 1-Ju 28-Ju 128-Ju 1-Ju 20-Ju 1-Ju 1-Ju 1-Ju 20-Ju 1-Ju 1-Ju 1-Ju 20-Ju 1-Ju 1-Ju 1-Ju 20-Ju 1-Ju 1-Ju 20-Ju 1-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 20-Ju 1-Ju 20	y 6-Ja 1 11-Ja 26-Ja 1 22-Ja 1 12-Au 1 22-Au 1 22-Au 1 22-Au 1 28-J 1 5-Fi 1 5-Fi 1 3-Fi 2 3-Au	y 4(n) 25 2	1 23-Fe 3 16-Au 1 29-Au 5 5 7 23-Mi 9 19-Au 3 26-Ae	6 17-Ма 9 4-Se 9 10-Ос 10-ОСС	1 23 20 21 43 11 41 9 11 61 45	14-Ma	y 12-Jt	in 3) 15-Ju	13-Au			2 8-5e		8 14 14 14 14 14 14 16 16 16 16 16 16 16 16 16 16 16 16 16
Ingar iversen Jehn Longva Kap Farvet Myretek II Ocean Trawler Odd Erk Ole Nordgard Olympic Prawn Orion Romey Romeytrat Stetnes	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNOA LMJF JWOP JWYW JXOK LHVR	6-Ap 1-Ja 26-Ma 18-Ma 18-Ma 15-Ma 26-Ma 21-Ju 29-Ma 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 21-Ju 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 20-Ju 29-Ma 20-Ju 20-J	ir 21-Max n 9-Ja y 25-Ju y 1-Ja y 27-Ju y 27-Ju y 27-Ju y 27-Ju y 27-Ju y 27-Ju y 2-Au n 18-J y 12-Au ui 12-Au ui 10-M yb 4-F uin 4 uin 11-	n 45 n 3 n 3 l 4 n 4 l 4 n 4 l 4 l 4 l 4 l 4 l 4 l 4 l 4 l 4 l 4 l	25-Ma 11-Ja 29-Ju 29-Ju 29-Ju 3 4 1-Ja 29-Ju 3 3 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 20-Ju 20	y 6-Ja 1 11-Ja 26-Ja 1 22-Ja 1 12-Au 1 12-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 2-Au 1 2	yıl 43 n n 22 99 4 99 4 99 4 99 4 99 4 99 4 99 4 9	1 23-Fe 1 23-Fe 1 29-Au 1 29-Au 1 29-Au 5 5 7 23-Mi 9 19-Ai 3 26-Ai 4 21-Ji	6 17-Ма 9 4-Se 9 10-Ос 9 0-ОС 100-ОС 100-ОС 1	1 23 2 20 1 43 1 41 9 1 1 41 9 1 1 49	14-Ma	y 12-Jt	in 3) 15-Ju	13-Aug					8 14 6 4 10 10 10 10 10 10 10 10 10 10 10 10 10
Ingar kresjon John Longva Kap Farvel Myrefek II Ocean Trawler Odd Erk Ole Nordgard Olympic Prawn Orion Romey Romey Romey Statind I	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAR LNOA LNOA LMJF JWYW JXOK LHVR LKON	6-Ap 1-Ja 26-Ma 18-Ma 15-Ma 26-Ma 26-Ma 21-Ju 29-Ma 29-Ma 20-Ja 4-Ji 26-Ja 4-F 6 1-Ju 30-Ju	r 21-Ma n 9-Ja y 25-Ju y 1-J y 27-Ju y 27-Ju y 27-Ju n 18-J y 12-J r 6-Ju u 12-An in 10-M kb 4-F un 4 pr 6-M	n 45 n 3 n 3 y 44 n 45 n 4 y 44 n 4 y 44 n 4 y 44 n 4 y 44 y 4	25-Ma 11-Ja 28-Ju 28-Ju 28-Ju 4 1-Ju 28-Ju 128-Ju 1-Ju 20-Ju 1-Ju 1-Ju 1-Ju 20-Ju 1-Ju 1-Ju 1-Ju 20-Ju 1-Ju 1-Ju 1-Ju 20-Ju 1-Ju 1-Ju 20-Ju 1-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 1-Ju 20-Ju 20-Ju 1-Ju 20	y 6-Ja 1 11-Ja 26-Ja 1 12-Au 1 12-Au 1 12-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 23-Au 1 23-Au 1 23-Au 1 23-Au 1 23-Au	yıl 43 n n 22 99 4 99 4 99 4 99 4 99 4 99 4 99 4 9	1 23-Fe 3 16-Au 1 29-Au 5 5 7 23-Mi 9 19-Au 3 26-Ae	6 17-Ма 9 4-Se 9 10-Ос 9 0-ОС 100-ОС 100-ОС 1	1 23 2 20 1 43 1 41 9 1 1 41 9 1 1 49	14-Ma	y 12-Jt	in 3) 15-Ju	13-Au					8 14 6 4 10 10 10 10 10 10 10 10 10 10 10 10 10
Ingar kressen John Longva Kap Farvel Myrefek II Ocean Trawler Odd Erk Ole Nordgard Olympic Prawn Orion Remey Remey Remey Remey Islatnes Stattind I Stattor	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNGA LMJF JWVP JWVW JXOK LHVR LKON LARD	8-Ap 1-Ja 26-Ma 18-Ma 15-Ma 26-Ma 21-Ju 29-Ma 29-Ma 29-Ma 29-Ma 20-Ja 20-Ja 20-Ja 20-Ja	r 21-Ma n 9-Ja y 25-Ju y 1-J y 27-Ju y 27-Ju y 27-Ju n 18-J y 12-J r 6-Ju u 12-An in 10-M kb 4-F un 4 un 11 pr 6-M	n 45 n 3 n 3 n 3 n 45 n 45 n 45 n 40 n 4	25-Ma 11-Ja 29-Ju 29-Ju 29-Ju 3 4 1-Ja 29-Ju 3 3 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 4 1-Ja 29-Ju 20-Ju 20	y 6-Ja 1 11-Ja 26-Ja 1 22-Ja 1 12-Au 1 12-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 22-Au 1 2-Au 1 2	yıl 43 n n 22 99 4 99 4 99 4 99 4 99 4 99 4 99 4 9	1 23-Fe 1 23-Fe 1 29-Au 1 29-Au 1 29-Au 5 5 7 23-Mi 9 19-Ai 3 26-Ai 4 21-Ji	6 17-Ма 9 4-5е 9 10-Ог 9 10-Ог 9 10-Ог 10 -Ог 10 -	r 23	14-Ma	y 12-tt		0 5	• 13-Au			3 8-5e		
Ingar kresen John Longva Kap Farvel Myrelesk II Ocean Trawler Odd Erik Ole Nordgard Olympic Prawn Orion Romey Romey Romey Romey Statind 1 Station Sytehjord	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNGA LMJF JWOP JWOP JWOP JWOR LKON LKON LARD LARD LNYG	6-App 1-Jaa 26-Ma 19-Ma 19-Ma 19-Ma 26-Ma 21-Ju 29-Ma 20-Ma 20-Ju 29-Ma 20-Ju 20-Ju 20-Ju 20-Ju 20-Ju 20-Ju 20-Ju 20-Ma 20	it 21 Mit n 9 Jit y 25 Jit y 1-J y 27 Jit y 26 A	n 45 y 42 n 3 n 3 n 3 n 3 n 4 n 4 n 4 n 4 n 4 n 4 n 4 n 4	25-Ma 11-Ja 28-Jun 28-Jun 5 23-Jun 5 23-Jun 5 23-Jun 5 23-Jun 5 23-Jun 5 23-Jun 5 23-Jun 5 24-Jun 1 24-Jun 5 23-Jun 7 24-Jun 7 24-Jun 7 24-Jun 7 24-Jun 7 24-Jun 7 24-Jun 7 22-Jun 7 22-Jun 2 22-Jun 22-Jun 2 22-Jun 2 22-Jun 2 22-Jun 2 22-Jun 2 22-Jun 2 22-Jun 2 2 22-Jun 9 15-Ma 8 31-Au	y 6-Jh 11-Ja 26-Jh 12-Au 11-Ja 26-Jh 11-Ja 26-Jh 11-Ja 26-Jh 11-Ja 26-Jh 11-Ja 20-Jh 11-Ja 11-Ja 12-Au 11-Ja 12-Au 11-Ja 11-Ja 12-Au 11-Ja 12-Au 11-Ja 12-Au 11-Ja 12-Au 11-Ja 12-Au 11-Ja 12-Au 11-Ja 12-Au 11-Ja 11-Ja 12-Au 11-Ja 11	и л л 11 22 13 29 4 19 19 4 19 19 19 19 19 19 19 19 19 19	1 23-Fe 1 23-Fe 1 29-Au 1 29-Au 1 29-Au 5 5 7 23-Mi 9 19-Ai 3 26-Ai 4 21-Ji	6 17-Ma 9 4-Se 9 10-Ot 9	r 23	14-Ma	y 12-tt) 15-Ju	• 13-Au			3 8-5e		
Ingar kresen John Longva Kap Farvel Myrelsk II Ocean Trawler Odd Erik Ole Nordgard Olympic Prawn Orion Romey Romey Romey Romey Romey Statind 1 Station Statind 1 Station	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNGA JWOP JWVW JWVW JKOK LHVR LKON LARD LARD LARD LMYG LHSK	6.App 1.Ja 26-Ma 15-Ma 15-Ma 26-Ma 21-JUU 29-Ma 7.Aj 29-Ma 4.J. 20-Ja 8.A. 20-Ja 20-Ja 20-Ja 21-JUUU 21-JUUU 21-JUUU 21-JUUU 21-JUUU 21-JUUUUUUUUUUUUUUUU	12 21 M2 17 21 M2 17 27 JL 17 J 17 J	n 45 y 4 n 3 n 3 n 3 n 3 n 4 n 4 g 6 ul 2 ul 4 n 4 n 4 ul 4 n 4 n 4 n 3 ul 44 n 4 n 3 ul 44 n 4 n 3 ul 44 n 4 n 3 ul 44 n 3 ul 44 n 4 n 3 ul 44 n 4 n 4 n 3 ul 44 n 4 n 4 n 4 n 4 n 4 n 4 n 4	25-Ma 25-Ma 28-Jun 28-Jun 5 5 5 5 1 24-Jun 6 7 4 4-Jun 7 7 7	y 6-Jh 11-Ja 26-Ji 11-Ja 22-Au 11 12-Au 1	JI 43 JI 23 JI 25 JI 30 JIII	23-Fe 23-Fe 24 24 25 5 5 5 5 5 5 5 5 5 7 23-Mi 9 19-Au 20-Au	6 17-Ma 9 4-Se 9 10-Ot 9	r 23	14-Ma	y 12-tt		0 5	13-Au			2 8-5e		
Ingar kresten John Longva Kap Farvel Myretesk II Ocean Trawler Odd Erik Ole Nordgard Olympic Prawn Orion Romey Romey Romey Romey Romey Romey Romey Statind I Station Statind I Station Sytelyord Sasvking Tromsbas	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNGA JWOP JWWW JAOK LHVR LKON LARD LARD LHSK LFMR	6.App 1.Ja 26-Ma 18-Ma 19-Ma 19-Ma 21-Juu 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 29-Ma 20-Ja 29-Ma 20-Ja 29-Ma 29	12 21 M2 17 21 M2 17 25 Ju 17 J 17 J	n 452 y 44 n 3 n 3 n 3 y 44 n 4 y 44 n 44 g 66 y 4 y 44 y 1 4 y 44 y 1 4 y 1 4 y 44 y 1 4 4 y 1 4 y 1 y 1 y 1 y 1 y 1 y 1 y 1 y 1	25-Ma 11-Ja 28-Jul 28-Jul 28-Jul 28-Jul 28-Jul 28-Jul 28-Jul 28-Jul 28-Jul 28-Jul 29-Jul 20-Jul	y 6-Jh 11-Ja 11-Ja 128-Jh 28-Jh 28-Jh 12-AL	и п п 11 25 29 4 29 20 20 20 20 20 20 20 20 20 20	23-Fe 23-Fe 24 25-Fe 25 55 55 55 57 72 23-Mu 4 21-Ju 72 20-Su 20	6 17-Ma 9 4-Se 9 10-Ot 9	r 23	14-Ma	y 12-tt		0 5	13-Au			2 8-5e		8 14 14 14 16 4 10 6 12 12 14 10 6 12 12 12 12 12 12 12 12 12 12
Ingar kresten John Longva Kap Farvel Myrelsek II Ocean Trawler Odd Erik Ole Nordgard Otympic Prawn Orion Remey Remey Remey Remey Remey Statind I Station Statind I Station Statind I Station	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNGA LMJF JWOP JWWW JXOK LMVR LKON LARD LNYG LHSK LFMR JXDH	6.App 1-Ja 26-Ma 18-Ma 19-Ma 26-Ma 20-Ma 29-Ma 29-Ma 29-Ma 29-Ma 20-Ja 4-Fe 1-Je 30-Ju 20-Ja 4-Fe 20-Ja 12-D 21-A 30-Ju 20-Ja 12-D 21-A 30-Ju 20-Ja 12-D 21-Ju 20-Ma 20-M	it 21 MM n 9 J25 JU y 25 JU y 1 J J y 27 JU y 26 A y 26 A y 27 A	n 452 y 44 n 3 n 3 n 3 g 6 y 44 n 44 g 6 y 44 y 1 44 y 2 10 10 10 10 10 10 10 10 10 10	25-Ma 9 11-Ja 28-Jul 28-Jul 28-Jul 28-Jul 9 11-Ja 28-Jul 9 11-Ja 9 10-J 9 11-Ja 9 22-Jul 1 24-Jul 1 24-Jul 1 24-Jul 1 24-Jul 1 24-Jul 9 11-Ja 9 22-Jul 1 24-Jul 1 24-Jul 9 22-Jul 1 24-Jul 1 3 1-Au 1 3 1-	y 6-Jh 1 11-Ja 1 28-Jh 2 28-Jh 1 12-AL 1 12	JI 42 JI 21 JI 22 JI 22 JI 22 JI 23 JI 23 JI 24 JI 23 JI 24 JI 24 JI 25 JI 30 JIII	23-Fe 23-Fe 24 29-Au 20-Au	b 17-Ma g 4-Se g 4-Se g 10-Oc g 10-	r 23 20 20 20 20 20 20 20 20 20 20 20 20 20	14-Ma	y 12-tt		0 5						
Ingari Versian John Longva Kap Farvel Myretek II Ocean Trawler Odd Erik Ole Nordgard Olympic Prawn Orion Remay Remaytrål Stetnes Stattind I Stattor Stattind I Stattor Stattind I Stattor Stattind I Stattor Stattind I	LAVJ JXXJ LGSO LCKT LGBZ LKR JXAX LNGA LMBR JXAX LNGA LMJF JWOP JWYW JXOK LHVR LKON LARD LHYR LKON LARD LHSK LFMR JXDH LAH	6.App 1-Ja 26-Ma 16-Ma 16-Ma 26-Ma 20-	it 21 MM n 9 J25 JU y 25 JU y 1 J J y 27 JU y 26 A y 26 A y 27 A y 27 A y 26 A y	n 45 y 4 n 3 n 3 u 4 n 4 n 4 n 4 n 4 n 4 n 4 n 4 n	25-Ma 211-Ja 28-Jun 28-Jun 4 1-JJ 28-Jun 5 5 5 5 124-Jun 9 5 5 124-Jun 9 15-Ma 4 4-JL 19-Fe 5 5 124-Jun 9 15-Ma 8 13-Ma 14-J 15-Ma 9 9 15-Ma 14-J 15-Ma 14-J 15-Ma	y 6-J-1 11-J-2 26-J-1 11-Z-Autor 11-Z-Autor 11-Z-Autor 11-Z-Autor 12	1 4 1 4 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 3 1 3 1 3 1 3 1 3 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	23-Fe 23-Fe 24-54 29-Au 20-Au 20	b 17-Ma g 4:Se g 10-Oi g 10	r 23 20 20 20 20 20 20 20 20 20 20 20 20 20	14-Ma	y 12-tt		0 5						
Ingari Versien John Longva Kap Farvel Myrelesk II Ocean Trawler Odd Erik Ole Nordgard Oympic Prawn Orion Romey Romey Romey Romey Romey Romey Statind I Station	LAVJ JXXJ LGSO LCKT LGBZ LNBR JXAX LNOA LNOA LNOA LNOA LNOA LNOA LNOA LNOA	6.ApP 1-Ja 26-Ma 19-Ma 19-Ma 19-Ma 19-Ma 20-Ma 20-Ja 20-Ma 4-Ja 20-J	x 21 MM n 9 J5 y 25 J y 1-J y 27-JL y 26-A y 24-D	n 433 y 4 n 3 n 3 u 4 n 4 n 4 n 4 n 4 n 4 n 4 n 4 n	25-Ma 21-28-Jun 28-Jun 28-Jun 28-Jun 29- 23-Jun 23-Jun 23-Jun 23-Jun 23-Jun 24-Jun 24-Jun 24-Jun 24-Jun 24-Jun 25-Ma 21-22-Jun 22-22-Jun	Y 6-J-1 11-2- 26-J-1 11-2- 26-J-1 11-2- 26-J-1 11-2- 26-J-1 12-2- 26-J-1 12-2- 26-J-1 12-2- 26-J-1 12-2- 26-J-1 22- 26-J-1 22- 26-J-1 22- 26-J-1 27-AL 27	III 4 III 22 IIII 23 IIII 23 IIII 23 IIII 23 IIII 23 IIII 23 IIII 24 IIII 5 IIII 5 IIII 5 IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	23-Fe 1	b 17-Ma g 4:Se g 10-Oi g 10	1 23 2 20 2 20 1 20	14-Ma	y 12-tt		0 5						55 88 14 14 14 16 4 10 6 13 4 10 6 13 4 10 6 13 4 10 10 10 10 10 10 10 10 10 10

1996		T	rip 1		٦	Frip 2		1	rip 3		. 7	Frip 4		1	Frip 5		1	rip 6		
Vesseiname	Radiosign	IN	OUT	Day6	IN	OUT	Days	IN	OUT	Days	N	OUT	Days	IN	OUT	Days	IN	ουτ	Days	Total days
Hekktind	LAVJ	16-Apr	19-May	34	23-May	8-Jul	47	14-Jul	4-Sep	53										134
logar iversen	JXXJ	23-May	30-Jun	39	4-Jul	26-Aug	54	29-Aug	27-Oct	60	2-Nov	21-Dec	50							203
John Longva	LGSO	31-May	27-Jun	28	30-Jun	31-Jul	32													60
Myrefisk II	LGBZ	24-May	6-Jul	44	11-Jul	23-Aug	44													86
Ole Nordgård	LNQA	30-May	6-Jul	38																38
Olympic Prawn	LMJF	3-Jun	14-Jul	42	19-Jul	30-Aug	43													85
Петлеу	JWYW	7-Jun	10-Jul	34																34
Remeytrål	JXOK	15-Jun	21-ປມ	37	26-Jul	24-Aug	30													67
Spitsbergen	LHZR	29-Jun	4-Jul	6	7-Jul	21-Aug	46	25-Aug	6-Sep	13	10-Sep	11-Oct	32	14-Oct	5-Nov	23	10-Nov	1-Dec	22	142
Statind I	LKONAHWY	6-Apr	20-May	45	25-May	1-Jul	36	8.Jul	31-Aug	55										136
Station	LARD	15-Apr	26-May	42	ЭФ-Мау	15-Ju	47	20-Jul	21-Aug	33		Ĺ	1			Ì		<u> </u>]	12
Sæviking	LHSK	1-Jul	25-Aug	56	31-Aug	12-Oct	43			[L	99
Tromsland	JXDH	7-Apr	11-May	35	15-May	4-Jun	21	9-Jun	8-Jul	30	12-Jul	18-Aug	38		_			L		124
Vestind	LHLU	21-Apr	9-Jun	50	15-Jun	27-Ju	43	1-Aug	21-Sep	52	26-Sep	1-Nov	37							182
Vima	LEMR	29-May	30-May	2	4-Jun	4-Ju	31													33
TOTAL				532			519			296		<u> </u>	157			23			22	1,549

•

1997		Tr	rip 1		Ţ	Frip 2		T	Frip 3		Trip 4	4		Trip 5	с С		Trip 6			Trip 7			Trip 8	-	
Vessetname R	Radiosign	<u>z</u>	OUT D	Days	N	OUT Days		N N	0UT DO	Days	R N	OUT Days	N N	DUT	r Days	Z	Ъ О	OUT Days	z	OUT	Days	z	- INO	Days T	Total days
Ingar Iversen JXXJ	5	17-Mar	21-Apr	36 2	23-Apr 1	11-May	19 15	19 16-May 24-Jun	4-Jun	40 27	27-Jun 17-	17Jul	21 2-A	2-Aug 25-Aug		4 28-Au	24 28-Aug 12-Oct	46	15-Oci	11-Nov		28. 20-Nov 16-Dec	16.Dec	27	241
Volstad Viking LAIR	E	1-Apr	4-May	34 7	7-May 2	26-Jun	51																		88
TOTAL				70			70			4 0		-	21		24	-	 	46	Ĺ					┢	329
1998		1	Lip 1	┢		Trip 2		E	Trip 3	┢	Trip 4	4		Trip 5	2			F	Trip 6						
Vesseiname R	Radiosign	ş	OUT	Days	N	OUT Days		z	а 170	Days IN	l	OUT Days	Z.		OUT Days	₹ 2	OUT	Days	OUT Days Total days						
Ingar Iversen JXXJ	د ۱	30 May	27-Jun	29	1-Jul	1-Aug	32 5	5-Aug 30-Aug	0-Aug	26 1	1-Sep 15-Ocl		45		┝				132						
Volstad Viking LAIR	н	2-Jul	5-Aug	35 8	8-Aug 2	20-Sep	44				ļ								62						
TOTAL				64			76	┝	-	26	_	-	45		Ĺ			Ĺ	211						
1999		T,	rio 1			Trip 2	┢─	Ē	Trip 3	-	Trip 4	4		Trip 5	 G		Trip 6		Ţ	Trip 7		Trìp 8	8	┢	
			-	-		; F			,	-	-			-	,		, 2			-	-		>	_	-

183 211

Total days

Days

25-Nov OUT

21-Nov Ż Days 42

19-Nov: OUT

io o o Z

Days 5-001 ٩ ١

28 24

25 20-Nov 15-Dec 15-Sep z ņ Days

8-Aug 25-Oct 18-Nov

25-Jul Z

> 21-Jul 16-Ocr

OUT

Days ¥

ĝ R Days

17-Sep 19-Jun Ĕ

ĝ

3-Aug 16-Jun OUT

25-Jun 14-May Ľ

31 32 Days

9-Apr 10-May 22-May 21-Jun Ъ

z 59 3

Days

Radiosign rxxr LAIR

6-Apr 17-May DO

17.Apr 9-Mar z

Volstad Viking Ingar iversen Vesselname

TOTAL

74

ß

8

3

Ы

40

394

42

176

i

3M Shrimp Catch and Effort, 1993, 1999

In accordance with the Working Group on Allocation and Shrimp meeting (Washington, D.C., USA, March 27-30, 2000) recommendation and further to the STACTIC (Dartmouth, N.S., Canada, June 27-29, 2000) meeting discussion, this is to note that the Russian Federation could not completely verify its data on shrimp fishery at present stage. As the Russian delegation had explained during previous annual NAFO meetings, the catches/effort statistics of Russian vessels in NAFO Regulatory Area during 1993-1995 have not been accurately monitored properly by many newly individual companies in Russia and State Committee of the Russian Federation for fisheries did not have complete reports of all vessels catching in this period in NRA. Also, there were a large number of Russian vessels conduction all time mixed - redfish & shrimp fishery in 3M during 1995. For preparing the 1995 divide total fishing days between redfish and shrimp fishery. We have not official statistics about the effort of Russian Vessels during 1995 on 3M shrimp fishery are 2800 fishing days. Considering above, the Russian Federation have established limitation of number of fishing vessels - 17 for 1996, and 1997-1998 number of fishing days 3M shrimp fishery - 2600, 1999-2000 number of fishing days 2100.

The Russian Federation will be trying to verify these data further, if possible, and any new information available will be advised to the NAFO Secretariat.

(original signed by A. Okhanov, Representative of the Russian Federation in Canada on Fisheries)

Annex 13. Statement from the Representative of Norway

Agenda Item 6 (a) - Review of submissions on shrimp catches and effort days

Prior to this meeting in STACTIC, Norway circulated the Working Paper, which we introduced earlier. In that paper we urged the other Contracting Parties to forward similar information regarding the activity of vessels flying their flag fishing for shrimp in 3M. Our inteOntion is of course to increase transparency regarding all figures on catch and effort in order to have a fruitful discussion at the annual meeting of NAFO, when the Fisheries Commission shall decide upon the future management measures for this stock.

At this meeting, Norway would like to stress the importance of this point. As a follow up to our Working Paper, we have asked the various Contracting Parties to disseminate information about catch and effort in the fishery. We must conclude, however, that for some Contracting Parties, this information is still not available. We would therefore, once again, urge these Contracting Parties to forward such information to the Executive Secretary of NAFO, Dr. Chepel, in due time before the Annual Meeting. We would also propose that the Executive Secretary of NAFO distribute these data to all Contracting Parties two weeks prior to the annual meeting.

Annex 14. Proposal (by European Union) to amend the NAFO Conservation and Enforcement Measures regarding "Part VII-Port Inspections" (STACTIC W.P. 00/9+Corr.)

Background

Part VII of the NAFO Conservation and Enforcement Measures requires Contracting Parties to ensure that port inspection take place on any occasion a fishing vessel having been fishing subject to NAFO Conservation and Enforcement Measures is discharging catch. According to the current measures, the results from port inspection shall be provided to the NAFO secretariat and shall be communicated to any other Contracting Party on request.

The content of port inspection should include verification of catches, of logbook records, mesh size and of inspection at sea. Sea inspection reports are sent to the Contracting Party without delay.

Communication of port inspection are sometimes delayed when vessels land in ports outside the Flag Contracting Party. In order to contribute to enhanced transparency and a better efficiency of the implementation of the NAFO Conservation and Enforcement Measures, it is proposed that the results of port inspection are communicated to the Flag Contracting Party without delay.

Furthermore, a standard report form would help to harmonise record of results of port inspection.

Proposal

1. Amend Part VII-1 of NAFO Conservation and Enforcement Measures to read ;

Part VII-1

- "(v) Results of port inspection shall be given in the "NAFO port inspection report", as defined in Part VII -Schedule I.
- (vi) The authorities of the Contracting Party of the port State shall, within 7 working days as from the date on which the inspection has been completed, transmit the "NAFO port inspection report" form to the Contracting Party of the flag State.
- (vii) Copy of the "NAFO port inspection report" shall be transmitted to the NAFO Executive Secretary within 30 days as from the date on which the landing has been completed and shall be provided to other Contracting Party on request."
- 2. Insert <u>Part VII-Schedule I : "NAFO port inspection report"</u> (see annex)

1	80
---	----

CONTENT

1.	INSF	INSPECTION INFORMATION									
	1.1	Format	of the data								
2.	TRIP	INFORM	IATION								
	2.1	Format	of the data								
3.	VES	VESSEL IDENTIFICATION									
	3.1	Format	of the data								
4.	RES	ULT OF I	NSPECTION OF LANDING								
	4.1	General	information								
		4.1.1	Format of the data								
	4.2	Quantit	y landed								
		4.2.1	Format of the data								
	4.3	Quantit	y staying on board the vessel								
		4.3.1	Format of the data								
5.	GEA	R INSPEC	CTION IN PORT								
	5.1	General data									
		5.1.1	Format of the data								
	5.2	Otter Ti	awl details								
		5.2.1	Format of the data								
Part VII-Schedule I: "NAFO port inspection report"

Page n°		of	
1. INSPECTION INFORMATION			
Inspection authority			
Date of the report			
Port and Country of inspection	Port Code:	Country Code:	

1.1 Format of the data

÷

Data Element	Code	M / 0	Туре	Content	Category ; Definition
Inspection authority	IA	М	Char*99	Text	Inspection detail : Name of the inspection authority
Date	DR	M	Num*8	YYYYM MDD	Inspection detail : Date the report is compiled
Country		М	FAO Code	Country Code	Vessel activity detail : Country where the vessel is discharging,
Port of inspection	LP	М	Char*99	Text/ ISO 3 alpha country code	Vessel activity detail : Place where the vessel is inspected : port followed by ISO –3 code of the country as "Boulogne-sur-mer / FRA"

Page n°			of	
	•	<u> </u>		

2. TRIP INFORMATION

To be filled in by the inspection authority as soon as the vessel land to port, based on logbook records.

Vessel name	
Trip number	
Date trip started	
Activity in the NAFO RA :	
Date Entry in the RA	
Date Exit from the RA	
Other areas visited	
Date trip ended	

2.1 Format of the data

Data Element	Code	M /0	Туре	Content	Category ; Definition
Vessel Name	NA	М	Char*30	ISO 8859.1	Vessel registration detail; name of the vessel
Vessel trip number	TN	М	Num*3	001-999	Vessel activity details : Number of the fishing trip in current year
Date trip started	TS	М	Num*8	YYYYMM DD	Vessel activity details : date started the current fishing trip
Date Entry in the RA	NE	М	Num*8	YYYYMM DD	Vessel activity details : Date the vessel entered the NRA for the current fishing trip
Date Exit from the RA	NX	М	Num*8	YYYYMM DD	Vessel activity details : Date the vessel exited from the NRA for the current fishing trip
Other areas visited	RF	0	Char*255	Text	Vessel activity detail : other area where vessel have been fishing during the current trip
Date trip Ended	TE	M	num*8	YYYYMM DD	Vessel activity details : date ended the current fishing trip

Page n°	 of	,

3. VESSEL IDENTIFICATION

To be filled in based on the licence information.

External Identification	 	
International Radio Call Sign	 	
Flag State		
NAFO Contracting Party	***,	
Home port	······································	
Vessel owner		· · ·
Vessel operator	 	
Master name		

Page n°

3.1 Format of the data

Data Element	Code	M /O	Туре	Content	Category ; Definition
External Identification Number	XR	М	Char*14	ISO 8859.1	Vessel registration details : Side Number of the vessel
International Radio Call Sign	RC	М	Char*7	IRCS Code	Vessel registration details : International Radio Call Sign of the vessel
Flag State	FS	M	Char*3	ISO-3166	Vessel registration detail; State where the vessel is registered, 3- ISO country code
NAFO Contracting Party	СР	0 (1)	Char*3	ISO-3166	Vessel registration detail :NAFO contracting party of the vessel, as ISO code of the country, EUR for European Community, NCP for Non Contracting Party
Home port	PO	0	Char*20	ISO 8859.1	Vessel registration details : Port of registration of the vessel or homeport
Vessel owner	vo	М	Char*60	ISO 8859.1	Vessel registration details : name and address of the vessel owner
Vessel operator	VC	M (2)	Char*60	ISO 8859.1	Vessel registration details : responsible for using the vessel
Master name	MA	0	Char*30	ISO 8859.1	Vessel activity details : name of the master

mandatory when use as single identification in other messages .
if different from vessel owner

of

Page n°			of	
4. RESULT OF PORT INSPECT	TION			
To be filled in after completion of land	ding			
4.1 General information				
Start of landing:	Date] Time	
End of landing :	Date		Time	
Has vessel landed all catches on board ?	YES] If YES, fill i	n table 4.2
	NO] IF NO, fill t	able 4.3
Comments		•		

185

4.1.1 Format of the data

.

- -

Data Element	Code	M /O	Туре	Content	Category ; Definition
Start date of landing	LS	М	num*8	YYYYM MDD	Landing detail : date the vessel started landing
End date of landing	LE	М	Char*1	T, S, P	Landing detail : date the vessel finished landing
Has vessel landed all catches on board ?	QQ	М	Char*1	Y, N	Landing detail : Has vessel landed all catches on board ?, answer Y if yes, N if not
Comments	со	0	Char*25 5	Text	Landing detail : comments as necessary.
					If landing has not been completed, please give an estimation on catch still on board

Page n°

		of
--	--	----

4.2. Quantity landed

Species (FAO Code)	Presentation	Live Weight (Log Book, Kg)	Conversion factor	Landing Processed Wt	Equivalent live weight (kg)	Diff (Kg)	Diff (%)
				(kg)	<u>_</u>		ļ
					· · · · · · · · · · · · · · · · · · ·		
	<u> </u>	<u></u>	l 				
	l		·			<u> </u>	
Comments	•	<u> </u>					•

4.2.1 Format of the data

Note : Quantities should be mention in regard to the species concerned and with reference to the nature of the information, e.g. : COD/OB350/PW320/D150/BC8,2.

Data Element	Code	M/0	Type	Content	Category ; Definition
Species	FI	М	Char*3	FAO species code	Landing detail : FAO 3-alpha code (Part V, Schedule II, Attachment II)
Presentation	FP	М	Char*5	Product form code	Landing detail : Product form code, as mention in attachment Z, codes being associated were necessary, i.e : gutted (G) head off (H) skin off (P)-frozen (F) : GHPF
Live Weight	↓	м	Num*5	0-99999	Quantities determined from the log-book.
Conversion factor	CF	0	Num*3	0,00-9,99	Product detail : Conversion factor as define by the master for the corresponding species, size and presentation, optional if already mention in table B
Process weight	PW	М	Num*5	0-99999	Landing detail : Quantities landed by species and presentation, in kilograms of product, rounded to the nearest 10 kg
Equivalent live weight	LW	М	Num*5	0-99999	Landing detail : Quantities landed in equivalent live weight, as "product weight x conversion factor", in kilograms, rounded to the nearest 10 kg
Comments	MS		Char*25 5	ISO 8859.1	Landing Details : free text area

Page n°	of	
l		

4.3 Quantity staying on board the vessel

,

Tol	be.	filled where	part of the	catches stav	on board aft	er completion of l	anding
-----	-----	--------------	-------------	--------------	--------------	--------------------	--------

Species	Presentation	Conversion factor	Process weight (kg)	Equivalent live weight (kg)
·····			· · · · · · · · · · · · · · · · · · ·	
·		. <u> </u>		
			•	
		· ·		

Comments	·	

Page n°

of

4.3.1 Format of the data

Note : Quantities should be mentioned in regard to the species concerned and with reference to the nature of the information, e.g. : COD/OB350/PW320/DI50/BC8,2.

Data Element	Code	M /O	Туре	Content	Category ; Definition
Species	FI	M	Char*3	FAO species code	Landing detail : FAO 3-alpha code (Part V, Schedule II, Attachment II)
Presentation	FP	M	Char*5	Product form code	Landing detail : Product form code, as mention in attachment Z, codes being associated were necessary, i.e : gutted (G) head off (H) skin off (P)-frozen (F) : GHPF
Conversion factor	CF	0	Num*3	0,00-9,99	Product detail : Conversion factor as define by the master for the corresponding species, size and presentation, optional if already mention in table B
Process weight	PW	M .	Num*5	0-99999	Landing detail : Quantities landed by species and presentation, in kilograms of product, rounded to the nearest 10 kg
Equivalent live weight	LW	M	Num*5	0-99999	Landing detail : Quantities landed in equivalent live weight, as "product weight x conversion factor", in kilograms, rounded to the nearest 10 kg
Comments	MS		Char*25 5	ISO 8859.1	Landing Details : free text area

Page n°	of	

189

5. GEAR INSPECTION IN PORT

Verification shall be done when non compliance have been cited / observed during inspection at sea.

To be filled in when port inspection will also concerned inspection of gears on board. A detail form shall be filled in for every gear having been subject to port inspection

5.1 General data

Number of gear inspected	
Date gear inspection	
Has the vessel been cited ?If Yes, complete the full "verification of inspection in port" form.If No, complete the form with the exception of the NAFO Seal Details.	Yes No

5.1.1 Format of the data

Data Element	Code	M /O	Туре	Content	Category ; Definition
Date of inspection	DR	М	Num*8	YYYYM MDD	Inspection detail : Date of current gear inspection
Inspected gear	IG	М	Num*2	00-99	Inspection detail : number of gear checked during port inspection

Ĵ

Page n°				of		
5.2 Otter Trawl details			r [•]			
NAFO Seal number	·····			 	·	
Is seal undamaged ?	Yes	[ю	
Gear Type:						
Attachments:				 	,	
Grate Bar Spacing (mm)						
Mesh Type:				 		

Average mesh sizes (mm)

TRAWL PART		· · ·	
Wings:			
Body:		· · · · · · · · · · · · · · · · · · ·	
Lengthening. Piece:	,		
Codend:		· · · · · · · · · · · · · · · · · · ·	

Page n°

of

191

5.2.1. Format of the data

Data Element	Code	M /O	Туре	Content	Category ; Definition
NAFO seal number	NS	M (1)	Num*8		Inspection detail (If required) : Number of the NAFO seal attached to the gear after inspection at sea
Is Seal Undamaged ?			Char*1	'Y' or 'N'	Whether NAFO inspection seal is intact.
Gear type	GE	М	Char*3	FAO Code	International Standard Statistical Classification of the Fishing Gear , OTB for otter trawl
Attachments					Otter trawl detail : attachment to footrope
Grade bar spacing	GB	М	Num*2	01-99	Otter trawl detail : grade bar spacing in millimetres
Mesh type	GT	М	Char*30	SQ, DI,	Otter trawl detail : respectively mesh type: SQ for square mesh , DI for diamant mesh
Mesh size average	GS	М			Otter trawl detail : average mesh size in the trawl part, by pair
Trawl part		М	Char*3	Wng, bod, lep, cod	Trawl part measured
Mesh size		М	Num*3	001-999	Mesh size in millimetres

:

. .

SECTION VI (pages 193 to 214)

Report of the STACTIC Technical Working Group on Communications 30 June 2000 Dartmouth, N.S., Canada

Report of the l	Meeting	195	
1.	Opening of the Meeting	195	
2.	Election of Chairman		
3.	Appointment of Rapporteur 1		
4.	Adoption of the Agenda		
5.	Consideration of a More Effective Hail System	195	
6.	Other Matters	199	
7.	Adjournment	199	
	Annex 1. List of Participants	200	
	Annex 2. Agenda		
	Annex 3. Presentation for an automated NAFO hail message processing system as proposed by Software Kinetics		
	for use in providing a fully automated hail system	203	
	Annex 4. Draft amendment to the Conservation and Enforcement		
	Measures	211	



Report of the STACTIC Technical Working Group on Communications

(FC Doc. 00/5)

30 June 2000 Dartmouth, N.S., Canada

1. Opening of the Meeting

The meeting was opened by the Executive Secretary at 1010 on 30 June. The Contracting Parties represented were Canada, Denmark (In respect of Greenland and the Faroe Islands), Estonia, the European Union (EU), Iceland, Japan and Russia. (Annex 1)

2. Election of Chairman

Mr. F. Wieland (EU) was elected by consensus as Chairman

3. Appointment of Rapporteur

Dr. K. Patterson (EU) was appointed Rapporteur

4. Adoption of the Agenda

After discussion the Agenda at Annex 2 was adopted.

5. Consideration of a More Effective Hail System

5.1 Background

The Executive Secretary briefly reviewed the history of the NAFO hail system noting that after an extended consultation process, the existing hail report format had been agreed on in 1991and incorporated in the Conservation and Enforcement Measures. In 1998 standardized formats for the electronic transmission of hails and satellite tracking reports for reporting from satellite systems were agreed for implementation in the NAFO Regulatory Area. At present, however, few Contracting Parties provide information in this latter format: principally Iceland and Norway. Other Contracting Parties still use previous hail system format.

The hail system was reported to be working effectively, but is a manual system based on oldfashioned technology which is causing an excessive workload for the NAFO Secretariat. An automated, internet-based system would be preferred, and a proposal was prepared (Annex 3). The Secretariat was aware that other projects and proposals from European users exist. Overall however, the Secretariat stresses its desire to move to a modern automated system to ensure accuracy of transmission, appropriate data storage and good handling of the data.

Responding to this review, the Chair asked for comments as to whether the Working Group could make a decision on this issue. The Danish representative was of the opinion that it could only prepare a proposal for the NAFO annual meeting. The EU representative thought that there was sufficient technical expertise present at the Working Group to make firm, well-founded proposals, and pointed out that an appropriate technology and associated systems already exist and thought that there was no need to develop new systems.

5.2. Presentation of Working Documents

Following a procedural discussion it was concluded that issues of communication, security, and compliance could be presented together and then discussed together, as the issues were so interdependent. The Working Group proceeded to presentations of working papers.

Working Paper 00/1 (Annex 3) was presented by a representative of the company "Software Kinetics" This proposal involves communication of encrypted data (using a Pretty Good Privacy (PGP) protocol) over the internet, protecting the content and origin of hail messages. A firewall would also be used for the Secretariat site. An MS-Access database would be designed to store and access the data.

The claimed advantages of this system were :

- The general availability of Internet access
- The low cost of communication
- The minimal dependence upon particular technologies as no proprietary protocols are used.

The representative from Denmark asked whether X25 and X400 communication protocols were considered. The representative answered that X25 was not in common use in North America and has higher costs, and SMTP was thought to provide all the necessary facilities for the proposal.

The EU representative asked whether the analysis of requirements was based on user requirements or on current practices. The representative answered that the starting point for the proposal was the existing system at the NAFO Secretariat, although it was understood that a variety of systems are in current use. None of the systems proposed required more than a properly-configured PC. The PGP programme is freely available. The generation of originating messages is possible from any modern hardware platform.

The EU representative further asked whether existing fishing fleet has many vessels using systems based on X25 and North Atlantic Format, questioned the need to use an internet based system, and asked how control of information into and out of the Secretariat would be managed.

A response was made by Canada to the effect that the internet is now a very common, widely used communication medium which is still suitable for secure communications (and is currently used by banks).

The Chair also questioned whether security considerations were adequately addressed, considering that the internet is a public communications medium. The Software Kinetics representative considered that PGP systems based on public and private keys provides sufficient and appropriate security. The concept of secure protocols based on public and private keys was explained, which affords:

- Privacy of transmission
- Security of origin
- Security of access (message originators cannot read each others' messages, whilst the authorised message recipients alone can read messages).

Denmark accepted that secure communications are possible in the internet environment. However, it was stressed that that a Vessel Monitoring System (VMS) already exists, based on X25 and X400 protocols, and that the system works already. There was no wish to introduce a third protocol, and the X25 could be retained for NAFO use. VMS provides close monitoring of

fishermen, and so confidentiality is an extremely sensitive issue. Denmark stressed its preference for a system based on already-existing protocols on secured lines. The Icelandic and EU representatives concurred.

Canada also expressed concern about asking vessels to provide messages in different formats. It was noted that the NAFO/Software Kinetics proposal involves separation of formatting issues from the issues of communication. One possibility was that flexibility of transmission methods between fishing vessel and Contracting Party could be maintained, while maintaining standardisation of transmission of reports between Contracting Parties and the NAFO Secretariat, based on internet protocols. This proposal would not affect the transmission of data from vessels to Contracting Parties. The format of the records of data transmitted to NAFO would be standardised in text format, regardless of mode of transmission between the vessels and the Contracting Parties' administrations. This would require no change to current vessel procedures.

The Executive Secretary briefly presented a general overview of the current NAFO procedures of receiving reports (by facs., e-mail, and new file transfer protocol from EU) at the NAFO Secretariat. Incoming reports are processed manually. Once daily, the NAFO secretariat compiles and circulates reports to parties with inspection presence (by e-mail to Canada, by CUTE FTP to EU (lately, temporary arrangements due to Y2K problems with X25). Most vessels in NAFO area are EU vessels and so most data is presently sent by internet. It was stressed that was an important decision that the hail system (used for enforcement purposes) should be managed by an impartial party: the NAFO Secretariat. NAFO system is however old fashioned, too labour intensive and should be updated to use modern systems.

The EU presented Working Paper 00/3 (Annex 4). The EU paper indicates changes necessary to update regulations once the Vessel Management System becomes fully operational, and includes a list of changes to part III.E of NAFO Conservation and Enforcement Measures which will be required on replacing the present hail system with a satellite-based system. The system proposed is based on the existing North Atlantic VMS. A 6hr reporting frequency is proposed, as used in East Atlantic, and includes conditions for the security and storage of data. The transition would be simple, as is based on introducing technology now already used by the fleets. Fisheries Monitoring Centres (FMCs) of Contracting Parties are already established to use this system. This format is also being introduced in the South Atlantic, and the EU is very keen to have a unified system for the whole Atlantic for simplicity. EU stated its intention that Working Paper 00/3 will form the basis of a formal proposal at the Annual Meeting.

The representative of Denmark (in respect of Faroe Islands and Greenland) thought the EU paper was a very good starting point for discussions as it reflects Denmark's position. The NAFO system should be identical to the NEAFC system.

For the record, the Norwegian Representative noted (in comments to the draft report) that the Norwegian position concurs with the views expressed by EU and also by Denmark (on behalf of Greenland and the Faroe Islands), i.e. as EU points out according to the last paragraph on page 3 "... that an appropriate technology and associated systems already exist and thought there was no need to develop new systems".

Canada considered that the EU paper was comprehensive and had no substantive objections; however concerns were raised about some requirements. It was thought that the EU paper is not consistent with present hail system. Item 1c presently required hails to be transmitted <u>prior</u> to movement. Canada was concerned at the proposed change to real-time reporting to anticipatory reporting (where an intention to move is reported) in the present hail systems. Canada requires time (to clarify its position on this topic). Further concerns were:

- a) The length of time during which the vessel may fish without the satellite tracking operating
- b) The extension to "not later than 24-hr" in reporting requirement.
- c) Referring to vessels fishing in 3L, it is asked whether the specific hail requirement (Part I.K.10 of Conservation and Enforcement Measures) would be maintained.

The Japanese representative questioned why the EU proposal was to delete the whole of paragraph 4 of Part III E from the Conservation and Enforcement Measures. The EU representative responded that when VMS is in operation, certain hail messages and parts of hail messages will become redundant, however, there will still remain a need for the hail system in the context of entry and exit messages.

The Japanese representative asked for clarification of the "real-time" concept, in the context of the 24hr maximum reporting requirement. Also, the difficulty of providing data within 24hrs over a weekend was pointed out. The EU representative responded that under normal circumstances a vessel would provide its location automatically every 6hrs. If a breakdown occurred, a manual response would be required by the Fisheries Monitoring Centre (FMC) of the Contracting Party. The 24hr limit should be viewed in that context and not in the context of normal operations. The Icelandic representative stated that in the NEAFC automated VMS system, position reports are transmitted immediately (within about 2 mins.). However, a longer (e.g. 24hr) period is allowed in the event of mechanical failure and the case that data need to be sent manually. "Real-time" reports are reports sent with delay incurred only due to the communications link (ie no significant delay).

Canada agreed with EU focus on automating data transmission, but urged care in defining minimum delays to avoid deterioration of the timeliness of the hail reports. Excessive delays in reporting rendered the hail information nearly useless for control purposes. The Icelandic representative stressed that the NEAFC fully automatic system was already operative and asked NAFO to review this system attentively. A brief presentation of the protocols used in the system was made.

The EU representative noted that the Secretariat could develop a facility to receive reports in several different protocol. Denmark concurred, and urged the Secretariat to explore more widely the availability of commercial communications software. It was questioned whether the Secretariat has appropriate finances. The Chair noted that about CAN \$ 35 000 has been allocated to this topic. Denmark considered a much higher budget would be required, possibly up to CAN \$150 000. A trained computer expert such as a database administrator working part time for the Secretariat could be required to maintain the system at the NAFO Secretariat.

5.3. Conclusions

A consensus was reached that an automation of the hail system would be required. A distinction was made between data flows:

- from vessel to Contracting Party
- from Contracting Party to NAFO Secretariat
- from NAFO Secretariat to Contracting Parties with inspection presence:

It remained open whether all three data streams required automating.

There were different views, focussing on the relative desirability of X-25 based systems or of internet SMTP. The group concluded that further reflection on the issues was needed, but it was agreed that:

- Confidential treatment of data is an issue of main importance. In particular, Japan noted that 6hr reporting is more detailed then existing requirements and may involve greater confidentiality requirements. Japan also required further discussion of the issue of confidentiality of fishing positions, both with respect of fishing vessels and of the NAFO Secretariat. Denmark noted further concern about security and confidentiality and will make a proposal on this topic in due course.
- The NAFO Secretariat is to be asked to review the existing NEAFC communications and data distribution system with a view to its possible applicability in the NAFO regulatory area. This could be aided by a visit by the NAFO Secretariat to the NEAFC Secretariat to review the operation of the existing NEAFC system.
- A cost-benefit evaluation of the various systems might be necessary and should be raised in STACTIC at the 2000 NAFO Annual Meeting, where operational recommendations could be made to the Fisheries Commission.

Further proposals and contributions on this issue could be made at the Annual meeting.

6. Other Matters

No other matters were raised.

7. Adjournment of the Meeting

The Meeting was adjourned at 1255 on 30 June 2000.

Annex 1. List of Participants

CANADA

Head of Delegation

T. Blanchard, Chief, NAFO Unit, Fisheries Management, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland AlC 5X1

Phone: +709 772 0928 - Fax: +709 772 5983 - E-mail: blanchardt@dfo-mpo.gc.ca

Advisers

R. Cosh, Dept. of Fisheries and Oceans, Stn. 13W099, 200 Kent Street, Ottawa, Ontario K1A 0E6 Phone: +613 993 2055 - Fax +613 990- ; E-mail: <u>coshb@dfo-mpo.gc.ca</u>

L. Penney, Fishery Officer, Fisheries Management, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 3630 - Fax: +709 772 5983 - E-mail: penneyl@dfo-mpo.gc.ca

P-A. Richer, Senior Program Officer, Enforcement Br., Conservation & Protection Directorate, 13th Floor -13W111, 200 Kent St., Ottawa, Ontario K1A 0E6

Phone: +613 990 0094 - Fax: +613 941 2718 - E-mail: richerp@dfo-mpo.gc.ca

DENMARK (IN RESPECT OF THE FAROE ISLANDS & GREENLAND)

Head of Delegation

M. T. Nedergaard, Fiskerilicensinspektor, Head of Unit, Gronlands Fiskerilicenskontrol, Postbox 501, DK-3900 Nuuk, Greenland Phones 200 245277 For ± 200 222225 Franklin CEUK (201 al.

Phone: +299 345377 - Fax +299 323235 - E-mail: GFLK@gh.gl

ESTONIA

Head of Delegation

A. Soome, Ministry of the Environment, Fisheries Department, Marja Str. 4d, 10617 Tallinn Phone: +372 6112987 Fax: +372 6567599 E-mail: <u>ains@klab.envir.ee</u>

EUROPEAN UNION

Head of Delegation

F. Wieland, Deputy Head of Unit, International Fisheries Organizations and Fisheries Agreements; Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi Wetstraat 200, B-1049 Brussels, Belgium

Phone: +32 2 296 3205 - Fax: +32 2 299 4802 -E-mail: Friedrich.Wieland@cec.eu.int

Advisers

K. Patterson, European Commission, Directorate General for Fisheries, Rue de la Loi 200, B-1049 Brussels, Belgium

Phone: + 32 2 299 2179 - Fax: +32 2 295 5621 - E-mail: kenneth.patterson@cec.eu.int

B. O'Shea, European Commission, Directorate General for Fisheries, 200 Rue Joseph II 99, Rm 1/27, B-1049, Brussels, Belgium

Phone: +32 2 296 6748 - Fax: +32 2 296 2338 - E-mail: Brendan.O'Shea@DG14.be

Y. Auffret, Ministere de l'Agriculture et de la Peche, Direction des Peches Maritimes, 3 Place de Fontenoy, 75007 Paris, France

Phone: +33 1 49558245 - Fax: +33 1 49558200 - E-mail: vves.auffret@agriculture.gouv.fr

H. Pott, Bundesministerium fur Ernahrung, Landwirtschaft und Forsten, Rochusstr. 1, D-53125 Bonn, Germany Phone: +49 228 529 4124 - Fax: +49 228 529 4410 - E-mail: <u>hermann.pott@bml.bund.de</u>

J. Del Hierro, Subdireccion General de Inspeccion Pesquera, Secretaria General de Pesca Maritima, c/Castellama 112, Madrid, Spain

Phone: +34 91 3471847 - Fax: +34 91 3471512

ICELAND

Head of Delegation

G. Geirsson, Icelandic Coast Guard, P. O. Box 7120, 127 Reikjavik Phone: +354 511 2222 - Fax: +354 511 2244 - E-mail: gylfi@lhg.is

Adviser

G. Hannesson, Directorate of Fisheries, Ingilfsstraeti 1, 101 Reykjavik Phone: +354 569 7900 - Fax: +354 569 7991 - E-mail: greta@hafro.is

JAPAN

Head of Delegation

Y. Sakamoto, Deputy Director, Far Seas Fisheries Div., Resources Management Dept., Fisheries Agency, Government of Japan, 1-2-1Kasumigaseki, Chiyoda-ku, Tokyo 100-8907 Phone: +81 33 591 6582 - Fax: +81 33 591 5824

Adviser

N. Takagi, Director Executive Secretary, Japan Deep Sea Trawlers Association, Ogawacho-Yasuda Bldg.,
6 Kanda-Ogawacho, 3-Chome, Chiyoda-ku, Tokyo 101-0052
Phone: +81 33 291 8508 - Fax: + 81 33 233 3267 - E-mail: <u>nittoro@mx3.mesh.ne.jp</u>

RUSSIA

Head of Delegation

A. Okhanov, Representative of the Russian Federation in Canada on Fisheries, 47 Oceanview Dr., Bedford, Nova Scotia, Canada B4A 4C4 Phone: +902 832 9225 - Fax: +902 832 9608

NAFO SECRETARIAT

L. I. Chepel, Executive Secretary

G. M. Moulton, Statistical /Conservation Measures Officer

B. Cruikshank, Senior Secretary

F. E. Perry, Desktop Publishing/Documents Clerk

Annex 2. Agenda

- 1. Opening of Meeting
- 2. Election of Chairman
- 3. Appointment of Rapporteur
- 4. Adoption of Agenda
- 5. Consideration of a more effective hail system or a satellite-based vessel monitoring system
 - (a) Background Summary
 - (b) appropriate means on communication of reports
 - (c) security and confidentiality of reports
 - (d) means to ensure compliance with applicable measures
- 6. Other matters
- 7. Adjournment

Annex 3. Presentation for an automated NAFO hail message processing system as proposed by Software Kinetics for use in providing a fully automated hail system

(Technical W.G. Working Paper 00/1)

1. OVERVIEW

The Northwest Atlantic Fisheries Organization (NAFO) Secretariat receives reports (Hail Messages) on fishing vessel activities within the NAFO Convention Area from the NAFO Member Nations. The Hail Messages follow a well-defined and documented standard. The received Hajl Messages are checked and collated by the NAFO Secretariat and entered into a database. Newly received Hail Messages are passed onto third party organizations that are contracted to perform inspection duties following the NAFO Convention.

1.1 Current practice

The current practice for handling the Hail Messages is largely manual. The Hail Messages are received primarily by FAX, with some sent in the clear by e-mail, or file transfer over X.25. The Hail Messages are processed and entered into a Hail Message database manually. Once the messages are processed and saved the new position data is transmitted to the contracted inspection organizations. Periodically reports are generated by the database based upon the saved data.

2. DESIRED FUTURE PRACTICE

The intent is to automate the entire Hail message handling process in order to remove operator intervention and to provide the required level of security for the data. In order to meet this intent and achieve the desired goals the following processing steps would need to be performed by the Automated NAFO Hail message Processing System:

- 1. The NAFO Member Nations would send their Hail Messages to the NAFO Secretariat using e-mail with the Hail Message e-mails being digitally signed and encrypted for security reasons.
- 2. A computer system located at the NAFO Secretariat in Dartmouth, Nova Scotia, would receive the Hail Message e-mail and automatically verify the digital signature and decrypt accepted messages.
- 3. Verified Hail Message e-mails would then be checked for validity and completeness as per the defined Hail Message standard.
- 4. Hail Messages that are complete and free of errors would then be stored in the Hail Message database.
- 5. Once a Hail Message is saved in the database, a Forward Hail Message e-mail will be generated (based on the original hail message, but with the fields put in a normalized order), digitally signed, encrypted and made available to the third party inspection organizations.
- 6. A Return Hail Message would automatically be generated and sent to the original hail message e-mail sender when an error in format or content is detected (including an appropriate error code as applicable).

3. AUTOMATED PROCESSING SYSTEM CAPABILITIES

In order to address automated hail Message processing, a system is required with the capability to:

- provide appropriate security protection,
- receive Hail messages via e-mail,
- parse the received e-mail to verify Hail message validity and completeness,
- store the Hail message in a database,
- forward necessary details to inspection agencies, and
- generate error notifications for Hail Messages that do not meet format and content standards.

The following figure provides a graphical representation of the system.



Figure 3-1. Graphical Representation of the Proposed System

In addition to the capabilities noted above, the automated system would also have to satisfy the following requirements:

- 1. All Hail Message e-mails shall be protected against unauthorized modification or access.
- 2. Hail Messages transmitted via a public medium (e.g. the Internet) shall be encrypted to ensure confidentiality and authenticity.
- 3. Only Hail Messages that are complete and have validated data shall be entered into the Hail Message database.
- 4. Only Hail Messages that are complete and have validated data shall be transmitted to the appropriate inspection organizations.
- 5. Hail message sending parties shall be notified via e-mail regarding invalid Hail Messages.
- 6. The system hosting the database and processing the Hail messages shall aim to meet the criteria of a C-2-level trusted system.
- 7. The Hail message database shall be capable of producing reports from the collected data (the three reports currently being produced by the NAFO Secretariat shall be continued).
- 8. Hail Messages electronically received shall be processed automatically by the system.
- 9. Forward Hail Messages generated for the third party inspection organizations shall be accumulated on the system for later retrieval by those third party inspection organizations.
- 10. Return Hail Messages for electronically received Hail Messages shall be sent to the appropriate return address.

3.1 C2-Level Trusted System

C2 refers to a set of security policies that define how a secure system operates. The C2 evaluation process is separate from the C2 certification process. Certification applies to a particular installation, including hardware, software, and the environment that the system is in. It is up to an individual site to become C2 certified.

The security policy in C2 is known as Discretionary Access Control (DAC). C2 classification does not define a substantive security system in the sense of classified or unclassified data. In a C2 (DAC) system, owners have absolute discretion about whether or not others have access to their objects. In the Windows NT implementation, the basic idea is that users of the system own objects, have control over the protection of the objects they own, and are accountable for all their access-related actions. For example, in Windows NT, every object (file, Clipboard, window, and so on) has an owner; any owner can give or not give other users access to its objects. The system tracks (audits) your actions for the administrators (that is, the system administrator can track the objects you accessed, both successes and failures).

The requirements for A-, B-, C- and D-level secure products are outlined in the Trusted Computer System Evaluation Criteria (TCSEC) published by the National Computer Security Center (NCSC), This publication is commonly referred to as the "Orange Book", and is part of NSA's security "rainbow series". Security level requirements are open to interpretations that change over time. When undergoing evaluation, each vendor negotiates with the NSA about whether or not the details of its particular system implementation conform with the abstract security policy concepts in the NSA's books. The vendor must provide evidence that the requirements are being met.

4. PROPOSED SOLUTION

4.1. Hail Message Processing System

A PC workstation/server is the proposed hardware platform. This hardware platform would be a typical name brand Pentium PC (including 17" monitor, keyboard, mouse, graphics card, and 200W power supply, CD-ROM) with a 3 year warranty and include:

- 2 hard drives (at lest 2 Gb each) to allow mirroring of data,
- Tape backup device
- Uninteruptable Power Supply (UPS), and
- Appropriate hardware for Internet connectivity (this may be supplied by the ISP depending upon the type of Internet access selected).
 - A permanent Internet connection is preferred to allow real-time handling of Hail Messages and to ensure the availability of Forward Hail Messages to the inspection organizations. A high speed Internet connection is not required, as the actual amounts of data being transferred are very small.

The software needed to run on the proposed hardware platform would include:

- Microsoft NT Server
- Microsoft Office Pro (includes Microsoft Word, PowerPoint, Excel, Access)
- InterMail Post Office (mail server software)
- PGP (e-mail encryption and digital signature software)

Custom application software would be developed for the following processing tasks:

- Decrypting and verifying digital signatures using PGP
- Validating received Hail Messages
- Generating Forward and Return Hail Messages
- Entering validated Hail Message information into the database
- Encrypting and digitally signing outgoing e-mail messages
- Porting existing Access database to new Access database

4.2 Security Recommendation – Firewall

Additionally, the use of a firewall would improve security. Software Kinetics recommends that the NAFO Secretariat consider including a firewall as part of the solution. The firewall would be a commercially available software application that would run on a dedicated PC host running Unix or NT. The advantages of using a commercially available firewall are that the product is technically supported and easy to set-up and maintain.

A packet filtering firewall should be sufficient for the NAFO Secretariat's needs. The packet filtering firewall is the easiest and least expensive implementation of a commercially available firewall. The amount and type of traffic to be checked is minimal and thus the packet filtering firewall is preferred over a proxy filter firewall.

4.2.1 Additional Firewall Considerations

A low-end solution would be to use a router and implement a router control list. The Internet Service Provider typically controls the routers. This most likely mean that the NAFO Secretariat would have little or no control of or access to the router configuration. In addition, routers provide little or no customization capabilities.

A medium-end solution would be to procure a PC running Linux with a free-ware packet filter software application. This would allow the NAFO Secretariat to implement a cost effective method to implement a firewall but this solution adds additional manual maintenance support requirements over the long term.

5. COST ESTIMATES

All prices noted in this section are estimates and are not to be considered as a quote. These cost estimates are provided for discussion purposes only. A firm fixed price will be quoted at a later date following discussions with the NAFO Directorate regarding their preferences. These cost estimates do not include applicable taxes.

ITEM	ESTIMATED COST
Basic Pentium PC (including Windows NT Server OS and 3 year warranty	\$4000
UPS	\$500
Printer	\$500
Microsoft Office Professional	\$800
InterMail Post Office	\$1500 (see NOTE 1)
PGP software (encryption, digital signatures)	\$200
Internet conection (on-demand 28.8 communication line	\$1000 (See NOTE 2)
Firewall Hardware (basic Pentium PC)	\$2500
Firewall Software (Linux – Red Hat)	\$80 (see NOTE 3)
TOTAL:	\$11,080.00

The following table provides cost estimates for the hardware and software components of the system.

NOTE 1: This includes a license for > 10 mail accounts and software support and maintenance.

NOTE 2: Unlimited on-demand access to the Internet through an Internet Service Provider (ISP) requires the use of a normal business phone line (supplied by the phone company) and a router (supplied by the ISP). The cost of the router and its setup are included in the estimated cost noted in the table. The business phone line charges are not included in the cost estimate noted in the table. There is an ongoing service charge paid on a monthly basis for the unlimited access connection. Other communication options exist including ISDN and ASDL. This solution assumes that e-mail sent to the Inspection Organizations does not need to be stored locally on the NAFO mail server for pickup – the e-mail will be immediately delivered to the Inspection Organizations mailbox (wherever they chose to host their mail server). **Refer to section 6 of this document for additional connectivity discussions details**.

NOTE 3: Linux is a Unix like operating system that includes firewall software suitable for this application.

On-going charges will apply for such items as the business phone line (approximately \$60/month) and ISP unlimited on-demand service charges (approximately \$200/month). These on-going charges have not been included in the cost estimates noted in the table above. These on-going costs must be considered for future expenditure considerations.

Software Kinetics has assumed that the proposed system hardware and operating system software will be ordered by the NAFO Directorate and delivered to Software Kinetics. The installation, development and configuration of application software will occur at Software Kinetics using the procured system as a development platform. Once the complete system has been built and fully tested (i.e., a Factory Acceptance Test), it will then be installed and configured for use at the NAFO Directorate.

The following tasks will be performed:

- System configuration (system assembled at Software Kinetics site for testing and development purposes),
- Security Implementation (NT, mail accounts, user accounts, etc.)
- Design, develop and test the Hail Message handling application software,
- Perform Factory Acceptance Test, Site Acceptance Test (including development of test procedures),
- Port existing Access database to new platform,
- Create System User Manual,
- System installation at NAFO Directorate,
- Project management,
- Firewall configuration, set up and installation, and
- User training.

Software Kinetics estimates that these tasks will involve a Project Manager, up to 2 Software Specialists and a Systems Management specialist. The Project Manager would be responsible for managing the entire effort and ensuring Customer satisfaction. The Software Specialists would design, develop and test the application, port the existing Access database to the new platform, and create the user documentation. The Systems Management Specialist would setup and configure the computer hardware, the operating systems and the firewall. One of the Software Specialists would provide the user training.

User training will be a day in duration and will be performed on the NAFO site.

The estimated labour cost to perform the work noted above is \$35,000.00.

The estimated schedule to complete this work would be 3 months after receipt of order.

6. SUPPORTING INFORMATION AND COMMENTS

6.1 Internet Connectivity Issues

Software Kinetics believes that the NAFO Directorate needs to consider the use of a dedicated Internet connection to provide a complete and fully secure service (i.e., a permanent ISDN connection, an ASDL connection, or cable access). The on-demand approach included in the cost estimates can provide appropriate Internet connectivity but:

- requires software intervention to automatically connect to the ISP for sending and receiving email, and
- potentially poses some e-mail security concerns (e-mail residing on ISP mail servers).

Alternative methods do exist based upon the service provided by Dartmouth/Halifax area ISP's, MT&T and Shaw Cablevision. The Internet services provided by these organizations have advantages and disadvantages given the NAFO Secretariat requirements. These possibilities require further investigation and discussion. Costs for the various alternatives have not been fully determined.

For on-demand service comparison purposes, a 64Kbps on-demand ISDN connection would cost approximately \$1300 to setup and configure with a monthly charge of \$300. A 128 Kbps on-demand ISDN connection would cost approximately \$1200 to setup and configure with a monthly charge of \$450.

The ISP provides complete access to Internet services including e-mail accounts, newsgroups, file transfer and the WWW as part of their basic service. The ISP will also register your unique domain name, supply both primary and secondary domain name resolution, and unlimited telephone and e-mail customer support.

ISP's recommend that a firewall be used to provide increased security.

Should additional PC's within the NAFO Secretariat need to be connected to Internet communications line then a hub would need to be purchased and configures. The hub would reside behind the firewall (on the NAFO side) and allow for multiple LAN's or PC's to be connected to it.

In addition, some ISP's will provide and support Firewalls. This aspect of their service provision capabilities is still under investigation.

6.2 Microsoft Access

The current database is implemented using Microsoft Access. Continuing to use Access allows the NAFO Secretariat staff to make use of their existing skills.

The reports currently required by the NAFO Directorate are already implemented using Access. The effort required to re-implement these reports on the new system is minimal.

6.3 Operating Systems and Security

Microsoft NT is not C2 rated but still is a logical choice for the operating system even though the aim is to be C2 rated. Currently there are no interactive operating systems that are C2 rated while connected to a network.

6.4 Mail Server Software

InterMail Post.Office allows incoming mail to be sent to a running application and allows for the creation of e-mail accounts that are independent of NT login accounts. This e-mail package includes a POP server and an integrated mail list manager. Consideration needs to be given to purchasing a 100 user license in order to receive software maintenance and support privileges.

6.5 Encryption Software

PGP is a logical choice for the encryption application since it is well tested, readily available on a variety of platforms, and compatible versions are available from sources not subject to US encryption software export restrictions.

6.6 UPS

The UPS, the multiple disk drives (for database replication) and the backup tape drive are included to increase system reliability and data integrity. A mid-range UPS allowing for the graceful shutdown of the Computer system was used for cost estimation purposes. A variety of UPS models are available with varying costs and features.

Annex 4. Draft amendment to the Conservation and Enforcement Measures (NAFO/FC Doc. OO/1)

(Technical W.G. Working Paper 00/3 presented by the European Union)

Introduction

Contracting Parties have agreed to require all vessels fishing in the Regulatory Area to be equipped with satellite tracking devices not later than January 1, 2001. The purpose of the amendment is to adopt detailed rules for satellite tracking and to adjust the Hail System requirements since certain report types become superfluous with satellite tracking. The proposed detailed measures for satellite tracking are identical to the ones contained in the Scheme of control and enforcement applicable in the NEAFC regulatory area. The amendment should be applicable as from January 1, 2001.

Draft amendment (enters into force as from January 1, 2001)

- In Part III.E (- Vessel requirements / <u>Hail System</u>):

- sub-paragraph 1(c) is deleted
- sub-paragraph 1(d) is deleted
- sub-paragraph 1(e) becomes sub-paragraph 1(c)
- sub-paragraph 4 is deleted

- In Part III – Annex I – Hail System message format

- sub-paragraph 1.2 is deleted
- sub-paragraph 1.3 is deleted
- sub-paragraph 1.4 becomes sub-paragraph 1.2
- sub-paragraph 1.5 becomes sub-paragraph 1.3
- in Example 1 point 1.2 is deleted
- in Example 1 point 1.3 is deleted
- in Example 1 point 1.4 becomes point 1.2
- in Example 1 point 1.5 becomes point 1.3
- Example 2 is deleted

- At the end of Part III - Vessel requirements, a new section F. is added:

F. Vessel monitoring system (VMS)

1. Each Contracting party shall ensure that each of its vessels operating in the Regulatory Area is equipped with a satellite tracking device allowing the continuous tracking of its position by the Contracting party.

To that end the satellite tracking device shall ensure the automatic communication at least once every six hours when operating in the Regulatory Area to a land-based fisheries monitoring centre (hereafter referred to as FMC) of data relating to:

- the vessel identification;
 - the most recent geographical position of the vessel (longitude, latitude) with a position error which shall be less than 500 metres, with a confidence interval of 99%;
- the date and time of the fixing of the said position of the vessel.

Each Contracting Party shall take the necessary measures to ensure that its FMC receives these data.

2. The FMC of each Contracting Party shall be equipped with computer hardware and software enabling automatic data processing and electronic data transmission. Each Contracting Party shall provide for back-up and recovery procedures in case of system failures.

3. Each Contracting Party shall take the necessary measures to ensure that the VMS data received from its fishing vessels are recorded in computer readable form for a period of three years.

4. The masters of fishing vessels shall ensure that the satellite tracking devices are at all times fully operational and that the information in paragraph 1. is transmitted. In the event of a technical failure or non-operation of the satellite tracking device fitted on board a fishing vessel, the device shall be repaired or replaced within one month. After this period, the master of a fishing vessel shall not be authorised to commence a fishing trip with a defective satellite tracking device. Where a device stops functioning and a fishing trip lasts more than one month, the repair or the replacement has to take place as soon as the vessel enters a port, the fishing vessel shall not be authorised to continue or commence a fishing trip without the satellite tracking device having been repaired or replaced.

5. Each Contracting Party shall ensure that a fishing vessel with a defective satellite tracking device shall communicate, at least daily, reports containing the information in paragraph 1. to the FMC, by other means of communication (radio, facsimile or telex).

6. Each Contracting Party shall communicate reports and messages pursuant to paragraph 1. and paragraph 5. to the Secretary as soon as possible, but not later than 24 hours after receipt of those reports and messages. If the Contracting Party so desires, it shall ensure that each of its fishing vessels shall communicate reports (by satellite, radio, facsimile or telex) to the Secretary.

7. Each Contracting Party shall ensure that the reports and messages transmitted between the Contracting Party and the Secretary or if the Contracting Party so desires, between its fishing vessels and the Secretary, shall be in accordance with the data exchange format set out in Annex II.

8. The Secretary shall make available as soon as possible the information received under paragraph 7. to other Contracting Parties with an active inspection presence in the Regulatory Area. All reports and messages shall be treated in a confidential manner.

9. Each Contracting Party shall notify the name, address, telephone, telex and facsimile numbers as well as the addresses for electronic communication of their relevant authorities to the Secretary before 1 January 2001 and thereafter any changes without delay.

Data Element:	Code:	Mandatory / Optional	Remarks:
Start record	SR	M	System detail; indicates start of record
Address	AD	М	Message detail; destination; XNS" for NAFO Secretariat
Sequence Number	SQ	0	Message detail; message serial number in current year
Type of Message	ТМ	М	Message detail; message type, "POS" as Position report/message to be communicated by VMS or other means by vessels with a defective satellite tracking device
Radio call sign	RC	М	Vessel registration detail; international radio call sign of the vessel
Trip Number	TN	0	Activity detail; fishing trip serial number in current year
Vessel Name	NA	0	Vessel registration detail; name of the vessel
Contracting Party Internal Reference Number	IR	0	Vessel registration detail. Unique Contracting Party vessel number as ISO-3 flag state code followed by number
External Registration · Number	XR	0	Vessel registration detail; the side number of the vessel
Latitude	LA	M	Activity detail; position at time of transmission
Longitude	LO	M	Activity detail; position at time of transmission
Date	DA	M	Message detail; date of transmission
Time	· TI	M	Message detail; time of transmission
Record Date	RD	M	Year, month and date
Record Time	RT	<u>M</u>	Hours and minutes in UTC
Record Number	RN	M	Serial number of the record in the relevant year
From	FR	M	Address of the transmitting party (Contracting Party)
End of record	ER	M	System detail; indicates end of the record

After Part III Annex I Hail System Message Format, a new annex is added: Part III Annex II VMS position report format

Each data transmission is structured as follows:

- double slash ("//") and the characters "SR" indicate the start of a message;
- a double slash ("//") and field code indicate the start of a data element;
- a single slash ("/") separates the field code and the data;
- pairs of data are separated by space;
- the characters "ER" and a double slash ("//") indicate the end of a record
- In Part VI.B (Satellite tracking)
- Sub-paragraph 1c is deleted
- Sub-paragraph 1d is deleted.

· .

·

· ·

SECTION VII

(pages 215 to 276)

Report of the General Council and its Subsidiary Bodies (STACFAD and STACFAC), 22nd Annual Meeting 18-22 September 2000 Boston, Massachusetts, USA

PART I.	Report of the General Council				
	1.	Opening of the Meeting	7		
	2.	Supervision and Coordination of the Organizational, Administrative			
		and Other Internal Affairs	3		
	3.	Coordination of External Relations			
	4.	Fishing Activities in the Regulatory Area Adverse to the Objectives			
		of the NAFO Convention)		
	5.	Finance			
	6.	Closing Procedures	2		
		Annex 1. List of Participants	3		
	·	Annex 2. Opening Statement by the President of NAFO –			
		E. Oltuski (Cuba) 236	5		
		Annex 3. Speech of the Honorable Rolland A. Schmitten,			
		Deputy Assistant Secretary, International Affairs, U.S.			
		Dept. of Commerce 237	7		
		Annex 4. Opening Statement by the Representative of Canada (P. Chamut))		
		Annex 5. Opening Statement by the Representative of the			
		European Union (E. Mastracchio) 241	I		
		Annex 6. Opening Statement by the Representative of the			
		United States of America (P. Kurkul) 242	2		
		Annex 7. Opening Statement by the Representative of France			
		on behalf of St. Pierre et Miquelon (G. Grignon) 242			
		Annex 8. Agenda 244			
		Annex 9. NAFO Schedule of Meetings, 2001 246			
		Annex 10. Press Release			
		Annex 11. List of Decisions and Actions by the General Council 250)		
PART II.	Rep	port of the Standing Committee on Finance and Administration (STACFAD) 252	1		
	1.	Opening by the Chairman	1		
	2.	Appointment of Rapporteur			
	3.	Adoption of Agenda			
	4.	Auditors' Report for 1999 25			
	5.	Meeting of the Pension Society			
	6.	Review of Cost Implications of the Hail and Satellite Tracking			
		Systems at the NAFO Secretariat 252			
	7.	Administrative and Financial Statements for 2000 (July) 252	3		
	8.	Review of the Accumulated Surplus Account			
	9.	Budget Estimate for 2001 255			
	10.	Preliminary Budget Forecast for 2002 257	7		

		Time and Place of 2002-2003 Annual Meetings	
	13.	Council during the current Annual Meeting	
		Annex 1. List of Participants 25 Annex 2. Agenda 25 Annex 3. Travel and Meeting Costs Estimate for the NAFO Secretariat 26 Annex 4. Schedule of Outstanding Contributions from Bulgaria 26 Annex 5. Budget Estimate for 2001 26 Annex 6. Preliminary Calculation of Billing for 2001 26 Annex 7. Preliminary Budget Forecast for 2002 26	59 50 52 53 54
PART III.		ort of the Standing Committee on Fishing Activities of non- ontracting Parties in the Regulatory Area (STACFAC)	66
	1.	Opening by the Chairman	
	2.	Appointment of Rapporteur	
	3.	Adoption of Agenda 26	6
	4.	Review of 2000 information on activities of non-Contracting	
		Party Vessels in the Regulatory Area 26	6
	5.	Review of 2000 information on landings and transhipments of fish	
		caught by non-Contracting Party vessels in the Regulatory Area 26	6
	6.	Review of 2000 information on imports of species regulated by NAFO	
		from non-Contracting Parties whose vessels have fished in the	
		Regulatory Area 26	57
	7.	Reports by Contracting Parties on diplomatic contacts with non-	
		Contracting Party Governments concerning fishing in the Regulatory	
		Area	7
	8.	Review of performance of the Scheme to deal with non-Contracting	,,
	0.	Parties fishing in the NAFO Regulatory Area	8
	9.	Report and Recommendations to the General Council	
		Election of Chairman and Vice-Chairman	
		Other Matters	
		Adjournment	
	12.		υ
		Annex 1. List of Participants	71
		Annex 2. Agenda	
	•	Annex 3. NAFO Letters	
			0

•
PART I

Report of the General Council Meeting (GC Doc. 00/7)

22nd Annual Meeting, 18-22 September 2000 Boston, Massachusetts, USA

1. Opening of the Meeting (items 1-5 of the Agenda)

- 1.1 The meeting was opened by the Chairman of the General Council, Enrique Oltuski (Cuba).
- 1.2 The Representatives of sixteen (16) Contracting Parties were present: Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland-DFG), Estonia, European Union, France (in respect of St. Pierre et Miquelon), Iceland, Japan, Republic of Korea, Latvia, Lithuania, Norway, Poland, Russia, Ukraine and the United States of America (Annex 1). Two (2) Contracting Parties Bulgaria and Romania were absent.
- 1.3 The Chairman welcomed the delegates to the 22nd Annual Meeting taking place in the historical city of Boston. He encouraged the delegates to work in the constructive spirit of cooperation (Annex 2).
- 1.4 The Honorable Rolland A. Schmitten, Deputy Assistant Secretary of International Affairs of the United States Department of Commerce, welcomed the NAFO delegates to Boston and the United States of America on behalf of his Government (Annex 3). Mr. Schmitten emphasized that the United States attaches great importance to NAFO as a successful regional fisheries management organization, both in its present conservation and management activities and its future ones.
- 1.5 The Heads of Delegations from (in order of presentation): Canada, European Union, United States and France (in respect of St. Pierre et Miquelon) addressed the Meeting (Annexes 4-7).
- 1.6 The meeting appointed the Executive Secretary as Rapporteur.
- 1.7 The Provisional Agenda was adopted as attached (Annex 8). The two items regarding Reports on Allocation of Fishing Rights and Shrimp Stocks were transferred to the Fisheries Commission agenda with the consent of Peter Gullestad (Norway), the Fisheries Commission Chairman, and two new sub-items e) and f) were added to Item 10 of the Agenda.
- 1.8 Admission of Observers was addressed by the Executive Secretary reporting on his invitations to FAO, ICCAT, ICES, NAMMCO, NASCO, NEAFC and NPAFC in accordance with new Rules of Procedure (9 and 10). These organizations acknowledged NAFO's invitations and all, except NAMMCO, advised that they would not be sending their observers to this NAFO meeting. NAMMCO was represented by Mr. K. Arnason (Iceland).

With regards to non-Contracting Parties harvesting fishery resources in the Regulatory Area, invitations have been sent to Belize, Honduras, Sao Tome e Principe and Sierra Leone. No attendance was noted.

1.9 On the item "Publicity", the meeting agreed to the normal procedure that no statements should be made to the media until after the conclusion of the meeting, when the NAFO Secretariat would issue a Press Release.

2. Supervision and Coordination of the Organizational, Administrative and other Internal Affairs (items 6-7)

- 2.1 The membership and Contracting Parties' credentials were reported by the Executive Secretary. The total membership of 18 Contracting Parties and 16 members of the Fisheries Commission were noted. The credentials of 220 NAFO delegates were notified to the NAFO Secretariat before 18 September 2000.
- 2.2 With regards to Bulgaria and Romania participation in NAFO business, it was noted that two Contracting Parties, Canada and Ukraine, held communication with those members. There was indication from Bulgaria to start their process of reinstatement in NAFO, and Romania advised on its disinterest in NAFO participation and intention of eventual withdrawal.
- 2.3 Item 7, "Administrative Report", was referred to STACFAD. At the closing session, the STACFAD Chairman reported its recommendation of adoption of the Report, which was accepted by the General Council.

3. Coordination of External Relations (items 8-10)

- 3.1 Under item 8, "Communication with the United Nations", the Chairman noted the NAFO paper (GF/00-344 of 11 June 00) presented to the UN Secretariat regarding "Large-scale pelagic drift-net fishing, unauthorized fishing in zones of national jurisdiction and on high seas, fisheries by-catch and discards, and other developments". The NAFO paper was developed through consultations and approval by NAFO members.
- 3.2 Under item 9, "FAO International Plans of Action on the Management of Fishing Capacity, Shark Fishery and Incidental Catch of Seabirds in Longline Fisheries", the European Union Representative and several delegations questioned the procedures for adoption of the USA proposal on this item by a mail vote intersessionally. The vote was conducted in accordance with NAFO Convention and Rules of Procedure. The European Union Representative thought that such voting decision to deliver copies of national FAO Reports to the NAFO Secretariat was in contradiction with the previous decision by the General Council to discuss this matter at the 22nd Annual Meeting (and then take a decision). In principle, the EU delegation was not objecting against cooperation with NAFO regarding FAO reports on this item.

The Representative of the United States and Estonia supported the decision to deliver copies of FAO reports to the NAFO Secretariat.

The heads of delegations decided at a separate meeting to recommend to Contracting Parties that their reports to FAO on this subject should be copied to the NAFO Secretariat, and the NAFO Secretariat, in turn, would circulate those reports to all Contracting Parties.

3.3 Item 10, "NAFO participation at other International Organizations", was presented by the Executive Secretary informing that Canadian observers represented NAFO at the FAO Consultations on CITES (1975 Convention on International Trade in Endangered Species of Flora and Fauna) and NASCO (North Atlantic Salmon Conservation Organization). The observer reports were circulated to the meeting.

218

Assistant Executive Secretary, T. Amaratunga, took part in FAO Working Party on Status and Trends of Fisheries (November 1999). His report was delivered to the Scientific Council June Meeting (SCS Doc. 00/15).

The NAMMCO Annual Meeting, 26-29 September 2000, will be attended by a delegate from Norway, and its report will be circulated to Contracting Parties.

The meeting decided to delegate NAFO observers to two additional meetings in 2000-2001. The Technical Consultations on Illegal, Unregulated, and Unreported Fishery (2-6 October 2000) will be covered by a Canadian delegate. The FAO Fisheries Committee Meeting (26 February - 02 March 2001) will be attended by observer from Iceland. Their reports will be circulated to Contracting Parties intersessionally.

4. Fishing Activities in the Regulatory Area Adverse to the Objectives of the NAFO Convention (items 11-13)

- 4.1 Under item 11, "Consideration of non-Contracting Parties activities in the NAFO Regulatory Area and agreement on the task of STACFAC at the current meeting", the Chairman of STACFAC, J-Pierre Plé (USA) briefed the Meeting on the status of STACFAC activity and actions. He was optimistic on the latest non-activity by former non-Contracting Parties (NCPs) in the NAFO Regulatory Area, but urged the meeting to maintain NAFO's steady actions against illegal fishing by NCPs. He noted a new trend of re-flagging/double flagging that posed danger of revival of NCP fishing in the Regulatory Area. The meeting agreed that STACAFC should continue its work and then report to the General Council.
- 4.2 Item 12, "STACFAC Report", was presented to the meeting by Mr. J-Pierre Plé, STACFAC Chairman, underlining the following major issues and recommendations:
 - a) There were no sightings of NCP vessels in the NAFO Regulatory Area in 2000 (and only two vessels in 1999); the Committee thought that "no sightings" does not necessarily imply an absence of such NCP activity;
 - b) There were no reports of landings or imports of species regulated by NAFO from NCPs in 2000;
 - c) During 1999-2000, NAFO diplomatic démarches were delivered to Honduras (by Canada); Sao Tome e Principe (by European Union); Belize and Sierra Leone (by USA). The results of those actions and communications with NCP authorities are presented in STACFAC Report (Part III). New diplomatic démarches were developed by STACFAC to Belize, Honduras, Panama and Sierra Leone.
 - d) The issue of Illegal, Unregulated and Unreported (IUU) Fishing should be under review of STACFAC pending on-going discussions at FAO;
 - e) The matter of possible double flagging arrangements to carry out fishing activities under two flags was a great concern of the Committee. Reference was made to Article 92 of UNCLOS, 1982.
- 4.3 STACFAC recommended the following actions and measures to the General Council.
 - Démarches, in the form of letters signed by the President of NAFO be made to Belize, Honduras, Panama, and Sierra Leone.

- The Executive Secretary circulate, as soon as possible to the secretariats of ICCAT, NEAFC, and CCAMLR reports of non-Contracting Party fishing activity in the NAFO Regulatory Area, and request such organizations to share reports of non-Contracting Party fishing activity in their respective areas with NAFO.
- The Executive Secretary circulate NAFO/GC Doc. 00/1 to ICCAT, NEAFC, and CCAMLR.
- The Contracting Parties submit a report at the next Annual Meeting on what legal, administrative and practical action they have taken to implement the Scheme.
- STACFAC undertake to study the impact of the proposed International Plan of Action on Illegal, Unregulated and Unreported (IUU) Fishing on NAFO at its next Annual Meeting.
- 4.4 The General Council adopted the STACFAC Report and its recommendations. The Chairman of the General Council signed the diplomatic démarches. The diplomatic démarches to Belize and Sierra Leone will be delivered by the USA delegation, and démarches to Honduras and Panama will be delivered by Canada.
- 4.5 The General Council was informed on new officers elected by STACFAC: Chairman, Mr. Daniel Silvestre (France-SPM), and Vice-Chairperson, Nadia Bouffard (Canada).
- 4.6 The General Council Representatives thanked Mr. J-Pierre Plé for his effective work as the STACFAC Chairman and attributed STACFAC progress to Mr. Plé's able chairmanship, hard work and vision. The Chairman wished him well on his new assignment.
- 4.7 Item 13, "Report of Working Group on Dispute Settlement Procedures (GC Doc. 00/4)", was presented to the meeting by the W.G. Chairman, Mr. Stein Owe (Norway). He summarized the discussions in a consolidated text of a proposal for a future DSP. The proposal contains text and alternatives in brackets reflecting the current level of views expressed to-date in the Working Group. He thought that progress was made to move further discussions. The European Union Representative noting the priority of this issue for the EU delegation, supported the DSP Report and thought that Contracting Parties should continue discussions to resolve this subject.

The Representative of Canada briefly reviewed the five (5) year history of DSP Working Group discussions noting that not much progress has been made as opposing views remained on significant issues concerning implementation of DSP within NAFO.

The Representative of Denmark (DFG) noted that the DSP Working Group moved far ahead and had good progress as there were not so many "square brackets" in the W.G. Report. He proposed to develop further guidelines to the Working Group by Heads of Delegations.

The Chairman summarized the discussions and proposed that the DSP W.G. Chair should undertake unofficial consultations with all interested parties and then report back to the General Council. The DSP Chair briefed the Heads of Delegations and then reported to the General Council. To his opinion, there was a high degree of support for the work of the DSP Working Group. However, it was recognized that there are still outstanding issues as witnessed by the brackets in the consolidated text of the proposal for a future DSP. The DSP Working Group Report (GC Doc. 00/4) was formally adopted by the General Council.

(

220

At the closing session, the General Council resolved to continue DSP discussions and call an intersessional Working Group in Dartmouth, N.S., Canada, 24-26 April 2001 (Annex 9). The Working Group will have to elect a new Chairman.

5. Finance (items 14-15)

- 5.1 Items 14-15 were referred to the Standing Committee on Finance and Administration (STACFAD). The STACFAD proceedings are attached in Part II of this General Council Report.
- 5.2 The Chairman of STACFAD, F. Kingston (EU) delivered its report to the General Council on 22 September and noted the following:
 - a) The Auditors Report 1999 was circulated to Heads of Delegations on 15 April 2000, and this report was recommended by STACFAD for adoption;
 - b) The NAFO Secretariat was represented at the Pension Society Meeting (the International Fisheries Commissions Society, IFCPS, of North America) by two staff members, and its report was reviewed at STACFAD. It was noted with satisfaction that a new set of investment guidelines for an improved asset unit of the Canadian Pension Plan Funds has been introduced during 1999/2000. These guidelines see a switch from 100% Guaranteed Investment Certificates (GIC's) to an asset unit of 50% equities and 50% bonds providing for a more stable investment base and return on investments.
 - c) The Executive Secretary was instructed to provide to the General Council a proposal not later than 31 December 2000 regarding the financing of the Automated Hail and Satellite Tracking System. A full report on the status of the spending under this budget item and implementation of the Automated Hail and Satellite Tracking System should be delivered at the 23rd Annual Meeting 2001.
 - d) The basic budgetary items of the NAFO Secretariat were agreed as follows:
 - the budget for 2001 to be adopted in the amount of \$1,389,000 Cdn.;
 - the Accumulated Surplus Account be maintained at a level not less than \$75,000 Cdn in order to fulfil NAFO's financial obligations in early 2001 until contributions are received;
 - the contributions from Bulgaria and Romania be deemed uncollectible and these amounts be applied against the Accumulated Surplus;
 - \$200,000 Cdn be allocated for the proposed Automated hail and Satellite Tracking System with a note that STACFAD was extremely concerned with the lack of information (from STACTIC and Fisheries Commission) upon which the Committee could properly evaluate this amount.
 - e) STACFAD recommended that Contracting Parties continue attempts to contact Bulgaria and Romania in order to ascertain whether they intend to participate in NAFO and to inform them of their outstanding contributions. Such information should be exchanged to all Contracting Parties through the NAFO Secretariat.
 - f) The Committee recommended to the General Council that Contracting Parties be urged to submit their fishery statistics (STATLANT 21A and 21B) to the NAFO Secretariat on time to ensure the ongoing integrity of the NAFO statistical database and scientific recommendations on fish stocks.

2001	- - -	Scientific Council General Council Fisheries Commission	- - -	12-21 September 17-21 September 17-21 September
2002	-	Scientific Council General Council Fisheries Commission	- - -	11-20 September 16-20 September 16-20 September
2003	-	Scientific Council General Council Fisheries Commission	- -	10-19 September 15-19 September 15-19 September

g) The dates of next Annual Meetings were recommended as follows:

- 5.3 The venue of the 23rd Annual Meeting 2001 will be in Cuba.
- 5.4 The venue of the 24th Annual Meeting 2002 was proposed by the European Union Delegation in concurrence with Spanish delegates to be held in Spain. The meeting accepted this proposal and the Chairman thanked the European Union and the Government of Spain for this invitation.
- 5.5 The General Council reviewed the STACFAD recommendations and adopted the 2001 budget and report as a whole.

The Representative of the United States noted their reservation regarding the allocation of \$200,000 Cdn to the Automated Hail and Satellite Tracking System without proper evaluation and justification of the amount of funds.

6. Closing Procedures (items 16-19)

- 6.1 Item 16, "Time and Place of Next Annual Meeting", was reported by STACFAD (above).
- 6.2 Under item 17, "Other Business", the Chairman introduced a "Schedule of NAFO Meetings 2001", which was adopted by the General Council (Annex 9).
- 6.3 The draft Press Release was circulated to the Heads of Delegations and then finalized by the Secretariat incorporating the comments from Contracting Parties (Annex 10).

6.4 The meeting adjourned at 1330 on 22 September 2000.

Annex 1. List of Participants

CANADA

Head of Delegation

P. S. Chamut, Assistant Deputy Minister, Dept. of Fisheries and Oceans, Fisheries Management, 200 Kent Street, Ottawa, Ontario K1A 0E6 Phone: +613 990 9864 - Fax: +613 990 9557

Representative

P. Chamut (see address above)

Advisers

C. J. Allen, Chief, Groundfish, Pelagics and Foreign Fisheries, Resource Management, Dept. of Fisheries and Oceans, Ottawa, Ontario K1A 0E6

Phone: +613 990 0105 - Fax: +613 990 7051 - E-mail: allenc@dfo-mpo.gc.ca

J. Angel, Canadian Association of Prawn Producers, P. O. Box 1C1, Head of St. Margarets Bay, N.S. B0J 1R0 Phone: +902 826 7765 - Fax: +902 826 7065 - E-mail: jangel@navnet.net

D. B. Atkinson, Northwest Atlantic Fisheries Centre, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 2052 - Fax: +709 772 4188 - E-mail: atkinsonb@dfo-mpo-gc.ca

J. W. Baird, Director, Resource Management Div., Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 4497 - Fax: +709 772 3628 - E:mail: bairdj@dfo-mpo.gc.ca

T. Blanchard, Chief, NAFO Unit, Fisheries Management, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland AIC 5X1

Phone: +709 772 0928 - Fax: +709 772 0008 - E-mail: blanchardt@dfo-mpo.gc.ca

D. R. Bollivar, Seafreez Foods Inc., 32 Beckfoot Drive, Dartmouth, N.S. B2Y 4C8

Phone: +902 469 5004 - Fax: +902 461 9689 - E-mail: bollivar@seafreez.com

N. Bouffard, Senior Advisor, Legal Issues, International Affairs Directorate, Dept. of Fisheries and Oceans,

200 Kent St., 13th Floor, Stn. 13159, Ottawa, Ontario K1A 0E6

Phone: +613 993 1860 Fax: +613 993 5995 E-mail: <u>bouffardn@dfo-mpo.gc.ca</u>

W. R. Bowering, Northwest Atlantic Fisheries Centre, P. O. Box 5667, St. John's, Newfoundland A1C 5X1 Phone: +709 772 2054 - Fax: +709 772 4188 - E-mail: boweringr@dfo-mpo-gc.ca

W. B. Brodie, Northwest Atlantic Fisheries Centre, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 3288 - Fax: +709 772 4105 - E-mail: brodieb@dfo-mpo.gc.ca

B. Chapman, 1388 River Road, Manotick, Ontario K4M 1B4

Phone: +613 692 8249 - Fax: +613 692 8250 - E-mail: bchapman@sympatico.ca

L. Dean, Dept. of Fish and Aquaculture, Government of Newfoundland and Labrador, P. O. Box 4750, St. John's, Newfoundland

Phone: +709 729 3722 - Fax: +709 729 0360

T. Dooley, Dept. of Fisheries, P. O. Box 8700, St. John's, Newfoundland A1L 5H4

Phone: +709 729 0335 - Fax: +709 729 6082 - E-mail: tdooley@matl.gov.nf.ca

E. Dussault, Director, Atlantic Affairs Div., International Affairs Directorate, Dept. of Fisheries and Oceans, 200 Kent Street, Ottawa, Ontario K1A 0E6

Phone: +613 993 5316 - Fax: +613 993 5995 - E-mail: dussaulte@dfo-mpo.gc.ca

W. Evans, Supervisor, Offshore Surveillance, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland AIC 5X1

Phone: +709 772 4412 - Fax: +709 772 0008 - E-mail: evansw@dfo-mpo.gc.ca

S. Firko, Dept. of Fisheries and Oceans, 200 Kent Street, Ottawa, Ontario KIA 0E6

Phone: +613 993-1852 - Fax: +613 993 5995

D. Forsythe, Mission of Canada to the European Union, Avenue de Tervuren, 2, Brussels 1040, Belgium Phone: +32 2 741 0688 - Fax: +32 2 741 0629 - E-mail: douglas.forsythe@dfait-maeci.gc.ca

G. Gregory, Fishery Products International Ltd., P. O. Box 550, Station A, St. John's, Newfoundland A1C 5L1 Phone: +709 570 0427 - Fax: +709 570 0436 - E-mail: ggregory@fpil.com N. Greig, P. O. Box 1058, Kuujjuaq, Quebec JOM 1C0

Phone: +819 964 2925 - Fax: +819 964 2613 - E-mail: n.greig@makvik.org

J. M. Kelsey, Department of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1 Phone: +709 772 7935 – Fax: + 709 772 6306 – E-mail: <u>kelseyj@dfo-mpo.gc.ca</u>

D. W. Kulka, Groundfish Div., Science Br., Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 2064 - Fax: +709 772 4188 - E-mail: kulkad@dfo-mpo.gc.ca

E, McCurdy, c/o FFAW/CAW, P. O. Box 10, St. John's, Newfoundland A1C 5H5

Phone: +709 576 7276 - Fax: +709 576 1962 - E-mail:

P. McGuinness, Vice-President, Fisheries Council of Canada, 38 Antares Drive, Suite 110, Nepean, Ontario K2E 7V2

Phone: +613 727 7450 - Fax: +613 727 7453 - E-mail: pmcguinness@fisheriescouncil.org

B. J. McNamara, Newfoundland Resources Ltd., 90 O'Leary Avenue, St. John's, Nfld. A1B 3P7

Phone: +709 579 7676 - Fax: +709 579 7668 - E-mail: nrl@nfld.com

J. Mercer, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 4494 - Fax: +709 772 5983 - E-mail: mercerja@dfo-mpo.gc.ca

A. Noseworthy, Deputy Minister, Intergovernmental Affairs, Government of Newfoundland and Labrador, P. O. Box 8700, St. John's, Newfoundland A1B 4J6

Phone: +709 729 2131 - Fax: +709 729 5038 - E-mail: anoseworthy@mail.gov.nf.ca

M. O'Connor, National Sea Products, 100 Battery Point, P. O. Box 910, Lunenburg, N.S. B0J 2C0 Phone: +902 634 5200 - Fax: +902 634 4926 - E-mail: <u>oconnorm@natsca.ca</u>

A. O'Rielly, President, Fisheries Association of Nfld. and Labrador Ltd., P. O. Box 8900, St. John's, Newfoundland A1B 3R9

Phone: +709 726 7223 - Fax: +709 754 3339 - E-mail: aorielly@nfld.com

F. G. Peacock, Director, Resource Mgmt. Br., Dept. of Fisheries and Oceans, P. O. Box 550, Halifax, Nova Scotia B3J 2S7

Phone: +902 426 3625 - Fax: +902 426 9683 - E-mail: peacockg@dfo.mpo.gc.ca

C. Reardon, Nova Scotia Dept. of Fisheries and Aquaculture, P. O. Box 2223, Halifax, N.S. B3J 3C4

Phone: +902 424 0349 - Fax: +902 424 4671 - E-mail: reardonc@ns.gov.ca

D. Rivard, Fisheries Research Br., Fisheries and Oceans, 200 Kent St., Ottawa, Ontario K1A 0E6

Phone: +613 990 0281 - Fax: +613 954 0807 - E-mail: rivardd@dfo-mpo.gc.ca

A. Saunders, Dept. of Foreign Affairs and International Trade (JLO), 125 Sussex Drive, Ottawa, Ontario K1A 0G2

Phone: +613 996 2643 - Fax: +613 992 6483 - E-mail: allison.saunders@dfait-maeci.gc.ca

M. Short, Dept. of Fisheries and Oceans, Baine Johnston Center, Suite 801, 10 Fort Williams Place, St. John's, Newfoundland A1C 1K4

Phone: +709 772 5238 - Fax: +709 772 5244

R. Steinbock, Advisor, Atlantic Affairs Div., International Directorate, Dept. of Fisheries and Oceans, 200 Kent St., Ottawa, Ontario K1A 0E6

Phone: +613 993 1836 - Fax: +613 993 5995 - E-mail: steinbob@dfo-mpo.gc.ca

R. Stirling, SPANS, P. O. Box 991, Dartmouth, N. S. B2Y 3Z6

Phone: +902 463 7790 - Fax: +902 469 8294 - E-mail: span@fox.nstn.ca

H. Strauss, Director, Oceans, Environmental and Economic Law Div., Dept. of Foreign Affairs and

International Trade Canada, 125 Sussex Drive, Ottawa, Ontario K1A 0G2

Phone: +613 992 2104 Fax: +613 992 6483 E-mail: howard.strauss@dfait-maeci.gc.ca

L. Strowbridge, Area Director, Eastern/Southern Newfoundland, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 4010 - Fax: +902 772 2659 - E-mail: strowbridgel@dfo-mpo.gc.ca

E. Wiseman, Director-General, International Directorate, Dept. of Fisheries and Oceans, 200 Kent Street, 13th Floor N, Stn 13-159, Ottawa, Ontario K1A 0E6

Phone: +613 993 1873 - Fax: +613 993 5995 - E-mail: wisemane@dfo-mpo.gc.ca

F. Woodman, Chairman, Fisheries Resource Conservation Council, P. O. Box 2001, Station D, Ottawa, Ontario K1P 5W3

Phone: +613 998 0433 - Fax: +613 998 1146 - E-mail: costah@dfo-mpo.gc.ca

CUBA

Head of Delegation

E. Otuski, Ministerio de la Industria Pesquera, 5th Avenue y 246, Sta Fe, Barlovento Phone: +537 297117

Alternate

J. Bajsre Alvarez, Ministerio de la Industria Pesquera, Barlovento, Santa Fe 19 100, Playa la Habana Phone: +537 297253 - Fax: +537 249168 - E-mail: baisre@fishnavy.inf.cu Representative

J. Baisre Alvarez (address above)

Advisers

V. E. Sarda Espinosa, Ministerio de la Industria Pesquera, 5th Avenue y 246, Sta Fe, Barlovento Phone: +537 297034 - Fax: +537 249168 - E-mail: abogados@fishnavy.inf.cu R. Espinosa, Dragnets, Asociacion Pesport, Puerto Pesquero de la Habana, Ave la Pesquera y Atares, Habana Vieja, Ciudad de La Habana Phone: +537 619090 L. Albelo Leon, Ministerio Industria Pesquera, Cuban Fishing Fleet Representative, 1881 Brunswick St., Apt. 906, Halifax, Nova Scotia, Canada B3J 3J7 Phone: +902 425 5773 - Fax: +902 423 8871

DENMARK (IN RESPECT OF FAROES AND GREENLAND)

Head of Delegation

E. Lemche, Head of Representation, Gronlands Hjemmestyre, Pilestraede 52, Box 2151, DK-1016 Copenhagen K, Denmark Phone: +45 33 69 34 35 Fax: +45 33 69 34 01 E-mail: Einar.Lemche@ghsdk.dk

Alternate

A. Kristiansen, Ministry of Fisheries and Maritime Affairs, Yviri við Strond 17, P. O. Box 347, FO-100 Torshavn, Faroe Islands Phone: + 298 35 30 30 - Fax: +298 35 30 35 - E-mail: andrask@fisk.fl.fo

Representatives

E. Lemche (see address above) A. Kristiansen (see address above)

Advisers

J. E. Hansen, Bondaheygur 9, FO-100 Torshavn, Faroe Islands Phone: +298 312990/210810 - Fax: +298 33 35 95 - E-mail: hogi@post.olivant.fo J. Joensen, (address please) Phone: +298 42 14 48 - Fax: +298 42 15 84 O. A. Jørgensen, Greenland Institute of Natural Resources, Pilestræde 52, Box 2151, Copenhagen, DK-1016 Phone: +45 33 69 34 61 - Fax: +45 33 69 34 06 - E-mail: grfioaj@inet.uni2.dk M. Kruse, Vaktar-og Bjargingartaenastan, P. O. Box 347, FR-110 Torshavn, Faroe Islands Phone: +298 311065 - Fax: +298 383981 - E-mail: vb@vb.fo L. Lind, Head of Section, Royal Danish Ministry of Foreign Affairs, 2 Asiatisk Plads, DK-1448, Copenhagen K. Denmark

Phone: +45 33 92 00 00 - Fax: +45 32 54 05 33 - E-mail: lilind@um.dk

226

M. T. Nedergaard, Fiskerilicensinspektor, Head of Unit, Gronlands Fiskerilicenskontrol, Postbox 501, DK-3900 Nuuk, Greenland

Phone: +299 345377 - Fax: +299 323235 - E-mail: mads@grl.gl

A. Nicolajsen, Fiskirannsoknarstovan, Noatun, P. O. Box 3051, FR-110 Torshavn, Faroe Islands

Phone: +298 1 5092 - Fax: +298 1 8264 - E-mail: aminic@frs.fo

J. Persson, Head of Section, Greenland Home Rule, Department of Industry, Box 269, 3900 Nuuk, Greenland Phone: +299 34 53 24 Fax: +299 32 47 04 E-mail: jpe@gh.gl

E. Rosing, Greenland Home Rule, Dept. of Industry, Box 269, 3900 Nuuk, Greenland

Phone: +299 34 53 32 - Fax: +299 32 47 04 - E-mail: emanuel@gh.gl

J. H. Toftum, Ministry of Fisheries, P. O. Box 64, FO-100 Torshavn, Faroe Islands

Phone: +298 353030 - Fax: +298 353035 - E-mail: jenst@fisk.fl.fo

ESTONIA

Head of Delegation

K. Mühlbaum, Ministry of the Environment, Fisheries Department, Marja 4d, 10617 Tallinn Phone: +372 6566720 - Fax: +372 6567599 - E-mail: kristiina@klab.envir.ee

Representative

K. Mühlbaum (see address above)

Advisers

M. Harjak, Dagomar Ltd., Sadama 15, Kardla

Phone: +372 4632031 - Fax: +372 4632039 - E-mail: marek@huukalur.ee

R. Kulla, E-Traal Ltd., 9 Narva st., Tallinn 10017

Phone: +372 5128888 - Fax: +372 6109244 - E-mail: traal@anet.ee

J. Pollu, Reyktal Ltd., Paljassaare Road 28-426, 10313 Tallinn

Phone: +372 6512066 - Fax: +372 6512055 - E-mail: reyktal@trenet.ee

T. Roose, Deputy Director General, Estonian Environmental, Inspectorate, Kopli 76, 10416 Tallinn, Estonia Phone: +3726603333 - Fax: +3726603350, E-mail: tarvo.roose@kki.ee

T. Saat, Director, Estonian Marine Institute, 18b Viljandi Road, 11216, Tallinn

Phone: +372 6281 570 - Fax: +372 6281 563 - E-mail: tsaat@sea.ee

A. Soome, Officer, Ministry of the Environment, Fisheries Department, Marja 4d, 10617 Tallinn

Phone: +372 6112 987 - Fax: +372 6567 599 - E-mail: ains@klab.envir.ee

L. Vaarja, Fisheries Adviser, Ministry of Environment, Fisheries Department, Marja 4d, 10617 Tallinn Phone: +372 6112 987 - Fax: +372 6567 599 - E-mail: laurivaarja@hot.ee

EUROPEAN UNION (EU)

Head of Delegation

E. Mastracchio, Director, European Commission, Directorate General for Fisheries, 200 Rue de la Loi, B-1049 Brussels, Belgium

Alternate

O. Tougaard, European Commission, Directorate General for Fisheries, 200 Rue de la Loi, B-1049 Brussels, Belgium

Phone: +32 2 295 2209 - Fax: +32 2 299 4802

Representatives

E. Mastracchio (see address above)

O. Tougaard (see address above)

Advisers

F. Wieland, Deputy Head of Unit, International Fisheries Organizations and Fisheries Agreements; Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi 200, B-1049 Brussels, Belgium

Phone: +32 2 296 3205 Fax: +32 2 299 4802 E-mail: Friedrich.Wieland@cec.eu.int

A. Thomson, Principal Assistant, International Fisheries Organizations and Fisheries Agreements; Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loj/Wetstraat 200, B-1049 Brussels, Belgium

Phone: +32 2 299 0180 - Fax: +32 2 299 4802 - E-mail: Andrew, Thomson@cec.eu.int

B. O'Shea, European Commission, Directorate General for Fisheries, Rue Joseph II 99, Rm 1/27, B-1049, Brussels, Belgium

Phone: +32 2 296 6748 - Fax: +32 2 296 2338 - Email: brendan.o'shea@cec.eu.int

K. Patterson, European Commission, Directorate General for Fisheries, Rue de la Loi 200, B-1049 Brussels, Belgium

Phone: + 32 2 299 2179 - Fax: +32 2 295 5621 - Email: kenneth.patterson@cec.eu.int

V. Angot, European Commission, Rue de la Loi 200, B-1049 Brussels, Belgium

Phone: +32 2 296 6406 - Fax: +32 2 296 2338 - Email: Veronique.Angot@ cec.eu.int

A. Gray, Director General I, External Relations: Commercial Policy and Relations with North America, the Far East America, Australia and New Zealand, Rue de la Loi, Wetstraat 170, B-1040 Brussels

Phone: + 32 2 2990077 - Fax # 32 2 2991046 – E-mail: alan-gray@dg1.cec.be

B. Prince, Policy Officer in charge of International Affairs, Ministere de l'Agriculture et de la Peche, Direction des Peches Maritimes, 3 Place de Fontenoy, 75007 Paris, France

Phone: +45 33 1 49 55 82 38 - Fax: +45 33 1 49 55 82 00 - E-mail: <u>berengere.prince@agriculture.gouv.fr</u> S. Segura, Conseiller des Affaires Etrangeres, Direction des Affaires Juridiques, Ministere des Affaires Etrangeres, 37 Ouai d'Orsay, 75700 Paris, France

Phone: +33 1 43 17 53 26 - Fax: +33 1 43 17 43 59 - E-mail: serge.segura@diplomatie.gouv.fr

G. F. Kingston, Senior Adviser, Economic and Commercial Affairs, Delegation of the European Commission, 45 O'Connor Street, Suite 1900, Ottawa, Ontario, Canada K1P 1A4

Phone: +613 238 6464 - Fax: +613 238 5191 - E-mail: fred.kingston@delcan.cec.eu.int

D. Cross, Eurostat, European Commission, Jean Monnet Bldg., BP 1907, L-2920 Luxembourg

Phone: +352 4301 37249 - Fax: +352 4301 37318 - E-mail: david.cross@cec.eu.int

T. Heaton, Director, DG BIII-Fisheries, Council of the European Union, Rue de la Loi 175, B-1048 Brussels, Belgium

Phone: +32 2 285 6486 - Fax: +32 2 285 8261 - E-mail: Trevor.Heaton@consilium.eu.int

S. Feldthaus, Head of Section, Ministry of Food, Agriculture and Fisheries, Holbergsgade 2, 1057 Copenhagen K, Denmark

Phone: +45 33 92 35 60 - Fax: +45 33 11 82 71 - Internet: sfe@fvm.dk

R: Akesson, Ministry of Agriculture and Fisheries, 10333 Stockholm, Sweden

Phone: +46 08 405 1122 - Fax: +46 08 10 5061 - E-mail: rolf.akesson@agriculture.ministry.so

H. Pott, Bundesministerium fur Ernahrung, Landwirtschaft und Forsten, Rochusstr. 1, D-53125 Bonn, Germany Phone: +49 228 529 4124 - Fax: +49 228 529 4410 - Email: <u>hermann.pott@bml.bund.de</u>

J. Manuel de Castro Santiago, Counsellor, Embassy of Portugal, 645 Island Park Drive, Ottawa, Ontario K1Y 0B8

Phone: +613 729 0883 - Fax: +613 729 4236

E. Monteiro, Director-General, Direccao Geral Pescas Aquicultura, Edificio Vasco da Gama, Alcantara, 1350 Lisbon, Portugal

Phone: +351 21 3914387 Fax: +351 21 3957858 E-mail: euricom@dg-psces.pt

M. H. Figueiredo, Directora de Servicos, Dept. de Relacoes Comunitarias, Internacionais e de Cooperacao, Direccao Geral das Pescas e Aquicultura, Edificio Vasco da Gama, Alcantara Mar, 1350 Lisbon, Portugal Phone: +351 21 3914350 Fax: +351 21 3979790 E-mail: <u>hfigueir@dg-pescas.pt</u>

A. Leite, Inspeccao geral deas Pescas, Av. Brasilia, 1400-038 Lisboa, Portugal

Phone: +351 21 3025170 - Fax: +351 21 3025101 - E-mail: albertoleite@igp.pt

L. M. Esteruelas, Counselor for Agriculture & Fisheries, Embassy of Spain, 2375 Pennsylvania Ave., N.W., Washington, D.C. 20037

Phone: +202-728-2339 - Fax: +202-728-2320 - E-mail: <u>Imesteruclas@erols.com</u>

F. Curcio, Subdirector General de Organismos Multilaterales de Pesca, Secretaria General de Pesca Maritimes, Jose Ortega y Gasset, 57, 28006 Madrid, Spain

Phone: +34 914027404 - Fax: +34 913093967 - E-mail: fcurcio@mapya.es

M. I. Aragon, Jefa Seccion de la Subdireccion General de Organismos Multilaterales de Pesca, Secretaria General de Pesca Maritima, Jose Ortega y Gasset, 57, 28006 Madrid, Spain

J. Del Hierro, Subdireccion General de Inspeccion Pesquera, Secretaria General de Pesca Maritima, c/Castellama 112, 5ª Plto, Madrid, Spain

Phone: +34 91 3471645 - Fax: +34 91 3471512

A. Hermida Trastoy, Director Xeral de Estructures Pesqueiras e Mercados, Xunte de Galicia, C/Sar, 75, Santiago 15702, A Coruna, Spain

Phone: + 34981546347 - Fax: +34981546288 - E-mail: andres.hermida.trastoy@xunte.es

H. J. Ratz, Institut fur Seefischerei, Palmaille 9, D-22767, Hamburg, Germany

Phone: +49 40 389 05169 - Fax: +49 40 389 05263 - E-mail: raetz.ish@bfa-fisch.de

M. Stein, Institut fur Seefischerei, Palmaille 9, D-22767, Hamburg, Germany

Phone: +49 40 389 05174 - Fax: +49 40 38905 263 E-mail: stein.ish@bfa-fisch.de

D. Briand, IFREMER, B. P. 4240, 97500 St. Pierre et Miquelon, France

Phone: +508 413083 Fax:+508 41 49 36 - E-mail: brianspm@cancom.net

R. Alpoim, Inst. de Investigacao das Pescas e do Mar (IPIMAR), Av. de Brasilia, 1400 Lisbon, Portugal

Phone: +3511 302 7000 - Fax: +3511-301-5948 - E-mail: ralpoim@ipimar.pt

A. Avila de Melo, Inst. de Investigação das Pescas e do Mar (IPIMAR), Av. de Brasilia, 1400 Lisbon, Portugal

Phone: +3511 302 7000 Fax: +3511-301-5948 E-mail: amelo@ipimar.pt

E. De Cardenas, Institute Espanol de Oceanografia, Avenida de Brasil 31, 28020 Madrid, Spain

Phone: +34 91 5974443 - Fax: +34 91 5974770 - E-mail: e.decardenas@md.ieo.es

S. Junquera, Instituto Espanol de Oceanografia, Cabo Estay-Canido, Aptdo. 1552, E-36280 Vigo (Pontevedra), Spain

Phone: +34 9 86 49 2111 - Fax: +34 9 86 49 2351 - E-mail: susana.junquera@vi.ieo.es

H. Murua, AZTI, Instituto para la Ciencia y Tecnologia Pesquera, Av. Satrustegi 8, 20008 Donostia – San Sebastian, Spain

Phone: +34 9 43 316731 - Fax: +34 9 43 212162 - E-mail: hmurua@azti.es

A. Vazquez, Instituto de Investigaciones Marinas, Eduardo Cabello 6, 36208 Vigo, Spain

Phone: +34 9 86 23 1930 - Fax: +34 9 86 29 2762 - E-mail: avazquez@iim.csic.es

P. Franca, ADAPI – Associacao Armadores das Pescas Industriais, Edificio Dos Armadores 13-A, Docapesca 1400 Lisbon, Portugal

Phone: +213015020 - Fax: +213019438 - E-mail: adapi.pescas@mail.telepac.pt.

M. Paiao, ADAPI – Associacao Armadores das Pescas Industriais, Edificio Dos Armadores 13-A, Doca Pesca 1400-038 Lisbon, Portugal

Phone: +213015020 - Fax: +213019438 - E-mail: adapi.peseas@mail.telepac.pt

R. Gordejuela Aguilar, Presidente de "ANAVAR".

J. R. Fuertes Gamundi, ANAMER-ANAVAR-AGARBA, Puerto Pesquero, Vigo, Spain Phone: +34 986 433844 - Fax: +34 986 439218

Filolie: +34 980 433844 - Fax: +34 980 439218

M. Liria Franch, Presidente de ANAMER, Pto Pesquero, Spain

C. Real Rodriguez, Presidente de "ASPE", Vigo, Spain

FRANCE (in respect of St. Pierre and Miquelon)

Head of Delegation

G. Grignon, 4C Rue Albert Briand, 97500 Saint Pierre et Miquelon Phone: +508 414 219 - Fax: +508 414 806 – E-mail: <u>archipel@cancom.net</u>

Alternate

D. Silvestre, Secretariat General de la Mer, 16 Boulevard Raspail, 75007 Paris Phone: +33142840876 - Fax: +33153634178 - E-mail: <u>daniel.silvestre@sgmer.premier-ministre.gouv.fr</u>

Representatives

G. Grignon (address above) D. Silvestre (address above)

Advisers

V. Sinquin, International Affairs Division, Overseas Ministry, 27 rue Oudinot, 75007, Paris Phone: +0153692746 – Fax: +0153692197 – E-mail: valerie.sinquin@outre_mer.gouv.fr M. Tremblay (Interpreter), 3124 Needham St., Halifax, N.S. B3K 3N9 Phone: +902 420 9158 - Fax: +902 455 2992 - E-mail: <u>mti@hfx.eastlink.ca</u>

ICELAND

Head of Delegation

T. Asgeirsson, Director of Fisheries, Ingolfsstraeti 1, 150 Reykjavik Phone: +354 569 7900 - Fax: +354 569 7991 - E-mail: thordur@hafro.is

Representative

T. Asgeirsson (see address above) Advisers

K. Arnason, Head of Division, Ministry of Fisheries, Skulagata 4, 150 Reykjavik Phone: +354 560 9670 - Fax: +354 562 1853 - E-mail: <u>kolbeinn.arnason@sjr.stjr.is</u>
H. Steinarsson, The Icelandic Directorate of Fisheries, , Ingilfsstraeti, 150 Reykjavik Phone: +354 5097938 - Fax: +354 5697991 - E-mail: <u>hostein@hfro.is</u>
K. Ragnarsson, Federation of Icelandic Fishing Vessel Owners, P. O. Box 893, 121 Reykjavik Phone: +354 550 9500 - Fax: +354 550 9501 - E-mail: kristjan@liu.is

JAPAN

Head of Delegation

K. Yonezawa, c/o Fishery Division, Economic Affairs Bureau, Ministry of Foreign Affairs, 2-2-1 Kasumigaseki, Chiyoda-ku, Tokyo

Representatives

K. Yonezawa (see address above)

Advisers

S. Kawahara, Director, Oceanic Resources Division, National Research Institute of Far Seas Fisheries, 5-7-1 Orido, Shimizu-shi 424, Sizuoka, 424

Phone: +81 543 36 6051 - Fax: +81 543 35 9642 - E-mail: kawahara@envo.affrc.go.jp

Y. Kashio, Representative, Japan Fisheries Association, Suite 1209 Duke Tower, 5251 Duke St. Tower, Halifax, N.S., Canada B3J 1P3

Phone: +902 423 7975 - Fax: +902 425 0537 - E-mail: jfa-hfx@ns.sympatico.ca

M. Miyashita, Far Seas Fisheries Div., Resources Management Dept., Fishery Agency Government of Japan, 1-2-1 Kasumigaseki, Chiyoda-ku, Tokyo 100

Phone: +03 3502 8111 ext. 7239/03 3591 6582 - Fax: +03 3591 5824

S. Nagase, Fisheries Div., Economic Affairs Bureau, Ministry of Foreign Affairs, 2-2-1 Kasumigaseki, Chiyoda-ku, Tokyo

Phone: +03-3580-3311 ext. 3351 - Fax: 03-3503-3136 - E-mail: saori.nagase@mofa.go.jp

Y. Ochi, Development Dept., Japan Marine Fishery Resources Research Center, Godo Kaikan Bldg. 3-27 Kioi -cho, Chiyoda-ku, Tokyo 102-0094

Phone: +03 3265 8301 - Fax: +03 3262 2359 - E-mail: ochi@jamarc.go.jp

N. Takagi, Director Executive Secretary Japan Deep Sea Trawlers Association, Ogawacho-Yasuda Bldg., 6 Kanda-Ogawacho, 3-Chome, Chiyoda-ku, Tokyo 101-0052

Phone: +81 33 291 8508 - Fax: + 81 33 233 3267 - E-mail: nittoro@mx3.mesh.ne.jp

K. Tanaka, Deputy Director, International Affairs Div., Fisheries Policy Planning Dept. Fisheries Agency, Government of Japan, 1-2-1 Kasumigaseki, Chiyoda-ku, Tokyo

Phone: +81 3 3591 1086 - Fax: +81 3 3502 0571 - E-mail: kengo tanaka@nm.maff.go.jp

REPUBLIC OF KOREA

Head of Delegation

S.-J. Yoo, Director, International Cooperation Division, Ministry of Maritime Affairs and Fisheries, 139 Chungiong-No. 3, Seodaemun-Gu, Seoul, 120-715 Phone: +82-2-3148-6990~1 - Fax: +82-2-3148-6996 -E-mail: <u>ussi@momaf.go.kr</u>

Representative

S.-J. Yoo (see address above)

Adviser

Y.-S. Jung, First Secretary for Maritime Affairs & Fisheries, Embassy of Korea, 2450 Massachusetts Avenue, N. W., Washington, D.C. 20008 Phone: +202 939 5676 - Fax: +202 387 0402

LATVIA

Head of Delegation

N. Riekstins, Director, National Board of Fisheries, Ministry of Agriculture, 2, Republikas laukums, LV-1010 Riga Phone: +371 732 3877 - Fax: +371 733 4892 - E-mail: <u>fish@com.latnet.lv</u>

Alternate

R. Derkacs, Head of International Agreements and Legal Div., National Board of Fisheries, 2, Republikas laukums, LV-1010 Riga Phone: +371 732 3877 - Fax: +371 733 4892 - E-mail: fish@com.latnet.ly

Representatives

N. Riekstins (see address above) R. Derkacs (see address above)

Advisers

I. Voits, President, Latvian Fisherics Association, Ganibu Dambis 24a-502, Riga, LV-1005 Phone: +371 7383197 - Fax: +371 7383197 - Mob. Phone 371 9363094

LITHUANIA

Head of Delegation

A. Raudonius, Vice-Minister of Agriculture, Ministry of Agriculture, 19 Gedimino pr., LT-2025 Vilnius Phone: +370 2 391306 Fax: +370 2 391308 E-mail: <u>albinasr@zum.lt</u>

Alternate

V. Vaitiekunas, Director, Fisheries Dept. under the Ministry of Agriculture, 19 Gedimino str., Vilnius 2600 Phone: +370 02 391174 - Fax: 37002 341176 - E-mail: <u>vytautasv@zum.lt</u>
A. Rusakevicius, Fisheries Department under the Ministry of Agriculture, 19 Gedimino St., 2600 Vilnius Phone: +370 2 391183 - Fax: +370 2 391176 - E-mail: <u>algirdasr@zum.lt</u>

Representatives

A Raudonius (see address above) A. Rusakevicius (see address above)

Advisers

B.Urboniene, JSC Vigomeras, Poilsio str. 20-33, 5810 Klaipeda Phone: +3706 345518 - Fax: +3706 344429 - E-mail: vigomeras@takas.lt

NORWAY

Head of Delegation

P. Gullestad, Directorate of Fisheries, P. O. Box 185, N-5002 Bergen Phone: +47 55 23 80 00 Fax: +47 55 23 80 90 E-mail: <u>peter.gullestad@fiskeridir.dep.telemax.no</u>

Alternate

T. Lobach, Directorate of Fisheries, P. O. Box 185, N-5002 Bergen Phone: +47 55 23 80 00 Fax: +47 55 23 80 90 E-mail: terje.lobach@fiskeridir.dep.telemax.no

Representatives

P. Gullestad (see address above)

T. Lobach (see address above)

Advisers

W. Barstad, Norwegian Fishing Vessel Owners Association, P.O. Box 67 Sentrum, 6001 Aalesund Phone: +47 70 10 14 60 - Fax: +47 70 10 14 80 - E-mail: webjorn@fiskebatreder.no
H. P. Johansen, Fisheries Counsellor, Royal Norwegian Embassy, 2720 34th St. N.W., Washington, D.C. 20008 Phone: +202 944 8981 - Fax: +202 337 0870 - E-mail: counselor@fish.norway.org
S. Owe, Director General, Ministry of Fisheries, P. O. Box 8118 Dep., 0032 Oslo
Phone: +47 22 24 64 71 Fax: +47 22 24 95 85 E-mail: stein.owe@fid.dep.telemax.no
E. K. Viken, Fiskeridepartementet, Postboks 8118 Dep., 0032 Olso
Phone: +22 24 6482 - Fax: +22 24 9585 - E-mail: ellen.viken@fid.dep.no

POLAND

Head of Delegation

Z. Gandera, Director, Fisheries Dept., Ministry of Agriculture and Rural Development, 30 Wspolna Str. 00-930 Warsaw Phone: +48 22 6280826 Fax: +48 22 6232204 – E-mail: <u>z.gandera@minrol.gov.pl</u>

Representative

Z. Gandera (see address above)

Advisers

M. Kucharski, Embassy of the Republic of Poland, 443 Daly Avenue, Ottawa, Ontario K1N 6H3
Phone: +613 789 0468 - Fax: +613 789 1218
B. Szemioth, Parkowa 13/17/123, Warszawa
Phone: +48228508420 - Fax: 48228908920 - E-mail: atlantex@alpha.net.pl

RUSSIA

Head of Delegation

V. Izmailov, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031

V. K. Babayan, Head of Laboratory for System Analysis of Fishery Resources, VNIRO, 17, V. Krasnoselskaya,

Phone: +7095 - Fax: +7095 9213463

Representative

V. Izmailov (see address above)

Advisers

Moscow 107140 Phone: +70 95 264 6985 - Fax: +70 95 264 9187 - E-mail: <u>babayan@vniro.msk.su</u> K. A. Bekyashev, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031 Phone: - Fax: +7095 921 3463 M. G. Botvinko, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031 Phone: +7095 924 7611 - Fax: +7095 921 3463 G. V. Gusev, State Committee for Fisheries of the Russian Federation, Fisheries Dept., 12 Rozhdestvensky Boul., Moscow 103031 Phone: +7095 921 9880 - Fax: +7095 921 3463 V. M. Kolesnikov, Deputy of Head of Resource Department, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow Phone: +7095 924 3372 - Fax: +7095 9243372 I. Mikhno, Fisheries Attache, Embassy of Russia, 1609 Decatur Street, Washington, D.C. 20011 Phone: +202 726 3838 - Fax: +202 726 0090 - E-mail: rusfishatt@starpower.net V. M. Mishkin, General Director, Scientific and Technical Firm "Complex Systems", 5, Kominterna str., P. O. Box 183038, Murmansk Phone: +78152 476080 - Fax: +47 7891 0098 A. Okhanov, Russian Representative on Fisheries in Canada, 47 Oceanview Drive, Bedford, Nova Scotia, Canada B4A 4C4 Phone: +902 832 9225 - Fax: +902 832 9608 V. A. Rikhter, ATLANTNIRO, 5 Dmitry Donskoy St., Kaliningrad, 236000 Phone: +70 112 22 5547 - Fax: +70 112 21 9997 - E-mail: atlant@baltnet.ru A. Rodin, Horlovsky St. 3-13, Moscow 9162881 Phone: +7095 9162381 - Fax: +7095 9162460 E. Samoilova, PINRO, 6 Knipovich St., Murmansk 183763 Phone: +7 8152473461 - Fax: +47 78910518 - E:mail - inter@pinro.murmansk.ru V. N. Shibanov, PINRO, 6 Knipovich St., Murmansk 183763 Phone: +7 8152 47 34 61 – Fax: +47 789 10 518 – E-mail: inter@pinro.murmansk.ru F. M. Troyanovsky, PINRO, 6 Knipovich St., Murmansk Phone: +7 8152 47 34 61 - Fax: +47 789 10 518 - E-mail: inter@pinro.murmansk.ru

232

UKRAINE

Head of Delegation

V. B. Chernik, Deputy Chairman, State Department for Fisheries of Ukraine, 82A Turgenivska str., Kiev, 04050

Phone: +38044 226 2405 - Fax: +380 44 226 2405 - E-mail: nauka@i.kiev.ua

Representative

V. B. Chernik (see address above)

Advisers

V. Litvinov, Senior Expert, Div. for International Fishing Policy, State Department for Fisheries of Ukraine, 82A Turgenivska str., Kiev, 252053
Phone: +38044 216 6883 - Fax: +38044 216 6883 - E-mail: nauka@i.kiev.ua
A. Anastasov, Head of Commercial Fishing Dept., "Yugrybpoisk" State Fishing Company, 6, I. Kozlova str., 334500 Kerch, Republic of Crimea
Phone: +06561 2 01 82 - Fax: +06561 2 13 35 / 23549

UNITED STATES OF AMERICA

Head of Delegation

 P. Kurkul, Regional Administrator, Northeast Region, National Marine Fisheries, 1 Blackburn Drive, Gloucester, MA 01930
 Phone: +978 281 9250 - Fax: +978 281 9371 - E-mail: <u>pat.kurkul@noaa.gov</u>

Representative

P. Kurkul (see address above)

Alternates

J. D. O'Malley, Executive Director, East Coast Fisheries Federation. Inc., P. O. Box 649, Narragansett, RI02879 Phone: +401 782 3440 - Fax: +401 782 4840

J. Pike, Director, Government Relations, Scher and Blackell, Suite 200, 1850 M Street NW, Washington, DC 20036

Phone: +202 463 2511 - Fax: +202 463 4950 - E-mail: jpike@shebla.com

Advisers

J. Anderson, Fisheries Management Specialist, Northeast Region, National Marine Fisheries Service, U.S. Department of Commerce, 1 Blackburn Dr., Gloucester, MA 01930
Phone: +978 281 9226 - Fax: 978-281-9135 - E-mail: jennifer.anderson@noaa.gov
N. Anderson, National Marine Fisheries Service, 1 Blackburn Drive, Gloucester, MA 01930
Phone: +978 281 9383 - Fax: +978 281 9371 - E-mail: nick.anderson@noaa.gov
K. Beal, National Marine Fisheries Service, 1 Blackburn Drive, Gloucester, MA 01930
Phone: +978 281 9267 - Fax: +978 281 9371 - E-mail: nick.anderson@noaa.gov
K. Beal, National Marine Fisheries Service, 1 Blackburn Drive, Gloucester, MA 01930
Phone: +978 281 9267 - Fax: +978 281 9117 - E-mail: ken.beal@noaa.gov
N. Brajevich, International Affairs Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520
Phone: +202 647 3228 - Fax: +202 736 7350 - E-mail: BrajevichNM@state.gov
P. Burns, National Marine Fisheries Service, 1 Blackburn Dr., Gloucester, MA 01930
Phone: +978 281 9144 - Fax: - E-mail: peter.burns.@noaa.gov
W. Chan, National Marine Fisheries Service, 1 Blackburn Dr., Gloucester, MA 01930
Phone: +978 281 9328 - Fax: +978 281 9394 - E-mail: winnie.chan@noaa.gov

234

C. Coogan, National Marine Fisheries Service, 1 Blackburn Dr., Gloucester, MA 01930

Phone: +978 281 9118 - Fax: +978 281 9137 - E-mail: ccoogan@noaa.gov

G. H. Darcy, Assistant Regional Administrator, Sustainable Fisheries, National, Marine Fisheries Service, Northeast Regional Office, One Blackburn Drive, Gloucester, MA 01930

Phone: +978 281 9331 - Fax: +978 281 9135 -- E-mail: George.Darcy@noaa.gov

D. Ellenton, President, World Wide Trading, Inc., 20 Locust Street, Suite 201, Danvers, MA 01923

Phone: +978 762 4665 - Fax: +978 750 6743 - E-mail: wwt@tiac.net

S. V. Fordham, Fisheries Project Manager, Center for Marine Conservation, 1725 DeSales Street, NW Suite 600, Washington, DC 20036

Phone: +202 429 5609 - Fax: +202 872 0619 - E-mail: sonja@dccmc.org

L. C. Hendrickson, Northeast Fisheries Science Center, 166 Water St., Woods Hole, MA 02543

Phone: +508 495 2285 - Fax: +508 495 2393 - E-mail: lisa.hendrickson@noaa.gov

W. P. Jensen, Chair, Management Board, Atlantic States Marine Fisheries Commission and Deputy Director, Fisheries Service, Maryland Dept. of Natural Resources, Annapolis, MD 21401

Phone: +410 260 8261 - Fax: +410 260 8279 - E-mail: pjensen@state.md.us

D. Warner-Kramer, International Affairs Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

Phone: - Fax: +202 736 7350 - E-mail: warner-kramerdm@state.gov

G. S. Martin, Attorney, Office of the General Counsel, Northeast Region, National Oceanic and Atmospheric Administration, U.S. Dept. of Commerce, 1 Blackburn Dr., Gloucester, MA 01930

Phone: +1 978 281 9242 Fax: +1 978 281 9389 E-mail: gene.s.martin@noaa.gov

D. T. Mathers, Lieutenant Commander, Coast Guard Liaison Officer, Office of Marine Conservation (Rm

5806), U.S. Department of State, 2201 C Street NW, Washington, D.C. 20520

Phone: +202 647 3177 - Fax: +202 736 7350 - E-mail: dmathers@comdt.uscg.mil

R. Mayo, Northeast Fisheries Science Center, NMFS, 166 Water St., Woods Hole, MA 02543

Phone: +508 495 2310 - Fax: +508 495 2393 - E-mail: ralph.mayo@noaa.gov

M. Mooney-Seus, New England Aquarium, Central Wharf, Boston, MA 02110-3399

Phone: +617 973 6587 - Fax: +617 973 0242 - E-mail: mmooneys@aol.com

P. Moran, International Fisheries Div., F/SF4, National Marine Fisheries Service, U.S. Dept. of Commerce, 1315 East-West Hwy., Silver Spring, MD 20910

Phone: +301 713 2276 - Fax: +301 713 2313 - E-mail: pat.moran@noaa.gov

S. Olsen, National Marine Fisheries Service, 1 Blackburn Drive, Gloucester, MA 01930

Phone: +978 281 9330 - Fax: +978 281 9117 - E-mail: <u>susan.olsen@noaa.gov</u>

J.-P. Plé, Senior Atlantic Fisheries Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

Phone: +202 647 2335 - Fax: +202 736 7350 - E-mail: jple@state.gov

W. Quinby, Director, Mayflower Shipping Ltd., 5 Yeamans Road, Charleston, SC 29407

Phone: +843 577 0560 - Fax: +843 577 6644 - E-mail: mayflower@mindspring.com

A. Rosenberg, Dean, College of Life Sciences and Agriculture, University of New Hampshire, Taylor Hall, 59 College Road, Durham, NH 03824

Phone: +603 862 2020 – E-mail: andy.rosenberg@unh.edu

F. M. Serchuk, Chief, Resource Evaluation and Assessment Division, Northeast Fisheries Science Center, NMFS, 166 Water St., Woods Hole, MA 02543-1097

Phone: +508 495 2245 - Fax: +508 495 2258 - E-mail: fred.serchuk@noaa.gov

D. E. Swanson, Chief, International Fisheries Div., F/SF4, National Marine Fisheries Service, U.S. Dept. of Commerce, 1315 East-West Highway, Silver Spring, MD 20910

5

Phone: +301 713 2276 - Fax: +301 713 2313 - E-mail: dean.swanson@noaa.gov

SECRETARIAT

L. I. Chepel, Executive Secretary

T. Amaratunga, Assistant Executive Secretary

F. D. Keating, Administrative Assistant B. J. Cruikshank, Senior Secretary

S. Goodick, Accounting Officer

D. C. A. Auby, Word Processing Secretary G. Moulton, Statistical/Conservation Measures Officer F. E. Perry, Desktop Publishing/Documents Clerk

B. Crawford, Graphic Arts/Printing Technician

Annex 2. Opening Statement by the NAFO President – E. Oltuski (Cuba)

Dear NAFO representatives, distinguished observers and guests.

I have the honor of addressing you for the first time since my election as President of the Northwest Atlantic Fisheries Organization. I wish to thank you for the trust you have deposited on my country and myself.

The coming millennium requires urgently that we think more about the future and not only about today. To protect the resources of the sea is not just a duty but an obligation. Seventy percent of the commercial marine species are overfished or at maximum exploitation levels. NAFO statistics show that the catches have dropped considerably in the last 20 years.

The welfare of the future generations demand that we make mutual concessions for the benefit of all. Sustainable fishing is only possible through a compromise between necessity and possibility in the exploitation of the resources based on scientific evidence.

In the conservation and best utilization of those resources lie the objectives of our work at this annual meeting. I am sure that we will all work in that direction.

In the names of Vice President - Patrick Chamut, Executive Secretary – Leonard Chepel and myself, I wish to thank the authorities of the United States for the organization and warm welcome we have all received in this beautiful and historical city of Boston.

In the hope that our meetings will take place in an atmosphere of frank cooperation and understanding, I declare officially inaugurated the Twenty-second Annual Meeting of the Northwest Atlantic Fisheries Organization.

236

Annex 3. Speech of the Honorable Rolland A. Schmitten Deputy Assistant Secretary for International Affairs United States Department of Commerce

Good afternoon, Mr. President, Distinguished Delegates, Ladies and Gentlemen, and a special hello to the many friends and colleagues that I recognize at this NAFO meeting.

As you continue your tour of the world's great seaports of the North Atlantic at annual meetings of NAFO, it is my honor and sincere privilege to welcome you to Boston and the United States of America. I am delighted that we are able to host the Twenty-second Annual Meeting in city with a 400 year history and tradition of seafaring and fishing. Legend has it that long ago one could nearly walk across the bay on the backs of cod, a condition that today is but a memory as we work toward rebuilding stocks and achieving sustainable fisheries. But Boston is not only a feature in America's seafaring and fishing history, it is the cultural center that was home to Ralph Waldo Emerson, Henry Wadsworth Longfellow, James Russell Lowell, and Henry David Thoreau. It is presently home to more than forty colleges and universities, including Harvard, Boston University, Tufts, and MIT—in fact, it has the largest concentration of institutions of higher learning in the United States. With cobblestones underfoot and skyscrapers overhead, Boston welcomes the world as the gateway to New England and the United States. It is, in the words of Charles Dickens, "what the whole United States should be."

A second reason it gives me such pleasure to welcome you here is the great importance we attach to NAFO as a successful regional fisheries management organization, both in its present conservation and management activities and its future ones. It is true that we have encountered our share of problems and challenges, but the more important point is that we are addressing them responsibly. It is with some degree of pride that the United States has joined NAFO. It is often too easy to stay outside a regional fisheries management body and criticize its efforts without taking the responsible step to join and support the organization from within and seek improvements as a member.

NAFO's Convention Area extends south to Cape Hatteras, North Carolina, and its Regulatory Area to the same point seaward of the U.S. exclusive economic zone. With its fishing history, the United States has a real interest in the stocks presently managed by NAFO as well as those NAFO may decide to manage in the future. This interest is manifested in the human and financial resources we invest in NAFO's activities and the commitment I give you today that we will continue making these investments.

My third reason for welcoming you is that NAFO is stepping up to meeting the challenges posed by an unprecedented suite of global fisheries agreements reached in the last decade that collectively prescribe a new ethic for fisheries management. Starting with the imperatives articulated in 1992 by Agenda 21 of the United Nations Conference on Environment and Development, we have collectively responded with: the Food and Agriculture Organization of the United Nations (FAO) Compliance Agreement of 1993; the Code of Conduct for Responsible Fisheries of 1995; the United Nations Fish Stocks Agreement of 1995; the FAO's international plans of action implementing its Code of Conduct in respect of sharks, seabirds, and fishing capacity in 1999; the Convention on the Conservation and Management of Highly Migratory Fish Stocks in the Western and Central Pacific Ocean in 2000; and the emerging Convention on the Conservation and Management of Fishery Resources in the South East Atlantic Ocean. Yes, fisheries managers throughout the world have been active in the quest for improved fishing conditions and sustainability of our fisheries resources. NAFO has taken up a number of the principles contained in these agreements and given them, or is in the process of giving them, expression in a manner consistent with its mission, including: the duty to cooperate to achieve effective conservation and management, standard requirements for the collection and sharing of data, the precautionary approach, measures taken by port states, dispute settlement, and transparency.

Finally, I welcome you because this meeting will grapple with issues in a manner that will again advance NAFO's reputation as an organization that meets challenges and succeeds. Here we will continue making decisions that take serious account of future as well as present generations in charting a course toward achieving and maintaining sustainable and responsible fisheries. In a time when the world focuses on measures to address illegal, unregulated, and unreported (IUU)fishing, NAFO will point to its innovative and successful efforts to address this problem. I hope I will see many of you in Rome next month as we bring our collective focus and wisdom to bear at the FAO's meeting on IUU fishing. Concerning IUU fishing, I envision that, in a context in which the international community is taking steps to ensure that all fishing vessels, from whatever home port, are bound to abide by the same rules, NAFO will continue striving to make its rules equitable, predictable, transparent, and responsive to the intereests of all relevant partners.

On a personal note, for those who do not know, we have bid our farewells to Andy Rosenberg. It was my pleasure to have Andy as my deputy when I headed the National Marine Fisheries Service, and no one could ask for a more talented and competent partner. Even though Andy has returned to the halls of the academy, he will nevertheless join the U.S. delegation later this week. I am equally proud of our new head of delegation Patricia Kurkul, who I had the privilege to recommend for her current position as Regional Administrator of our Northeast Region, for the National Marine Fisheries Service. Yes, the United States is in good hands and committed to the challenges of NAFO.

And finally, to you, Mr. Chairman, we compliment you in your new position and we wish you good luck in your term. I again welcome all of you to Boston and the United States, wishing you fair weather and winds, and extending every good wish for a successful and productive meeting. Welcome.

Thank you, Mr. Chairman.

238

Annex 4. Opening Statement by the Representative of Canada (P. Chamut)

Mr. Chairman, distinguished Representatives, it is a pleasure for Canada to participate at this year's NAFO Annual Meeting in the historic and beautiful city of Boston.

On behalf of the Canadian delegation, I wish to thank the United States for their hospitality and the opportunity to visit the historic sites in Boston.

Against the backdrop of this area, with its maritime heritage, we are here again, as members of NAFO, to give effect to our obligations to conserve fish stocks, and ensure their effective management. The importance of individually and collectively meeting our obligations has never been so high.

Many of the stocks of fish under NAFO stewardship continue to be at historically low levels. The work we do here to protect and rebuild those stocks has very real consequence for the hundreds of Canadian communities, and thousands of facilities which have relied upon these stocks for generations. These families, and communities, have nowhere else to go, and few economic alternatives.

Their present outlook is bleak, and their future outlook will be more promising only if this Organization maintains and strengthens its commitment to conservation, and stock rebuilding. Canada's objective at this meeting will be to ensure that we build on our achievements and we continue to build towards a brighter future.

Five years ago NAFO adopted strengthened Conservation and Enforcement Measures and two years ago NAFO adopted on a permanent basis the program for 100% observer coverage and satellite tracking on NAFO member vessels. These new strengthened measures were rightly hailed as a milestone towards enhanced international cooperation to achieve a common purpose. That purpose is to ensure that high seas fishing activities are conducted in a rational, sustainable and responsible manner.

We need to build on the achievements of recent years to sustain the progress which has been made in the establishment of sound conservation measures and in controlling overfishing and deterring unsustainable fishing practices. The implementation of these measures has laid the groundwork for the recovery and rebuilding of stocks in the Northwest Atlantic. This is of fundamental importance and benefit to all NAFO Parties who wish to see renewed fishing possibilities in the NAFO Regulatory Area.

The collapse of important groundfish stocks has demonstrated the need for enhanced conservation and disciplined management. The past abundance is a reminder of the challenge which faces us - the conservation and rebuilding of the once plentiful stocks of the Northwest Atlantic.

This 22nd annual meeting of NAFO will be an especially critical one for the Organization, as we are facing a number of important issues which will have significant implications for the future.

The recommendations of the Scientific Council underline the need for continuing restraint and vigilance in surveillance and enforcement of the NAFO conservation measures in order to ensure that the path to rebuilding of stocks is not compromised.

The Scientific Council recommends continuing in 2001 most of the moratoria on fishing for groundfish stocks. However, there are some positive signs. It is heartening to see continued rebuilding of 3LNO yellowtail flounder and Greenland halibut.

While enjoying the benefits of these encouraging signs, we must also be vigilant to the danger signals reported by the Scientific Council, in particular the high proportion of catches of young, immature fish. More effective bycatch and discard rules, increased minimum fish and mesh sizes as well as area and seasonal closures must be considered to protect juvenile fish and non-target species that are at low abundance and require protection if they are to rebuild.

Canada's objective is that all Parties can once again enjoy the bounty of sustainable fisheries in the Northwest Atlantic. I am sure this objective is shared by all. However with most of the NAFO stocks currently under moratoria, it is clearly too early to benefit from the restraint we have practised over the past several years nor consider that the challenges we face have been overcome. In fact, continued progress will depend upon our willingness to modify or extend some conservation measures or to introduce new ones if we are to achieve our goal.

As Contracting Parties to the NAFO Convention, we all share the responsibility to conserve the resources in the NAFO Regulatory Area. We must ensure that achievement of that responsibility remains our primary objective, rather than accommodating short term economic interests.

I am looking forward to a constructive and positive dialogue at this session which will help advance the interests of this Organization and all its members, and provide for a brighter future for all those who are dependent on resources under our stewardship. Thank you.

240

Annex 5. Opening Statement by the Representative of the European Union (E. Mastracchio)

Mr. Chairman, distinguished delegates, ladies and gentlemen,

It is once again a great pleasure for my delegation and I to be here at a NAFO Annual Meeting, this time in the United States in this beautiful city of Boston for the very first time. We know the significance of fisheries for this part of the Northwest Atlantic and particularly for one of our recent members, the USA.

We cannot but state our continued belief in the future of fisheries, in the Northwest Atlantic, in that it lies with the sound conservation and management of fishery resources based on the best possible scientific advice available. It is our common interest that NAFO remains at the forefront of good management in international fisheries.

There are many issues to be discussed this week and the commitment of the Community to the principles I mentioned above will be very evident throughout. We insist that there is consistency and compatibility between action taken both within areas of national jurisdiction and beyond. I am therefore deeply concerned that once again, Canada has taken regulatory measures for 2000 for cod in area 2J3KL despite the action taken in the 1999 Annual Meeting in the context of NAFO. Our whole basis for decision this week must remain sound conservation and management of fish stocks throughout their entire area of distribution.

Of particular interest to the Community this week will be a number of issues, not least of which are the future of Dispute Settlement Procedures in NAFO and the next steps to take on the Program for observers and satellite tracking.

On Dispute Settlement Procedures, the Community would like to see some real progress. I reiterate what I said last year. We strongly support compulsory and binding dispute settlement procedures, the prevention of disputes and, where necessary, the existence of mechanisms that entail binding decisions with due regard to the peculiarities of the NAFO Convention. Progress must be made on this issue, preferably in the NAFO context.

With regard to observers and satellite tracking, there is clearly a need for us to re-examine the relevance of observer coverage before any permanent scheme can be contemplated. Furthermore, it is the preference of the Community that 100% satellite tracking should become the cornerstone of any future permanent scheme within NAFO waters.

Mr. Chairman, there are of course other issues to discuss this week, but they are very many and they have great importance for us all. At this point, I must wish all delegates present today the very best towards achieving a successful meeting. We, in the Community Delegation, look forward to working with you and all those who participate in a constructive manner so that our common objectives can be reached and our challenges fulfilled in the interest of this Organization. This will enable NAFO to remain at its prominent position in the lead for international fisheries.

Finally, Mr. Chairman, may I welcome you here in your new capacity. You will have a daunting but, nevertheless, I feel a satisfying week's work ahead. Thank you.

Annex 6. Opening Statement by the Representative of the United States of America (P. Kurkul)

Mr. Chairman, Distinguished Delegates, Ladies and Gentlemen,

The United States Delegation adds a warm welcome to all participants in this 22^{nd} Annual Meeting of NAFO. Our delegation is very excited to be hosting the first of NAFO's annual meeting to be held in the US. We hope you will enjoy your stay in Boston, and we intend to work with you and all delegations to make this meeting a success.

NAFO has taken up a number of exciting challenges in recent years, including: the precautionary approach, quota allocation, improved data collection and management systems, transparency, and dispute settlement. All of these issues spring from the FAO Code of Conduct for Responsible Fisheries and the UN Fish Stocks Agreement, the latter of which should enter into force in the very near future. We have more work to do on many of these issues, but no regional fisheries management organization can take more pride than NAFO for having tackled them and made progress. I hope that this meeting continues this progress. I also wish to note that NAFO may also have important contributions to make in the implementation of the FAO's international plans of action, particularly those concerning the conservation and management of sharks and the management of fishing capacity.

Although NAFO may not currently be setting TACs for a large amount of fish, we are the stewards for managing a very important and productive area of the world's oceans. We must therefore adhere conservatively to the best scientific advice available for all stocks, and we must continue to improve our management practices by, among other things, being precautionary in our management of new, existing, and recovering fisheries.

Again, the U.S. Delegation welcomes the opportunity at this meeting to work with friends and colleagues to continue making NAFO a model regional fisheries management organization for the world.

Thank you very much.

Annex 7. Opening Statement by the Representative of France on behalf of St. Pierre et Miquelon (G. Grignon)

Mr. Chairman, Dear Colleagues,

May we, first of all, express our thanks to the U.S. government and to the U.S. delegation for their hospitality and warm welcome in this beautiful city of Boston.

France in respect of Saint-Pierre and Miquelon is now an old new Contracting Party since we have been a member of the organization for five years.

Of course, Saint-Pierre and Miquelon has been in the Convention Area for a much longer period, since the first French settlements in the island of Saint Pierre and Miquelon go back to the seventeen century.

As a coastal state in the area under Article I.3 of the Convention, Saint Pierre and Miquelon is highly dependent on sea ressources; the last figure of catch records used to set up the draft budget shows that more than 5000 tonnes were caught by the French fishermen of Saint Pierre and Miquelon in the convention area in 1998 for a population of 6000. Such a figure illustrates the fact that France on behalf of Saint-Pierre and Miquelon is a Contracting Party whose economy is overwhelmingly dependent on the fishery, with a small coastal comunity dependent on stocks present in the convention area.

For these reasons, we are fully aware of the necessity of ensuring appropriate sustainable management of the stocks, and stringent rules concerning the fishing activity. As you know, Mr Chairman, fishing vessels can always leave for others fishing grounds, but coastal communities like ours do not have that option.

We know that the organization has done a lot already and is working hard to set up and improve the rules of management and cooperation taking into account the interests of all Contracting Parties.

We are looking forward to fully participate in this work in order to reach this goal, which is th cornerstone of international cooperation.

Annex 8. Agenda

I. Opening Procedure

- 1. Opening by Chairman, E. Oltuski (Cuba)
- 2. Appointment of Rapporteur
- 3. Adoption of Agenda
- 4. Admission of Observers
- 5. Publicity

II. Supervision and Coordination of the Organizational, Administrative and Other Internal Affairs

- 6. Review of Membership
 - a) General Council
 - b) Fisheries Commission
 - c) Reports from Contracting Parties on their communication with Bulgaria and Romania
- 7. Administrative Report

III. Coordination of External Relations

- 8. Communication with the United Nations (Resolution 53/33 24 November 1998)
- 9. FAO International Plans of Action on the Management of Fishing Capacity, Shark Fisheries and Incidental Catch of Seabirds in Longline Fisheries
- 10. NAFO Participation at other International Organizations
 - a) FAO Consultation on the Suitability of the CITES Criteria for Listing Commercially Exploited Aquatic Species (28-30 June 2000).
 - b) FAO ACFR (Committee on Fisheries Research) Working Party on Status and Trends of Fisheries (29 November 1999)
 - c) NASCO Meeting, 2000
 - d) NAMMCO Annual Meeting, 2000
 - e) The Technical Consultation on Illegal, Unregulated, and Unreported Fishing, 2-6 October 2000, Rome, Italy
 - f) FAO Committee on Fisheries Meeting, 26 February to 2 March 2001, Rome, Italy

IV. Fishing Activities in the Regulatory Area Adverse to the Objectives of the NAFO Convention

- 11. Consideration of non-Contracting Party activity in the NAFO Regulatory Area and agreement on the task of STACFAC at the current meeting
- 12. Report of STACFAC at the Annual Meeting and decisions on actions
- 13. Report of the Working Group on Dispute Settlement Procedures (DSP)

244

14. Report of the Working Group on the Allocation of Fishing Rights

15. Report of the Meeting on Shrimp Stocks

V. Finance

16. Report of STACFAD at the Annual Meeting

17. Adoption of the Budget and STACFAD recommendations for 2001

VI. Closing Procedure

18. Time and Place of Next Annual Meeting

19. Other Business

20. Press Release

21. Adjournment

Annex 9. Schedule of NAFO Meetings, 2001

<u>Meetings</u>	Date (full days)	Place
Working Group of Fisheries Commission on oceanic Redfish (in co-operation with NEAFC)	20-21 February	Reykjavik, Iceland
Working Group of the Fisheries Commission on Shrimp Data (catch and effort statistics) Special Fisheries Commission Meeting (Shrimps)	27-29 March	Copenhagen, Denmark
Working Group of the General Council on Dispute Settlement	24-26 April	Dartmouth, N.S.,
Dispute Settlement	Canada	
STACTIC Meeting (VMS/confidentiality Juveniles/by-catch)	26-28 June (at NEAFC)	London, U.K.
Working Group (small) of STACTIC to overhaul Conservation and Enforcement Measures (FC Doc 00/1 and Supplement 2001)	Canada to arrange/ advise dates (Spring 2001)	Ottawa or St. John's, Canada
Working Group (small) of the Fisheries Commission on Precautionary Approach	EU to arrange/advise dates (Spring 2001)	Brussels, Belgium

Annex 10. Press Release

- 1. The 22nd Annual Meeting of the Northwest Atlantic Fisheries Organization (NAFO) was held in Boston, Massachusetts, USA, during 18-22 September 2000, under the chairmanship of Enrique Oltuski (Cuba), President of NAFO. The NAFO constituent bodies - General Council, Fisheries Commission and Scientific Council convened their sessions at the Boston Back Bay Hilton.
- 2. The meeting was attended by 220 participants from sixteen Contracting Parties Canada, Cuba, Denmark (in respect of Faroe Islands and Greenland), Estonia, European Union, France (in respect of St. Pierre et Miquelon), Iceland, Japan, Republic of Korea, Latvia, Lithuania, Norway, Poland, Russia, Ukraine and United States of America.
- Prior to the 22nd Annual Meeting, the following NAFO meetings were held during 2000: (1) Joint Scientific Council and Fisheries Commission Working Group on Precautionary Approach (Brussels, Belgium, 29 February-02 March); (2) Working Group on Allocation of Fishing Rights (Washington, D.C., USA, 28-30 March); (3) Meeting on Shrimp Stocks in the Regulatory Area (Washington, D.C., USA, 27 March); (4) Working Group on Dispute Settlement Procedures (DSP) (Copenhagen, Denmark, 29-31 May); (5) Standing Committee on International Control (Dartmouth, Canada, 27-29 June); (6) STACTIC Technical Working Group on Communications (Dartmouth, Canada, 30 June); (7) Scientific Council Meeting (Dartmouth, Canada, 1-15 June); (8) Workshop on Assessment Methods (Boston, Mass., USA, 13-15 September).
- 4. The Scientific Council, under the chairmanship of Bill Brodie (Canada), reviewed and assessed the status of 19 fish stocks in the NAFO Regulatory and Convention Areas. The scientific advice and recommendations from the Scientific Council were presented to the Fisheries Commission with recommendation that major groundfish stocks are continuing to be at low abundance and there should not be a direct fishery for those stocks in 2001. The Scientific Council reported biomass increases for Greenland halibut in Divisions 2J+3KLMNO and Yellowtail flounder in Div. 3LNO. The Joint Scientific Council/Fisheries Commission Working Group on Precautionary Approach evaluated and discussed a precautionary approach (PA) to NAFO-managed stocks and proposed PA steps in the fields of "harvest strategies", "conservation and enforcement measures" and "research/monitoring" for two model stocks Cod in Div. 3NO, Yellowtail flounder in Div. 3LNO.
- 5. The Fisheries Commission, under the chairmanship of P. Gullestad (Norway), considered the Scientific Council recommendations and agreed on joint international measures and actions for the conservation and utilization of the fishery resources in the Regulatory Area. For all stocks scientific advice on levels of harvest was adopted.

The Commission agreed to impose "no direct fishery" in 2001 on the following stocks: Cod in Divisions 3M and 3L (that portion within the Regulatory Area) and 3NO, Redfish in Div. 3LN, American plaice in Divisions 3M and 3LNO, Witch flounder in Div. 3NO and 3L (that portion within the Regulatory Area) and Capelin in 3NO. The Quota Table for 2001 was adopted (Attachment 1).

With respect to management measures for cod in 2J3KL, Contracting Parties other than Canada expressed their serious concern that management measures for this stock may not be consistent throughout its range in the Convention Area in the year 2000.

New conservation and enforcement measures were agreed:

- To avoid excessive incidental catch a new fishing strategy will apply by changing fishing areas, etc.
- Concerning the shrimp fishery on the Flemish Cap in Division 3M, it was decided that the existing effort allocation Scheme in the shrimp fishery and all other regulatory measures should continue, and that the fishing days should be 90% of maximum number of those observed by Contracting Parties for their vessels in one of the years during 1993-1995.
- The Program for Observers and Satellite Tracking was modified to integrate the hail system with the satellite tracking and vessel monitoring systems (VMS), which will be in force as of January 1, 2001.
- Regulatory measures for chartering vessels between Contracting Parties were modified, including more precise guidelines for notification and charter arrangements, and a provision that the catch of the chartered vessel be added to the catch statistics of the Contracting Party to whom fishing possibilities have been allocated.
- 6. The General Council, under the chairmanship of E. Oltuski (Cuba), deliberated several outstanding issues regarding internal and external NAFO policy on the following terms:
 - Standing Committee on non-Contracting Party Fishing Activity in the Regulatory Area (STACFAC) shall undertake the study of the impact of the FAO International Plans of Action on Illegal, Unregulated and Unreported (IUU) Fishing on NAFO at its next Annual Meeting.
 - The issue of Dispute Settlement Procedures will be deliberated at the General Council in the near future.
 - Although there were no sightings of non-Contracting Party (NCP) fishing in the NAFO Regulatory Area in 2000, the consensus was to continue appropriate actions to deter NCP activity in the NAFO Area.
 - The President of NAFO signed diplomatic démarches to the Non-Contracting Party flag-States whose vessels have fished in the NAFO Regulatory Area in recent years, namely Belize, Honduras, Panama and Sierra Leone.
- 7. The following elections of NAFO officers took place:

Chairman of Standing Committee on Fishing Activities of non-Contracting Parties in the Regulatory Area (STACFAC)

Vice-Chairman of Standing Committee on Fishing Activities of non-Contracting Parties in the Regulatory Area (STACFAC)

Chairman of the Standing Committee on Fisheries Environment (STACFEN)

NAFO General Council 22 September 2000

- Daniel Silvestre (France in respect of St. Pierre et Miquelon)
- Nadia Bouffard (Canada)
- Manfred Stein (EU)

NAFO Secretariat Dartmouth, N.S., Canada

248 ·

QUOTA TABLE. Total allowable catches (TACs) and quotas (metric tons) for 2001 of particular stocks in Subarcas 3 and 4 of the NAFO Convention Area. The values listed include quantities to be taken both inside and outside the 200-mile fishing zone, where applicable.

	Cod		Rec	Redfish	Americ	American plaice	Yellowtail	Witch	Capelin	G. halibut	G. halibut Squid (IIIex) ^{2,3}	Shrimp
Contracting Party	Div. 3M	Div. 3M Div. 3NO*	Dív. 3M	Div. 3M Div. 3LN*	Div. 3M*	Div. 3M* Div. 3LNO*	Div. 3LNO	Div. 3NO*	Div. 3NO*	Div. 3LMNO	Subareas 3+4	Div. 3L
1 Canada	c		200	0	0	0	126757	0	0	4 446	N.S.	5 000
2. Cuba	0) 1	1750	0		,	۰.	ł	0	•	510	67
	ds											ŗ
and Greenland)	0	ı	69	•	,	•	.,	•			- -	5
 European Union 	0	0	3100	0	c	0	260	,	0	16 406	<u>2.0</u>	10
 France (St. Pierre et 											634	13
Miquelon)	'	,	69	•	•	•		•	•	k	604	5
6. Iceland	I	1	•	•	'	•	1	•	њ.			6
7. Japan	'	•	400	ı	,	,	•	,	0	3 038		10
8. Korea	ſ	•	69	•	1	•	1	•	••	•	504	5
9. Norway	0		,	•	'	•	•	•	0	•		6
10. Poland	0	,		ı	,		٠	,	0		177	10
II. Estonia												0
12. Latvia 🖊	0	c	13 250	o	C	,		0	0		1 133	10
13. Lithuania	7	5		\$						ULL (10
14. Russia								-		671 C		62
15. Ukraine 16. Huitad Statas of												
America Status of			69				•		•		453	67
7. Others	0	0	124	0	0	0	.69	0	I	1 9715	794	0
Track Allowing the second			s noné	*	*	•	13 0008	-	* 	29 640	34 000	6 000

Quotas to be fished by vessels from Estonia, Latvia, Lithuania and the Russian Federation. The provisions of Part I, Section ¹ The opening date for the Squid (Illex) fishery is 1 July.

³ Any quota listed for squid may be increased by a transfer from any "coastal state" as defined in Article 1, paragraph 3 of the NAFO Convention, provided that the TAC for squid is not exceeded. Transfers made to Contracting Parties conducting fisheries for squid in the Regulatory Area shall be reported to the Executive Secretary, and the report shall be made as promptly as possible.

'Not specified because the allocation to these Contracting Parties are as yet undetermined, although their sum shall not exceed the difference between the total of allocations to other Contracting Parties and the TAC. ⁵ Of which no more than 40% (788 t) may be fished before 1 May 2001 and no more than 80% (1577 t) may be fished before 1 October 2001.

⁶ Each Contracting Party shall notify the Executive Secretary bi-weekly of catches taken by its vessels from this stock. Not more than 2500 tons may be fished before July 1, 2001. The Executive Secretary shall notify without delay all Contracting Parties of the Gate on which, for this stock, accumulated reported catch taken by vessels of the Contracting Parties is estimated to equal 50 and then 100 percent of the TAC for that stock. ⁷ Contracting Parties shall inform the NAFO Executive Secretary before 1 December 2000 of the measures to be taken to meet the advice of the NAFO Scientific Council.

⁸ The provisions of Part 1, Section A.5c) of NAFO Conservation and Enforcement Measures shall apply.

*No directed fishing – The provisions of Part I, Section A.5a and c of NAFO Conservation and Enforcement Measures shall apply.

Annex 11. Decisions and Actions by the General Council (22nd Annual Meeting, 18-22 September 2000)

Substantive Issue	Decision/Action (GC Doc. 00/7, Part I: item)
 FAO International Plans of Action on the Management of Fishing Capacity, Shark Fishery and Incidental Catch of Seabirds in Longline Fisheries 	Agreed: Contracting Parties report to FAO shall be sent in copy to the NAFO Secretariat, which in turn would circulate the reports to other Contracting Parties; item 3.2
2. NAFO observers at international meetings.	Represented by:
FAO, Rome: - Technical Consultation on Illegal,	- Delegate of Canada, October 2000
Unregulated and Unreported Fishing - FAO Fisheries Committee Meeting	- Delegate of Iceland, Feb-March, 2001; item 3.3.
 Report of STACFAC New diplomatic demarches to Belize, Honduras, Panama and Sierra Leone 	Discussed: items 4,1-4.6 Agreed/signed; item 4.4.
 4. Working Group on Dispute Settlement Procedures (DSP) - Intersessional meeting of the Working Group 	Discussed: item 4.7. Decided to meet in Dartmouth, N.S., Canada, 24-26 April 2001; item 4.7.
5. Election of OfficersChairman of STACFACVice-Chairperson of STACFAC	Daniel Silvestre (France-SPM) Nadia Bouffard (Canada)
6. Budget for 2001- Hail and VMS Automated System	Adopted \$1,389,000 Cdn; item 5.2d, 5.5 - \$200,000 Cdn
7. Venue of 24th Annual Meeting, 2002	Agreed: Spain, 11-20 September 2002; item 5.4

PART II

Report of the Standing Committee on Finance and Administration (STACFAD)

1. Opening by the Chairman

The first session of the Standing Committee on Finance and Administration (STACFAD) was opened by Mr. Fred Kingston (European Union) at 1015 hrs on 18 September 2000.

The Chairman welcomed all delegates and in his opening remarks noted that the agenda had a number of additional issues that would result in further costs for the Organization. Delegates from the following Contracting Parties were present: Canada, Cuba, Denmark (in respect of Faroe Islands and Greenland), European Union, Japan, Latvia, Norway, Russian Federation, Ukraine and United States of America. (Annex 1)

2. Appointment of Rapporteur

Deirdre Warner-Kramer (USA), Nikki Brajevich (USA), and Stan Goodick (NAFO Secretariat) were appointed Rapporteurs.

3. Adoption of Agenda

The provisional agenda was adopted as circulated to the Contracting Parties (Annex 2).

4. Auditors' Report for 1999

The Executive Secretary presented the Auditors' Report and Financial Statements of the Northwest Atlantic Fisheries Organization for the Year Ended 31 December 1999. The Executive Secretary indicated that the Auditors' Report, signed by Deloitte & Touche, was circulated to the Heads of Delegation on 25 April 2000 and no comments had been received on the report.

As stated in Note 4 of the Auditors' Report entitled "Provision for Employee Termination Benefits", the Committee noted the Organization's practice of funding this liability at the rate of \$10,000 per annum as approved by the General Council at the 20th Annual Meeting in 1998.

The EU inquired about the existence of a headquarters agreement. In response to the question, the Executive Secretary stated that there was no headquarters agreement as such and provided copies of Order in Council P.C. 1980-132 dated January 1980 as referred to in the Auditors Report, and also copies of the NAFO Staff Rules dated March 1999.

STACFAD recommended to the General Council that the 1999 Auditors' Report be adopted.

5. Meeting of the Pension Society

The International Fisheries Commissions Pension Society (IFCPS) administers the pension plans and benefits for employees of NAFO and other international fisheries commissions based in North America. The annual meeting was held during 26-28 April 2000 in Seattle, Washington, USA. The next annual meeting of the IFCPS is scheduled to take place in Vancouver, British Columbia, Canada during 25-27 April 2001. The Executive Secretary presented STACFAD Working Paper 00/4 summarizing the annual meeting, which was attended by the NAFO Secretariat staff Forbes Keating and Stan Goodick. The following items were noted:

- a) The IFCPS renewed its contract with Eckler Partners Limited for a two year period to provide pension consulting, actuarial and administrative services. The Society and representatives from the Commissions have been pleased with the services provided by . Eckler Partners Limited and NAFO's annual share of the contract is approximately \$1,700.
- b) A new set of investment guidelines for an improved asset mix of the Canadian Pension Plan Funds was established during 1999. The new guidelines see a switch from 100% Guaranteed Investment Certificates (GIC's) to an asset mix of 50% equities and 50% bonds providing for a more stable investment base and return on investment. As GIC's are maturing, the proceeds are being reinvested into the new asset mix.

6. Review of Cost Implications of the Hail and Satellite Tracking Systems at the NAFO Secretariat

STACFAD Working Paper 00/3 was presented by the Executive Secretary. He noted that research for the development of an automated system at the NAFO Secretariat for handling all hail and satellite tracking reports and their transmission to Contracting Parties with inspection presence in the NAFO Regulatory Area was ongoing. A STACTIC Working Group on Technical Communications met to discuss this issue on 30 June 2000.

The General Council, at the 21st Annual Meeting of NAFO, allocated \$35,000 to the 2000 budget for the automation of the hail system. Of this amount, only \$330 was used to finance a computer consultant at the STACTIC Working Group in June 2000. Unless a resolution on this issue is made for the current year, the remaining 2000 budgetary amount of \$34,670 will be returned to the Accumulated Surplus.

The Chair presented STACTIC Working Paper 00/14 - Overview of NEAFC Scheme re Automated Communications – which amplified the request from STACTIC for funding in the amount of \$200,000 for the Automated Hail and Satellite Tracking System. In response to a request from STACFAD, a representative from STACTIC came and answered technical and cost questions. The representative from the United States noted that while the Committee supported the Automated Hail and Satellite Tracking System in principle, there were concerns about how the requested funds would be spent and what safeguards were in place to ensure transparency in the bidding process. The representative from Canada concurred, further noting that the Committee also required a detailed workplan. STACFAD members noted that the lack of adequate and specific information greatly hampered their ability to ensure transparency, accountability and the presence of safeguards.

After a lengthy discussion, STACFAD recommended to the General Council that it authorize and approve a budget including an expense not to exceed \$200,000 for the proposed Automated Hail and Satellite Tracking System. However, STACFAD was extremely concerned with the lack of information upon which the Committee could properly evaluate this requested amount. STACFAD recommended that the General Council instruct the Executive Secretary to provide on or before October 31, 2000 a more detailed report regarding the NEAFC System. STACFAD further recommended to the General Council that it instruct the Executive Secretary, in conjunction with the Fisheries Commission, to provide to the General Council a proposal not later than December 31, 2000 regarding the financing of the
Automated Hail and Satellite Tracking System. This proposal should, at a minimum, consist of the following:

- Clearly defined procedures for selecting consultants/experts (with proven experience with similar type projects) to prepare a Call for Tender and to advise on the most suitable tender,
- An implementation workplan for the Automated Hail and Satellite Tracking System,
- Procedures guaranteeing transparency to the Contracting Parties in expenditures, decision-making and the selection process,
- An evaluation process, including periodic progress reports.

The representative from the European Union noted that the Fisheries Commission might find it necessary to appoint an Ad Hoc Working Group that would meet from time to time to advise the Executive Secretary on this matter.

STACFAD further recommended that the General Council instruct the Executive Secretary, in conjunction with the Fisheries Commission, to provide a full report on the status of the spending under this budget item and the implementation of the Automated Hail and Satellite Tracking System at the next Annual Meeting.

7. Administrative and Financial Statements for 2000 (July)

The Committee reviewed the Administrative Report and Financial Statements (NAFO GC Doc. 00/5) for 2000.

The representative of the Scientific Council noted that the term of Manfred Stein (European Union), Chair of STACFEN, had been extended until September 2001.

The Secretariat reviewed the financial statements for 2000 contained in Statement 1 of the Administrative and Financial Statements.

- the total 2000 projected expense obligations are \$1,160,500, which is \$3,500 over the approved budget of \$1,157,000.
- salaries are projected to be \$5,000 over budget. According to Rule 6.1 of the Financial Regulations, NAFO follows the salary scale of the Public Service of Canada. Salary levels were updated pursuant to contracts ratified by the Treasury Board of Canada and the Public Service Alliance of Canada. The contracts were for the period June 1999 to June 2000 and salary increases were retroactive to June 1999.
 - The costs of the Annual General Meeting and Scientific Council Meetings are projected to be \$9,000 over budget. The Committee noted that due to the practice of establishing budgetary amounts prior to receiving the final list of meetings, this budget item had been consistently underestimated for several years.
- Computer Services is projected to be \$34,000 under budget. From the \$35,000 allocated for the automation of the hail system, \$330 was expended and from the \$15,000 allocated to computer services, \$15,670 was expended, \$670 over-budget.

STACFAD recommended to the General Council that in accordance with Rule 6.1 of the Financial Regulations it approve the 2% salary increase for Secretariat employees, in line with contracts negotiated between the Public Service Alliance of Canada and the Treasury Board of Canada. This increase will be applied retroactively from June 1999. This salary increase is independent from the recent pay equity settlement granted to the Public Service Alliance of Canada.

The representative of the European Union noted that expenditures for intersessional meetings were significantly over budget for a number of years running. For information purposes the Executive Secretary provided STACFAD W/P 00/8 (Annex 3) outlining certain travel and meeting cost estimates for the NAFO Secretariat. The Committee requested the Executive Secretary to provide such reports on annual basis. The Committee agreed to address this issue in more detail under Agenda item 9, Preliminary Budget Estimate for 2001. The representative of the Scientific Council requested that the costs of the November Scientific Council shrimp meeting be moved from under the intersessional line item to the Scientific Council/Annual Meeting line item, as this was a regular meeting occurring each year. The Committee agreed to this request.

Contributions Receivable from Contracting Parties for 2000 fiscal year are \$71,489. These outstanding contributions are due from Cuba (\$17,459.98), France (in respect of St. Pierre et Miquelon) (\$19,171.13), Republic of Korea (\$16,842.79) and the Russian Federation (\$18,015.45). The representative from France (in respect of St. Pierre et Miquelon) noted that its payment was in the process of being issued. It was also noted that contributions were not received from Bulgaria and Romania.

The representative from the Russian Federation informed participants that a change in the approval process of payments to international organizations within his government has delayed its payment. Efforts are however being made to make the contribution in the near future.

Canada informed the Committee that their communications with Romania have indicated that Romania has put in motion procedures to withdraw from NAFO. The withdrawal process within the Romanian Government is a lengthy process, although, indications are that most of the approvals have gone forward. Any estimates on the completion of this process would be premature at this time.

USA noted that it had in previous years forwarded démarches to the Governments of Bulgaria and Romania with no positive information on their intentions. This however, was the first year that the USA had not attempted such contacts.

The Executive Secretary noted that Bulgaria had contacted the NAFO Secretariat (GC W/P 00/2) with intentions to confirm its participation to NAFO and continue its membership as well as to negotiate future quotas and other conditions of this participation, including payment of outstanding contributions.

The Committee discussed how the payment of any outstanding contributions would be applied. The Committee instructed the Executive Secretary to look into this matter and report to STACFAD at the next Annual Meeting.

As in prior years, STACFAD recommended to the General Council that contributions from Bulgaria and Romania be deemed uncollectible and recommended that these amounts be applied against the Accumulated Surplus.

A schedule of outstanding contributions detailing the total amounts due from Bulgaria and Romania is attached. (Annex 4).

The Committee recommended that Contracting Parties continue attempts to contact Bulgaria and Romania in order to ascertain whether they intend to participate in NAFO and to inform them of their outstanding contributions. The Committee further recommended that Contracting Parties exchange information about such contacts through the NAFO Secretariat.

The Committee noted that fishery statistics (STATLANT 21A & 21B) were outstanding from a number of Contracting Parties. The Committee also wished to note the recommendation made by Scientific Council at its June 2000 Meeting that 'the Scientific Council should prepare a document for submission to the General Council and the Fisheries Commission on the adverse effect the absence of the STATLANT 21A and 21B data was having on the work of the Scientific Council'.

STACFAD recommended to the General Council that Contracting Parties be urged to submit their fishery statistics (STATLANT 21A & 21B) to the NAFO Secretariat on time to ensure the ongoing integrity of the NAFO Statistical database.

8. Review of the Accumulated Surplus Account

The accumulated surplus account was reviewed and it was noted that the year-end balance is estimated to be \$234,814 provided that all outstanding member contributions (excluding Bulgaria/Romania) are received.

As in past years, STACFAD recommended that \$75,000 be maintained as a minimum balance in this account in order to fulfill NAFO's financial obligations in early 2001 until contributions are received.

The question was raised whether the \$75,000 balance maintained in the accumulated surplus in order to finance appropriations pending receipt of annual payments by Contracting Parties is adequate. The Committee requested that the Executive Secretary ask the Auditors to review this issue and report on what would be the minimum/maximum level at the next Annual Meeting.

The remaining estimated accumulated surplus balance (\$159,814) at the end of 2000 would be used to reduce contributions due from Contracting Parties in 2001.

9. Budget Estimate for 2001

The Executive Secretary presented the budget estimate for 2001 (Annex 5).

The Committee reviewed the preliminary budget estimate of \$1,389,000 that noted the following:

- salary levels include a 2% economic increase. This is independent from the recent pay equity settlement granted to the Public Service Alliance of Canada.
- the meeting account has been itemized to reflect the reclassification of the November Scientific Council Shrimp Meeting from the inter-sessional meetings line item to the Annual/Scientific Meetings line item, as well as an increase in the inter-sessional meeting line item, to reflect a more realistic estimate.

a new line item for the development of an automated system at the NAFO Secretariat for handling all hail and satellite tracking reports and their transmission to Contracting Parties with inspection presence in the NAFO Regulatory Area.

The Committee discussed in length the proposal to apply the pay equity settlement granted Canadian public service employees by the Treasury Board of Canada, following a 1998 ruling by the Canadian Human Rights Tribunal to certain clerical employees on the basis of Rule 6.1 of the Financial Regulations. As reflected in STACFAD WP 00/1, this would result in

- signing bonuses in the amount of \$6,000,
 - clerical pay scale increases of approximately \$17, 933 paid to four clerical staff, and
 - pay equity adjustment dating back to 1985 in the amount of \$190,651.

The representative from Canada questioned the applicability of these pay equity increases proposed for certain Secretariat clerical employees. Canada informed the Committee that the pay equity settlement only applies to specific federal government employees in specific job categories. It does not apply to agencies, contractors and the like. For information purposes Canada distributed STACFAD W/P 00/5 concerning Information on Pay Equity Agreement between the Treasury Board of Canada (TB) and the Public Service Alliance of Canada (PSAC) and STACFAD W/P 00/6 concerning information on Implementation of signing bonus re: Program and Administrative Services between TB and PSAC. The representative of Canada mentioned that the issue of pay equity also raises questions with respect to the classification of NAFO employees. Canada suggested that a review of the job descriptions be undertaken to determine fairness of remuneration. The representatives from the United States and the European Community concurred that, although the Committee acknowledged the Secretariat's discretion to grade and remunerate its employees as it saw fit, given the large amount of money involved in the settlement, the Committee would prefer as much transparency as possible.

Canada will undertake to review the job descriptions that will be provided by the NAFO Secretariat. The review will apply to those positions identified as being in the CR Category in STACFAD Working Paper 00/1. Canada will prepare a report outlining the review of its findings for the 2001 Annual Meeting and will circulate that report in advance of the Meeting, if possible.

The Committee remains committed to providing priority to resolving this issue at the next Annual Meeting.

Included in the budget estimate for 2001 are requests from Scientific Council for financial support (STACFAD W/P 00/2) which were discussed and the Committee amended as follows:

- a) Joint ICES/NAFO Symposium, August 2001 (\$8,000).
- b) NAFO Symposium on Deep-sea Fisheries, September 2001 (\$8,000).
- c) Assistant Executive Secretary at FAO & non-FAO Regional Bodies February 2001 (\$3,500).
- d) Assistant Executive Secretary and STACREC Chairman, CWP Session, New Caledonia, July 2001 (\$10,000).
- e) Website work and scanning of Scientific Council Journals (21 Volumes) and Scientific Council Studies (21 Volumes) (\$8,000).

The Committee in reviewing the requests emphasized that the funding for the attendance of the STACREC Chairman at the CWP Session is of an exceptional nature. STACFAD notes that this is

not a precedent for future funding and that Contracting Parties should ensure that appropriate funding is made available for such attendance.

Regarding the funding of Symposia in general, the Committee was concerned that proper procedures be put in place to ensure that i) funding could be recovered in the event that a Symposium was cancelled, after payment was made and that ii) an accounting be submitted by the recipient. The Committee requested that NAFO Secretariat develop such procedures and report back to STACFAD at the next Annual Meeting.

The Committee requested that any future requests for funding be supported by adequate documentation for STACFAD to make an informed decision.

The preliminary calculation of the 2001 billing is \$1,229,186 (Annex 6).

i

STACFAD recommended to the General Council that the budget of \$1,389,000 be adopted for 2001.

10. Preliminary Budget Forecast for 2002

STACFAD reviewed the preliminary budget forecast for 2002 of 1,190,000 (Annex 7) and approved the forecast in principle. It was noted that the preliminary budget forecast for 2002 would be reviewed in detail during the 23^{rd} Annual Meeting.

11. Time and Place of 2002 and 2003 Annual Meetings

The location of the Annual Meeting for 2001 is Cuba. The 2002 and 2003 Annual Meetings will be held in Halifax, Nova Scotia, Canada, unless an invitation to host the Annual Meeting is extended by a Contracting Party and accepted by the Organization.

The dates of the next Annual Meetings are as follows:

2001	-	Scientific Council	-	12-21 September
	-	General Council	-	17-21 September
	-	Fisheries Commission	-	17-21 September
2002	-	Scientific Council	-	11-20 September
	-	General Council	-	16-20 September
	-	Fisheries Commission	-	16-20 September

STACFAD recommended that the dates of the 2003 Annual Meeting be as follows:

2003	-	Scientific Council	•	10-19 September
	-	General Council	-	15-19 September
	-	Fisheries Commission	-	15-19 September

12. Other issues including any questions referred from the General Council during the current Annual Meeting

There were no items referred over from the General Council for the consideration of STACFAD.

13. Adjournment

The final session of the STACFAD meeting adjourned at 1600 hrs on 21 September 2000.

Annex 1. List of Participants

<u>Name</u>

Edith Dussault

Victor Sarda Espinosa

Lise Lind

Fred Kingston Serge Segura Manfred Stein Andrew Thomson

Yousuke Ochi Yoshiharu Kashio

Richard Derkacs

Halvard P. Johansen

M.G. Botvinko Gennady Gusev

Vasyl Chernik

Nikki Brajevich Jean-Pierre Plé Deirdre Warner-Kramer

Leonard I. Chepel Stan M. Goodick Forbes D. Keating

Contracting Party

Canada

Cuba

Denmark (in respect of Faroe Islands and Greenland)

European Union European Union European Union

Japan Japan

Latvia

Norway

Russian Federation Russian Federation

Ukraine

United States of America United States of America United States of America

NAFO Secretariat NAFO Secretariat NAFO Secretariat

Annex 2. Agenda

- 1. Opening by the Chairman, G.F. Kingston (EU)
- 2. Appointment of Rapporteur
- 3. Adoption of Agenda
- 4. Auditor's Report
- 5. Meeting of the Pension Society
- 6. Review of Cost Implications of the Hail and Satellite Tracking Systems in the Regulatory Area
- 7. Administrative and Financial Statements for 2000 (July)
- 8. Review of Accumulated Surplus Account
- 9. Preliminary Budget Estimate for 2001
- 10. Preliminary Budget Forecast for 2002
- 11. Time and Place of 2002-2003 Annual Meeting
- 12. Other issues including any questions referred from the General Council during the current Annual Meeting
- 13. Adjournment

Annex 3. Travel and Meeting Costs Estimate for the NAFO Secretariat (for information purposes only) (STACFAD Working Paper 00/8)

(Canadian \$)

Travel Costs Estimate for NAFO Secretariat

Meeting of FAO and Non-FAO Regional Fishery Bodies FAO Headquarters, Rome	<u>Costs</u>	# of <u>Staff</u>		Total <u>Cost</u>
Airfare Hotel Per Diem & Misc Total Travel Costs Estimate	\$1,500 \$ 900 <u>\$ 900</u> <u>\$3,300</u>	x 1	=	<u>\$3,300</u>
Inter-sessional Meeting – Brussels				
Airfare Hotel Per Diem & Misc Total Travel Costs Estimate	\$1,500 \$1,000 <u>\$900</u> <u>\$3,400</u>	x 2	=	<u>\$6,800</u>
Inter-sessional Meeting - Copenhagen				
Airfare Hotel Per Diem & Misc Total Travel Costs Estimate	\$1,500 \$1,200 <u>\$_900</u> <u>\$3,600</u>	x 2	=	<u>\$7.200</u>
Inter-sessional Meeting - Washington				
Airfare . Hotel Per Diem & Misc Total Travel Estimate	\$ 600 \$1,300 <u>\$ 700</u> <u>\$2,600</u>	x 2		<u>\$5,200</u>
Inter-sessional Meeting - Spain				
Airfare Hotel Per Diem & Misc Total Travel Estimate	\$1,500 \$1,200 <u>\$_900</u> <u>\$3,600</u>	x 2	=:	<u>\$7,200</u>

Meeting Costs Estimate for NAFO Secretariat

Intersessional Meeting - NAFO Headquarters

Airfare	\$ 0			
Hotel	\$ 0			
Per Diem & Misc	\$ 0			
Total Meeting Costs Estimate	\$ 0	=	<u>\$</u>	0

Intersessional Meeting - Dartmouth Holiday Inn

Hotel Rental	\$4,000	
Equipment Rental	\$ 500 [°]	
Phones/Fax Lines	<u>\$ 500</u>	
Total Meeting Costs Estimate	<u>\$5,000</u>	= .

261

<u>\$5,000</u>

The following is a summary of outstanding contributions from Bulgaria and Romania:

	<u>Bulgaria</u>	Romania
1 January – 31 December 1982		\$2,700.75
1 January – 31 December 1983	· ·	11,000.00
1 January - 31 December 1984		11,483.06
1 January – 31 December 1985		12,688.81
1 January – 31 December 1986		11,784.09
1 January – 31 December 1987		15,273.97
1 January – 31 December 1988		14,189.50
1 January – 31 December 1989		16,618.05
1 January – 31 December 1990		17,875.65
1 January – 31 December 1991		20,060.56
1 January – 31 December 1992		18,702.14
1 January – 31 December 1993	18,109.12	17,473.10
1 January – 31 December 1994	14,893.10	14,893.10
1 January – 31 December 1995	16,614.28	16,614.28
1 January – 31 December 1996	15,944.93	15,944.93
1 January – 31 December 1997	15,002.75	15,002.76
1 January – 31 December 1998	16,121.90	16,121.89
1 January – 31 December 1999	16,267.88	16,267.87
1 January – 31 December 2000	16,842.79	16,842.79
	\$129,796.75	\$281,537.30

Annex 5. Budget Estimate for 2001

NORTHWEST ATLANTIC FISHERIES ORGANIZATION

Budget Estimate for 2000

	(Canadiai	n Dollars)		
			Preliminary	
	Approved	Projected	Budget	Budget
	Budget	Expenditures	Forecast	Estimate
	for 2000	for 2000	for 2001	for 2001
1. Personal Services				
a) Salaries	\$ 677,500	\$ 682,500	\$691,000	\$699,500 [°]
b) Superannuation and Annuities	73,500	73,500	74,000	76,000
c) Additional Help	1,000	1,000	1,000	1,000
d) Group Medical and Insurance Plans	52,000	57,000	52,000	57,500
e) Termination Benefits	23,000	34,000	21,500	23,000 ¹
f) Accrued Vacation Pay	1,000	1,000	1,000	1,000
g) Termination Benefits Liability	10,000	10,000	10,000	10,000
2. Travel	20,000	20,000	10,000	19,000
3. Transportation	1,000	1,000	1,000	1,000
4. Communications	60,000	57,000	60,000	60,000
5. Publications	28,000	29,000	37,000	37,000
Other Contractual Services	43,000	45,000	43,000	44,000
Materials and Supplies	30,000	30,000	30,000	30,000
8. Equipment	5,000	5,000	5,000	5,000
9. Meetings				
Annual General Meeting and	•			
Scientific Council Meetings	62,000	71,000	61,000	64,000
Inter-sessional Meetings	20,000	27,500	20,000	30,000
Symposium	50,000	16,000	15,000	15,000
10. Computer Services	50,000	16,000	15,000	15,000
11. Automated Hail System	-		-	200,000
	\$1,157,000	\$1,160,500	\$1,132,500	\$1,389,000

^a Contracts between the Treasury Board of Canada and the Public Service Alliance of Canada negotiated during the year included retro-active salary increases to June 1999. 1999. If approved by General Council, NAFO's projected salaries for the year 2000 will be 682,500. NAFO's salaries budget estimate for 2001 includes a 2% economic increase.

^b This figure is for 2001 credits and conforms with NAFO Staff Rule 10.4(a).

^c Travel costs for 2001 include () the Assistant Executive Secretary to the February 2001 meeting of the Food and Agriculture Organization of the United Nations (FAO) and Non-FAO Regional Fishery Bodies or Arrangements and the associated Co-ordinating Working Party on Fishery Statistics (CWP) Intersessional Meeting at FAO Headquarters, Rome, Italy, ii) the Assistant Executive Secretary and the STACREC Chairman to the CWP 19th Session in Noumea, New Caledonia (9-13 July 2001) and iii) two staff members to the meeting of Directors and Executive Secretaries of the International Fisheries Commissions Pension Society re: pension scheme, April 2001, Vancouver, Canada.

^dThis figure includes the cost for Annual Meeting, September 2001, Havana, Cuba and the Scientific Council Meeting, June 2001, Halifax, Canada, and the Scientific Council Shrimp Meeting, November 2001, venue to be determined.

^eGeneral provision for inter-sessional meetings during 2001.

^f Contribution to the Joint ICES/NAFO Symposium, August 2001 and NAFO Symposium on Deep-sea Fisheries, September 2001.

^gThe 2000 budget included \$35,000 for the automation of the hail system at NAFO Headquarters.

Annex 6. Preliminary Calculation of Billing for 2001

Preliminary calculation of billing for Contracting Parties against the proposed estimate of \$1,389,000 for the 2001 financial year (based on 18 Contracting Parties to NAFO). (Canadian Dollars)

> 60% of funds required = \$737,511.66 30% of funds required = 368,755.74 10% of funds required = 122,918.60

Contracting Parties	Nominal Catches for 1998	% of Total Catch in the Convention Area	10%	30%	60%	Amount Billed
Bulgaria	-	-	-	\$ 20,486.43	-	\$ 20,486.43
Canada ¹	493,576	59.28	\$77,195.83	20,486,43	\$437,196.88	534,879,14
Cuba	-	-	-	20,486.43	-	20,486.43
Denmark (in respect of Faro	es					
and Greenland)1.2	97,555	11.72	15,257,71	20,486.43	86,436.36	122,180.50
Estonia	5,533	0.66	-	20,486.43	4,867.58	25,354.01
European Union	23,209	2.79	-	20,486.43	20,576.57	41,063.00
France (in respect of St. Pier	те					
et Miquelon)	5,394	0.65	843.63	20,486,43	4,793.83	26,123.89
Iceland	6,572	0.79	-	20,486.43	5,826.34	26,312.77
Јарал	3,000	0.36	-	20,486.43	2,655.04	23,141.47
Republic of Korea	-	-	-	20,486.43	-	20,486.43
Latvia	1,191	0.14	-	20,486.43	1,032.52	21,518.95
Lithuania	3,107	0.37	-	20,486.43	2,728.80	23,215.23
Norway	1,340	0.16	-	20,486.43	1,180.03	21,666.46
Poland	148	0.02	-	20,486.43	147.51	20,633.94
Romania	-	-	-	20,486.43	-	20,486.43
Russian Federation	2,601	0.31	-	20,486.43	2,286.30	22,772.73
Ukraine	-	-	-	20,486.43	-	20,486.43
United States of America ¹	189,394	22.75	29,621.43	20,486.43	167,783.90	217,891.76
	832,620	100.00	\$122,918.60	\$368,755.74	\$737,511.66	\$1,229,186.00
Funds required to meet 1 Jan	nuary - 31 Dec	ember 2001 Ad	Iministrative Bu	dget		\$1,229,186.00

¹ Provisional Statistics used when calculating 1998 nominal catches which have not been reported from some Contracting Parties.

² Faroe Islands = 8,345 metric tons Greenland = 89,210 metric tons

Annex 7. Preliminary Budget Forecast for 2002 (Canadian Dollars)

NORTHWEST ATLANTIC FISHERIES ORGANIZATION

1.	Personal Services	
	 a) Salaries b) Superannuation and Annuities c) Additional Help d) Group Medical and Insurance Plans e) Termination Benefits f) Accrued Vacation Pay g) Termination Benefits Liability 	\$ 707,000 80,000 1,000 59,000 20,000 ^a 1,000 10,000
2.	Travel	20,000 ^b
3.	Transportation	1,000
4.	Communications	60,000
5.	Publications	29,000
6.	Other Contractual Services	45,000
7.	Materials and Supplies	. 30,000
8.	Equipment	5,000
9.	Meetings Annual General Meeting and Scientific Council Meetings Inter-sessional Meetings	65,000 [°] 30,000
10.	Computer Services	15,000
11.	Automated Hail System	12,000 ^d
		\$1,190,000

а This figure is for 2002 credits and conforms with NAFO Staff Rule 10.4(a). b

d Annual communication charge for X.25 line.

Travel costs for 2002 include i) two persons to meeting of Directors and Executive Secretaries of the International Commissions Pension Society re discussion of pension scheme for employees, April 2002 ii) the Assistant Executive Secretary's attendance at a sessional meeting of CWP and iii) the Executive Secretary's home leave to the Ukraine.

с This figure includes the cost for Annual Meeting, September 2002 and the Scientific Council Meeting, June 2002, in Halifax, N.S., Canada and for Scientific Council Shrimp Meeting, November 2002.

PART III

Report of the Standing Committee on the Fishing Activities of non-Contracting Parties in the Regulatory Area (STACFAC)

1. Opening by the Chairman

The meeting was called to order by the Chairman, Dr Jean-Pierre Plé (USA) at 10.15 on 18 September 2000. The following Contracting Parties were present: Canada, Denmark (in respect of the Faroe Islands and Greenland), the European Community (EC), France (in respect of St. Pierre and Miquelon), Iceland, Japan, Korea, Norway, and the United States of America (USA) (Annex 1). The Chairman warmly welcomed all delegates to the meeting.

2. Appointment of Rapporteur

Mr. Alan Gray (EC) was appointed rapporteur.

3. Adoption of Agenda

The Provisional Agenda was presented for adoption by delegates. Denmark (in respect of the Faroe Islands and Greenland) proposed to add discussion on the potential effect of the outcome of the discussions on Illegal, Unregulated and Unreported (IUU) Fishing, being held in the FAO, would have on NAFO. France (in respect of St. Pierre and Miquelon) proposed that relations between NAFO and other regional fisheries management organisations (RFMOs) should also be discussed. The EU wanted to raise the issue of double flagging of vessels. With the agreement to the inclusion of these items, the agenda was adopted (Annex 2).

4. Review of 2000 information on activities of non-Contracting Party vessels in the Regulatory Area

The Contracting Parties reported no sightings of non-Contracting Party vessels in the Regulatory Area thus far in 2000. In reply to a question from the Chairman on the overall level of surveillance, the EU and Canadian representatives noted that their respective surveillance efforts had not diminished.

The Committee agreed that the fact that there had been no sightings of non-Contracting Party fishing vessels in the Regulatory Area does not necessarily imply an absence of such activity. The EU representative noted that there was reason to believe that such activities were being undertaken and being reported as catches taken inside the Regulatory Area of other regional organizations. This could be addressed by the strengthening of co-operation and information exchange between regional organizations.

The Chairman concluded that although there were no reported sightings of non-Contracting Party vessels fishing in the Regulatory Area, the Contracting Parties must remain vigilant and continue efforts to deter non-Contracting Party fishing in the Regulatory Area.

5. Review of 2000 information on landings and transshipments of fish caught by non-Contracting Party vessels in the Regulatory Area

No Contracting Party reported any landings or transhipments by non-Contracting Party vessels in the Regulatory Area. However, the EU representative noted that there had been indications of landings where there were doubts on the veracity of the origin of the fish landed. This could be due to misreporting of where the fish were caught. The Chairman again complimented Denmark (in respect of the Faroe Islands and Greenland) for the action that it had taken in 1999 to deny the attempted landing of fish from non-Contracting Party vessels. This action sent a strong message to these vessels of the measures they now faced.

6. Review of 2000 information on imports of species regulated by NAFO from non-Contracting Parties whose vessels have fished in the Regulatory Area

No Contracting Party reported any information on imports of species regulated by NAFO from non-Contracting Parties whose vessels fished in the Regulatory Area. The EU representative reported that although there had been no imports from non-Contracting Party vessels that fished in the NAFO area, investigations were underway in the EU regarding imports of uncertain origin, possibly from or via African states.

7. Reports by Contracting Parties on diplomatic contacts with non-Contracting Party Governments concerning fishing in the Regulatory Area

In accordance with the decisions taken by the General Council, diplomatic démarches were made to the non-Contracting Party governments whose vessels were sighted fishing in the Regulatory Area in 1999. Such démarches were made to Honduras, Belize, Sierra Leone, and Sao Tomé and Principe.

Canada reported on the contacts it had with Honduras regarding two vessels that were identified as being registered in that country. Honduras replied to this démarche, which was forwarded to the NAFO Secretariat (STACFAC Working Paper 00/1). In its reply, Honduras confirmed that the two vessels concerned, the "High Sierra" and the "Albri II" are not registered in Honduras.

The U.S. representative reported that the United States made the NAFO démarches to Belize and Sierra Leone. The reply from Belize confirmed that the "High Sierra" and the "Albri II" are not registered in Belize. No reply has been received from Sierra Leone.

The EU representative reported that a démarche was made to Sao Tomé and Principe regarding the vessels "Austral" or "Australia" in July, following the completion of the legislative process concerning the EU regulation implementing the NAFO Scheme. No reply has been received. However, inspection reports indicate that the registration of this vessel in Sao Tomé and Principe was of a temporary nature and has not been renewed. Nonetheless, this should not stop further contacts with Sao Tomé and Principe requesting that all vessels registered there should comply with the conservation measures applied in various RFMOs. The EU representative undertook to follow-up on the démarche made to Sao Tomé and Principe.

The Chairman noted that there are reasons to believe that Belize and Honduras fully understand the seriousness of the matter. The Chairman added there seemed to be improved co-operation with Belize, Honduras, Sierra Leone, and Panama and this seemed to indicate that the démarches made to these countries had been successful. France noted that although these countries may now be cooperating with NAFO, there is still concern regarding similar cooperation with other RFMOs, in particular with respect to non-Contracting Party fishing vessels flying the flag of Panama in the Convention Area of the Commission for the Conservation of Antarctic Marine Living Resources (CCAMLR).

France (in respect of St. Pierre and Miquelon) raised a question about the flag status of the non-Contracting Party vessel, the "Austral", that was sighted in the North-East Atlantic Fisheries Commission (NEAFC) area in 1999. The Chairman stated that at the 1999 Annual Meeting, STACFAC concluded that there were reasonable grounds to suspect that this vessel was without nationality. The Chairman then explained U.S. national measures regarding the treatment of stateless vessels engaged in large-scale driftnet fishing. He noted that under U.S. law such vessels could be assimilated and considered to be subject to U.S. jurisdiction. After boarding and inspection, such vessels may be arrested if it is determined that the vessel violated U.S. law. Current U.S. law does not extend to dealing with stateless fishing vessels that undermine NAFO conservation and management measures, but the need for such legislation was raised with Congress. Canada informed the Committee that it had national legislation that permitted it to board, inspect and arrest stateless non-Contracting Party vessels operating in the Regulatory Area. Norway reported that it was developing legislation to deal with stateless fishing vessels.

The Chairman prepared letters for signature by the NAFO President, to be sent to Belize, Honduras, Panama, and Sierra Leone. The letters note that there have been no sightings of non-Contracting Party vessels from any country. The letters go on to express satisfaction for their cooperation in this matter and urges them to co-operate in similar fashion with other organizations. The letters also bring to the attention of these countries the current information available on vessels previously sighted in the NAFO Regulatory Area. The letters are attached in Annex 3.

Regarding the request from France (in respect of St. Pierre and Miquelon) concerning contact and information exchange between NAFO and other RFMOs, the Chairman referred the Committee to STACFAC Working Paper 00/2, Information on non-Contracting Party Fishing Activities (by other International Organizations), which contained letters from the International Commission for the Conservation of Atlantic Tunas (ICCAT) and NEAFC. The Chairman noted that co-operation between NAFO and other RFMOs, and in particular with NEAFC, is a problem faced by RFMOs and highlighted the need to continue and improve information exchange and co-operation among RFMOs. ICCAT and NEAFC have requested that NAFO maintain an open exchange of information. The Chairman noted that such exchanges will benefit all three organizations.

Denmark (in respect of the Faroe Islands and Greenland) welcomed this action by other RFMOs and noted the problem of reconciling reported sightings in one RFMO and the reporting of catches from another regulated area. Such information exchange should be carried out when sightings are made and not left until the annual meeting of the respective organizations. This point was supported by Canada. The Committee agreed that the NAFO Secretariat should communicate with ICCAT, NEAFC, and CCAMLR and request that they share with NAFO information on sightings of non-Contracting Party vessels in their respective areas as soon as possible. The Committee also agreed that the NAFO Secretariat should share with these organizations such sightings in the NAFO Regulatory Area as soon as possible. The Chairman asked that members of NAFO that are also members of NEAFC to support better communication between these two organizations.

8. Review of the performance of the NAFO Scheme to deal with non-Contracting Parties fishing in the NAFO Regulatory Area

The Chairman noted that NAFO/GC Doc. 00/1, the 1999 Report to the General Council, presented a positive picture regarding the actions of Contracting Parties in the implementation of NAFO Scheme. It was also noted that the performance of the Scheme was aided by improvements in the information exchange with non-Contracting Party governments and other RFMOs.

The representative of the EU agreed that this gave a fair description of the situation and that NAFO should take pride in the Scheme as it set the precedent for other organizations.

The Chairman noted that point 4 from last year's recommendation had not been carried out and should be resubmitted to the General Council.

The Chairman also reminded the Committee that in 1999 the Committee recommended to the General Council that Contracting Parties should submit annual reports under paragraph 13 of the Scheme, including negative reports if appropriate.

9. Report and Recommendations to the General Council

STACFAC recommends to the General Council that:

- Démarches, in the form of letters signed by the President of NAFO be made to Belize, Honduras, Panama, and Sierra Leone.
- The Executive Secretary circulate, as soon as possible to the secretariats of ICCAT, NEAFC, and CCAMLR reports of non-Contracting Party fishing activity in the NAFO Regulatory Area, and request such organizations to share reports of non-Contracting Party fishing activity in their respective areas with NAFO.
- The Executive Secretary circulate NAFO/GC Doc. 00/1 to ICCAT, NEAFC, and CCAMLR.
- The Contracting Parties submit a report at the next Annual Meeting on what legal, administrative and practical action they have taken to implement the Scheme.
- STACFAC undertake to study the impact of the proposed International Plan of Action on Illegal, Unregulated and Unreported (IUU) Fishing on NAFO at its next Annual Meeting. (See discussion under Section 11.)

10. Election of Chairman and Vice-Chairman

The Chairman brought to Committee's attention that the terms of service of both the Chairman and the Vice-Chairman would soon expire. Mr Daniel Silvestre (France) (in respect of St. Pierre and Miquelon) was elected Chairman for the next two years. Ms Nadia Bouffard (Canada) was elected Vice-Chairman for the next two years. The Committee thanked the current Chairman for his work during the past five years and wished him well on his new assignment.

11. Other matters

1) Illegal, Unregulated and Unreported (IUU) Fishing - effect of work in FAO on NAFO

It was agreed that STACFAC should review the outcome of the FAO meetings concerning the development of a proposed International Plan of Action on IUU Fishing with a view to determining whether follow-up action within NAFO is appropriate. The Committee is confident that the FAO is well aware of developments in NAFO on this subject.

2) Double flagging agreements

The EU representative informed the Committee of information that his delegation had received regarding commercial operators within a Contracting Party actively seeking approval to operate under a double flagging arrangement. This would permit the operators to carry out fishing activities under two flags in order to make the most possible use of quotas of different Contracting Parties. The EU representative expressed concern about any such practice, which, if it were to

emerge, would lead to these vessels being considered stateless under international law. In such cases, these vessels may be subject to the NAFO Scheme, as amended in 1999.

All other members of the Committee shared this concern and shared the view that greater vigilance and clarity of the facts are required. The Chairman concluded that the double flagging issue, if confirmed, would violate international law. This practice raised the possibility of undermining NAFO Conservation and Enforcement Measures. It was also noted that this is an issue that might go beyond the remit of STACFAC. The Committee notes that all Contracting Parties should respect the provisions of Article 92 of the 1982 United Nations Convention on the Law of the Sea (UNCLOS).

12. Adjournment

The Committee adjourned at 13.45 on Wednesday, 20 September 2000.

Annex 1. List of Participants

Name of Participant

Nadia Bouffard Allison Saunders

Einar Lemche

Į

ł

į

Emanuel Rosing Andras Kristiansen

Friedrich Wieland Alan Gray Isabel Aragon Sonja Feldthaus

Daniel Silvestre

Valèrie Sinquin

Kolbeinn Árnason

Kengo Tanaka Saori Nagase

You-Sub Jung Sang-Jung Yoo

Stein Owe

Gene Martin Jean-Pierre Plé

Contracting Party

[,] Canada

Denmark (in respect of the Faroe Islands and Greenland).

European Community

**

France (in respect of St. Pierre and Miquelon)

Iceland

Japan

Korea

Norway

United States of America

Annex 2. Agenda

- 1. Opening by the Chairman, Dr J-P Plé (USA)
- 2. Appointment of Rapporteur
- 3. Adoption of Agenda
- 4. Review of 2000 information on activities of non-Contracting Party vessels in the Regulatory Area
- 5. Review of 2000 information on landings and transhipments of fish caught by non-Contracting Party vessels in the Regulatory Area
- 6. Review of information on imports of species regulated by NAFO from non-Contracting Party whose vessels have fished in the NAFO Regulatory Area
- 7. Reports by Contracting Parties on diplomatic contacts with non-Contracting Party Governments concerning fishing the Regulatory Area
- 8. Review of the performance of the NAFO Scheme to deal with non-Contracting Parties fishing in the NAFO Regulatory Area
- 9. Report and Recommendations to the General Council
- 10. Election of Chairman and Vice-Chairman
- 11. Other matters
 1) Illegal, Unregulated and Unreported (IUU) Fishing effect of work in FAO on NAFO
 - 2) Double flagging agreements
- 12. Adjournment

Annex 3. NAFO Letters

Address Foreign Minister of Belize

Dear Mr. Minister:

I have been instructed by all members of the Northwest Atlantic Fisheries Organization (NAFO) present at its 22nd Annual Meeting to note that NAFO is encouraged that no new vessels registered in Belize, or any other Non-Contracting Party, have thus far been observed fishing in the NAFO Regulatory Area during 2000.

I wish also to express the satisfaction of all NAFO members for the cooperation of your government to counter Non-Contracting Party fishing activities in the NAFO Regulatory Area. NAFO would appreciate your continued cooperation to deter any such fishing activity in the future by vessels flying the flag of Belize, and urges you to cooperate in similar fashion with other regional fisheries management organizations. In particular, NAFO asks that Belize remain vigilant concerning any attempt to misuse the flag of Belize in order to undermine NAFO conservation and management measures.

In this regard, NAFO draws your particular attention to the vessels "Austral" (also known as the "Australia"), the "High Sierra" (also known as the "Albri II"), "Porto Santo", and "Santa Princesa", all of which had previously been identified by NAFO as flying the flag of a Non-Contracting Party while fishing in the NAFO Regulatory Area. In addition, as a result of conflicting information regarding their nationality, NAFO determined at its 21st Annual Meeting that there are reasonable grounds for suspecting that the vessels "Austral"/"Australia" and the "Albri II"/"High Sierra" are vessels without nationality. NAFO would be grateful that you notify the organization if any of the above mentioned vessels attempt to register in Belize.

On behalf of the Contracting Parties to NAFO present at its 22nd Annual Meeting, September 18-22, 2000: Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, European Union, France (in respect of St. Pierre and Miquelon), Iceland, Japan, Republic of Korea, Latvia, Lithuania, Norway, Poland, the Russian Federation, Ukraine and the United States of America.

(DATE)

E. Oltuski President and Chairman of the General Council

Address Foreign Minister of Honduras

Dear Mr. Minister:

I have been instructed by all members of the Northwest Atlantic Fisheries Organization (NAFO) present at its 22nd Annual Meeting to note that NAFO is encouraged that no new vessels registered in Honduras, or any other Non-Contracting Party, have thus far been observed fishing in the NAFO Regulatory Area during 2000.

1 wish also to express the satisfaction of all NAFO members for the cooperation of your government to counter Non-Contracting Party fishing activities in the NAFO Regulatory Area. NAFO would appreciate your continued cooperation to deter any such fishing activity in the future by vessels flying the flag of Honduras, and urges you to cooperate in similar fashion with other regional fisheries management organizations. In particular, NAFO asks that Honduras remain vigilant concerning any attempt to misuse the flag of Honduras in order to undermine NAFO conservation and management measures.

In this regard, NAFO draws your particular attention to the vessels "Austral" (also known as the "Australia"), the "High Sierra" (also known as the "Albri II"), "Porto Santo", and "Santa Princesa", all of which had previously been identified by NAFO as flying the flag of a Non-Contracting Party while fishing in the NAFO Regulatory Area. In addition, as a result of conflicting information regarding their nationality, NAFO determined at its 21st Annual Meeting that there are reasonable grounds for suspecting that the vessels "Austral"/"Australia" and the "Albri II"/"High Sierra" are vessels without nationality. NAFO would be grateful that you notify the organization if any of the above mentioned vessels attempt to register in Honduras.

On behalf of the Contracting Parties to NAFO present at its 22nd Annual Meeting, September 18-22, 2000: Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, European Union, France (in respect of St. Pierre and Miquelon), Iceland, Japan, Republic of Korea, Latvia, Lithuania, Norway, Poland, the Russian Federation, Ukraine and the United States of America.

(DATE)

E. Oltuski President and Chairman of the General Council

Address Foreign Minister of Panama

Dear Mr. Minister:

I have been instructed by all members of the Northwest Atlantic Fisheries Organization (NAFO) present at its 22nd Annual Meeting to note that NAFO is encouraged that no new vessels registered in Panama, or any other Non-Contracting Party, have thus far been observed fishing in the NAFO Regulatory Area during 2000.

I wish also to express the satisfaction of all NAFO members for the cooperation of your government to counter Non-Contracting Party fishing activities in the NAFO Regulatory Area. NAFO would appreciate your continued cooperation to deter any such fishing activity in the future by vessels flying the flag of Panama, and urges you to cooperate in similar fashion with other regional fisheries management organizations. In particular, NAFO asks that Panama remain vigilant concerning any attempt to misuse the flag of Panama in order to undermine NAFO conservation and management measures.

In this regard, NAFO draws your particular attention to the vessels "Austral" (also known as the "Australia"), the "High Sierra" (also known as the "Albri II"), "Porto Santo", and "Santa Princesa", all of which had previously been identified by NAFO as flying the flag of a Non-Contracting Party while fishing in the NAFO Regulatory Area. In addition, as a result of conflicting information regarding their nationality, NAFO determined at its 21st Annual Meeting that there are reasonable grounds for suspecting that the vessels "Austral"/"Australia" and the "Albri II"/"High Sierra" are vessels without nationality. NAFO would be grateful that you notify the organization if any of the above mentioned vessels attempt to register in Panama.

On behalf of the Contracting Parties to NAFO present at its 22nd Annual Meeting, September 18-22, 2000: Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, European Union, France (in respect of St. Pierre and Miquelon), Iceland, Japan, Republic of Korea, Latvia, Lithuania, Norway, Poland, the Russian Federation, Ukraine and the United States of America.

(DATE)

E. Oltuski President and Chairman of the General Council

Address Foreign Minister of Sierra Leone

Dear Mr. Minister:

I have been instructed by all members of the Northwest Atlantic Fisheries Organization (NAFO) present at its 22nd Annual Meeting to note that NAFO is encouraged that no new vessels registered in Sierra Leone, or any other Non-Contracting Party, have thus far been observed fishing in the NAFO Regulatory Area during 2000.

I wish also to express the satisfaction of all NAFO members for the cooperation of your government to counter Non-Contracting Party fishing activities in the NAFO Regulatory Area. NAFO would appreciate your continued cooperation to deter any such fishing activity in the future by vessels flying the flag of Sierra Leone, and urges you to cooperate in similar fashion with other regional fisheries management organizations. In particular, NAFO asks that Sierra Leone remain vigilant concerning any attempt to misuse the flag of Sierra Leone in order to undermine NAFO conservation and management measures.

In this regard, NAFO draws your particular attention to the vessels "Austral" (also known as the "Australia"), the "High Sierra" (also known as the "Albri II"), "Porto Santo", and "Santa Princesa", all of which had previously been identified by NAFO as flying the flag of a Non-Contracting Party while fishing in the NAFO Regulatory Area. In addition, as a result of conflicting information regarding their nationality, NAFO determined at its 21st Annual Meeting that there are reasonable grounds for suspecting that the vessels "Austral"/"Australia" and the "Albri II"/"High Sierra" are vessels without nationality. NAFO would be grateful that you notify the organization if any of the above mentioned vessels attempt to register in Sierra Leone.

On behalf of the Contracting Parties to NAFO present at its 22nd Annual Meeting, September 18-22, 2000: Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, European Union, France (in respect of St. Pierre and Miquelon), Iceland, Japan, Republic of Korea, Latvia, Lithuania, Norway, Poland, the Russian Federation, Ukraine and the United States of America.

(DATE)

E. Oltuski President and Chairman of the General Council

SECTION VIII

(pages 277 to 386)

Report of the Fisheries Commission and its Subsidiary Body (STACTIC), 22nd Annual Meeting 18-22 September 2000 Boston, Massachusetts, USA

Rej	port of the Fisheries Commission 279
1.	Opening Procedures
2.	Administrative
3.	Conservation and Enforcement Measures
4.	Conservation of Fish Stocks in the Regulatory Area
6.	Closing Procedures
	Annex 1. List of Participants
	Annex 2. Agenda
	Annex 3. Amendment to the Conservation and Enforcement
	Measures regarding Observers
	Annex 4. Proposal to Amend the NAFO Conservation and Enforcement
	Measures with a view to introducing satellite based
	vessel monitoring and related measures
	Annex 5. Amendment to the NAFO Conservation and Enforcement
	Measures regarding Shrimp in Division 3M 312
	Annex 6. Working Group on 3M Shrimp Fishery Data, 27 March 2001,
	Copenhagen - Provisional Agenda
	Annex 7. Special Fisheries Commission Meeting on Shrimp
	Allocation and Management in the NAFO Regulatory
	Area, 28-29 March, Copenhagen - Provisional Agenda 315
	Annex 8. Paper on Chartering 316
	Annex 9. Formats for the Electronic Transmission of NAFO Hails
	from Contracting Parties to the NAFO Secretariat
	Annex 10. Proposal to amend the NAFO Conservation and Enforcement
	Measures regarding Incidental Catch Limits
	Annex 11. Proposal to amend the NAFO Conservation and Enforcement
	Measures regarding Part VII – Port Inspections
	Annex 12. Statement by the Representative of the European Union
	on 2J3KL Cod
	Annex 13. Proposal to amend the NAFO Conservation and Enforcement
	Measures Part I.HOther Measures-Management Measures
	for Shrimp in Div. 3LNO
	Annex 14. Amendment to NAFO Conservation and Enforcement
	Measures regarding 3L Cod 338
	Annex 15. Amendment to NAFO Conservation and Enforcement
	Measures regarding Witch Flounder in Div. 3L 339
	Annex 16. Quota Table for 2001 340
	Annex 17. Fisheries Commission's Request for Scientific Advice on
	Management in 2002 of Certain Stocks in Subareas 3
	and 4, including supplementary questions on Division 3M
	Shrimp for 2001
	Annex 18. List of Decisions and Actions by the Fisheries Commission
	1. 2. 3. 4.

PART II.	Rep	ort of the Standing Committee on International Control (STACTIC)	8
	1. 2.	Opening of the Meeting	8
	3.	Adoption of Agenda	
	4.	Review of Annual Returns of Infringements	8
	5.	Review of Surveillance and Inspection Reports	
	6.	Review of Operation of the Hail System	8
	7.	Observer Program and Satellite Tracking 34	
	8.	Discussion of Other Conservation and Enforcement Measures	0
	9.	Discussion of possible amendments to the Conservation and	
		Enforcement Measures	
		Time and Place of the Next Meeting	
		Other Matters	
		Adoption of Report	
	13.	Adjournment	3
		Annex 1. Agenda	4
		Annex 2. Report of STACTIC <i>Ad hoc</i> Working Group on Comparison of the NAFO Hail System with the NEAFC Communication	
		System	5
		Annex 3. Proposal to amend the NAFO Conservation and Enforcement Measures regarding Incidental Catch Limits	2
		Annex 4. Proposal to amend the NAFO Conservation and Enforcement Measures regarding Part VII-Port Inspections	
		Annex 5. Paper on Chartering	
		Annex 6. Proposal to amend the NAFO Conservation and Enforcement Measures with a view to introducing new rules concerning	
		obligatory inspection presence in the Regulatory Area	36

PART I

Report of the Fisheries Commission Meeting (FC Doc. 00/21)

22nd Annual Meeting, 18-22 September 2000 Boston, Massachusetts, USA

1. **Opening Procedures** (items 1-5 of the Agenda)

- 1.1 The meeting was called to order by the Chairman, Mr. P. Gullestad (Norway) at 0915 hrs. on 19 September 2000. Representatives from the following Contracting Parties were present: Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, the European Union (EU), France (in respect of St. Pierre et Miquelon), Iceland, Japan, the Republic of Korea, Latvia, Lithuania, Norway, Poland, Russia, Ukraine and the United States of America (Annex 1).
- 1.2 Mr. Patrick E. Moran (United States) was appointed Rapporteur.
- 1.3 The Provisional Agenda was reviewed and two changes were agreed. At the request of the General Council, the Report of the Meeting on Shrimp Stocks was inserted as Agenda item 10a. It was also agreed that the Representative of Latvia would present the results of the meeting on the bloc quota following this item. Additionally, an item was proposed by Norway with respect to pelagic redfish in Division 1F of the Regulatory Area. This item was identified as a new Agenda item 17.11. The Agenda was **adopted** as amended (Annex 2).
- 1.4 Admission of observers was discussed in the meeting of the General Council.
- 1.5 Publicity was discussed in the meeting of the General Council.

2. Administrative (item 6)

2.1 Review of Membership was discussed at the opening session of the General Council (under provisions of Article XIII.1 of the NAFO Convention).

3. Conservation and Enforcement Measures (items 7-14)

- 3.1 With respect to Agenda item 7, Scientific Council Chair W.B. Brodie (Canada) presented the Report of the Joint Fisheries Commission/Scientific Council Working Group Meeting on the Precautionary Approach (PA). This meeting took place 29 February 2 March 2000, in Brussels, Belgium (NAFO/FC Doc 00/2).
- 3.2 Regarding the issue of harmonization of concepts and terminology, the Working Group examined the results of the February 2000 ICES CWP meeting (SCS Doc. 00/7) and a paper on harmonization submitted by the EU. The Working Group concluded that no formulations of the precautionary approach have been accepted by international fisheries organizations, although some elements of the approach have been implemented by various management authorities. It was agreed that broad similarities exist between the ICES and NAFO versions of the precautionary approach (i.e., biomass limits and biomass buffers), but noted that harvest control rules differ. It was generally agreed that

determination of harvest control rules should be the responsibility of the Fisheries Commission. There was no agreement on the recommendations found in the EU paper, and there was considerable debate regarding the potential relationship (if any) between Flim and Fmsy.

- 3.3 Regarding operationalizing the precautionary approach into management plans for three model stocks, the Working Group reviewed a discussion paper submitted by Canada. This document outlined progress made on cod in Div. 3NO and yellowtail flounder in Div. 3LNO, and proposed additional steps for implementation of the precautionary approach with regard to these stocks. It was noted that there was a need to address harvest control rules in an implementation plan. The Working Group agreed on the next steps in implementation of the precautionary approach for two of the two model stocks. It was noted that work by the Scientific Council relating 3M shrimp is ongoing and will be reviewed again in November 2000, prior to the 2001 fishing season.
- 3.4 The Joint Working Group also agreed on the next steps for implementing the precautionary approach for American Plaice in Div. 3LNO. It was suggested similar detailed implementations plans (such as those outlined for the two model stocks and American Plaice in Div. 3LNO) might be developed for other NAFO stocks. It was also agreed that, for other stocks, management objectives should include rebuilding and maintenance of stock biomass at a level that can support sustainable fisheries and produce stable yields. Additionally, it was agreed that the Fisheries Commission should specify management strategies, ensure that data collection and analysis is carried out, and supply additional technical management measures (such as to address bycatch issues) when necessary.
- 3.5 At the Joint Working Group meeting, two Contracting Parties tabled proposals for modification to the Fisheries Commission's Request for Advice from the Scientific Council for 2001. After no agreement could be reached regarding the inclusion of references to the UN Fish Stocks Agreement in such revised requests, it was agreed that no revisions should take place to the current request for advice. Instead, it was agreed that five items pertaining to advice under the precautionary approach would be submitted to the Scientific Council for consideration.
- 3.6 Regarding the consideration of criteria for re-opening a fishery in light of the precautionary approach, four technical measures were identified by the working group and recommended for consideration by the Fisheries Commission. These measures seek to address: protection of spawners; protection of pre-recruits; concerns with bycatch; and concerns with bycatch of other species. The Joint Working Group also noted a number of additional supporting management measures to complement the application of the precautionary approach during discussions on the model stocks. These additional measures are included in Annexes 6-8 of the Joint Working Group Report, dealing with two of the model stocks (Div. 3NO cod and Div. 3LNO yellowtail flounder) and one additional stock (Div. 3LNO American plaice). Additionally, the Working Group considered a great many other possible supportive management measures.
- 3.7 The report of the Joint Working Group was **adopted**. Discussion followed on whether the working group should continue its work. The Representative of Canada, supported by the United States, strongly supported continued work and adoption of the recommendations of the working group. Canada proposed that NAFO adopt a three-year pilot project (beginning in 2001) during which the work already done relevant to the three model stocks would be operationalized and more stocks would be considered for future

implementation. The Representative of the European Union (EU) and others, however, stated that much more work needed to be done before decisions could be taken regarding implementation of the precautionary approach. He noted inconsistencies between the NAFO model and that of NEAFC, and the lack of agreement among Contracting Parties regarding fundamental elements of the precautionary approach. It was suggested that the Working Group should not meet in 2001, so that some of these issues might be addressed by the Fisheries Commission and bilaterally.

With a view to making further progress on the implementation of the Precautionary 3.8 Approach, it was agreed that a small group of technical experts will meet in the first half of 2001 to advance future work in the Fisheries Commission Working Group. The small meeting will be organized by the European Community. A report from this meeting will be circulated to all Contracting Parties, with a recommendation whether the Working Group should meet prior to the 23rd Annual Meeting, and if so, provide an agenda for the meeting. Any recommendation that the Working Group meet shall be the subject of a mail vote.

- With respect to Agenda item 8, Report of the STACTIC June Meeting, Mr. C. Allen 3.9 (Canada) reported the results of the 27-29 June STACTIC Meeting in Dartmouth, Nova Scotia, Canada (see NAFO/FC Doc. 00/4). This meeting was held to begin work on the scientific requirements for the observer program, amendments to the existing program, and the observer manual. STACTIC also considered possible amendments to the Conservation and Enforcement Measures regarding juvenile fish. Other matters addressed by STACTIC at this meeting included: a review of submissions on shrimp catches and effort days; possible follow-up to the Working Group on the Precautionary Approach; consideration of rule for chartering and the issue of flag hopping; possible harmonization of port inspection reports; preparation of the review and, as appropriate, the revision of the Program for Observers and Satellite Tracking; and new development and/or possible overhaul of the Conservation and Enforcement Measures.
- In discussions following the STACTIC intersessional report, the STACTIC 3.10 recommendation regarding the objectivity of observers (FC Doc. 00/8) was adopted (Annex 3). Although there was general support for a review and clarification of the Conservation and Enforcement Measures, it was agreed that decision on this issue, and others addressed at the STACTIC intersessional meeting, should be deferred pending further discussion during the annual meeting. The report of the June 2000 STACTIC Intersessional Meeting was adopted. However, several delegates expressed reservations regarding Div. 3M shrimp catch and effort data attached as Annex 10 to the STACTIC Report.
- With respect to Agenda item 9, Inspection and Control Measures in the NAFO 3.11 Regulatory Area, Contracting Parties generally supported continued use and enhancement of the NAFO vessel monitoring system (VMS). The EU tabled a proposal in STACTIC that amended the current program for VMS and observers, outlined detailed rules for satellite tracking, and adjusted hail system requirements. While the EU proposal was adopted after brief discussion (NAFO FC Doc. 00/13 - see Annex 4), the Representative from Iceland (supported in principle by Denmark and Norway) expressed dissatisfaction with the 100% level of observer requirement that remained in the program. Iceland noted that such requirements are expensive and unnecessary in fisheries such as that for 3M shrimp. Thus, Iceland stated its intention to formally object to the revised text of the

program. Denmark and Norway clarified that they would not formally object to this revision.

- 3.12 Additionally, the Fisheries Commission **agreed** that provisions on secure and confidential treatment of the electronic reports and messages transmitted in accordance with the NAFO Conservation and Enforcement Measures should be addressed at the STACTIC intersessional meeting and that these provisions as described in STACTIC Working Paper 00/19 are taken into account in the NAFO Secretariat's Call for Tender, the acquisition and implementation of the Automated Hail and Satellite Tracking System.
- 3.13 With Respect to Agenda item 10, Mr. F. Wieland (EU) provided the report of the March 2000 meeting of the Working Group on the Allocation of Fishing Rights in Washington, D.C. (NAFO GC Doc. 00/2). He noted that discussions at this meeting were both challenging and complicated. In discussions relating to the qualifying criteria for stocks not currently allocated, Mr. Wieland stated that there was some agreement that such criteria should be listed in no order of priority and that such a list should not be limiting. Additionally, there was agreement that qualifying Parties must be Fisheries Commission members in good standing. However, there was less consensus regarding the issue of allocation criteria for stocks not allocated. Although there was support for the use of reference fishing patterns in establishing allocations, questions relating to coastal State status/zonal attachment and the use of "others" and "cooperating Party" quotas were not resolved. Additionally, discussions regarding reallocation of already allocated quotas (including stocks currently under moratoria) produced no consensus. The Working Group agreed that guidance should be sought from the Fisheries Commission regarding steps to be taken in the future.
- 3.14 Discussions following the report of the March 2000 allocation intersessional focused on the utility of continued work by the Working Group. The Representative of Denmark (in respect of the Faroe Islands and Greenland) noted that his delegation was among those who have pressed for an early review of the present allocation key. While discussions in the Working Group have been fruitful thus far, there is a lack of political will among Contracting Parties to move the issue forward. He therefore suggested that once stocks begin to recover, allocative issues should be addressed in due time. Thus, he suggested that the Working Group should not meet in 2001. This view was supported by the Representatives of Iceland, the EU, Latvia, Russia, and Norway. The Representatives of the United States, Canada, France (in respect of St. Pierre and Miquelon), and Korea. on the other hand, expressed strong support for continued work. Particular concern was noted that allocation issues pertaining to new stocks must be dealt with in a timely manner. Following further discussion, the Report of the March 2000 Allocation Working Group meeting was adopted and it was decided that the Working Group would not meet in 2001, recognizing the understandings identified in paragraph 3.18 below.
- 3.15 During his presentation of the report on the March 2000 intersessional on allocation, Mr. Wieland also noted that those Contracting Parties included in the "bloc quota" met to discuss possible scenarios for resolving this difficult issue. At the Working Group meeting, this group set a future meeting date and location and considered possible terms of reference for this future meeting. Mr. N. Riekstins (Latvia) reported on the outcome of this subsequent meeting (report at GF/00-566), noting that relevant Parties had agreed on a reference period beginning in 1992 and stocks to which this reference period should be applied. He then listed (in no particular order) some of the agreed criteria and principles of allocations, noting that relevant application and weighting had not yet been decided. Additionally, the Representative of Latvia stated that the group considered some criteria

for allocation of stocks not fished during the reference period. Mr. Riekstins stated that the time and location of the next meeting of the bloc Parties would be announced in the future.

3.16 Regarding Agenda item 10(a), Mr. Wieland (EU) then provided the report of the March 2000 Meeting on Shrimp Stocks in Washington, D.C. (NAFO GC Doc. 00/3), noting that there was general agreement among Contracting Parties that the current effort allocation system for Div. 3M shrimp is not achieving the conservation goals as outlined in the Scientific Council advice for this stock. However, there was not consensus regarding how the current situation might be improved. Mr. Wieland noted that some Parties continued to call for a move to TAC-based management of this stock, while others preferred to simply address relevant problems within the existing effort-based scheme. Additionally, discussions touched on possible new approaches to management of the Div. 3L shrimp stock. With regard to NAFO shrimp stocks, the Working Group agreed that guidance should be sought from the Fisheries Commission regarding steps to be taken in the future.

3.17 Following the report on the meeting on NAFO shrimp stocks, Contracting Parties expressed a variety of views regarding possible scenarios for future management of NAFO shrimp stocks. Some Parties supported a switch to TAC-based management, while others called for continuation of an effort-based management scheme. The Representative of Denmark (in respect of the Faroe Islands and Greenland) emphasized the importance of this issue to his country and called for a special meeting of the Fisheries Commission early next year so that decisions could be taken on 3L and 3M shrimp management in time for the 2001 shrimp fishing season. Canada, the United States, and others supported this proposal, citing strong conservation concerns relating to the 3M shrimp stock. However, other Parties supported the view that management issues relating to these stocks should be dealt with at this annual meeting, in order to ensure that measures are in place for the upcoming season. A great deal of dissatisfaction was also expressed by the Representatives of Iceland, Latvia, Norway, and the EU regarding the confusion surrounding the historical data for the 3M shrimp fishery. The Representative from the Ukraine noted that, under no circumstances, should any NAFO members be forced to accept zero TACs in this fishery. The report of the March 2000 meeting on NAFO shrimp stocks was adopted.

3.18 After further discussion, it was agreed that the current measures in place for 3M shrimp should be updated for use during the 2001 fishing season. Thus, the measures as outlined in FC Doc. 00/11 were **adopted** (Annex 5). Additionally, it was **agreed** that a working group should meet, possibly on 27 March 2001, in Copenhagen, Denmark, to review shrimp catch statistics according to the guidance provided in FC Doc. 00/19. It was also **agreed** that a special meeting of the Fisheries Commission should be called during 28-29 March 2001 in Copenhagen, Denmark, to examine alternatives for future management and allocation of NAFO shrimp stocks. Following a request for clarification by the Representative of the United States, it was generally agreed that the special fisheries commission meeting is to be part of the on-going broader allocation discussions. In addition, there was general agreement that further discussions on the broader allocation issue should take place during the 23rd Annual Meeting. The Fisheries Commission agreed on provisional agendas for these two meetings (attached as Annexes 6 and 7).

3.19 With respect to Agenda item 11, Review of the provisions on chartering operations in the NAFO Regulatory Area, there was general agreement that there was a need to clarify the rules relating to chartering operations in the NAFO Regulatory Area. Concern was

expressed regarding the confusion during 2000 over the use of allocated 3M shrimp fishing days through chartering operations (transfers of fishing days) and, in response to a question by the Representative of the EU, the NAFO Executive Secretary attempted to clarify the steps taken in approving requests for transfers of fishing days for use in chartering operations during 2000. There was general agreement that, in future cases where there is doubt regarding appropriate steps to be taken, the Executive Secretary should consult with the appropriate Chairman.

- 3.20 Concern was also expressed regarding the possible use of non-Contracting Party vessels reflagged through bareboat chartering operations. A number of Contracting Parties noted that effort in the 3M shrimp fishery continues to increase, creating levels of mortality beyond that recommended by the Scientific Council. After some discussion, the Fisheries Commission requested that STACTIC attempt to clarify the rules regarding chartering operations and report back at this meeting. The resulting document (FC Doc.00/12) was **adopted** for 2001 (Annex 8).
- 3.21 Regarding Agenda item 12, Increase in inspection presence in the NAFO Regulatory Area, the Representatives of Canada and the EU expressed concern regarding the lack of inspection presence of other Contracting Parties in the NAFO Regulatory Area. Although the EU (supported by Canada) tabled a proposal to introduce rules concerning obligatory inspection presence, no action was taken in this regard. It was requested that this issue be addressed more fully at the 2001 NAFO Annual Meeting. It was **agreed** that the current measures in place for inspection presence should be continued for 2001.
- 3.22 With respect to Agenda item 13, the acting Chairman of STACTIC, J.W. Baird (Canada), provided the report of STACTIC at the Annual Meeting. Regarding the STACTIC review of the annual returns of infringements, it was noted that there had been an overall improvement on the level of Contracting Party reporting on the disposition of apparent infringements. With regard to the STACTIC review of surveillance and inspection reports, Canada and the EU presented information on surveillance activities during 1999.
- 3.23 Discussions in STACTIC relating to the review of the operation of the hail system examined papers relating to: the NEAFC scheme for automated communications (STACTIC Working Paper 00/14); the current NAFO hail system (STACTIC Working Paper 00/18); confidentiality of information collected through automated hail reports and satellite tracking (STACTIC Working Paper 00/19). Topics addressed during this discussion related to modes of transmission of data, costs, and security. Additionally, an ad hoc STACTIC working group presented the results of a comparison between the NAFO and NEAFC systems. STACTIC agreed to pass on proposed format changes to the current NAFO hail system (found in STACTIC Working Paper 00/32) to the Fisheries Commission for consideration. This proposal was subsequently **adopted** by the Fisheries Commission as FC Doc. 00/14 (Annex 9).
- 3.24 Regarding the NAFO Observer and Satellite Tracking Programs, STACTIC examined the scientific requirements of the programs (as reflected in SCS Doc. 00/23 Harmonized NAFO Observer Program Data System Proposal). The Committee also considered an EU proposal for an observer manual (STACTIC Working Paper 00/10), and discussed possible amendments to the existing observer program (STACTIC Working Papers 98/03, 00/20 and 00/27). After considerable discussion, STACTIC recommended, and the Fisheries Commission **adopted**, the proposal put forth in SCS Doc. 00/23.

- 3.25 In STACTIC discussions relating to possible improvements in the procedures for gathering discard information, the Representatives of Canada and the EU reported some improvement in recording of discards in logbooks during 2000. STACTIC also considered fishing strategies to be employed to avoid excessive incidental catches and, after some discussion and revisions, agreed to forward a proposal by Canada on this subject (STACTIC Working Paper 00/23) to the Fisheries Commission for consideration. This paper was subsequently **adopted** by the Fisheries Commission as FC Doc. 00/15 (Annex 10). Additionally, STACTIC considered possible amendments to the Conservation and Enforcement Measures regarding juvenile fish proposed by Canada (STACTIC Working Papers 00/22 and 00/24). Although there was some support for these working papers, no action on these proposals was recommended due to concerns expressed by some Contracting Parties (such as Japan and the EU).
- 3.26 Regarding possible harmonization of port inspection reports, the EU presented a proposal to amend the Conservation and Enforcement Measures Regarding Part VII Port Inspections (STACTIC Working Paper 00/31). There was general support for the proposal although some concern was expressed regarding the requirements relating to transmission of reports. The paper was revised a number of times based on the comments of Contracting Parties and it was agreed that STACTIC would forward it to the Fisheries Commission for consideration. The revised working paper was subsequently **adopted** by the Fisheries Commission as FC Doc. 00/16 (Annex 11).
- 3.27 With respect to possible amendments to the Conservation and Enforcement Measures, there was agreement in STACTIC that an overhaul of the NAFO measures was necessary in order to ensure a cohesive document, clarify roles and responsibilities of those that would use the document, and reflect advancements in international fisheries agreements. No course of action was recommended by STACTIC regarding this issue. However, regarding the review of NAFO's Conservation and Enforcement Measures, Denmark (in respect of Faroe Islands and Greenland) suggested that a working paper be developed in which the present rules were split in four columns (rules for vessels, inspectors/observers, Contracting Parties and NAFO Secretariat). All present text and sequence should be retained. Canada and the European Community offered to make a preliminary review of these measures to identify redundancies and inconsistencies in the measures. The review shall be circulated to Contracting Parties by June 30, 2001. This activity will be organized by Canada. This course of action was adopted by the Fisheries Commission.
- 3.28 STACTIC also considered issues relating to chartering arrangements, reviewing Fisheries Commission Working Papers from the United States and Poland and a STACTIC Working Paper from Ukraine on the subject. The language recommended by STACTIC was subsequently adopted by the Fisheries Commission as noted in the section of this report concerning Chartering Operations. Additionally, STACTIC considered STACTIC Working Paper 00/29 regarding an increase of inspection presence in the NAFO Regulatory Area and STACTIC Working Paper 00/30 regarding satellite based vessel monitoring and related measures. These issues were both passed back to the Fisheries Commission for further consideration. The Fisheries Commission **adopted** the Report of STACTIC at the Annual Meeting.
- 3.29 With respect to Agenda item 14, Canadian Management Measures for 2J3KL Cod in 2000, the Representative of the EU strongly objected to the 7000 mt inshore fishery that took place in Canada in 2000. He noted his concern that, given the Canadian fishery, the management measures in place are not consistent throughout the range of this stock. The concerns of the EU were echoed by a number of the Contracting Parties present.

286

3.30 The Representative of Canada stated that Canada has the right to set TACs for the 2J3KL cod stock within Canadian waters and clarified that this was a small scale, highly regulated fishery. He noted the domestic process in place to recommend TACs and regulate this fishery, and pointed out that the data provided through this fishery is an important contribution to the conservation of this stock. Additionally, the Canadian delegate emphasized the current and historical importance of this fishery to the Canadian people and insisted that Canada would never do anything to endanger the stock. The EU Representative responded by making a statement on the management of this stock (Annex 12).

4. Conservation of Fish Stocks in the Regulatory Area (items 15-19)

4.1 With respect to item 15 of the Agenda, Summary of Scientific Advice, the Chairman of the Scientific Council, Dr. W.B. Brodie (Canada) presented a summary of NAFO SCS Doc 00/24 "Report of the Scientific Council, 1-15 June 2000" which provides the scientific advice for the management of stocks in the NAFO Regulatory Area for 2001 and 2002 and addresses special requests to the Scientific Council. He summarized this advice in the table below.

ADVICE FOR 2001

Shrimp 3M Redfish 3M Cod 3M American plaice 3M Witch flounder 3NO Cod 3NO American plaice 3LNO Redfish 3LN Witch Flounder 2J3KL Yellowtail flounder 3LNO Squid (*Illex*) 3+4 Greenland halibut 2+3KLMNO Not to exceed 30,000mt 3,000-5,000mt No directed fishery, lowest possible by-catch No directed fishery, lowest possible by-catch No directed fishery, lowest possible by-catch No directed fishery, lowest possible by-catch No directed fishery, lowest possible by-catch No directed fishery, lowest possible by-catch No directed fishery, lowest possible by-catch No directed fishery, lowest possible by-catch 13,000mt 19,000-34,000mt Not to exceed 40,000mt

ADVICE FOR 2002

Cod 3M	No directed fishery, lowest possible by-catch
American plaice 3M	No directed fishery, lowest possible by-catch
Witch flounder 3NO	No directed fishery, lowest possible by-catch

4.2 Special requests for advice were submitted for: 3NO Capelin; precautionary measures for NAFO stocks; Squid in Subareas 3&4; information on catches and/or discards of juvenile fish in various NAFO fisheries; elasmobranchs in SubAreas 0-6; and 3LN shrimp. With respect to 3M shrimp, Dr Brodie noted that some uncertainty exists with regard to the status of this stock and the Scientific Council would review its advice in November 2000.

Inquiries were made to the Chairman of the Scientific Council to clarify several questions regarding the scientific advice.

4.3 With respect to **3M redfish**, the Representative of Canada noted that the June 2000 Scientific Council Report stated that bycatch of age 1 fish is at about 20% of the total

number. He asked for clarification on the consequences of this level of mortality with regard to rebuilding of the stock, given the significance of this increase. Dr. Brodie noted that, although the Scientific Council previously did some analyses on potential losses of yield due to bycatch, since the introduction of new grates that reduce bycatch, no new examination has taken place. He noted that a new examination would have to be made in order to provide a comparison.

- 4.4 In reference to the question from Canada regarding bycatches of **3M redfish**, the Representative of the United States asked if recommended TACs for NAFO fisheries cover all sources of mortality and, if not, what were the consequences of not including these data. In response, the Scientific Council Chair noted that all removals are considered in recommending NAFO TACs.
- 4.5 Regarding the NEAFC-managed redfish stock that is now being found in NAFO Div. 1F, the Representative of Norway asked if the Scientific Council had any information on the distribution of this stock or advice on technical management measures given the deep, pelagic nature of this stock. The Scientific Council Chair noted that, although there is some knowledge among members of the Scientific Council regarding this stock, no formal discussion had yet taken place. Thus, no advice was possible at this time.
- 4.6 Regarding **3M shrimp**, the Representative of Norway noted that actual catches of this stock are estimated to be in the 40,000t range, while the Scientific advice for 2000 is based on estimates of 30, 000mt. He asked for comments from the Scientific Council Chair regarding how the actual catches in 2000 might affect the scientific advice for this stock in 2001. In response, the Scientific Council Chair noted the large degree of uncertainty associated with the status of this stock due to a lack of information. Although all available information was taken into account by the Scientific Council in recommending the TAC for 2001, certain assumptions had to be made. He noted that the Scientific Council would be considering this stock again in November 2000.
- 4.7 Regarding **3M shrimp**, the Representative of Iceland noted his county's intention to contribute to the Scientific Council evaluation regarding the effects of closed areas on this stock. He noted Iceland's particular concern regarding the effects of bycatch of very small shrimp in this fishery and expressed the desire to be part of the discussions on this issue at the November 2000 meeting of the Scientific Council.
- 4.8 With regard to **3M shrimp**, the Representative of the United States noted that, given the estimated catches in this fishery, it is clear that effort management has not been successful. She asked if the Scientific Council will be able to provide any additional advice on this stock without a direct survey. The Chair stated that, given the higher than estimated catches and the lack of appropriate recruitment indices, it is unlikely that things will improve for this stock. However, he noted that there may be some additional, initial data available this year from surveys conducted by Denmark.
- 4.9 With respect to **Greenland halibut**, the Representative of Canada expressed concern that current catches of juveniles might lead to a forgoing of future potential yield. He asked if the presence of these fish is a consequence of the mesh size used in the Greenland halibut fishery. The Scientific Council Chair noted that current estimates for maturity differs between males and females in the Greenland halibut fishery and that these differences present a problem for recommending appropriate mesh size. The Chair stated that the Council had looked at a number of different models and average retention rates resulting from the simulations, but that results were quite variable depending on assumptions used

in a given model. He noted that if the goal were to only catch fully mature Greenland halibut, the mesh size would have to be increased considerably.

- 4.10 Regarding **Greenland halibut**, the Representative of Canada noted the recommended increased TAC based on biomass increases and asked the Scientific Council Chair if there were any preliminary data from the survey series this summer that would confirm this biomass increase. The Chair responded that there was no information yet available from the EU survey series, although some preliminary indications are that there may be a slight reduction between 1999 and 2000. He noted that the information from the Canadian survey will be considered soon.
- 4.11 In response to a question from the Representative of the EU, the Scientific Council Chair noted that exploitable biomass and spawning stock biomass should increase if **2+3KLMNO Greenland halibut** mortality remains at the current level of 40,000t. Regarding a second question from the EU on the implications to the yield of this stock of an increase to a 145mm mesh size, the Chair noted that a study on this could be done. He noted that a new analysis would be required, and he could not indicate how long such a study might take.
- 4.12 In response to a question from the United States regarding the availability of data gathered in sentinel and index fisheries for **Greenland halibut** from fishery independent sources, the Scientific Council Chair noted that he was not aware of any fishery independent sources for such data.
- 4.13 Regarding **yellowtail flounder**, the Representative of the EU asked if the recommended 13,000mt TAC is consistent with NAFO's goal of keeping bycatches of stocks under moratoria at the lowest possible level. The Scientific Council Chair stated that there would be some implications associated with this TAC, but clarified that bycatch in this fishery is not currently detrimental.
- 4.14 With respect to 2J+3KL cod, the Representative of the EU asked for clarification regarding the use made of information from the inshore index, sentinel, and food/recreational fisheries for this stock. The Scientific Chair noted the value of inshore data, stating that such data (including catch rate, distribution, age composition, size, etc.) have been gathered from the index fishery in 1998, the commercial fishery in 1999, and sentinel surveys taken from varying sites around Newfoundland during 1995-2000. The Representative of the EU then requested information regarding the status of this stock and the impact of a fishery at the 7000mt level (for 2000) with respect to precautionary criteria as proposed by the Scientific Council and reference points previously used for management of this stock. The Chair of the Scientific Council responded that this issue had not yet been considered by the Council and that it would not be possible to do so at this meeting.
- 4.15 Regarding possible evaluation of this stock in the future relative to the precautionary approach, he noted that it is clear that the stock is well below the levels of the 1980s. However, he pointed out that early estimates of this stock were based on assessments of both the inshore and offshore components, while the remaining stock is primarily inshore. In response to requests from the Representative of the EU regarding evaluation of the effects of a 7000mt to 9000mt fishery on rebuilding of the inshore/offshore fisheries in the future, and estimating the proportion of juvenile fish taken in the inshore fishery, the Scientific Council Chair noted that these issues could be discussed at this meeting, but results would not be available since the necessary database and personnel were not
present. The Scientific Council Chair then asked that the EU provide clarification regarding its final question on this stock, which dealt with the implications to the genetic diversity of this stock of concentrated fishing on local aggregations.

- 4.16 The Representative of Canada noted that the bycatch for American plaice was high given its status as a stock under moratorium, noting that some bycatch was in the yellowtail flounder fishery but much more was in the Greenland halibut fishery. Noting that if the TAC for Greenland halibut is increased, bycatches of American plaice will also increase, he asked if this advice is consistent with recommendations in place for American plaice. The Scientific Council Chair agreed that increased bycatch would likely result, and cited the Scientific Council estimates of these bycatches in the Scientific Council Report.
- 4.17 In response to a question form the Representative of Iceland relating to the effects of harp seal predation on cod stocks, the Scientific Council Chair provided a brief summary of the Scientific Council discussions on this issue. This information can be found on page 158 of the Scientific Council Report.
- 4.18 The Chairman of the Fisheries Commission then summarized the outstanding questions to the Scientific Council and requested that these questions be put into writing for further consideration by the Council at this meeting as appropriate.
- 4.19 With respect to Agenda item 16, management and technical measures for fish stocks in the Regulatory Area in 2001, it was agreed that moratoria should remain in place for 3M cod and 3M American plaice. It was also agreed that a TAC of 5000mt should be set for 3M redfish. The representative of Latvia noted his country's intention to formally object to the block quota allocation for 3M redfish and also to further agenda items for Squid in Subareas 3+4 (block quota) and Greenland halibut in 3LMNO (others quota). With regard to 3M shrimp, it was decided that the effort scheme currently in place for 2000 should be rolled over for 2001 as indicated in FC Doc. 00/11. The Representative of Iceland noted his country's well established concern regarding effort-based management of this stock and stated that Iceland would once again formally object to this scheme. He also noted Iceland's support for a closed area for the protection of juvenile 3M shrimp.
- 4.20 Regarding Agenda item 17, management and technical measures for fish stocks straddling national fishing limits in 2001, its was generally **agreed that moratoria** should remain in place for **3NO cod**, **3LN redfish**, **3LNO American plaice**, **3NO witch flounder**, and **3NO capelin**. Regarding **3LNO yellowtail flounder**, the Representatives of Canada supported the proposed increase to 13,000mt, calling for a continuation of precautionary approach considerations and bycatch controls. While initially expressing concern regarding a possible TAC increase for this stock, following further discussion the United States removed its objection to this proposal. Additionally, the Representative of the United States expressed the desire to address appropriate allocation of this stock. The Representative of the EU expressed concern about the possible TAC increase.
- 4.21 Regarding **Sub-Areas 3+4 squid**, there was general support for a TAC of 34,000t with the protocol as expressed in FC Working Paper 00/10 for mid-season adjustment based on productivity indicators. However, the Representative of the United States noted that, given the imprecise advice on this stock, it should be discussed whether setting the TAC

at the top of the suggested range is appropriately precautionary. FC Doc.17, prohibiting any directed shrimp fishery in Div. 3NO was adopted (Annex13).

4.22 With respect to **3LMNO Greenland halibut**, the Representative of the EU expressed support, in principle, for an increase to 40,000mt. However, he noted that caution was advisable given the uncertainty surrounding year classes in the mid-1990s and the fact that much of the current catch is juvenile. The Representative of Canada noted that the good news concerning this stock should be tempered by the high level of juvenile catch. He also urged caution and noted that any TAC above 30,000t should include measures to address juvenile and bycatch concerns. The Representatives of Japan, Latvia, and Lithuania supported a TAC increase to 40,000t. Following further discussion, it was decided that the 40,000mt TAC for Greenland halibut stock in Subarea 2+3 be **adopted** for 2001. According to the decision (Resolution, FC Doc. 95/7) of the Special Fisheries Commission Meeting (Toronto, 1995), 25.9% (or 10,360 mt) of this amount is allocated to Canada, inside 200-mile zone, and other portion - 29,640 mt will be distributed to Contracting Parties (Quota Table). With regard to the issues of juveniles and bycatch, FC Doc. 00/15 was **adopted** (see Annex 10) and a statement was adopted as follows:

All NAFO Contracting Parties strongly support the establishment and full implementation of measures to protect juveniles and reduce bycatch.

Having agreed at its 22nd Annual Meeting to adopt a proposal to amend NAFO Conservation and Enforcement Measures regarding incidental catch limits, NAFO Contracting Parties hereby agree to consider the following measures to further protect juveniles and reduce bycatch at the June 2001 meeting of the Scientific Council and a 2001 intersessional meeting of STACTIC:

such as:

a) mesh size increases; b) depth restrictions; c) area closures; or d) other effective measures.

In the light of the outcome of the 2001 meeting of the Scientific Council and any other relevant scientific evidence, suitable measures will be examined and, as appropriate, designed and implemented in 2002 with due consideration of conservation requirements and the particularities of various NAFO fisheries.

- 4.23 The Representative of Canada noted that he agreed to the insertion of the above statement. However, he expressed strong disappointment with regard to the measures taken thus far for the protection of juveniles and reduction of bycatch for this fishery. This statement was fully supported by the Representative of the United States.
- 4.24 Regarding **2J3KL cod**, it was **agreed** that the measures in place for this fishery for 2000 should be extended for 2001 (as indicated in FC Doc. 00/10 see Annex 14). It was agreed that the moratorium in place for **3L cod** should continue. The European Union reiterated his Party's dissatisfaction regarding Canadian activities with regard to the inshore component of this fishery.
- 4.25 With respect to **2J3KL witch flounder**, the Representative of Canada noted that there is a moratorium on this stock in the Canadian 200-mile zone and asked that NAFO continue this moratorium in the Regulatory Area. It was **agreed** that this moratorium should be

continued and that the measures in place in the Regulatory Area for this stock should be updated to reflect this decision (FC Doc. 00/9 - Annex 15).

- 4.26 Regarding **Div. 1F redfish**, the Representative of Denmark (in respect of the Faroe Islands and Greenland) noted that this pelagic spillover stock is managed by the NEAFC and also falls within the national jurisdiction of some NAFO Contracting Parties. He stated that there was a need to discuss possible scenarios for management and suggested that NAFO might agree to implement management measures adopted by NEAFC for this stock in the NAFO Regulatory Area. At the very least, NAFO should adopt a 100mm mesh size for this fishery. The Representatives of Norway, Russia, Iceland, and the EU supported this proposal, but the Representative of Canada pointed out that there is a need to clarify all measures that might apply to this stock before this proposal could be considered properly. The United States clarified that, since this could be considered a new fishery, it must be understood that any agreements pertaining to this stock should not prejudice on-going discussions relating to allocation of new fisheries.
- 4.27 Since it was brought to the attention of Contracting Parties that recently oceanic redfish (*Sebastes mentella*) from the NEAFC Regulatory Area had crossed into Division 1F of the NAFO Regulatory Area, it was **agreed** to invite NEAFC to participate in a Joint NEAFC/NAFO Working Group to discuss various issues pertaining to this situation with a view to developing a compatible management approach to the pelagic *Sebastes mentella* stock. This Joint Working Group should meet during 13-14 February 2001 in Reykjavik, Iceland.
- 4.28 After a brief discussion regarding footnoting, the quota table for 2001 was adopted (Annex 16).
- 4.29 With respect to Agenda item 18, formulation of request to the Scientific Council, the Representative of the United States (supported by Canada and the EU) suggested that text be added to the request to address the issue of the implications of mesh size changes with regard to the 2+3KLMNO Greenland halibut fishery. A number of Contracting Parties suggested slight revisions to the initial draft. Additionally, the Representative of Russia called for the inclusion of text requesting advice regarding the methodology for scientific research on fish stocks under moratoria. The Representative of Iceland also asked that language from last year's request be included dealing with an evaluation of the possible results of closed areas on the 3M shrimp fishery. All three of these proposed additions were **adopted**.
- 4.30 In response to a call from the EU for the Scientific Council to provide advice on 2J3KL cod throughout its entire range, the Representative of Canada noted that they are responsible for this stock and this issue will be addressed in the Canadian request for advice from the Scientific Council. The (amended) request for advice from the Scientific Council for 2001 was **adopted** (Annex 17).
- 4.31 There was no discussion relating to Agenda item 19, transfers of quotas between Contracting Parties.

5. Closing Procedures (items 20-22)

5.1 Regarding Agenda item 20, Time and Place of Next Meeting, the Fisheries Commission's Annual Meeting in the year 2001 will be held in Cuba (location and date to be

determined). It was noted that the list of intersessional meetings would be forwarded to the General Council for consideration.

- 5.2 With respect to Agenda item 21, Other Business, the Chairman of the Fisheries Commission noted that a proposal had been circulated by the Ukraine for a chartering operation for 3M shrimp. The Representative of the Ukraine spoke to this proposal, noting that its text was based on the newly negotiated language guiding this process. He asked that, given the time constraints involved, Contracting Parties consider this proposal at this meeting and vote accordingly. However, following comments by Contracting Parties, it was decided that the proposal of the Ukraine should be considered according to established procedures.
- 5.3 Agenda item 22, Adjournment, the Annual Meeting of the Fisheries Commission was adjourned at 1250 hours on Friday, 22 September 2000.

292

Annex 1. List of Participants

CANADA

Head of Delegation

P. S. Chamut, Assistant Deputy Minister, Dept. of Fisheries and Oceans, Fisheries Management, 200 Kent Street, Ottawa, Ontario K1A 0E6 Phone: +613 990 9864 – Fax: +613 990 9557

Representative

P. Chamut (see address above)

Advisers

C. J. Allen, Chief, Groundfish, Pelagics and Foreign Fisheries, Resource Management, Dept. of Fisheries and Oceans, Ottawa, Ontario K1A 0E6

Phone: +613 990 0105 - Fax: +613 990 7051 - E-mail: allenc@dfo-mpo.gc.ca

J. Angel, Canadian Association of Prawn Producers, P. O. Box 1C1, Head of St. Margarets Bay, N.S. B0J 1R0 Phone: +902 826 7765 - Fax: +902 826 7065 - E-mail: jangel@navnet.net

D. B. Atkinson, Northwest Atlantic Fisheries Centre, P. O. Box 5667, St. John's, Newfoundland A1C 5X1 Phone: +709 772 2052 - Fax: +709 772 4188 - E-mail: atkinsonb@dfo-mpo-gc.ca

J. W. Baird, Director, Resource Management Div., Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 4497 - Fax: +709 772 3628 - E:mail: bairdj@dfo-mpo.gc.ca

T. Blanchard, Chief, NAFO Unit, Fisheries Management, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 0928 - Fax: +709 772 0008 - E-mail: blanchardt@dfo-mpo.gc.ca

D. R. Bollivar, Seafreez Foods Inc., 32 Beckfoot Drive, Dartmouth, N.S. B2Y 4C8

Phone: +902 469 5004 - Fax: +902 461 9689 - E-mail: bollivar@seafreez.com

N. Bouffard, Senior Advisor, Legal Issues, International Affairs Directorate, Dept. of Fisheries and Oceans,

200 Kent St., 13th Floor, Stn. 13159, Ottawa, Ontario K1A 0E6

Phone: +613 993 1860 Fax: +613 993 5995 E-mail: bouffardn@dfo-mpo.gc.ca

W. R. Bowering, Northwest Atlantic Fisheries Centre, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 2054 - Fax: +709 772 4188 - E-mail: boweringr@dfo-mpo-gc.ca

W. B. Brodie, Northwest Atlantic Fisheries Centre, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 3288 - Fax: +709 772 4105 - E-mail: brodieb@dfo-mpo.gc.ca

B. Chapman, 1388 River Road, Manotick, Ontario K4M 1B4

Phone: +613 692 8249 - Fax: +613 692 8250 - E-mail: bchapman@sympatico.ca

L. Dean, Dept. of Fish and Aquaculture, Government of Newfoundland and Labrador, P. O. Box 4750, St. John's, Newfoundland

Phone: +709 729 3722 - Fax: +709 729 0360

T. Dooley, Dept. of Fisheries, P. O. Box 8700, St. John's, Newfoundland A1L 5H4

Phone: +709 729 0335 - Fax: +709 729 6082 - E-mail: tdooley@matl.gov.nf.ca

E. Dussault, Director, Atlantic Affairs Div., International Affairs Directorate, Dept. of Fisheries and Oceans, 200 Kent Street, Ottawa, Ontario K1A 0E6

Phone: +613 993 5316 - Fax: +613 993 5995 - E-mail: dussaulte@dfo-mpo.gc.ca

W. Evans, Supervisor, Offshore Surveillance, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 4412 - Fax: +709 772 0008 - E-mail: evansw@dfo-mpo.gc.ca

S. Firko, Dept. of Fisheries and Oceans, 200 Kent Street, Ottawa, Ontario K1A 0E6

Phone: +613 993-1852 - Fax: +613 993 5995

D. Forsythe, Mission of Canada to the European Union, Avenue de Tervuren, 2, Brussels 1040, Belgium Phone: +32 2 741 0688 - Fax: +32 2 741 0629 - E-mail: douglas.forsythe@dfait-maeci.gc.ca

G. Gregory, Fishery Products International Ltd., P. O. Box 550, Station A, St. John's, Newfoundland A1C 5L1 Phone: +709 570 0427 - Fax: +709 570 0436 - E-mail: <u>ggregory@fpil.com</u> N. Greig, P. O. Box 1058, Kuujjuaq, Quebec J0M 1C0

Phone: +819 964 2925 - Fax: +819 964 2613 - E-mail: n.greig@makivik.org

J. M. Kelsey, Department of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1 Phone: +709 772 7935 – Fax: + 709 772 6306 – E-mail: <u>kelseyj@dfo-mpo.gc.ca</u>

D. W. Kulka, Science, Oceans and Environment Br., Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's,

Newfoundland AIC 5X1

Phone: +709 772 2064 - Fax: +709 772 4188 - E-mail: kulkad@dfo-mpo.gc.ca

E. McCurdy, c/o FFAW/CAW, P. O. Box 10, St. John's, Newfoundland A1C 5H5

Phone: +709 576 7276 - Fax: +709 576 1962 - E-mail:

P. McGuinness, Vice-President, Fisheries Council of Canada, 38 Antares Drive, Suite 110, Nepean, Ontario K2E 7V2

Phone: +613 727 7450 - Fax: +613 727 7453 - E-mail: pmcguinness@fisheriescouncil.org

B. J. McNamara, Newfoundland Resources Ltd., 90 O'Leary Avenue, St. John's, Nfld. A1B 3P7 Phone: +709 579 7676 - Fax: +709 579 7668 - E-mail: nrl@nfld.com

J. Mercer, Dept. of Fisheries and Oceans, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 4494 - Fax: +709 772 5983 - E-mail: mercerja@dfo-mpo.gc.ca

A. Noseworthy, Deputy Minister, Intergovernmental Affairs, Government of Newfoundland and Labrador, P. O. Box 8700, St. John's, Newfoundland A1B 4J6

Phone: +709 729 2131 - Fax: +709 729 5038 - E-mail: anoseworthy@mail.gov.nf.ca

M. O'Connor, National Sea Products, 100 Battery Point, P. O. Box 910, Lunenburg, N.S. B0J 2C0

Phone: +902 634 5200 - Fax: +902 634 4926 - E-mail: oconnorm@natsea.ca

A. O'Rielly, President, Fisheries Association of Nfld. and Labrador Ltd., P. O. Box 8900, St. John's, Newfoundland A1B 3R9

Phone: +709 726 7223 - Fax: +709 754 3339 - E-mail: aorielly@nfld.com

F. G. Peacock, Director, Resource Mgmt. Br., Dept. of Fisheries and Oceans, P. O. Box 550, Halifax, Nova Scotia B3J 2S7

Phone: +902 426 3625 - Fax: +902 426 9683 - E-mail: peacockg@dfo.mpo.gc.ca

C. Reardon, Nova Scotia Dept. of Fisheries and Aquaculture, P. O. Box 2223, Halifax, N.S. B3J 3C4

Phone: +902 424 0349 - Fax: +902 424 4671 - E-mail: reardonc@ns.gov.ca

D. Rivard, Fisheries Research Br., Fisheries and Oceans, 200 Kent St., Ottawa, Ontario K1A 0E6 Phone: +613 990 0281 - Fax: +613 954 0807 - E-mail: <u>rivardd@dfo-mpo.gc.ca</u>

A. Saunders, Dept. of Foreign Affairs and International Trade (JLO), 125 Sussex Drive, Ottawa, Ontario K1A 0G2

Phone: +613 996 2643 - Fax: +613 992 6483 - E-mail: allison.saunders@dfait-maeci.gc.ca

M. Short, Dept. of Fisheries and Oceans, Baine Johnston Center, Suite 801, 10 Fort Williams Place, St. John's, Newfoundland A1C 1K4

Phone: +709 772 5238 - Fax: +709 772 5244

R. Steinbock, Advisor, Atlantic Affairs Div., International Directorate, Dept. of Fisheries and Oceans, 200 Kent St., Ottawa, Ontario K1A 0E6

Phone: +613 993 1836 - Fax: +613 993 5995 - E-mail: steinbob@dfo-mpo.gc.ca

R. Stirling, SPANS, P. O. Box 991, Dartmouth, N. S. B2Y 3Z6

Phone: +902 463 7790 - Fax: +902 469 8294 - E-mail: span@fox.nstn.ca

H. Strauss, Director, Oceans, Environmental and Economic Law Div., Dept. of Foreign Affairs and International Trade Canada, 125 Sussex Drive, Ottawa, Ontario K1A 0G2

Phone: +613 992 2104 Fax: +613 992 6483 E-mail: howard.strauss@dfait-maeci.gc.ca

L. Strowbridge, Area Director, Eastern/Southern Newfoundland, P. O. Box 5667, St. John's, Newfoundland A1C 5X1

Phone: +709 772 4010 - Fax: +902 772 2659 - E-mail: strowbridgel@dfo-mpo.gc.ca

E. Wiseman, Director-General, International Directorate, Dept. of Fisheries and Oceans, 200 Kent Street, 13th Floor N, Stn 13-159, Ottawa, Ontario K1A 0E6

Phone: +613 993 1873 - Fax: +613 993 5995 - E-mail: wisemane@dfo-mpo.gc.ca

F. Woodman, Chairman, Fisheries Resource Conservation Council, P. O. Box 2001, Station D, Ottawa, Ontario K1P 5W3

Phone: +613 998 0433 - Fax: +613 998 1146 - E-mail: costah@dfo-mpo.gc.ca

CUBA

Head of Delegation

E. Otuski, Ministerio de la Industria Pesquera, 5th Avenue y 246, Sta Fe, Barlovento Phone: +537 297117

Alternate

J. Baisre Alvarez, Ministerio de la Industria Pesquera, Barlovento, Santa Fe 19 100, Playa la Habana Phone: +537 297253 - Fax: +537 249168 - E-mail: <u>baisre@fishnavy.inf.cu</u>

Representative

J. Baisre Alvarez (address above)

Advisers

V. E. Sarda Espinosa, Ministerio de la Industria Pesquera, 5th Avenue y 246, Sta Fe, Barlovento Phone: +537 297034 - Fax: +537 249168 - E-mail: <u>abogados@fishnavy.inf.cu</u>
R. Espinosa, Dragnets, Asociacion Pesport, Puerto Pesquero de la Habana, Ave la Pesquera y Atares, Habana Vieja, Ciudad de La Habana Phone: +537 619090
L. Albelo Leon, Ministerio Industria Pesquera, Cuban Fishing Fleet Representative, 1881 Brunswick St., Apt. 906, Halifax, Nova Scotia, Canada B3J 3J7

Phone: +902 425 5773 - Fax: +902 423 8871

DENMARK (IN RESPECT OF FAROES AND GREENLAND)

Head of Delegation

E. Lemche, Head of Representation, Gronlands Hjemmestyre, Pilestraede 52, Box 2151, DK-1016 Copenhagen K, Denmark Phone: +45 33 69 34 35 Fax: +45 33 69 34 01 E-mail: Einar.Lemche@ghsdk.dk

Alternate

A. Kristiansen, Ministry of Fisheries and Maritime Affairs, Yviri við Strond 17, P. O. Box 347, FO-100 Torshavn, Faroe Islands Phone: + 298 35 30 30 - Fax: +298 35 30 35 - E-mail: andrask@fisk.fl.fo

Representatives

E. Lemche (see address above) A. Kristiansen (see address above)

Advisers

J. E. Hansen, Bondaheygur 9, FO-100 Torshavn, Faroe Islands
Phone: +298 312990/210810 - Fax: +298 33 35 95 - E-mail: hogi@post.olivant.fo
J. Joensen, (address please)
Phone: +298 42 14 48 - Fax: +298 42 15 84
O. A. Jørgensen, Greenland Institute of Natural Resources, Pilestræde 52, Box 2151, Copenhagen, DK-1016
Phone: +45 33 69 34 61 - Fax: +45 33 69 34 06 - E-mail: grfioaj@inet.uni2.dk

M. Kruse, Vaktar-og Bjargingartaenastan, P. O. Box 347, FR-110 Torshavn, Faroe Islands Phone: +298 311065 - Fax: +298 383981 - E-mail: vb@vb.fo 296

L. Lind, Head of Section, Royal Danish Ministry of Foreign Affairs, 2 Asiatisk Plads, DK-1448, Copenhagen K. Denmark

Phone: +45 33 92 00 00 - Fax: +45 32 54 05 33 - E-mail: lilind@um.dk

M. T. Nedergaard, Fiskerilicensinspektor, Head of Unit, Gronlands Fiskerilicenskontrol, Postbox 501, DK-3900 Nuuk, Greenland

Phone: +299 345377 - Fax: +299 323235 - E-mail: mads@grl.gl

A. Nicolajsen, Fiskirannsoknarstovan, Noatun, P. O. Box 3051, FR-110 Torshavn, Faroe Islands Phone: +298 1 5092 - Fax: +298 1 8264 - E-mail: arninic@frs.fo

J. Persson, Head of Section, Greenland Home Rule, Department of Industry, Box 269, 3900 Nuuk, Greenland Phone: +299 34 53 24 Fax: +299 32 47 04 E-mail: jpe@gh.gl

E. Rosing, Greenland Home Rule, Dept. of Industry, Box 269, 3900 Nuuk, Greenland

Phone: +299 34 53 32 - Fax: +299 32 47 04 - E-mail: emanuel@gh.gl

J. H. Toftum, Ministry of Fisheries, P. O. Box 64, FO-100 Torshavn, Faroe Islands Phone: +298 353030 - Fax: +298 353035 - E-mail: jenst@fisk.fl.fo

ESTONIA

Head of Delegation

K. Mühlbaum, Ministry of the Environment, Fisheries Department, Maria 4d, 10617 Tallinn Phone: +372 6566720 - Fax: +372 6567599 - E-mail: kristiina@klab.envir.ee

Representative

K. Mühlbaum (see address above)

Advisers

M. Harjak, Dagomar Ltd., Sadama 15, Kardla Phone: +372 4632031 - Fax: +372 4632039 - E-mail: marek@huukalur.ee R. Kulla, E-Traal Ltd., 9 Narva st., Tallinn 10017 Phone: +372 5128888 - Fax: +372 6109244 - E-mail: traal@anet.ee J. Pollu, Reyktal Ltd., Paljassaare Road 28-426, 10313 Tallinn Phone: +372 6512066 - Fax: +372 6512055 - E-mail: reyktal@trenet.ee T. Roose, Deputy Director General, Estonian Environmental, Inspectorate, Kopli 76, 10416 Tallinn, Estonia Phone: +3726603333 - Fax: +3726603350, E-mail: tarvo.roose@kki.ee T. Saat, Director, Estonian Marine Institute, 18b Viljandi Road, 11216, Tallinn Phone: +372 6281 570 - Fax: +372 6281 563 - E-mail: tsaat@sea.ee A. Soome, Officer, Ministry of the Environment, Fisheries Department, Marja 4d, 10617 Tallinn Phone: +372 61 12 987 - Fax: +372 6567 599 - E-mail: ains@klab.envir.ee L. Vaarja, Fisheries Adviser, Ministry of Environment, Fisheries Department, Marja 4d, 10617 Tallinn

Phone: +372 6112 987 - Fax: +372 6567 599 - E-mail: laurivaarja@hot.ee

EUROPEAN UNION (EU)

Head of Delegation

E. Mastracchio, Director, European Commission, Directorate General for Fisheries, 200 Rue de la Loi, B-1049 Brussels, Belgium

Alternate

O. Tougaard, European Commission, Directorate General for Fisheries, 200 Rue de la Loi, B-1049 Brussels, Belgium

Phone: +32 2 295 2209 - Fax: +32 2 299 4802

Representatives

E. Mastracchio (see address above)

O. Tougaard (see address above)

Advisers

F. Wieland, Deputy Head of Unit, International Fisheries Organizations and Fisheries Agreements; Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi 200, B-1049 Brussels, Belgium

Phone: +32 2 296 3205 Fax: +32 2 299 4802 E-mail: Friedrich.Wieland@ccc.eu.int

A. Thomson, Principal Assistant, International Fisheries Organizations and Fisheries Agreements; Baltic, North Atlantic and North Pacific, European Commission, Directorate General for Fisheries, Rue de la Loi/Wetstraat 200, B-1049 Brussels, Belgium

Phone: +32 2 299 0180 - Fax: +32 2 299 4802 - E-mail: Andrew. Thomson@cec.eu.int

B. O'Shea, European Commission, Directorate General for Fisheries, Rue Joseph II 99, Rm 1/27, B-1049, Brussels, Belgium

Phone: +32 2 296 6748 - Fax: +32 2 296 2338 - Email: brendan.o'shea@cec.eu.int

K. Patterson, European Commission, Directorate General for Fisheries, Rue de la Loi 200, B-1049 Brussels, Belgium

Phone: + 32 2 299 2179 - Fax: +32 2 295 5621 - Email: kenneth.patterson@cec.eu.int

V. Angot, European Commission, Rue de la Loi 200, B-1049 Brussels, Belgium

Phone: +32 2 296 6406 - Fax: +32 2 296 2338 - Email: Veronique.Angot@ cec.eu.int

A. Gray, Director General I, External Relations: Commerical Policy and Relations with North America, the Far East America, Australia and New Zealand, Rue de la Loi, Wetstraat 170, B-1040 Brussels Phone: + 32 2 2990077 - Fax # 32 2 2991046 - E-mail: alan-gray@dg1.cec.be

B. Prince, Policy Officer in charge of International Affairs, Ministere de l'Agriculture et de la Peche, Direction des Peches Maritimes, 3 Place de Fontenoy, 75007 Paris, France

Phone: +45 33 1 49 55 82 38 - Fax: +45 33 1 49 55 82 00 - E-mail: berengere.prince@agriculture.gouv.fr

S. Segura, Conseiller des Affaires Etrangeres, Direction des Affaires Juridiques, Ministere des Affaires Etrangeres, 37 Quai d'Orsay, 75700 Paris, France

Phone: +33 1 43 17 53 26 - Fax: +33 1 43 17 43 59 - E-mail: serge.segura@diplomatie.gouv.fr

G. F. Kingston, Senior Adviser, Economic and Commercial Affairs, Delegation of the European Commission, 45 O'Connor Street, Suite 1900, Ottawa, Ontario, Canada K1P 1A4

Phone: +613 238 6464 - Fax: +613 238 5191 - E-mail: fred.kingston@delcan.cec.eu.int

D. Cross, Eurostat, European Commission, Jean Monnet Bldg., BP 1907, L-2920 Luxembourg

Phone: +352 4301 37249 - Fax: +352 4301 37318 - E-mail: david.cross@cec.eu.int

T. Heaton, Director, DG BIII-Fisheries, Council of the European Union, Rue de la Loi 175, B-1048 Brussels, Belgium

Phone: +32 2 285 6486 - Fax: +32 2 285 8261 - E-mail: Trevor.Heaton@consilium.eu.int

S. Feldthaus, Head of Section, Ministry of Food, Agriculture and Fisheries, Holbergsgade 2, 1057 Copenhagen K, Denmark

Phone: +45 33 92 35 60 - Fax: +45 33 11 82 71 - Internet: sfe@fvm.dk

R. Akesson, Ministry of Agriculture and Fisheries, 10333 Stockholm, Sweden

Phone: +46 08 405 1122 - Fax: +46 08 10 5061 - E-mail: rolf.akesson@agriculture.ministry.so

H. Pott, Bundesministerium fur Ernahrung, Landwirtschaft und Forsten, Rochusstr. 1, D-53125 Bonn, Germany Phone: +49 228 529 4124 - Fax: +49 228 529 4410 - Email: hermann.pott@bml.bund.de

J. Manuel de Castro Santiago, Counsellor, Embassy of Portugal, 645 Island Park Drive, Ottawa, Ontario K1Y 0B8

Phone: +613 729 0883 - Fax: +613 729 4236

E. Monteiro, Director-General, Direccao Geral Pescas Aquicultura, Edificio Vasco da Gama, Alcantara, 1350 Lisbon, Portugal

Phone: +351 21 3914387 Fax: +351 21 3957858 E-mail: euricom@dg-psces.pt

M. H. Figueiredo, Directora de Servicos, Dept. de Relacoes Comunitarias, Internacionais e de Cooperacao, Direccao Geral das Pescas e Aquicultura, Edificio Vasco da Gama, Alcantara Mar, 1350 Lisbon, Portugal Phone: +351 21 3914350 Fax: +351 21 3979790 E-mail: hfigueir@dg-pescas.pt

A. Leite, Inspeccao geral deas Pescas, Av. Brasilia, 1400-038 Lisboa, Portugal

Phone: +351 21 3025170 - Fax: +351 21 3025101 - E-mail: albertoleite@igp.pt

L. M. Esteruelas, Counselor for Agriculture & Fisheries, Embassy of Spain, 2375 Pennsylvania Ave., N.W., Washington, D.C. 20037

Phone: +202-728-2339 - Fax: +202-728-2320 - E-mail: <u>lmesteruelas@erols.com</u>

F. Curcio, Subdirector General de Organismos Multilaterales de Pesca, Secretaria General de Pesca Maritimes, Jose Ortega y Gasset, 57, 28006 Madrid, Spain

Phone: +34 914027404 - Fax: +34 913093967 - E-mail: fcurcio@mapya.es

M. I. Aragon, Jefa Seccion de la Subdireccion General de Organismos Multilaterales de Pesca, Secretaria General de Pesca Maritima, Jose Ortega y Gasset, 57, 28006 Madrid, Spain

J. Del Hierro, Subdireccion General de Inspeccion Pesquera, Secretaria General de Pesca Maritima, c/Castellama 112, 5^a Plto, Madrid, Spain

Phone: +34 91 3471645 - Fax: +34 91 3471512

A. Hermida Trastoy, Director Xeral de Estructures Pesqueiras e Mercados, Xunte de Galicia, C/Sar, 75, Santiago 15702, A Coruna, Spain

Phone: + 34981546347 - Fax: +34981546288 - E-mail: andres.hermida.trastoy@xunte.es

H. J. Ratz, Institut fur Seefischerei, Palmaille 9, D-22767, Hamburg, Germany

Phone: +49 40 389 05169 - Fax: +49 40 389 05263 - E-mail: raetz.ish@bfa-fisch.de

M. Stein, Institut fur Seefischerei, Palmaille 9, D-22767, Hamburg, Germany

Phone: +49 40 389 05174 - Fax: +49 40 38905 263 E-mail: stein.ish@bfa-fisch.de

D. Briand, IFREMER, B. P. 4240, 97500 St. Pierre et Miquelon, France

Phone: +508 413083 Fax:+508 41 49 36 - E-mail: brianspm@cancom.net

R. Alpoim, Inst. de Investigacao das Pescas e do Mar (IPIMAR), Av. de Brasilia, 1400 Lisbon, Portugal

Phone: +3511 302 7000 - Fax: +3511-301-5948 - E-mail: ralpoim@ipimar.pt

A. Avila de Melo, Inst. de Investigacao das Pescas e do Mar (IPIMAR), Av. de Brasilia, 1400 Lisbon, Portugal

Phone: +3511 302 7000 Fax: +3511-301-5948 E-mail: amelo@ipimar.pt

E. De Cardenas, Institute Espanol de Oceanografia, Avenida de Brasil 31, 28020 Madrid, Spain

Phone: +34 91 5974443 - Fax: +34 91 5974770 - E-mail: e.decardenas@md.ieo.es

S. Junquera, Instituto Espanol de Oceanografia, Cabo Estay-Canido, Aptdo. 1552, E-36280 Vigo (Pontevedra), Spain

Phone: +34 9 86 49 2111 - Fax: +34 9 86 49 2351 - E-mail: susana.junquera@vi.ieo.es

H. Murua, AZTI, Instituto para la Ciencia y Tecnologia Pesquera, Av. Satrustegi 8, 20008 Donostia – San Sebastian, Spain

Phone: +34 9 43 316731 - Fax: +34 9 43 212162 - E-mail: hmurua@azti.es

A. Vazquez, Instituto de Investigaciones Marinas, Eduardo Cabello 6, 36208 Vigo, Spain

Phone: +34 9 86 23 1930 - Fax: +34 9 86 29 2762 - E-mail: avazquez@iim.csic.es

P. Franca, ADAPI – Associacao Armadores das Pescas Industriais, Edificio Dos Armadores 13-A, Docapesca 1400 Lisbon, Portugal

Phone: +213015020 - Fax: +213019438 - E-mail: adapi.pescas@mail.telepac.pt.

M. Paiao, ADAPI – Associacao Armadores das Pescas Industriais, Edificio Dos Armadores 13-A, Doca Pesca 1400-038 Lisbon, Portugal

Phone: +213015020 - Fax: +213019438 - E-mail: adapi.peseas@mail.telepac.pt

R. Gordejuela Aguilar, Presidente de "ANAVAR".

J. R. Fuertes Gamundi, ANAMER-ANAVAR-AGARBA, Puerto Pesquero, Vigo, Spain Phone: +34 986 433844 - Fax: +34 986 439218

M. Liria Franch, Presidente de ANAMER, Pto Pesquero, Spain

C. Real Rodriguez, Presidente de "ASPE", Vigo, Spain

FRANCE (in respect of St. Pierre and Miquelon)

Head of Delegation

G. Grignon, 4C Rue Albert Briand, 97500 Saint Pierre et Miquelon Phone: +508 414 219 - Fax: +508 414 806 – E-mail: <u>archipel@cancom.nct</u>

Alternate

D. Silvestre, Secretariat General de la Mer, 16 Boulevard Raspail, 75007 Paris Phone: +0033142840876 - Fax: +0033142840790 - E-mail: <u>daniel.silvestre@sgmer.premier-ministre.gouv.fr</u>

Representatives

G. Grignon (address above) D. Silvestre (address above)

Advisers

V. Sinquin, International Affairs Division, Overseas Ministry, 27 rue Oudinot, 75007, Paris Phone: +0153692746 – Fax: +0153692197 – E-mail: valerie.sinquin@outre_mer.gouv.fr M. Tremblay (Interpreter), 3124 Needham St., Halifax, N.S. B3K 3N9 Phone: +902 420 9158 - Fax: +902 455 2992 - E-mail: <u>mti@hfx.eastlink.ca</u>

ICELAND

Head of Delegation

T. Asgeirsson, Director of Fisheries, Ingolfsstraeti 1, 150 Reykjavik Phone: +354 569 7900 - Fax: +354 569 7991 - E-mail: <u>thordur@hafro.is</u>

Representative

T. Asgeirsson (see address above)

Advisers

K. Arnason, Head of Division, Ministry of Fisheries, Skulagata 4, 150 Reykjavik Phone: +354 560 9670 - Fax: +354 562 1853 - E-mail: <u>kolbeinn.arnason@sjr.stjr.is</u>
H. Steinarsson, The Icelandic Directorate of Fisheries, , Ingilfsstraeti, 150 Reykjavik Phone: +354 5097938 - Fax: +354 5697991 - E-mail: <u>hostein@hfro.is</u>
K. Ragnarsson, Federation of Icelandic Fishing Vessel Owners, P. O. Box 893, 121 Reykjavik Phone: +354 550 9500 - Fax: +354 550 9501 - E-mail: kristjan@liu.is

JAPAN

Head of Delegation

K. Yonezawa, c/o Fishery Division, Economic Affairs Bureau, Ministry of Foreign Affairs, 2-2-1 Kasumigaseki, Chiyoda-ku, Tokyo

Representatives

K. Yonezawa (see address above)

Advisers

S. Kawahara, Director, Oceanic Resources Division, National Research Institute of Far Seas Fisheries, 5-7-1 Orido, Shimizu-shi 424, Sizuoka, 424

Phone: +81 543 36 6051 - Fax: +81 543 35 9642 - E-mail: kawahara@enyo.affrc.go.jp

Y. Kashio, Representative, Japan Fisheries Association, Suite 1209 Duke Tower, 5251 Duke St. Tower, Halifax, N.S., Canada B3J 1P3

Phone: +902 423 7975 - Fax: +902 425 0537 - E-mail: jfa-hfx@ns.sympatico.ca

M. Miyashita, Far Seas Fisheries Div., Resources Management Dept., Fishery Agency Government of Japan. 1-2-1 Kasumigaseki, Chivoda-ku, Tokyo 100

Phone: +03 3502 8111 ext. 7239/03 3591 6582 - Fax: +03 3591 5824

S. Nagase, Fisheries Div., Economic Affairs Bureau, Ministry of Foreign Affairs, 2-2-1 Kasumigaseki, Chiyoda-ku, Tokyo

Phone: +03-3580-3311 ext. 3351 - Fax: 03-3503-3136 - E-mail: saori.nagase@mofa.go.jp

Y. Ochi, Development Dept., Japan Marine Fishery Resources Research Center, Godo Kaikan Bldg. 3-27 Kioj -cho, Chiyoda-ku, Tokyo 102-0094

Phone: +03 3265 8301 - Fax: +03 3262 2359 - E-mail: ochi@jamarc.go.jp

N. Takagi, Director Executive Secretary Japan Deep Sea Trawlers Association, Ogawacho-Yasuda Bldg., 6 Kanda-Ogawacho, 3-Chome, Chivoda-ku, Tokyo 101-0052

Phone: +81 33 291 8508 - Fax: + 81 33 233 3267 - E-mail: nittoro@mx3.mesh.ne.jp

K. Tanaka, Deputy Director, International Affairs Div., Fisheries Policy Planning Dept. Fisheries Agency, Government of Japan, 1-2-1 Kasumigaseki, Chivoda-ku, Tokyo

Phone: +81 3 3591 1086 - Fax: +81 3 3502 0571 - E-mail: kengo_tanaka@nm.maff.go.jp

REPUBLIC OF KOREA

Head of Delegation

S.-J. Yoo, Director, International Cooperation Division, Ministry of Maritime Affairs and Fisheries, 139 Chungiong-No. 3, Seodaemun-Gu, Seoul, 120-715

Phone: +82-2-3148-6990~1 - Fax: +82-2-3148-6996 -E-mail: ussj@momaf.go.kr

Representative

S.-J. Yoo (see address above)

Adviser

Y.-S. Jung, First Secretary for Maritime Affairs & Fisheries, Embassy of Korea, 2450 Massachusetts Avenue, N. W., Washington, D.C. 20008 Phone: +202 939 5676 - Fax: +202 387 0402

LATVIA

Head of Delegation

N. Riekstins, Director, National Board of Fisheries, Ministry of Agriculture, 2, Republikas laukums, LV-1010 Riga

Phone: +371 732 3877 - Fax: +371 733 4892 - E-mail: fish@com.latnet.lv

Alternate

R. Derkacs, Head of International Agreements and Legal Div., National Board of Fisheries, 2, Republikas laukums, LV-1010 Riga

Phone: +371 732 3877 - Fax: +371 733 4892 - E-mail: fish@com.latnet.lv

Representatives

N. Riekstins (see address above) R. Derkacs (see address above)

Advisers

I. Voits, President, Latvian Fisheries Association, Ganibu Dambis 24a-502, Riga, LV-1005 Phone: +371 7383197 - Fax: +371 7383197 - Mob. Phone 371 9363094

LITHUANIA

Head of Delegation

A. Raudonius, Vice-Minister of Agriculture, Ministry of Agriculture, 19 Gedimino pr., LT-2025 Vilnius Phone: +370 2 391306 Fax: +370 2 391308 E-mail: <u>albinasr@zum.lt</u>

Alternate

V. Vaitiekunas, Director, Fisheries Dept. under the Ministry of Agriculture, 19 Gedimino str., Vilnius 2600 Phone: +370 02 391174 - Fax: 37002 341176 - E-mail: <u>vytautasv@zum.lt</u>
A. Rusakevicius, Fisheries Department under the Ministry of Agriculture, 19 Gedimino St., 2600 Vilnius Phone: +370 2 391183 - Fax: +370 2 391176 - E-mail: <u>algirdasr@zum.lt</u>

Representatives

A Raudonius (see address above) A. Rusakevicius (see address above)

Adviser

B.Urboniene, JSC Vigomeras, Poilsio str. 20-33, 5810 Klaipeda Phone: +3706 345518 - Fax: +3706 344429 - E-mail: vigomeras@takas.lt

NORWAY

Head of Delegation

P. Gullestad, Directorate of Fisheries, P. O. Box 185, N-5002 Bergen
Phone: +47 55 23 80 00 Fax: +47 55 23 80 90 E-mail: <u>peter.gullestad@fiskeridir.dep.telemax.no</u>

Alternate

T. Lobach, Directorate of Fisheries, P. O. Box 185, N-5002 Bergen Phone: +47 55 23 80 00 Fax: +47 55 23 80 90 E-mail: terje.lobach@fiskeridir.dep.telemax.no

Representatives

P. Gullestad (see address above)

T. Lobach (see address above)

Advisers

W. Barstad, Norwegian Fishing Vessel Owners Association, P.O. Box 67 Sentrum, 6001 Aalesund Phone: +47 70 10 14 60 - Fax: +47 70 10 14 80 - E-mail: webjorn@fiskebatreder.no
H. P. Johansen, Fisheries Counsellor, Royal Norwegian Embassy, 2720 34th St. N.W., Washington, D.C. 20008 Phone: +202 944 8981 - Fax: +202 337 0870 - E-mail: counselor@fish.norway.org
S. Owe, Director General, Ministry of Fisheries, P. O. Box 8118 Dep., 0032 Oslo
Phone: +47 22 24 64 71 Fax: +47 22 24 95 85 E-mail: stein.owe@fid.dep.telemax.no
E. K. Viken, Fiskeridepartementet, Postboks 8118 Dep., 0032 Olso
Phone: +22 24 6482 - Fax: +22 24 9585 - E-mail: ellen.viken@fid.dep.no

POLAND

Head of Delegation

Z. Gandera, Director, Fisheries Dept., Ministry of Agriculture and Rural Development, 30 Wspolna Str. 00-930 Warsaw

Phone: +48 22 6280826 Fax: +48 22 6232204 - E-mail: z.gandera@minrol.gov.pl

Representative

Z. Gandera (see address above)

Advisers

M. Kucharski, Embassy of the Republic of Poland, 443 Daly Avenue, Ottawa, Ontario K1N 6H3 Phone: +613 789 0468 - Fax: +613 789 1218
B. Szemioth, Parkowa 13/17/123, Warszawa Phone: +48228508420 - Fax: 48228908920 - E-mail: <u>atlantex@alpha.net.pl</u>

RUSSIA

Head of Delegation

V. Izmailov, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031
 Phone: +7095 - Fax: +7095 9213463

 r_{1000} , r_{1095} - r_{ax} , r_{1095} - r_{2x}

Representative

V. Izmailov (see address above)

Advisers

V. K. Babayan, Head of Laboratory for System Analysis of Fishery Resources, VNIRO, 17, V. Krasnoselskaya, . Moscow 107140

Phone: +70 95 264 6985 - Fax: +70 95 264 9187 - E-mail: <u>babayan@vniro.msk.su</u>

K. A. Bekyashev, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031

Phone: - Fax: +7095 921 3463

M. G. Botvinko, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow 103031

Phone: +7095 924 7611 - Fax: +7095 921 3463

G. V. Gusev, State Committee for Fisheries of the Russian Federation, Fisheries Dept., 12 Rozhdestvensky Boul., Moscow 103031

Phone: +7095 921 9880 - Fax: +7095 921 3463

V. M. Kolesnikov, Deputy of Head of Resource Department, State Committee for Fisheries of the Russian Federation, 12 Rozhdestvensky Boul., Moscow

Phone: +7095 924 3372 - Fax: +7095 9243372

I. Mikhno, Fisheries Attache, Embassy of Russia, 1609 Decatur Street, Washington, D.C. 20011

Phone: +202 726 3838 - Fax: +202 726 0090 - E-mail: rusfishatt@starpower.net

V. M. Mishkin, General Director, Scientific and Technical Firm "Complex Systems", 5, Kominterna str., P. O. Box 183038, Murmansk

Phone: +78152 476080 - Fax: +47 7891 0098

A. Okhanov, Russian Representative on Fisheries in Canada, 47 Oceanview Drive, Bedford, Nova Scotia, Canada B4A 4C4

Phone: +902 832 9225 - Fax: +902 832 9608

V. A. Rikhter, ATLANTNIRO, 5 Dmitry Donskoy St., Kaliningrad, 236000 Phone: +70 112 22 5547 – Fax: +70 112 21 9997 – E-mail: <u>atlant@baltnet.ru</u> A. Rodin, Horlovsky St. 3-13, Moscow 9162881

Phone: +7095 9162381 - Fax: +7095 9162460

E. Samoilova, PINRO, 6 Knipovich St., Murmansk 183763

Phone: +7 8152473461 - Fax: +47 78910518 - E:mail - inter@pinro.murmansk.ru

V. N. Shibanov, PINRO, 6 Knipovich St., Murmansk 183763

Phone: +7 8152 47 34 61 - Fax: +47 789 10 518 - E-mail: inter@pinro.murmansk.ru

F. M. Troyanovsky, PINRO, 6 Knipovich St., Murmansk

Phone: +7 8152 47 34 61 - Fax: +47 789 10 518 - E-mail: inter@pinro.murmansk.ru

UKRAINE

Head of Delegation

V. B. Chernik, Deputy Chairman, State Department for Fisheries of Ukraine, 82A Turgenivska str., Kiev, 04050

Phone: +38044 226 2405 - Fax: +380 44 226 2405 - E-mail: nauka@i.kiev.ua

Representative

V. B. Chernik (see address above)

Advisers

V. Litvinov, Senior Expert, Div. for International Fishing Policy, State Department for Fisheries of Ukraine, 82A Turgenivska str., Kiev, 252053

Phone: +38044 216 6883 - Fax: +38044 216 6883 - E-mail: nauka@i.kiev.ua

A. Anastasov, Head of Commercial Fishing Dept., "Yugrybpoisk" State Fishing Company, 6, I. Kozlova str., 334500 Kerch, Republic of Crimea

Phone: +06561 2 01 82 - Fax: +06561 2 13 35 / 23549

UNITED STATES OF AMERICA

Head of Delegation

 P. Kurkul, Regional Administrator, Northeast Region, National Marine Fisheries, 1 Blackburn Drive, Gloucester, MA 01930
 Phone: +978 281 9250 - Fax: +978 281 9371 - E-mail: pat.kurkul@noaa.gov

Representative

P. Kurkul (see address above)

Alternates

J. D. O'Malley, Executive Director, East Coast Fisheries Federation. Inc., P. O. Box 649, Narragansett, RI 02879 Phone: +401 782 3440 - Fax: +401 782 4840

J. Pike, Director, Government Relations, Scher and Blackell, Suite 200, 1850 M Street NW, Washington, DC 20036

Phone: +202 463 2511 - Fax: +202 463 4950 - E-mail: jpike@shebla.com

Advisers

J. Anderson, Fisheries Management Specialist, Northeast Region, National Marine Fisheries Service, U.S. Department of Commerce, 1 Blackburn Dr., Gloucester, MA 01930
 Phone: +978 281 9226 - Fax: 978-281-9135 - E-mail: jennifer.anderson@noaa.gov
 N. Anderson, National Marine Fisheries Service, 1 Blackburn Drive, Gloucester, MA 01930

Phone: +978 281 9383 – Fax: +978 281 9371 – E-mail: <u>nick.anderson@noaa.gov</u>

K. Beal, National Marine Fisheries Service, 1 Blackburn Drive, Gloucester, MA 01930

Phone: +978 281 9267 - Fax: +978 281 9117 - E-mail: <u>ken.beal@noaa.gov</u>

N. Brajevich, International Affairs Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

Phone: +202 647 3228 - Fax: +202 736 7350 - E-mail: BrajevichNM@state.gov

P. Burns, National Marine Fisheries Service, 1 Blackburn Dr., Gloucester, MA 01930

Phone: +978 281 9144 - Fax: - E-mail: peter.burns.@noaa.gov

W. Chan, National Marine Fisheries Service, 1 Blackburn Dr., Gloucester, MA 01930

Phone: +978 281 9328 - Fax: +978 281 9394 - E-mail: winnie.chan@noaa.gov

C. Coogan, National Marine Fisheries Service, 1 Blackburn Dr., Gloucester, MA 01930

Phone: +978 281 9118 - Fax: +978 281 9137 - E-mail: ccoogan@noaa.gov

G. H. Darcy, Assistant Regional Administrator, Sustainable Fisheries, National, Marine Fisheries Service, Northeast Regional Office, One Blackburn Drive, Gloucester, MA 01930

Phone: +978 281 9331 – Fax: +978 281 9135 – E-mail: George.Darcy@noaa.gov

D. Ellenton, President, World Wide Trading, Inc., 20 Locust Street, Suite 201, Danvers, MA 01923

Phone: +978 762 4665 - Fax: +978 750 6743 - E-mail: wwt@tiac.net

S. V. Fordham, Fisheries Project Manager, Center for Marine Conservation, 1725 DeSales Street, NW Suite 600, Washington, DC 20036

Phone: +202 429 5609 - Fax: +202 872 0619 - E-mail: sonja@dccmc.org

L. C. Hendrickson, Northeast Fisheries Science Center, 166 Water St., Woods Hole, MA 02543 Phone: +508 495 2285 - Fax: +508 495 2393 - E-mail: lisa.hendrickson@noaa.gov

W. P. Jensen, Chair, Management Board, Atlantic States Marine Fisheries Commission and Deputy Director,

Fisheries Service, Maryland Dept. of Natural Resources, Annapolis, MD 21401

Phone: +410 260 8261 - Fax: +410 260 8279 - E-mail: pjensen@state.md.us

D. Warner-Kramer, International Affairs Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

Phone: - Fax: +202 736 7350 - E-mail: warner-kramerdm@state.gov

G. S. Martin, Attorney, Office of the General Counsel, Northeast Region, National Oceanic and Atmospheric Administration, U.S. Dept. of Commerce, 1 Blackburn Dr., Gloucester, MA 01930

Phone: +1 978 281 9242 Fax: +1 978 281 9389 E-mail: gene.s.martin@noaa.gov

D. T. Mathers, Lieutenant Commander, Coast Guard Liaison Officer, Office of Marine Conservation (Rm 5806), U.S. Department of State, 2201 C Street NW, Washington, D.C. 20520

Phone: +202 647 3177 - Fax: +202 736 7350 - E-mail: dmathers@comdt.uscg.mil

R. Mayo, Northeast Fisheries Science Center, NMFS, 166 Water St., Woods Hole, MA 02543

Phone: +508 495 2310 - Fax: +508 495 2393 - E-mail: ralph.mayo@noaa.gov

M. Mooney-Seus, New England Aquarium, Central Wharf, Boston, MA 02110-3399

Phone: +617 973 6587 - Fax: +617 973 0242 - E-mail: mmooneys@aol.com

P. Moran, International Fisheries Div., F/SF4, National Marine Fisheries Service, U.S. Dept. of Commerce, 1315 East-West Hwy., Silver Spring, MD 20910

Phone: +301 713 2276 - Fax: +301 713 2313 - E-mail: pat.moran@noaa.gov

S. Olsen, National Marine Fisheries Service, 1 Blackburn Drive, Gloucester, MA 01930

Phone: +978 281 9330 - Fax: +978 281 9117 - E-mail: susan.olsen@noaa.gov

J.-P. Plé, Senior Atlantic Fisheries Officer, Office of Marine Conservation (Room 5806), U.S. Dept. of State, 2201 C Street NW, Washington, DC 20520

Phone: +202 647 2335 - Fax: +202 736 7350 - E-mail: jple@state.gov

W. Quinby, Director, Mayflower Shipping Ltd., 5 Yeamans Road, Charleston, SC 29407

Phone: +843 577 0560 - Fax: +843 577 6644 - E-mail: mayflower@mindspring.com

A. Rosenberg, Dean, College of Life Sciences and Agriculture, University of New Hampshire, Taylor Hall, 59 College Road, Durham, NH 03824

Phone: +603 862 2020 - E-mail: andy.rosenberg@unh.edu

F. M. Serchuk, Chief, Resource Evaluation and Assessment Division, Northeast Fisheries Science Center,

NMFS, 166 Water St., Woods Hole, MA 02543-1097

Phone: +508 495 2245 - Fax: +508 495 2258 - E-mail: fred.serchuk@noaa.gov

D. E. Swanson, Chief, International Fisheries Div., F/SF4, National Marine Fisheries Service, U.S. Dept. of Commerce, 1315 East-West Highway, Silver Spring, MD 20910

Phone: +301 713 2276 - Fax: +301 713 2313 - E-mail: dean.swanson@noaa.gov

SECRETARIAT

L. I. Chepel, Executive Secretary

T. Amaratunga, Assistant Executive Secretary

F. D. Keating, Administrative Assistant

B. J. Cruikshank, Senior Secretary

S. Goodick, Accounting Officer

ŗ

D. C. A. Auby, Word Processing Secretary G. Moulton, Statistical/Conservation Measures Officer

F. E. Perry, Desktop Publishing/Documents Clerk

B. Crawford, Graphic Arts/Printing Technician

Annex 2. Agenda

I. Opening Procedure

- 1. Opening by the Chairman, P. Gullestad (Norway)
- 2. Appointment of Rapporteur
- 3. Adoption of Agenda
- 4. Admission of Observers
- 5. Publicity

II. Administrative

6. Review of Commission Membership

III. Conservation and Enforcement Measures

- 7. Report of the Brussels Working Group on Precautionary Approach (PA)
- 8. Report of STACTIC June Meeting
- 9. Inspection and Control Measures in the NAFO Regulatory Area
- Report of the Working Group on the Allocation of Fishing Rights

 a) Report of the Meeting on Shrimp Stocks
- 11. Review of the provisions on chartering operations in the NAFO Regulatory Area
- 12. Increase of inspection presence in the NAFO Regulatory Area
- 13. Report of STACTIC at the Annual Meeting
- 14. Canadian Management Measures for 2J3KL Cod in 2000

IV. Conservation of Fish Stocks in the Regulatory Area

- 15. Summary of Scientific Advice by the Scientific Council
- 16. Management and Technical Measures for Fish Stocks in the Regulatory Area, 2001
 - 16.1 Cod in Div. 3M
 - 16.2 Redfish in Div. 3M
 - 16.3 American plaice in Div. 3M
 - 16.4 Shrimp in Div. 3M
- 17. Management and Technical Measures for Fish Stocks Straddling National Fishing Limits, 2000
 - 17.1 Cod in Div. 3NO
 - 17.2 Redfish in Div. 3LN

17.3 American plaice in Div. 3LNO

17.4 Yellowtail flounder in Div. 3LNO

- 17.5 Witch flounder in Div. 3NO
- 17.6 Capelin in Div. 3NO
- 17.7 Squid (Illex) in Subareas 3 and 4
- 17.8 Shrimp in Div. 3LNO
- 17.9 Greenland halibut in Div. 3LMNO
- 17.10 If available in the Regulatory Area:
 - i) Cod in Div. 2J3KL
 - ii) Witch flounder in Div. 2J3KL
- 17.11 Redfish in Division 1F

18. Formulation of Request to the Scientific Council for:

a) Scientific advice on the management of fish stocks in 2002

19. Transfer of Quotas Between Contracting Parties

V. Closing Procedure

20. Time and Place of the Next Meeting

21. Other Business

22. Adjournment

Annex 3. Amendment to the Conservation and Enforcement Measures regarding Observers (FC Doc. 00/8)

This proposal was discussed by STACTIC during June 2000 Meeting and formalized/adopted during current 22nd Annual Meeting, September 19/00 with recommendation to the Fisheries Commission to amend Part VI.A.1(a) to read (in brackets and bold):

A. Observers

1.

Each Contracting Party shall require all its vessels fishing in the Regulatory Area to accept observers on the basis of the following:

a) each Contracting Party shall have the primary responsibility to obtain, for placement on its vessels, independent and impartial observers. (Observers are not to perform duties, other than those described in Sections 3, 4 and 5 below.)

Annex 4. Proposal to amend the NAFO Conservation and Enforcement Measures with a view to introducing satellite based vessel monitoring and related measures (FC Doc. 00/13)

Introduction

Contracting Parties have agreed to require all vessels fishing in the Regulatory Area to be equipped with satellite tracking devices not later than January 1, 2001. The purpose of the amendment is to adopt detailed rules for satellite tracking and to adjust the Hail System requirements since certain report types become superfluous with satellite tracking. The proposed detailed measures for satellite tracking are identical to the ones contained in the Scheme of control and enforcement applicable in the NEAFC Regulatory Area. The amendment should be applicable no later than July 1st, 2001.

Proposal

Draft amendment (to enter into force no later than July 1, 2001)

- Part VI Programme for Observers and Satellite tracking
- Chapeau : the terms "for application in 2001" are replaced by the terms "for application in 2003"
- Section B :
- The title is replaced by : "Satellite tracking/ Vessel monitoring System "VMS" "
- The current text is replaced by the following text :

"1. Each Contracting Party shall ensure that each of its vessels operating in the Regulatory Area is equipped with a satellite tracking device allowing the continuous tracking of its position by the Contracting Party.

To that end the satellite tracking device shall ensure the automatic communication at least once every six hours when operating in the Regulatory Area to a land-based fisheries monitoring centre (hereafter referred to as FMC) of data relating to:

- the vessel identification;
- the most recent geographical position of the vessel (longitude, latitude) with a position error which shall be less than 500 metres, with a confidence interval of 99%;
 - the date and time of the fixing of the said position of the vessel.

Each Contracting Party shall take the necessary measures to ensure that its FMC receives these data.

2. The FMC of each Contracting Party shall be equipped with computer hardware and software enabling automatic data processing and electronic data transmission. Each Contracting Party shall provide for back-up and recovery procedures in case of system failures.

3. Each Contracting Party shall take the necessary measures to ensure that the VMS data received from its fishing vessels are recorded in computer readable form for a period of three years.

4. The masters of fishing vessels shall ensure that the satellite tracking devices are at all times fully operational and that the information in paragraph 1. is transmitted. In the event of a technical failure or non-operation of the satellite tracking device fitted on board a fishing vessel, the device shall be repaired or replaced within one month. After this period, the master of a fishing vessel shall not be authorised to commence a fishing trip with a defective satellite tracking device. Where a device stops functioning and a fishing trip lasts more than one month, the repair or the replacement has to take place as soon as the vessel enters a port, the fishing vessel shall not be authorised to commence a fishing trip without the satellite tracking device having been repaired or replaced.

5. Each Contracting Party shall ensure that a fishing vessel with a defective satellite tracking device shall communicate, at least daily, reports containing the information in paragraph 1. to the FMC, by other means of communication (radio, facsimile or telex).

6. Each Contracting Party shall communicate reports and messages pursuant to paragraph 1. and paragraph 5. to the Executive Secretary as soon as possible, but not later than 24 hours after receipt of those reports and messages. If the Contracting Party so desires, it shall ensure that each of its fishing vessels shall communicate reports (by satellite, radio, facsimile or telex) to the Executive Secretary.

7. Each Contracting Party shall ensure that the reports and messages transmitted between the Contracting Party and the Executive Secretary or if the Contracting Party so desires, between its fishing vessels and the Executive Secretary, shall be in accordance with the data exchange format set out in Annex II. (Part III "VMS position report format")

8. The Executive Secretary shall make available as soon as possible the information received under paragraph 7. to other Contracting Parties with an active inspection presence in the Regulatory Area. All reports and messages shall be treated in a confidential manner.

9. Each Contracting Party shall notify the name, address, telephone, telex and facsimile numbers as well as the addresses for electronic communication of their relevant authorities to the Executive Secretary before 1 July 2001 and thereafter any changes without delay.

10. Subject to any other arrangements between Contracting Parties, each Contracting Party shall pay all costs associated with this system."

VMS position report format (Annex II of Part III)

Data	Code:	Mandatory /	Remarks:
Element:		Optional	
Start record	SR	M	System detail; indicates start of record
From	FR	M	Address of the transmitting party (Contracting Party)
Address	AD	М	Message detail; destination; XNS" for NAFO Secretariat
Sequence Number	SQ	0	Message detail; message serial number in current year
Type of Message	TM	М	Message detail; message type, "POS" as Position report/message to be communicated by VMS or other means by vessels with a defective satellite tracking device
Radio call sign	RC	М	Vessel registration detail; international radio call sign of the vessel
Trip Number	TN	0	Activity detail; fishing trip serial number in current year
Vessel Name	NA	M	Vessel registration detail; name of the vessel
External Registration Number	XR	М	Vessel registration detail; the side number of the vessel
Latitude	LA	M	Activity detail; position at time of transmission
Longitude	LO	M	Activity detail; position at time of transmission
Date	DA	M	Message detail; date of transmission
Time	TI	M	Message detail; time of transmission
Record Date	RD	M	Year, month and date
Record Time	RT	M	Hours and minutes in UTC
Record Number	RN	М	Serial number of the record in the relevant year
End of record	ER	М	System detail; indicates end of the record

Each data transmission is structured as follows:

- double slash ("//") and the characters "SR" indicate the start of a message;
- a double slash ("//") and field code indicate the start of a data element;
- a single slash ("/") separates the field code and the data;
- pairs of data are separated by space;
- the characters "ER" and a double slash ("//")indicate the end of a record.

Annex 5. Amendment to NAFO Conservation and Enforcement Measures regarding Shrimp in Division 3M (FC Doc. 00/11)

Part I.G. to read (amendment in bold):

- G. Other Measures Management Measures for Shrimp in Div. 3M
 - 1. Vessels fishing for shrimps in Division 3M in 2001 shall use nets with a minimum mesh size of 40 mm.
 - 2. Vessels fishing for shrimp in Division 3M in 2001 shall use sorting grids or grates maximum spacing between the bars of 22 mm.
 - 3. In the event that total by-catches of all regulated groundfish species in any haul exceed 5 percent by weight, vessel shall immediately change fishing area (minimum of 5 nautical miles) in order to seek to avoid further by-catches of regulated groundfish.
 - a) Each Contracting Party shall limit in 2001 the number of vessels fishing for shrimp in Div. 3M to the number that have participated in this fishery in the period from 1 January 1993 to 31 August 1995.
 - b) Each Contracting Party shall, in 2001, limit the number of fishing days by its vessels fishing for shrimp in Div. 3M to 90% of the maximum number of fishing days observed for their vessels in one of the years 1993, 1994 or 1995 (until 31 August 1995). However, for Contracting Parties with a track record in the period from 1 January 1993 to 31 August 1995, a minimum level of 400 fishing days is permitted.
 - c) Contracting Parties with no track record in the shrimp fishery in the period from 1 January 1993 to 31 August 1995 may, in 2001, fish for shrimp with one vessel in 100 fishing days.
 - d) Each Contracting Party shall communicate the number of fishing days to the Executive Secretary before 1 November 2000, that are available to that Contracting Party for 2001. The number of days shall be counted from the hail reports of vessels fishing for shrimp in Div. 3M and shall include the days of entry and exit from the Regulatory Area. In the case where vessels fishing for shrimp and other species on the same trip the number of days shall be counted from the day the vessel entered the shrimp fishery to the day the vessel ceased that fishery.

The Executive Secretary shall scrutinize the communications from the Contracting Parties, work with the relevant Contracting Parties if discrepancies are revealed, and by 1 December 2000 notify the number of vessels and fishing days applicable to all Contracting Parties.

- e) Vessels fishing for 3M shrimp may fish this stock in 2001 in Division 3M and in the area defined by the coordinates in footnote 1¹. However, in the period from June 1, 2001 (00.01 GMT) to September 30, 2001 (24.00 GMT), fishing for shrimp in the area defined by the coordinates in footnote 2² is prohibited.
- f) Each Contracting Party shall, within 30 days following the calendar month in which the catches were made, report provisional monthly fishing days in Div. 3M and the

area defined in footnote 1 to the Executive Secretary. The Executive Secretary shall, within 10 days following the monthly deadlines for receipt of the provisional fishing days statistics, collate the information received and circulate it to Contracting Parties.

- g) For vessels conducting trans-zonal fishery for shrimps between Div. 3M and the area defined in footnote 1, the same regulations as in NAFO Conservation and Enforcement Measures, Part III – Annex I – Hail System Message Format, no. 1.3., shall apply.
- h) Each Contracting Party shall in 2001 closely monitor its vessels fishing for shrimp and close the fishery when the number of fishing days available to that Party is exhausted. The number of fishing days shall be counted from the hail reports of vessel fishing for shrimp and shall include the days of entry or moves into Div. 3M and the area defined in footnote 1 and the days of moves or exit from Div. 3M and the area defined in footnote 1.
- i) In the case where a vessel is fishing for shrimp and other species on the same trip, the change of fishery shall be hailed and the number of fishing days counted accordingly.
- j) Fishing days of a Contracting Party may only be utilized by a vessel flying the flag of another Contracting Party under the conditions provided in I.B.
- k) Fishing days are not transferable between Contracting Parties.
- This management plan for 2001 will be reviewed at the Special Meeting of the Fisheries Commission on Shrimp in light of the most recent advice from the Scientific Council and the outcome of the Special Meeting.

l		-	² (see the map belo	ow)	
Point No.	Latitude	Longitude	Point No.	Latitude	Longitude
1	47°20'0	46°40'0	l (same as no. 7)	47°55'0	45°00'0
2	47°20'0	46°30'0	2	47°30'0	44°15'0
3	46°00'0	46°30'0	3	46°55'0	44°15'0
4	46°00'0	46°40'0	4	46°35'0	44°30'0
			5	46°35'0	45°40'0
			6	47°30'0	45°40'0
			7 (same as no. 1)	47°55'0	45°00'0



Annex 6. Working Group on 3M Shrimp Fishery Data 27 March 2001, Copenhagen, Denmark

Provisional Agenda

- 1. Opening by the Chair, H. Koster (EU)
- 2. Appointment of Rapporteur
- 3. Adoption of Agenda
- 4. Review and validation of catch and effort data for 3M shrimp
- 5. Adjournment

Annex 7. Special Fisheries Commission Meeting on Shrimp Allocation and Management in the NAFO Regulatory Area 28-29 March 2001, Copenhagen, Denmark

Provisional Agenda

- . 1. Opening by the Chairman, P. Gullestad (Norway)
 - 2. Appointment of Rapporteur
 - 3. Adoption of Agenda
 - 4. Evaluation of existing effort management system in Division 3M
 - 5. Possible establishment of a TAC in Division 3M
 - 6. Allocation of fishing opportunities among Contracting Parties in the NAFO Regulatory Area
 - 7. Other business
 - 8. Adjournment

Annex 8. Paper on Chartering

(FC Doc. 00/12)

Proposal to Modify Part I.B. and I.G. of NAFO Conservation and Enforcement Measures (amendments underlined)

Amend Part I.B. as follows:

- **B.** Chartering Arrangements
- 1. Replace the wording by:

"Each Contracting Party may <u>utilize</u> partly or wholly quota and shrimp fishing days allocated to that Party under Schedule I and Part I.G by way of charter arrangement with a fishing vessel flying the flag of another Contracting Party notified in accordance with Part III.D, subject to:

- the consent of the flag Contracting Party;
 - a favourable proposal adopted through a mail vote in accordance with Article XI.2 of the Convention.
- Contracting Parties shall limit such <u>charter arrangements</u> to one fishing vessel per year and for a limited duration not exceeding 6 months.
- 3. Contracting Parties intending to have recourse to such charter arrangements shall together with a request for a mail vote notify the following information to the NAFO Executive Secretary:
 - the name and registration of the <u>chartered</u> vessel and the <u>relevant</u> flag Contracting Party
 - a copy of the charter
 - the fishing possibilities <u>concerned</u>
 - the date as from which the vessel is authorized to commence fishing on these fishing possibilities
 - the duration of the charter
- 4. The <u>relevant</u> flag Contracting Party shall notify in writing its consent to the NAFO Executive Secretary.
- 5. The NAFO Executive Secretary shall circulate the above information and the consent of the flag Contracting Party without delay to Contracting Parties.
- 6. The <u>relevant flag</u> Contracting Party is responsible for ensuring that the vessel complies with the requirements of the NAFO Conservation and Enforcement Measures. This does not nullify the obligations of the Contracting Party to which the quota and shrimp fishing days have been allocated under Part I of the Conservation and Enforcement Measures, as appropriate.
- 7. All catches and incidental catches from such chartering arrangements shall be recorded by the relevant flag Contracting Party separate from other national catch data recorded according to Part I.D., and shall be reported to the Contracting Party to which the fishing possibilities have been allocated and to the Executive Secretary separate from other national catch data according to Part I.D. The Executive Secretary shall add these catches to the catch statistics of the Contracting Party to which the fishing possibilities have originally been allocated.
- 8. As a pilot project, these provisions shall apply only to the year 2001.

Amend Part I.G. as follows:

- Insert a new point I.G.4.j) which would read:

"j) Fishing days of a Contracting Party may only be utilized by a vessel flying the flag of another Contracting Party under the conditions provided in I.B (chartering arrangements)."

- Renumber point I.G.4.j) as point I.G.4.k) which would read:

k) "Fishing days are not transferable between Contracting Parties" (deletion of the last part of the sentence)

Annex 9. Formats for the Electronic Transmission of NAFO Hails from Contracting Parties to the NAFO Secretariat (FC Doc. 00/14)

EXPLANATORY NOTES

- a) The formats herein conform with the requirements for the NAFO Hails System as set out in FC Document <u>00/1</u>, the NAFO Conservation and Enforcement <u>Measures</u> Part III and Part III Annex I Hail System Message Format.
- b) The formats consist of variable length delimited records, and are based on systems currently in use in <u>NEAFC</u>.
- c) The variable length record is preferred over a fixed length record as some Contracting Parties collect more information from their vessels than is required by NAFO, and are forwarding the entire record to NAFO. The format is conducive to extraction of the required data fields by the receiving parties.
- d) The following convention is used in this paper: //FIELD NAME/field value//, where the field name is shown in uppercase, followed by the character "/", followed by the field value in lowercase. Fields are separated by "//".
- e) Each record begins with the string //SR// to indicate the Start of the Record.
- f) Each record ends with the string //ER// to indicate the End of the Record.
- g) Character fields (CHAR) shall conform with the ISO 8859.1 character set standard.
- h) Country codes used for addressee (AD) and sender (FR) shall conform with the ISO 3166 (1993) standard. E/F 7.3 states that user-assigned country codes shall start with the character "X", therefore it is proposed that the code XNS be used to designate the NAFO Secretariat, the addressee for hail messages.

Example 1 (continued)

NAFO HAILS SYSTEM - Part III Annex I Hail System Message Format

1.1 ENTRY HAIL

//SR

j

//FR/Name of transmitting party

//AD/Destination "XNS" for NAFO

//SQ/sequence number

//NA/name of vessel

//RC/International radio call sign

//XR/external identification letters and numbers

//DA/date of transmission

//TI/timeof transmission

//LA/latitude at time of transmission

//LO/longitude at time of transmission

//TM/indication of type of message "ENT"

//DI/NAFO Division into which the vessel is about to enter.

//OB/total round weight of fish by species (3 alpha codes) on board <u>upon entry into the</u> <u>Regulatory Area</u>, in kilograms rounded to the nearest 100 kilograms. Allow several pairs of fields, consisting of species + weight, with each field separated by a space. e.g. //OB/species weight species weight species weight//

//MA/name of the Master

//<u>T</u>S/target species

Allow several species to be entered, with the values separated by spaces, e.g. //TS/species species species//

320

Example 1 (continued)

NAFO HAILS SYSTEM - Part III Annex I Hail System Message Format

1.2 MOVE HAIL

//SR

//FR/Name of transmitting party

//AD/Destination "XNS" for NAFO

//SQ/sequence number

//NA/name of vessel

//RC/International radio call sign

//XR/external identification letters and numbers

//DA/date of transmission

//TI/timeof transmission

//LA/latitude at time of transmission

//LO/longitude at time of transmission

//TM/indication of type of message

"MOV"

//DI/NAFO Division into which the vessel is about to enter.

//MA/name of the Master

//TS/target species

Allow several species to be entered, with the values separated by spaces, e.g. //<u>T</u>S/species species//

Example 1 (continued)

NAFO HAILS SYSTEM - Part III Annex I Hail System Message Format

1.3 TRANSZONAL HAIL (between NAFO Divisions)

//SR

//FR/Name of transmitting party

//AD/Destination "XNS" for NAFO

//SQ/sequence number

//NA/name of vessel

//RC/International radio call sign

//XR/external identification letters and numbers

//DA/date of transmission

//TI/timeof transmission

//LA/latitude at time of transmission

//LO/longitude at time of transmission

//TM/indication of type of message "ZON"

//MA/name of the Master

//TS/target species

Allow several species to be entered, with the values separated by spaces, e.g. //TS/species species species//

Example 1 (continued)

NAFO HAILS SYSTEM - Part III Annex I Hail System Message Format

1.4 EXIT HAIL

//SR

//FR/Name of transmitting party

//AD/Destination "XNS" for NAFO

//SQ/sequence number

//NA/name of vessel

//RC/International radio call sign

//XR/external identification letters and numbers

//DA/date of transmission

//TI/timeof transmission

//LA/latitude at time of transmission

//LO/longitude at time of transmission

//TM/indication of type of message "EXI"

//DI/NAFO Division into which the vessel is about to enter.

//CA/catch in round weight taken in the Regulatory Area by species (3 alpha codes) in kilograms (rounded to the nearest 100 kilograms). Allow several pairs of fields, consisting of species + weight, with each field separated by a space. e.g. //CA/species weight species weight species weight//

//MA/name of the Master

//ER//

322

. 323

Example 1 (continued)

NAFO HAILS SYSTEM - Part III Annex I Hail System Message Format

1.5 TRANSHIPMENT HAIL

//SR

//FR/Name of transmitting party

//AD/Destination "XNS" for NAFO

//SQ/sequence number

//NA/name of vessel

//RC/International radio call sign

//XR/external identification letters and numbers

//DA/date of transmission

//TI/time<u>of transmission</u>

//LA/latitude at time of transmission

//LO/longitude at time of transmission

//TM/indication of type of message "TRA"

//KG/total round weight by species (3 alpha codes) to be transhipped in kilograms (rounded to the nearest 100 kilograms). Allow several pairs of fields, consisting of species + weight, with each field separated by a space. e.g. //KG/species weight species weight species weight//

//MA/name of the Master

lements	
f data el	
nition o	
Defii	

Data Element	Field Code	Type	Mandatory/Optional	Definition/Remarks
Start of Record	SR		W	Indicates start of the record
From	FR	<u>Char*3</u>	W	ISO-3166 Address. Address of the transmitting Contracting Party
Addressee	AD	<u>Char*4</u>	W	ISO-3166 Address. Address of the party receiving the message, "XNS" for NAFO
Sequence Number	SQ	9*muN	O	<u>Message serial number</u>
Name	AN	Char*30	X	ISO 8859.1. Name of vessel
International radio call sign	RC	<u>Char*7</u>	W	IRCS Code. International Radio Call sign of the vessel
External identification	XR	Char*14	M	ISO 8859.1. Side number of the vessel
Date	DA	Num*8	W	YYYYMMDD, Year, month and day
Time	IL	Num*4	W	HHMM, Hours and minutes in UTC
Latitude	LA	Char*5	M	S/NDDMM (WGS-84) e.g. //LA/N4600 = 46°00' North
Longitude	го	Char*6	M	E/WDDDMM (WGS-84) e.g. //LO/W04631 = 46°31' West
Type of Message	TM	Char*3	Μ	Indication of type of message ENT/MOV/ZON/TRA/EXI
NAFO Division	DI	Char*2	M	NAFO Division
Name of Master	· MA	Char*30	Μ	Name of the master of the fishing vessel
Quantity	OB	Char*3 Num*7	W	FAO Code. Total round weight of fish by species (3 alpha codes) on board upon entry into the Regulatory Area, in kiloprams rounded to the nearest 100
				kilograms. Allow several pairs of fields, consisting of species + weight, with each field separated by a space.
Catch	CA	Char*3 Num*7	W	FAO Code. Total catch in round weight taken in the Regulatory Area by species (3 alpha codes) in kilograms (rounded to the nearest 100 kilograms). Allow several pairs of fields, consisting of species + weight, with each field separated by a space.
Transhipped	<u>KG</u>	<u>Char*3</u> <u>Num*7</u>	W	FAO Code. Total round weight by species (3 alpha codes) to be transhipped in kilograms (rounded to the nearest 100 kilograms). Allow several pairs of fields, consisting of species + weight, with each field separated by a space
Target Species	TS	<u>Char*3</u>	M	FAO Code: allowance for multiple main species
End Record	ER		W	End of record
Annex 10. Proposal to amend the NAFO Conservation and Enforcement Measures Regarding Incidental Catch Limits (FC Doc. 00/15)

Proposal:

Amend the NAFO Conservation and Enforcement Measures to add the following paragraph (f) to Part I A 5 Incidental Catch Limits

- (f) To avoid excessive incidental catch the following fishing strategy shall be implemented;
 - (i) If the amount of incidental catch of any one species listed in Schedule I for which no quota has been allocated in that division to that Contracting Party, in any one haul exceeds 10% of the total catch of the other species in that haul, the vessel shall immediately change fishing area to reduce the incidental catch. The vessel must move a minimum 5 nautical miles from any position of the previous haul.
 - (ii) In cases where a ban on fishing is in force for any particular species or an "Others" quota for any species has been fully utilized, and the amount of incidental catch of this species in any one haul exceeds 5% of the total catch of other species in that haul, the vessel shall immediately change fishing area to reduce the incidental catch. The vessel must move a minimum 5 nautical miles from any position of the previous haul.
 - (iii) If any future haul exceeds the permitted incidental catch limit outlined in (i) or (ii) above, whichever is applicable, the vessel shall again immediately change fishing area to reduce the incidental catch. The vessel must move a minimum 5 nautical miles from any position of the previous hauls and shall not return to the area for at least 48 hours.

Annex 11. Proposal to Amend the NAFO Conservation and Enforcement Measures Regarding Part VII - Port Inspections

(FC Doc. 00/16)

Background

Part VII of the NAFO Conservation and Enforcement Measures requires Contracting Parties to ensure that port inspection take place on any occasion a fishing vessel having been fishing subject to NAFO Conservation and Enforcement Measures is discharging catch. According to the current measures, the results from port inspection shall be provided to the NAFO secretariat and shall be communicated to any other Contracting Party on request.

The content of port inspection should include verification of catches, of logbook records, mesh size and of inspection at sea. Sea inspection reports are sent to the Contracting Party without delay.

Communication of port inspection is sometimes delayed when vessels land in ports outside the Flag Contracting Party. In order to contribute to enhanced transparency and a better efficiency of the implementation of the NAFO Conservation and Enforcement Measures, it is proposed that the results of port inspection are communicated to the Flag Contracting Party without delay.

Furthermore, a standard report form would help to harmonise record of results of port inspection.

Proposal

1. Amend Part VII-1 of NAFO Conservation and Enforcement Measures to read : Part VII-1

- "(v) Results of port inspection shall include at least the information listed in Part VII Schedule I -B.
- (vi) The authorities of the port State shall, on request, transmit the results of the port inspection to the flag State of the vessel, within 14 working days from the date on which the inspection has been completed.

(vii) The copy of the results of the port inspection shall be transmitted to the NAFO Executive Secretary within 30 days as from the date on which the landing has been completed and shall be provided to other Contracting Party on request."

(viii) Where possible, Contracting Parties should transmit the results of the port inspection as required in (v) to (vii) in the format defined in Part VII-Schedule I-Part A.

2. Insert <u>Part VII-Schedule I : "port inspection report"</u> (see annex)

Part VII-Schedule I

B. Information to be inserted in the report

1. INSPECTION REFERENCES

ļ

1

Data Element	M /O	M /O Category ; Definition			
Inspection authority	 	Inspection detail : Name of the inspection authority or of the alternate body nominated by the authority			
Date	М	Inspection detail : Date the report is compiled			
Port of inspection	M	Vessel activity detail : Place where the vessel is inspected : port followed by ISO –3 code of the country as "St Johns / CAN"			
,	M	Vessel registration detail; name of the vessel			

2. TRIP INFORMATION

Data Element M /O		Category ; Definition			
Date trip started	М	Vessel activity details : date started the current fishing trip			
Vessel trip number	0	Vessel activity details : Number of the fishing trip in current year			
Date Entry in the RA	М	Vessel activity details : Date the vessel entered the NRA for the current fishing trip			
Date Exit from the RA	М	Vessel activity details : Date the vessel exited from the NRA for the current fishing trip			
Other areas visited	0	Vessel activity detail : other area where vessel have been fishing during the current trip			
Date trip Ended	М	Vessel activity details : date ended the current fishing trip			

;

- -----

3. **VESSEL IDENTIFICATION**

Data Element	М /О	Category ; Definition				
External Identification Number	М	Vessel registration details : Side Number of the vessel				
International Radio Call Sign	М	Vessel registration details : International Radio Call Sign of the vessel				
Flag State	М	Vessel registration detail; State where the vessel is registered, 3 SO country code				
NAFO Contracting Party	O (1)	Vessel registration detail :NAFO contracting party of the vesse as ISO code of the country, EUR for European Community, NC for Non Contracting Party				
Home port	0	Vessel registration details : Port of registration of the vessel or homeport				
Vessel owner	М	Vessel registration details : name and address of the vessel owner				
Vessel operator	M (2)	Vessel registration details : responsible for using the vessel				
Master name	0	Vessel activity details : name of the master				

if different from the flag state
 if different from the vessel owner

RESULT OF INSPECTION ON DISCHARGE 4.

4.1 General mormation					
Data Element	M /O	Category ; Definition			
Start date of M discharge		Discharge detail : date the vessel started discharge			
End date of discharge	М	Discharge detail : date the vessel finished discharge			
Has vessel landed all catches on board ?	М	Discharge detail : Has vessel landed all catches on board ?, answer Y if yes, N if not			
Comments	0	Discharge detail : comments as necessary. If discharge as not been completed, please give an estimation of catch still on board			

4.1 General information

4.2 Quantity discharged

Data Element	M /O	Category ; Definition
Species	М.	Discharge detail : FAO 3-alpha code (part V, schedule II, attachment II)
Presentation	М	Discharge detail : Product form
Live Weight	М	Quantities determined from the logbook.
Conversion factor	0	Product detail : Conversion factor as define by the master for the corresponding species, size and presentation, optional if already mention in table B
Process weight	M	Discharge detail : Quantities landed by species and presentation, in kilograms of product, rounded to the nearest 10 kg
Equivalent live weight	M	Discharge detail : Quantities landed in equivalent live weight, as "product weight x conversion factor", in kilograms, rounded to the nearest 10 kg
Comments	0	Discharge Details : free text area

4.3 Quantities staying on board the vessel

Data Element	M /O	Category ; Definition
Species	М	Discharge detail : FAO 3-alpha code (part V, schedule II, attachment II)
Presentation	М	Discharge detail : Product form
Conversion factor	0	Product detail : Conversion factor as define by the master for the corresponding species, size and presentation, optional if already mention in table B
Process weight	М	Discharge detail : Quantities landed by species and presentation, in kilograms of product, rounded to the nearest 10 kg
Equivalent live weight	М	Discharge detail : Quantities landed in equivalent live weight, as "product weight x conversion factor", in kilograms, rounded to the nearest 10 kg
Comments	0	Discharge Details : free text area

5. **RESULT OF GEAR INSPECTION¹**

Data Element	М /О	Category ; Definition		
Date of inspection	М	Inspection detail : Date of current gear inspection		
Inspected gear	М	Inspection detail : number of gear checked during port inspection		

5.1 General information

5.2 Otter trawl details

Data Element	M /O	Category ; Definition				
NAFO seal number	М	Inspection detail (if required) : Number of the NAFO seal attached to the gear after inspection at sea				
Is Seal Undamaged ?	M	Whether NAFO inspection seal is intact "yes" or "no"				
Gear type	М	International Standard Statistical Classification of the Fishing Gear, OTB for otter trawl				
Attachments		Otter trawl detail : attachment to footrope				
Grade bar spacing	M	Otter trawl detail : grade bar spacing in millimetres				
Mesh type	M	Otter trawl detail : respectively mesh type: SQ for square mesh , DI for diamant mesh				
Mesh size average	М	Otter trawl detail : average mesh size in the trawl part, by pair				
Trawl part	М	Trawl part measured				
Mesh size	М	Mesh size in millimetres				

¹ Verification shall be done when non-compliance have been cited / observed during inspection at sea.

To be filled in when port inspection also concerns inspection of gears on board. A detail form shall be filled in for every gear having been subject to port inspection

A. "Port inspection report" form

"Port inspection report"

Page n°	Of
1. INSPECTION REFERENCE	
Inspection authority	
Date of the report	
Port of inspection	
Vessel name	
	· ·
2. TRIP INFORMATION ¹	
Date trip started	
Trip number ²	
Activity in the NAFO RA :	
Date Entry in the RA	
Date Exit from the RA	
Date Exit from the RA Other areas visited	

1 To be filled in by the inspection authority or any alternate body nominated by the authorities as soon as the vessel land to port, based on logbook records. ² Where applicable

3. VESSEL IDENTIFICATION³

External Identification	·
International Radio Call Sign	
Flag State	
NAFO Contracting Party	
Home port	
Vessel owner	
Vessel operator	
Master name	

4.

RESULT OF INSPECTION OF DISCHARGE⁴

4.1 General information

Starting of discharge :	Date	Time
Ending of discharge :	Date	Time
Has vessel discharged all catches on board ?	YES	If YES, fill in table 4.2
	NO	IF NO, fill table 4.3
Comments		

4.2 Quantity discharged

Species (FAO Code)	Presentation	Live Weight (Log Book, Kg)	Conversion factor	Landing Processed Wt (kg)	Equivalent live weight (kg)	Diff (Kg)	Diff (%)
		L	<u> </u>	<u> </u>	<u>.</u>		

³ To be filled in based on the license information.

⁴ To be filled in after completion of discharge

Species (FAO Code)	Presentation	Live Weight (Log Book, Kg)	Conversion factor	Landing Processed Wt (kg)	Equivalent live weight (kg)	Diff (Kg)	Diff (%)
		· · · · · · · · · · · · · · · · · · ·					
							· · · · ·
Commer		<u></u>				L	

4.3 Quantity staying on board the vessel

To be filled where part of the catches stay on board after completion of discharge

Species	Pres	entation	Conversi	on factor	Process weigh (kg)	t Equivalent live weight (kg)
		-				
			· .			
			: 			
ļ						
ļ						

Comments	 	

5. GEAR INSPECTION IN PORT ⁵

5.1 General data

Number of gear inspected

Date gear inspection

Has the vessel been cited ?

If Yes, complete the full "verification of inspection in port" form.

If No, complete the form with the exception of the NAFO Seal Details.

		• •	
	Yes		
D	No		

⁵ Verification shall be done when non-compliance have been cited / observed during inspection at sea.

To be filled in when port inspection also concerns inspection of gears on board. A detail form shall be filled in for every gear having been subject to port inspection

5.2 Otter Trawl details

NAFO Seal number		· ·	
Is seal undamaged ?	Yes	No	
Gear Type:			
Attachments:			
Grate Bar Spacing (mm)			
Mesh Type:			

Average mesh sizes (mm)

 	 •

_ . _... _ ...

Annex 12. Statement by the Representative of the European Union on 2J3KL Cod

2J3KL cod has been and continues to be one of the key fish stocks in the Northwest Atlantic. The stock has been close to collapse and consequently has been kept under moratoria for many years to protect the stock in its entirety. The EU must therefore reiterate its grave concern at Canada's action in repeating its irresponsible behaviour as seen in 1999, whereby it has taken a decision to allocate to itself a so-called "index" TAC of 7,000 tonnes for a commercial fishery for 2000. The stock has thus become subject once again to conflicting and inconsistent conservation and management measures. As in 1999, there is neither scientific justification for the decision in question nor are there any indications to allow one to distinguish between different stock components for the inshore and offshore fisheries.

This situation is, therefore, contrary to both the consistency requirements laid down in Article XI(3) of the NAFO Convention and the Precautionary Approach. It also falls short of the conservation and compatibility standards reflected in the 1995 UN Agreement on Straddling Fish Stocks. Due to the biological unity of the stock, there is a danger that efforts which aim at ensuring the long-term sustainability of the stock are being seriously undermined and that the recovery of the stock itself is in jeopardy. The EU, therefore, strongly urges Canada to adopt consistent conservation and management measures for the year 2001 for this stock.

Annex 13. Proposal to amend the Conservation and Enforcement Measures Part I.H – Other Measures – Management Measures for Shrimp in Div. 3LNO (FC Doc. 00/17)

Background

During the 1999 meeting, the Fisheries Commission adopted a proposal for a 3L shrimp fishery (Part I.K). However, when doing so, Part I.H should also have been amended as it prohibited any directed shrimp fishery in Divisions 3LNO.

<u>Proposal</u>

Amend the title of Part I.H and the text of the prohibition to refer to Divisions 3NO only.

Annex 14. Amendment to NAFO Conservation and Enforcement Measures regarding 3L Cod (FC Doc. 00/10)

Part I.

F. Other Measures - No Directed Fishery for Cod in Div. 3L in the Regulatory Area

Noting differences that have been expressed on the subject of 2J3KL cod by Contracting Parties,

Noting the need to avoid prejudice to the legal position of any Contracting Party on this subject,

Noting the provisions of Article XI(3) of the NAFO Convention, which aim at ensuring consistency between the measures established for the Regulatory Area and the measures adopted by the relevant coastal State;

Noting that the advice from the Scientific Council strongly suggests a continuation of the moratorium for the entire stock;

Directed fisheries for cod in Division 3L in the Regulatory Area shall not be permitted in **2001**.

Contracting Parties other than Canada expressed their serious concern that management measures for this stock may not be consistent throughout its range in the Convention Area in the year **2001**.

Annex 15. Amendment to NAFO Conservation and Enforcement Measures regarding Witch Flounder in Div. 3L (FC Doc. 00/9)

Part I.

I. Other Measures - No Directed Fishery for Witch flounder in Division-3L in the Regulatory Area

Noting the available scientific advice, and

Noting the current moratorium that is being applied by Canada to the directed fishing of this stock inside the Canadian 200 mile zone,

Directed fisheries for witch in Division 3L in the Regulatory Area shall not be permitted in 2001.

OTA TABLE. Total allowable catches (TACs) and quotas (metric tons) for 2001 of particular stocks in Subareas 3 and 4 of the NAFO Convention Are both inside and outside the 200-mile fishing zone, where applicable.	 The values listed include quantities to be taken 	-
-	TA TABLE. Total allowable catches (TACs) and quotas (metric tons) for 2001 of particular stoc	d outside the 200-mile fishing zone, where appl

	Cod	1	Re	Redfish	Ameri	American plaice	Yellowtail	Witch	Capelin	G. halibut	G. halibut Squid (IIIex) ^{2,3}	Shrimp
Contracting Party	Div. 3M	Div. 3M Div. 3NO*	Div. 3M	Div. 3M Div. 3LN*	Div. 3M*	Div. 3LNO+	Div. 3LNO	Div. 3NO*	Div. 3NO*	Div. 3LMNO	Subareas 3+4	Div. 31
l. Canada	0	c	200	0			17675			4 446	1 U U	000 \$
2. Cuba	0	, '	1750	• c	, ,	•		> '				80 r
Denmark (Faroe Islands	spi			•					•	1		à
and Greenland)	0	ı	69		'	•	,		•			67
4. European Union	0	0	3100	0	0	0	260^{7}		0	16 406	N.S.*	61
France (St. Pierre et												•
Miguelon)	'	•	. 69	•	,	•	,	•	•	1	453	67
6. Iceland	•		,		'	ı	•			•	. •	67
7. Japan	1		400		'				0	3.038	510	67
8. Korea	'		69		•		ŀ	,			453	67
9. Norway	0	•	,	'	'	1		,	0			5.5
10. Poland	0	ı		•	'	•		,			766	5
11. Estonia									>			6)
12. Latvia 🖊	c	c	ו ז פּצען	c				c	c			61
 Lithuania 	>	0	000 01	5		I	•	Ċ.	Ð		551 1	67
14. Russia							•			3 779		67
15. Ukraine 16. Huited States of												67
America			99	1							121	5
17. Others	0	0	124	0	0	• 0	65'	0	. 1	1 971	794	0
Total Allowshie Catch	•	•	\$ 000		•		1.1 0008	•	,	017.02		000 1
TOTAL VILLOWADILL CALL			000 5				000 01		•	040 67	34 000	0 0 0

¹ The opening date for the Squid (illex) fishery is 1 July.

Airy quota listed for squid may be increased by a transfer from any "coastal state" as defined in Article 1, paragraph 3 of the NAFO Convention, provided that the TAC for squid is not exceeded. Transfers made to Contracting Parties conducting fisheries for squid in the Regulatory Area shall be reported to the Executive Secretary, and the report shall be made as promptly as possible.

'Not specified because the allocation to these Contracting Parties are as yet undetermined, although their sum shall not exceed the difference between the total of allocations to other Contracting Parties and the TAC. Of which no more than 40% (788 t) may be fished before I May 2001 and no more than 80% (1577 t) may be fished before 1 October 2001.

Each Contracting Party shall notify the Executive Secretary bi-weekly of catches taken by its vessels from this stock. Not more than 2500 tons may be fished before July 1, 2001. The Executive Secretary shall notify

without delay all Contracting Parties of the date on which. for this stock, accumulated reported catch taken by vessels of the Contracting Parties is estimated to equal 50 and then 100 percent of the TAC for that stock. Contracting Parties shall inform the NAFO Executive Secretary before 1 December 2000 of the measures to be taken to meet the advice of the NAFO Scientific Council.

¹ The provisions of Part I, Section A.5c) of NAFO Conservation and Enforcement Measures shall apply. *No directed fishing – The provisions of Part I, Section A.5a and c of NAFO Conservation and Enforcement Measures shall apply.

ទ

Annex 16. Quota Table for 2001

Annex 17. Fisheries Commission's Request for Scientific Advice on Management in 2002 of Certain Stocks in Subareas 3 and 4, including supplementary questions on Division 3M Shrimp for 2001 (FC Doc. 00/20)

1. The Fisheries Commission with the concurrence of the Coastal State as regards the stocks below which occur within its jurisdiction, requests that the Scientific Council, at a meeting in advance of the 2001 Annual Meeting, provide advice on the scientific basis for the management of the following fish and invertebrate stocks or groups of stocks in 2002:

Redfish (Div. 3M) Yellowtail flounder (Div. 3LNO) Squid (Sub-areas 3 and 4) Shrimp (Div. 3M, 3LNO) Greenland halibut (Sub-areas 2 and 3KLMNO) Capelin (Div. 3NO)

2. The Fisheries Commission with the concurrence of the Coastal State as regards the stocks below which occur within its jurisdiction, requests that the Scientific Council, at a meeting in advance of the 2001 Annual Meeting, provide advice on the scientific basis for the management of the following fish stocks on an alternating year basis:

Cod (Div. 3NO; Div. 3M) Redfish (Div. 3LN) American plaice (Div. 3LNO; Div. 3M) Witch flounder (Div. 3NO)

To implement this system of assessments in alternating years, all stocks were assessed in 1999 but advice pertained to different time periods to allow the introduction of the new scheme over time. Consequently:

- In 2000, advice was provided for 2001 and 2002 for cod in 3M, American plaice in 3M and witch flounder in 3NO. These stocks will then next be assessed in 2002.
- In 2001, advice will be provided for 2002 and 2003 for American plaice in 3LNO, cod in 3NO and redfish in 3LN. The next assessment of these stocks will thus be conducted in 2003.

The Fisheries Commission requests the Scientific Council to continue to monitor the status of these stocks annually and, should a significant change be observed in stock status (e.g. from surveys) or in by-catches in other fisheries, provide updated advice as appropriate.

- 3. The Commission and the Coastal State request the Scientific Council to consider the following in assessing and projecting future stock levels for those stocks listed above:
 - a) The preferred tool for the presentation of a synthetic view of the past dynamics of an exploited stock and its future development is a stock assessment model, whether agebased or age-aggregated.
 - b) For those stocks subject to analytical-type assessments, the status of the stocks should be reviewed and management options evaluated in terms of their implications for fishable stock size in both the short and long term. As general reference points, the implications

of fishing at $F_{0.1}$ and F_{2000} in 2002 and subsequent years should be evaluated. The present stock size and spawning stock size should be described in relation to those observed historically and those expected in the longer term under this range of options.

- c) For those stocks subject to general production-type assessments, the time series of data should be updated, the status of the stock should be reviewed and management options evaluated in the way described above to the extent possible. In this case, the general reference points should be the level of fishing effort or fishing mortality (F) which is calculated to be required to take the MSY catch in the long term and two-thirds of that effort level.
- d) For those resources for which only general biological and/or catch data are available, few standard criteria exist on which to base advice. The stock status should be evaluated in the context of management requirements for long-term sustainability and the advice provided should be consistent with the precautionary approach.
- e) Spawning stock biomass levels considered necessary for maintenance of sustained recruitment should be recommended for each stock. In those cases where present spawning stock size is a matter of scientific concern in relation to the continuing reproductive potential of the stock, management options should be offered that specifically respond to such concerns.
- f) Information should be provided on stock size, spawning stock sizes, recruitment prospects, fishing mortality, catch rates and TACs implied by these management strategies for the short and the long term in the following format:
- I. For stocks for which analytical-type assessments are possible, graphs of all of the following for the longest time-period possible:
 - historical yield and fishing mortality;
 - spawning stock biomass and recruitment levels;
 - catch options for the year 2002 and subsequent years over a range of fishing mortality rates (F) at least from F_{0.1} to F_{max};
 - spawning stock biomass corresponding to each catch option;
 - yield-per-recruit and spawning stock per recruit values for a range of fishing mortalities.
- II. For stocks for which advice is based on general production models, the relevant graph of production as a function of fishing mortality rate or fishing effort. Age-aggregated assessments should also provide graphs of all of the following for the longest time-period possible:
 - exploitable biomass (both absolute and relative to B_{MSY})
 - yield/biomass ratio as proxy for fishing mortality (both absolute and relative to F_{MSY})
 - estimates of recruitment from surveys, if available.
- III. Where analytical methods are not attempted, the following graphs should be presented, for one or several surveys, for the longest time-period possible:
 - time trends of survey abundance estimates, over:
 - an age or size range chosen to represent the spawning population
 - an age or size-range chosen to represent the exploited population
 - recruitment proxy or index for an age or size-range chosen to represent the recruiting population.

• fishing mortality proxy, such as the ratio of reported commercial catches to a measure of the exploited population.

For age-structured assessments, yield-per-recruit graphs and associated estimates of yield-per-recruit based reference points should be provided. In particular, the three reference points, actual F, $F_{0.1}$ and F_{max} should be shown.

- g) For squid (<u>Illex</u>) in Sub-areas 3 and 4, the Scientific Council is requested to advise on the level of TAC in high abundance years and on the criteria which could be reliably used to forecast changes in productivity under an annual management regime. Scientists are encouraged to further analyze available data toward developing other possible indicators that could be used under an in-season management regime for squid, recognizing that the practical use of such indicators would require that they be available as early in the season as possible.
- h) For shrimp in 3M, the Fisheries Commission notes that information to date from the commercial fishery in 2000 is showing relatively high catch rates. In light of this apparent change in stock status, the Scientific Council is requested to review information from the 2000 fishery at its November 2000 meeting and to evaluate the impact on this resource of removals in year 2001 and 2002 corresponding to 25,000 t, 30,000 t, 35,000 t and 40,000 t respectively. Furthermore, the Scientific Council is requested at its November 2000 meeting to evaluate, on the basis of the best data available, whether the provision for a Div. 3M shrimp closure in FC Working Paper 99/16 (Rev.) would be a precautionary approach-based measure and, if so, whether proposed area and timing of the closure are appropriate.
- 4. The results described in Section 3 should include information about the reliability of the results. To this end, the following information should be included in a synoptic form:
 - Parameter uncertainty in assessments, possibly as confidence intervals
 - Robustness of assessments to alternative assumptions or data series
 - Illustration of conflicts in data series

This information may be accompanied by quality statements giving the opinion of the Scientific Council about the reliability of the various data series for particular purposes.

- 5. Noting the progress made by the Scientific Council on the development of a framework for implementation of the Precautionary Approach, the Fisheries Commission requests that the Scientific Council provide the following information for the 2001 Annual Meeting of the Fisheries Commission for stocks under its responsibility requiring advice for 2002, or 2002 and 2003:
 - a) the limit and target precautionary reference points described in Annex II of the UN Fisheries Agreement indicating areas of uncertainty (when precautionary reference points cannot be determined directly, proxies should be provided);
 - b) information including medium term considerations and associated risk or probabilities which will assist the Commission to develop the management strategies described in paragraphs 4 and 5 of Annex II in the Agreement;
 - c) information on the research and monitoring required to evaluate and refine the reference points described in paragraphs 1 and 3 of Annex II of the Agreement; these research requirements should be set out in the order of priority considered appropriate by the Scientific Council;
 - any other aspect of Article 6 and Annex II of the Agreement which the Scientific Council considers useful for implementation of the Agreement's provisions regarding the precautionary approach to capture fisheries;

- e) propose criteria and harvest strategies for re-opening of fisheries and for new and developing fisheries; and
- f) to work toward the harmonization of the terminology and application of the precautionary approach within relevant advisory bodies.
- 6. In addition, the following elements should be taken into account by the Scientific Council when considering the precautionary approach:
 - a) Many of the stocks in the NAFO Regulatory Area are well below any appreciable level of B_{lim} or B_{buf} . For these stocks, the most important task for the Scientific Council is to inform on how to rebuild the stocks. In this context and building on previous work of the Scientific Council in this area, the Scientific Council is requested to evaluate various scenarios corresponding to recovery plans with timeframes of 5 to 10 years, or longer as appropriate. This evaluation should provide the information necessary for the Fisheries Commission to consider the balance between risks and yield levels, including information on the consequences and risks of no action at all.

References to "risk" and to "risk analyses" should refer to estimated probabilities of stock population parameters falling outside biological reference points.

- b) Where reference points are proposed by the Scientific Council as indicators of biological risk, they should be accompanied by a description of the nature of the risk incurred if the reference point is crossed (e.g. short-term risk of recruitment overfishing, loss of long-term yield, etc.)
- c) When a buffer reference point is proposed in order to maintain a low probability that a stock, measured to be at the buffer reference point may actually be at or beyond the limit reference point, the Scientific Council should explain the assumptions made about the uncertainty with which the stock is measured, and also the level of 'low probability' that is used in the calculation.
- d) Wherever possible, short and medium term consequences should be identified for various exploitation rates (including no fishing) in terms of yield, stability in yield from year to year, and the risk or probability of moving the stock beyond B_{lim} or B_{buf} . Whenever possible, this information should be cast in terms of risk assessments relating fishing mortality rates to the risks of falling below B_{lim} and B_{buf} , as well as of being above F_{lim} and F_{buf} , the risks of stock collapse and recruitment overfishing, as well as the risks of growth overfishing and the consequences in terms of both short and long term yields.
- e) When providing risk estimates, it is very important that the time horizon be clearly spelled out. By way of consequence, risks should be expressed in timeframes of 5, 10 and 15 years (or more), or in terms of other appropriate year ranges depending on stock specific dynamics. Furthermore, in order to provide the Fisheries Commission with the information necessary to consider the balance between risks and yield levels, each harvesting strategy or risk scenario should include, for the selected year ranges, the risks and yields associated with various harvesting options in relation to B_{tim} (B_{buf}) and B_{target}, and F_{lim} (F_{buf}) and F_{target}.
- 7. The Fisheries Commission, with the concurrence of the Coastal State, requests that the Scientific Council review available information, including any Canadian assessment

documentation on the stock status, and provide advice on catch levels for the 2J3KL witch flounder resource for 2002 and 2003. Any information pertaining to the relative distribution of the resource within the stock area, as well as changes in this distribution over time should also be provided.

- 8. The Scientific Council is requested to review all available information from both research vessel surveys and commercial catches on the relative biomass and geographic distribution of the following unregulated species/stocks occurring within the NAFO Regulatory Area: monkfish (Lophius americanus), wolffishes (Anarhichas lupus, A. minor, A. denticulatus), thorny skate (Amblyraja radiata), black dogfish (Centroscyllium fabricii), eelpouts (Lycodes spp.), longfin hake (Urophycis chesteri), and orange roughy (Hoplosthethus atlanticus).
- 9. The Scientific Council is requested to evaluate the distribution of the fishable biomass of the main commercial species of fish in relation to depth (in 100-m intervals). Separate values should be provided a) for fish above and below the length of 50% maturity and b) for fish above and below the current minimum landing size.
- 10. The Fisheries Commission also requests, with the concurrence of the Coastal State, that the Scientific Council evaluate the likely future medium-term development for Greenland halibut in 2+3KLMNO, Yellowtail flounder in 3LNO, American plaice in 3LNO (if possible) and cod in 3NO, under the following assumed constraints:
- a) Closure of targeted Greenland halibut fishery in depths less than 200, 500 and 800 meters or any other depths considered appropriate. These cases, which will have to make a reasonable assumption on the redirection of effort so removed onto the remaining depth strata, should be compared with evaluation of current fishing practices.
- b) Subject to the above, likely future medium-term consequences (5 to 10 years) for the yield, spawning biomass, exploitable biomass and recruitment, stating the relevant biological assumptions.
- c) The scenarios should be explored for a range of fishing effort assumptions corresponding to:
 - i) Maintaining overall fishing effort at the same levels as estimated in the last year for which good information is available.
 - ii) Increases or decreases of +/- 30% in fishing effort from this value.
 - iii) Additional scenarios as considered appropriate by the scientific Council.

In these scenarios, the Scientific Council should evaluate whether these fishing strategies provide adequate long-term protection to juvenile fish to allow maintenance of the spawning biomass at an appropriate level.

- 11. The Scientific Council is requested to review the distribution of juvenile American plaice and update the distribution of yellowtail flounder based on results from comprehensive research surveys. The Scientific Council is also requested to delineate further the areas of juvenile concentration in the Southeast Shoal area and its surroundings.
- 12. Regarding redfish in NAFO Division 1F, the Scientific Council is requested to review all available information on the distribution of this resource over time, as well as on the affinity of this stock to the pelagic redfish resource found in the ICES Sub-area XII, parts of SA Va and XIV or to the redfish found in NAFO Sub-areas 1-3.
- 13. With regard to shrimp in Divisions 3LNO, the Fisheries Commission, with the concurrence of the Coastal State, requests that the Scientific Council provide information on the geographical

distribution of this resource, as well as describe the relative and seasonal distribution inside and outside the NAFO Regulatory Area.

- 14. The Fisheries Commission requests the Scientific Council to provide information on the long-term effects of increasing mesh size from 130 mm to 145 mm in yield-per-recruit and stock spawning biomass-per recruit for Greenland halibut in 2+3KLMNO and in reducing by-catch of other species in that fishery. The Scientific Council is also requested to evaluate the medium term consequences in terms of yield and stock size of any such changes in mesh size.
- 15. The Fisheries Commission requests the Scientific Council to provide advice regarding the methodology for scientific research on fish stocks under moratoria.

Substantive Issue	Decision/Action (FC Doc. 00/21, Part I: item)
 Precautionary Approach (management NAFO stocks) 	Discussed: items 3.1-3.8 Decided to convene a group of technical experts in 2001, which will consider new recommendations to the Fisheries Commission
2. Allocation of Fishing Rights	Discussed: items 3.13-3.14 No further decision/action was taken.
3. Management of NAFO shrimp stocks	Discussed: items 3.16-3.18 Decided to convene a special meeting of the Fisheries Commission 28-29 March 2001 in Copenhagen.
4. Chartering Operations	Adopted: amendment to regulations, item 3.20 and Annex 8 (FC Doc. 00/12)
5. Conservation and Enforcement Measures: (STACTIC Report)	Discussed: items 3.22-3.28
- Formats for hail/VMS system	Adopted: item 3.23
- Scientific requirements for observers - Incidental catch	Adopted: item 3.24 Adopted: item 3.25
- Incidental catch - Harmonization of port inspection reports	Adopted: item 3.25 Adopted: item 3.26
- Overhaul of NAFO Conservation and Enforcement Measures	Agreed: item 3.27
TAC's and Regulatory Measures for major stocks in the Regulatory Area:	Discussed/Adopted: items 4.1-4.30
- Cod 2J3KL in the Regulatory Area	no directed fishery
- Cod 3M	no directed fishery
- Cod 3NO	no directed fishery
- Redfish 3M	5,000 mt no directed fishery
- Redfish 3LN - American plaice 3M	no directed fishery
- American plaice 3LNO	no directed fishery
- Yellowtail 3LNO	13,000 mt
- Witch 3NO	no directed fishery
- Witch 2J3KL (in the Reg. Area)	no directed fishery
- Capelin 3NO	no directed fishery
- Greenland halibut	29,640 mt
- Squid (Illex)	34,000 mt
- Shrimp in Div. 3L - Shrimp in Div. 3NO	6,000 mt no directed fishery
 Schedule I – Quota Table 2001 	Adopted: item 4.28
 Request to the Scientific Council for Scientific Advice on Management of Fish stocks in 2002 	Adopted: item 4.29

Annex 18. List of Decisions and Actions by the Fisheries Commission (22nd Annual Meeting, 18-22 September 2000)

PART II

Report of the Standing Committee on International Control (STACTIC)

22nd Annual Meeting, 18-22 September 2000 Boston, Massachusetts, USA

1. Opening of the Meeting

The Chairman, Jim Baird (Canada) opened the meeting at 1000 on 18 September 2000. Representatives from the following Contracting Parties were present: Canada, Cuba, Denmark (in respect of the Faroe Islands and Greenland), Estonia, the European Union (EU), France (St. Pierre & Miquelon), Iceland, Japan, Latvia, Lithuania, Norway, Russia, Ukraine and the United States.

2. Appointment of Rapporteur

Jennifer Anderson (United States) was appointed Rapporteur.

3. Adoption of Agenda

The provisional agenda, as circulated, was adopted with additional items referred from the Fisheries Commission (Annex 1).

4. Review of Annual Returns of Infringements

The Chairman commented that overall, improvements in reporting of the disposition of apparent infringements, that was noted for 1998 has continued in 1999. The representative from Canada agreed but pointed out that there were still several Contracting Parties that had not provided reports as noted in NAFO/FC Doc. 00/6. The representative from Canada asked that those Contracting Parties try to provide reports prior to the adjournment of the NAFO meetings on September 22. The representative from the European Union advised that for instances where its data were missing the cases were still pending, moreover, the European Union indicated that as data becomes available the reports will be provided to NAFO.

The representative from Canada pointed out that according to the Executive Secretary's report the outstanding reports on apparent infringements are from 1999 only. This is an improvement as in the past reports were outstanding for several years. The Contracting Parties were congratulated for addressing this long outstanding issue.

5. Review of Surveillance and Inspection Reports

The representatives from the European Union and Canada presented information (STACTIC Working Papers 00/25 and 00/26) on surveillance activities conducted during 1999.

6. Review of the Operation of the Hail System

The Chairman referred the Contracting Parties to STACTIC Working Paper 00/14 (Overview of the NEAFC Scheme re Automated Communications, 2000), STACTIC Working Paper 00/18 (Review of Operation of the Hail System, 2000) and STACTIC Working Paper 00/19 (Working Paper presented by Denmark (in respect of the Faroe Islands and Greenland) concerning

confidentiality towards information collected through automated hail reports and satellite tracking system.

The representative from Japan asked the Secretariat if NAFO would accept file transfers over X.25 connection. The Secretariat explained that the X.25 connection is being phased out in Canada and replaced by the Internet system. The Internet system can accept data from all mediums (e-mail, fax, etc.).

The representative from Denmark presented STACTIC Working Paper 00/19. He noted that Contracting Parties must have satellite tracking systems in place by January 1, 2001. When Vessel Monitoring Systems (VMS) are implemented, confidentiality and security issues must be addressed.

The representative from Denmark noted that suitable software is available for a fully functional VMS system and could be operational at NAFO at a cost of C\$200,000. The representative from Canada noted that the cost of implementing data transmission from the requirements of 100% satellite tracking is considerably less than that of a VMS system.

It was agreed at the June 2000 STACTIC Technical Working Group on Communications that an automated hail system was necessary. There were different views on the mode of data transmission.

The Chairman asked the Contracting Parties to form an ad-hoc Working Group to review and compare the NAFO and NEAFC systems and present a report to STACTIC. This report was adopted and is included in Annex 2. Additionally, proposed format changes to the current NAFO hail system found in Working Paper 00/32 were tabled.

7(a). Observer Program and Satellite Tracking; Scientific Requirements/Observer Manual

The Contracting Parties reviewed STACTIC Working Paper 00/13 (Provisional Account of Observer Reports received at the NAFO Secretariat). Contracting Parties with outstanding 1999 observer reports agreed to provide them as soon as possible. Contracting Parties should ensure that observer reports are transmitted to the Secretariat as required as the absence of such reports is a breach of the observer program. This could be interpreted as a failure of some Contracting Parties to deploy observers.

STACTIC agreed that SCS Doc. 00/23 (Harmonized NAFO Observer Program Data System Proposal), that was presented at the June 2000 STACTIC Intersessional, was the most appropriate listing of scientific requirements for observers. This paper was prepared after extensive discussion at the Scientific Council, in response to a request from STACTIC to define the scientific requirements for the observer program in a harmonized format. The Scientific Council, represented by Ralph Mayo (US) and Dave Kulka (Canada), explained that the SCS Document 00/23 included information found in the Working Paper 00/10 (EU-observer manual).

The European Union clarified that its Observer Manual proposal (STACTIC WP 00/10) was actually a set of forms to be used by the observer to complete their functions.

There was a discussion of previous versions of a potential Observer Manual. It was agreed that the Canadian Observer Manual, submitted to STACTIC at the June 2000 Intersessional, would be a good starting point for development of a NAFO Observer Manual.

7(b). Amendments to Existing Programs

At the STACTIC June 2000 Intersessional meeting, it was agreed that Contracting Parties would provide updated information on how they ensure impartiality and independence for observers. The representatives from Denmark and Japan summarized STACTIC Working Papers 00/20 and 00/27, respectively on this topic. The Chairman noted that the information provided was in addition to information previously tabled at STACTIC (STACTIC W.P. 98/03).

8(a). Discussion of Other Conservation and Enforcement Measures; Examination of Possible Improvements in the Procedures for Gathering Discard Information

Canada and the European Union reported marginal improvement in the recording of discards in logbooks during 2000. It was agreed by STACTIC that further improvements on the recording of discards are required.

8(b). Review of the Fishing Strategies to be Employed to Avoid Excessive Incidental Catches

The representative of Canada summarized STACTIC Working Paper 00/23 (Proposal to Amend Conservation and Enforcement Measures Regarding Incidental Catch Limits).

After some discussion a revised proposal was accepted by all but one Contracting Party and is included as Annex 3. The representative from Japan could not support a haul by haul assessment and suggested a longer fishing period was more appropriate (Example 48 hours).

8(c). Possible Amendments to the Conservation and Enforcement Measures Regarding Juvenile Fish

The representative from Canada summarized STACTIC Working Paper 00/24 Revision 2 (Protection of Pre-recruits in Nursery Areas).

After some discussion, Canada's proposal was tabled with some support. However, dissenting views were expressed by the representatives from Japan and the European Union. The representative from Japan suggested that the current minimum mesh size requirements offer sufficient protection for juvenile fish. The representative from the European Union voiced concern that by closing an area, there would be no collection of data inside the area, that the closure could be difficult to enforce and questioned if there were alternate measures that would produce the same result. The representative from Canada responded that scientific cruises could take place inside the closed area and that 100% observer coverage and 100% satellite tracking would ensure enforceability of the closure. He also pointed out that STACTIC had been explicitly tasked with developing recommendations for protection of juveniles. He noted that Canada had tabled this proposal as well as two other proposals, and that no other Contracting Party had made any proposals.

No final agreement between the Contracting Parties was reached on this proposal.

The representative from Canada summarized STACTIC Working Paper 00/22 (Proposal to amend the NAFO Conservation and Enforcement Measures Regarding the Protection of Juvenile Groundfish). The Canadian proposal recommended an increase in minimum mesh size for groundfish and an implementation of a depth restriction for fishing Greenland halibut. It was noted that the mesh size increase proposal was also tabled at the June Intersessional Meeting. There was no additional information provided from the Scientific Council on this issue during the course of the annual meeting. Several Contracting Parties indicated that they could not support this proposal.

There appeared to be some support for the proposal on depth restriction for Greenland halibut. However, one Contracting Party felt that further data from the Scientific Council was necessary in order to make a decision. Another Contracting Party indicated that additional consultation within their delegation was required.

8(d). Possible Harmonization of Port Inspection Reports

The representative from the European Union summarized STACTIC Working Paper 00/31 (Proposal to Amend the NAFO Conservation and Enforcement Measures Regarding Part VII – Port Inspections) and noted that it was the same working paper presented at the STACTIC June 2000 Intersessional meeting. Several Contracting Parties supported the proposal in principle but raised concerns regarding the requirement to transmit the report to the flag state within 7 working days. The representative from Canada also noted that due to the large volume of port inspections Canada performs and their inspection procedures, it would be difficult for Canada to complete the proposed NAFO port inspection report.

Based on comments from the Contracting Parties, the representative from the European Union presented a revision of Working Paper 00/31. The revised proposal would permit Contracting Parties to transmit the results of the port inspection to the flag state within 14 days, provided the flag state requested the report. It was noted that standing requests would be accepted. The revision also defined a list of mandatory information required for port inspection and also indicated that attached forms were optional.

The Contracting Parties agreed that the revised wording in the proposal was acceptable. However, the representative from Canada noted that it would support providing all the information requested by part 1 of the proposal, but could not use the data format suggested by the European Union. The representative from Denmark noted that it could not accept part C of the proposal before formal agreement of coding specifications in the North Atlantic format.

In conclusion the proposal was accepted and is included in Annex 4. STACTIC recommends the adoption of this proposal by the Fisheries Commission.

9. Discussion of Possible Amendments to the Conservation and Enforcement Measures

The Fisheries Commission asked STACTIC to consider if it was necessary to overhaul the Conservation and Enforcement Measures and if so, what process would be necessary for the overhaul.

The Contracting Parties agreed that an overhaul of the Conservation and Enforcement Measures was necessary in order to: develop a cohesive document to reflect changes that have occurred since the measure were originally drafted; to identify roles and responsibilities of vessel masters, contracting parties, inspectors and the Secretariat; to take account of advancements in other international fisheries agreements. However, the Contracting Parties were reluctant to recommend a specific structure and course of action and should seek guidance from the Fisheries Commission.

10. Time and Place of Next Meeting

The next meeting of STACTIC (VMS/confidentiality, Juveniles/by-catch) will be held in London (at NEAFC) from 26-28 June 2001.

11(a). Other Matters, Chartering Arrangements

FC Working Papers 00/6 (United States Draft Working Paper on Charters) and 00/7 (Polish Position on Charters) and STACTIC Working Paper 00/28 (Proposal to Amend Conservation and Enforcement Measures by the Ukraine) were reviewed.

Several Contracting Parties commented that the Conservation and Enforcement Measures require further clarification of charter arrangements. The Chairman suggested that STACTIC should only consider clarifying the Conservation and Enforcement Measures regarding charter arrangements and it was agreed by the Contracting Parties that the proposal by the United States to extend the pilot project and drop the mail vote provision should be considered by the Fisheries Commission.

The representative from France (in respect of St. Pierre and Miquelon) tabled STACTIC Working Paper 00/33 to provide clarification of the wording in the Conservation and Enforcement Measures respecting chartering. There was general support from the Contracting Parties for the proposal, however several Contracting Parties suggested revisions to the document's wording. The Contracting Parties agreed on the revised Working Paper 00/33.

If the Fisheries Commission decides to extend the pilot project on charters STACTIC recommends the proposal outlined in Annex 5 be adopted by the Fisheries Commission.

11(b). Increase of Inspection Presence in the NAFO Regulatory Area

The representative from the European Union summarized STACTIC Working Paper 00/29 (Proposal to Amend Conservation and Enforcement Measures with a view to Introducing New Rules Concerning Obligatory Inspection Presence in the Regulatory Area). The European Union's paper sought to share the burden connected with providing an adequate inspection presence in the NAFO Regulatory Area.

With the exception of Iceland, the Chairman noted there was general support for the proposal by the Contracting Parties. The representative from the United States suggested language to help clarify some points in the European Union's proposal. The European Union presented Working Paper 00/29 Revised. There was support for the amended proposal (Annex 6) by all but one Contracting Party. Iceland did not support this proposal and indicated in their opinion there was sufficient inspection presence in the NAFO Regulatory Area.

11(c). Integration of Vessel Monitoring System

The representative from the European Union presented STACTIC Working Paper 00/30 (Proposal to Amend the NAFO Conservation and Enforcement Measures with a view to Introducing Satellite Based Vessel Monitoring and Related Measures).

Some Contracting Parties noted that although they support enhancements to satellite tracking, they believe the current hail system must remain in place for a transition period.

The European Union presented a revised Working Paper 00/30 and proposed a 2-year transition period. The revision also proposed an implementation date of July 1, 2001 for VMS. It must be re-iterated that 100% satellite tracking is still required by January 1, 2001.

The representatives from Denmark (in respect of Faroe Islands and Greenland) and Japan stressed concern over the confidentiality of vessel reports. The representative from Canada noted several concerns including their desire to see polling increased every 4 hours instead of the European Union's proposed 6 hour polling intervals. Canada also noted concern that there was no provision in the proposal making it an apparent infringement for a master to interfere with the VMS.

The Chairman noted that there was agreement on this proposal in principle however, there was no conclusion by STACTIC. The issues of confidentiality and the use of regulatory or convention area were not resolved. The Chairman also noted that the Contracting Parties proposed an allocation of C\$200,000 for the automatic hail and satellite tracking system.

12. Adoption of Report

The report was adopted by STACTIC with the following recommendations:

STACTIC recommends to the Fisheries Commission that:

- A proposal to amend the NAFO Conservation and Enforcement Measures Regarding Part VII

 Port Inspections (STACTIC W.P. 00/31-Revision 2) be adopted.
- 2. Formats for the Electronic Transmission of NAFO Hails from Contracting Parties to the NAFO Secretariat be adopted. (STACTIC W.P. 00/32)

13. Adjournment

STACTIC adjourned on 22 September 2000 at 1030 hrs.

- 1. Opening by the Acting Chairman, J. Baird (Canada)
- 2. Appointment of Rapporteur
- 3. Adoption of Agenda
- 4. Review of Annual Returns of Infringements
 - a) review of disposition of outstanding infringements by Contracting Parties
- 5. Review of Surveillance and Inspection Reports
- 6. Review of Operation of the Hail System
- 7. Observer Program and Satellite Tracking
 - a) Scientific Requirements/Observer Manual
 - b) Amendments to existing Programs
- 8. Discussion of Other Conservation and Enforcement Measures:
 - a) examination of possible improvements in the procedures for gathering discards information;
 - b) review of the fishing strategies to be employed to avoid excessive incidental catches;

c) possible amendments to the Conservation and Enforcement Measures regarding juvenile fish;

- d) possible harmonization of port inspection reports.
- 9. Discussion of possible amendments to the Conservation and Enforcement Measures (request from the Fisheries Commission)
- 10. Time and Place of the Next Meeting
- 11. Other Matters
 - a) Chartering Arrangements
 - b) Increase of Inspection Presence in the NAFO Regulatory Area
 - c) Integration of Vessel Monitoring System
- 12. Adoption of Report
- 13. Adjournment

Annex 2. Report of STACTIC Ad Hoc Working Group on Comparison of the NAFO Hail System with the NEAFC Communication System

1. Opening of the Meeting

The Chairman, Mr. M. T. Nedergaard (Denmark in respect of Faroe Islands and Greenland), opened the meeting on 18 September 2000. The following Contracting Parties were present: Canada, Denmark (in respect of Faroe Islands and Greenland), European Union, Iceland, Russia and United States.

2. Comparisons of NAFO Hails and NEAFC Scheme

The differences and similarities as found between the NAFO Hail Reports and the NEAFC Scheme are presented in Appendix 1. Amendments to the NEAFC and NAFO scheme are considered necessary in order to create the necessary compatibility between the two systems. Main discrepancies identified between NAFO and NEAFC message formats were as follows. When transposing the NEAFC message format with the NAFO hail system the following amendments should be addressed:

- 1. "FR" Contracting Party sending the message, to be added
- 2. "DI" NAFO Division, to be added
- 3. "MA" Name of Master to be added
- 4. "TS" Target species (was previously DS directed species)
- 5. Making sequence number optional
- 6. Vessel Name should be mandatory
- 7. External Registration number should be mandatory
- 8. Days fished should be added as optional
- 9. Transhipped To should be added as optional
- 10. Transhipped From should be added as optional

3. Hail-VMS Connectivity

Pursuant to the introduction of VMS system 1 Jan 2001 the group foresaw that the automatic hail system should take account of the VMS requirement because both elements are technically interconnected. To this end it was noted that the cost of the hail system may have to be increased to take account of the VMS requirements.

4. Aspects relating to Inspection and Surveillance

Both the Hail and VMS systems aim at providing fishery patrol vessels with information regarding the location of fishing vessels operating in the area. NAFO and NEAFC handle this information in different ways.

Under the current NAFO hail system, the NAFO secretariat forward the hail messages to the Contracting Parties having notified inspection presence in the area, throughout the year, irrespective of whether the patrol vessel is active or not.

Under the NEAFC system, information on active fishing vessels is sent only to inspection vessels operating in the RA. Communication of the list of active vessels based on reception of the surveillance entry (SEN) and exit (SEX)messages which are sent by inspection craft (vessels, airplane...) when they enter or exit the RA. Lists of fishing vessels operating in the RA are elaborated by the Secretariat, based on position messages received from fishing vessels. These lists are sent daily only to the active patrol vessels i.e. having sent SEN message.

NAFO

NEAFC

inspectors etc.

Information regarding inspection activity

Start/end of surveillance		SEN and SEX message are sent to the Secretary. Messages
	No message required for air surveillance	include codified information on date, time, name, position,

Information regarding activity of fishing vessels

Nature of the information	Copy of the hail messages	List of active fishing vessels (i.e. having sent ENT message) and list of last POS report sent by those vessels.
Preparation of list of active vessels	Handled by the CP's inspection services, not handled by the secretariat	Handled by the Secretariat. Automated procedure has been set up.
Distribution	Sent to CP (inspection department) with a notified inspection presence.	Sent to the fishery patrol vessels which have notified their SEN message.
Frequency	Throughout the year	Daily
Security and confidentiality	Kept by inspection services (no specific confidentiality requirements)	Subject to confidentiality requirements, data to be destroyed after specified delay

It should be stressed that information on position of fishing vessels must be regarded as confidential under the NEAFC system and thus is subject to specific data security and confidentiality requirements. A VMS system presupposes the existence of such data security and confidentiality requirements.

<u>Appendix 1</u>

· · ·			RY" report		· · · · · · · · · · · · · · · · · · ·
NAFO (data element)	NEAFC (data element)	Code	Mandatory/ Optional NAFO	Mandatory/ Optional NEAFC	Remarks:
Start record	Start record	SR	М	М	System detail; indicates start of record
FROM		FR	М	M	Address of transmitting Party
Address	Address	AD	М	М	Message detail; destination "XNS" for NAFO
Sequence Number	Sequence Number	SQ	М	М	Message detail; serial number in current year
Type of Message	Type of Message	ТМ	М	M	Message detail; message type, "ENT" as Entry report
Radio call sign	Radio call sign	RC	М	М	Vessel registration detail; international radio call sign of the vessel
	Trip Number	TN	0	0	Activity detail; fishing trip serial number in current year
Vessel Name	Vessel Name	NA	М	0	Vessel registration detail; name of the vessel
	Contracting Party Internal Reference Number	IR	0	0	Vessel registration detail. Unique Contracting Party vessel number as ISO-3 flag state code followed by number
External Registration Number	External Registration Number	XR	М	0	Vessel registration detail; the side number of the vessel.
Latitude	Latitude	LA	М	М	Activity detail; position at time of transmission
Longitude	Longitude	LO	М	М	Activity detail; position at time of transmission

DIVISION		· DI	М		Division into which the vessel is about to enter
Quantity on board	Quantity on board	HO (Code used by NEAFC - OB			Activity detail; quantity by species on board, in pairs as needed.
Species	Species		М	М	FAO species code
live weight	live weight		M ·	М	Live weight in kilograms, rounded to the nearest 100 kilograms
TARGET SPECIES		DS ¹ Proposed TS	М		FAO species code
MASTERS NAME		MA	М		Name of the master
Date	Date	DA	М	M	Message detail; date of transmission
Time	Time	ΤI	М	M	Message detail; time of transmission
End of record	End of record	ER	М	М	System detail; indicates end of the record

¹ Under NEAFC Scheme DS means prohibited species

"EXIT" report

NAFO (Data Element)	NEAFC (Data Element)	Code	Mandatory/ Optional NAFO	Mandatory/ Optional NEAFC	Remarks:			
Start record	Start record	SR	М	М	System detail; indicates start of record			
FROM		FR	М	М	Address of transmitting Party			
Address	Address	AD	M .	М	Message detail; destination "XNS" for NAFO			
Sequ e nce Number	Sequence Number	SQ	M .	M	Message detail; message serial number in current year			
Type of Message	Type of Message	TM	М	М	Message detail; "EXI" as Exit report			
Radio call sign	Radio call sign	RC	М	М	Vessel registration detail; international radio call sign of the vessel			
	Trip Number	TN	0	0	Activity detail; fishing trip serial number in current year			
Vessel Name	Vessel Name	NA	М	0	Vessel registration detail; name of the vessel			
Contracting Party Internal Reference Number	Contracting Party Internal Reference Number	IR	0	. 0	Vessel registration detail. Unique Contracting Party vessel number as ISO-3 flag state code followed by number			
External Registration Number	External Registration Number	XR	М	0	Vessel registration detail; the side number of the vessel			
Latitude	Latitude	LA	M	М	Activity detail; position at time of transmission			

Longitude	Longitude	LO	М	М	Activity detail; position at time of transmission
DIVISION	1	DI	М		Division from which the vessel is about to leave
CATCH	Weekly Catch	CA			Activity detail; Cumulative catch retained on board by species, since commencement of fishing in the R.A
Species	Species	. 1	М	М	FAO species code
live weight	live weight	,	М	Μ	Live weight in kilograms, rounded to the nearest 100 kilograms
	Days Fished	DF	0	М	Activity detail; number of fishing days in the Regulatory Area either since commencement of fishing or last "Catch" report
MASTERS NAME		МА	М		Name of the master
Date	Date	DA	М	М	Message detail; date of transmission
Time	Time	TI	М	М	Message detail; time of transmission
End of record	End of record	ER	M	М	System detail; indicates end of the record
		F	T;	port	1
--	--	------	--------------------------------	---------------------------------	---
NAFO (Data Element)	NEAFC (Data Element)	Code	Mandatory/ Optional NAFO	Mandatory/ Optional NEAFC	Remarks
Start record	Start record	SR	М	M	System detail; indicates start of record
FROM		FR	М	М	Address of transmitting Party
Address	Address	AD	М	M	Message detail; destination "XNS" for NAFO
Sequence Number	Sequence Number	SQ	М	M	Message detail; message serial number in current year
Type of Message	Type of Message	ТМ	М	M	Message detail; message type, "TRA" as Transshipment report
Radio call sign	Radio call sign	RC	M	M	Vessel registration detail; international radio call sign of the vessel
	Trip Number	TN	0	0	Activity detail; fishing trip serial number in current year
Vessel Name	Vessel Name	NA	М	0	Vessel registration detail; name of the vessel
	Contracting Party Internal Reference Number	ſR	0	0	Vessel registration detail. Unique Contracting Party vessel number as ISO-3 flag state code followed by number
External Registration Number	External Registration Number	XR	М	0	Vessel registration detail; the side number of the vessel
Latitude	Latitude	LA	М	М	Activity detail; position at time of transshipment
Longitude	Longitude	LO	М	М	Activity detail; position at time of transshipment
Quantity on- loaded or off- loaded	Quantity on- loaded or off- loaded	KG			Quantity by species on- loaded or off-loaded in the R.A., in pairs as needed.
Species	Species		M ·	M	FAO species code

"TRANSHIPMENT" report

live weight	live weight		М	М	Live weight in kilograms, rounded to the nearest 100 kilograms
	Transshipped To	TT	0	M ⁱ	Vessel registration detail; International radio call sign of the receiving vessel
	Transshipped From	TF	0	M'	Vessel registration detail; International radio call sign of the donor vessel
MASTERS NAME		МА	М		Name of the master
Date	Date	DA	М	М	Message detail; date of transmission
Time	Time	ΤI	М	М	Message detail; time of transmission
End of record	End of record	ER	М	М	System detail; indicates end of the record

Whichever is appropriate.

"POSITION" "MOVE" "TRANSZONAL" report

r			" "TRANSZON	T- ••	
NAFO (Data Element)	NEAFC (Data Element)	Code	Mandatory/ Optional NAFO	Mandatory/ Optional NEAFC	Remarks:
Start record	Start record	SR	М	М	System detail; indicates start of record
FROM		FR	М	М	Address of transmitting Party
Address	Address	AD	М	М	Message detail; destination "XNS" for NAFO
Sequence Number	Sequence Number	SQ	М	М	Message detail; message serial number in current year
Type of Message	Type of Message	TM	Μ	M .	Message detail; message type, "POS" as Position report/message, "MOV' (as prescribed in Part II-Annex I, Para 1.2 Hail System message format), "ZON" (as prescribed in Part II-Annex I, Para 1.3 Hail System message format), to be communicated by VMS, or other means by vessels with a defective satellite tracking device
Radio call sign	Radio call sign	RC	М	М	Vessel registration detail; international radio call sign of the vessel
	Trip Number	TN	0	0	Activity detail; fishing trip serial number in current year
Vessel Name	Vessel Name	NA	М	0	Vessel registration detail; name of the vessel
	Contracting Party Internal Reference Number	IR	0	0	Vessel registration detail. Unique Contracting Party vessel number as ISO-3 flag state

		_			code followed by number
External Registration Number	External Registration Number	XR	М	· 0	Vessel registration detail; the side number of the vessel
Latitude	Latitude	LA	М	M	Activity detail; position at time of transmission
Longitude	Longitude	LO	M	М	Activity detail; position at time of transmission
DIVISION		DI	M		Division into which the vessel is about to enter
MASTERS NAME		MA	М		Name of the master
TARGET SPECIES		DS ¹ Proposed TS	М		FAO species code
Date	Date	DA	М	М	Message detail; date of transmission
Time	Time	TI	М	М	Message detail; time of transmission
End of record	End of record	ER	М	М	System detail; indicates end of the record

DATA EXCHANGE FORMAT AND PROTOCOLS

A. Data transmission format

Each data transmission is structured as follows:

- double slash ("//") and the characters "SR" indicate the start of a message;
- a double slash ("//") and field code indicate the start of a data element;
- a single slash ("/") separates the field code and the data;
- pairs of data are separated by space;
- the characters "ER" and a double slash ("//")indicate the end of a record.

B. Data exchange protocols NEAFC

Authorised data exchange protocols for electronic transmission of reports and messages between Contracting Parties and the Secretariat is X25 or X400

C. Data exchange protocols NAFO

Data exchange protocols for electronic transmission of reports and messages between Contracting Parties and the Secretariat is focusing on the relative desirability of X-25 based system or of internet SMTP.

oring, inspection and surveillance information	
of fisheries monito	
ormat for electronic exchange	The North Atlantic Format)
D. Fc	J

Category	Data Element	Field code	Type	Contents	Definitions
System	Start Record	SR			Indicates start of the record
Details	End Record	ER			Indicates end of the record
	Return Status	RS	Char*3	Codes	ACK / NAC = Acknowledged / Not Acknowledged
	Return Error Number	RE	Num*3	001 – 999	Codes indicating errors as received at operation center
Message Details	Address destination	AD	Char*3	ISO-3166 Address	Address of the party receiving the message, "XNE" for NEAFC, "XNS" for NAFO
	From	FR	Char*3	ISO-3166 Address	Address of the transmitting party
	Type of Message	TM	Char*3	Code	First three letters of the message type
	Sequence Number	SQ	Num*6	NNNNN	Message serial number
	Record Number	RN	Num*6	NNNNN	Serial number of the record in the relevant year
• .	Record Date	RD	Num*8	AYYYMMDD	Year, month and date
	Record Time	RT	Num*4	HHMM	Hours and minutes in UTC
	Date	DA	Num*8	ОДММҮҮҮҮ	Year, month and date
	Time	F	Num*4	MMHH	Hours and minutes in UTC

Category	Data Element	Field code	Type	Contents	Definitions
Vessel	Radio Call Sign	RC	Char*7	IRCS Code	International Radio Call Sign of the vessel
Registratio	Vessel name	NA	Char*30	ISO 8859.1	Name of the vessel
	Ext. registration	XR	Char*14	ISO 8859.1	Side Number of the vessel
	Flag State	FS	Char*3	ISO-3166	State of registration
	Contracting Party	Я	Char*3	ISO-3166 +max. 9N	Unique vessel number attributed by the flag State pursuant to
	internal reference number		Num*9		registration
	Port Name	- DO	PO - Char*20	ISO 8859.1	Port of registration of the vessel/homeport
	Vessel Owner	vo	Char*60	ISO 8859.1	Name and address of the vessel owner
	Vessel Charterer	VC	Char*60	ISO 8859.1	Name and address of the vessel charterer
Vessel	Vessel capacity	Γ	Char*2	"0C"/"LC"	According to: "OC" OSLO 1947 convention /"LC" LONDON
Characteris unit	unit		Num*4	Tonnage	ICTM-69
-tics Details					Capacity of the vessel in metric tons
	Vessel Power	٧P	Char*2	ISO 8859.1	Indication of which measurement unit applies "HP" or "KW"
	unit		Num*5	0-99999	Total main engine power
	Vessel Length	٨L	Char*2	"OA"/"PP"	unit "OA" length overall, "PP" length between perpendiculars
			Num*3	Length in meters	Total length of the vessel in meters, rounded to the nearest
	Vessel Type	Ē	Char*3	Code	As listed in Annex XII.C
	Fishing Gear	E	Char*3 ·	FAO Code	International Standard Statistical Classification of the Fishing Gear as (Annex XII.A)

Category	Data Element	Field code	Type	Contents	Definitions
Licence	Date of issuing	IS	Num*8	AYYYMMDD	Date of Authorisation to fish for one or more regulated species
Details	Regulated Resources	RR	Char*3	FAO Species Code	FAO Species Code for the regulated resource
	Start Date	SD	Num*8	ОДММУҮҮҮ	Date on which the validity of the authorisation/suspension commences
	End Date	ED	Num*8	D DDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDD	Expiry date of the validity of the authorisation to fish for the regulated resource
	Limited Authorisation	ΓΩ	Char*1	ISO 8859.1	"Y" or "N" to indicate whether a limited authorisation is valid or not
	Relevant Area	RA	Char*6	ICES Code	Area(s) prohibited
	Directed Species	DS	Char*3	FAO Species Code	Prohibited species
Activity	Latitude	LA	Char*5	S/NDDMM (WGS-84)	e.g. //LA/N6235 = 62°35' North
Details	Longitude	LO	Char*6	E/WDDDMM (WGS-84)	e.g. //LO/W02134 = 21°34' West
	Trip Number	IN	8*muN	666-100	Number of the fishing trip in current year
	Days Fished	DF	Num*3	1 – 365	Number of days the vessel spent in the Regulatory Area during the trip
	Weekly catch	CA			the cumulative catch retained on board by species, in kilograms live weight rounded to the nearest 100 kg since the vessel entered the R.A. or, in the event a previous has been transmitted during the same trip, since the last "Catch" report, in pairs as needed.
	Species		Char*3	FAO species code	
	Quantity		Num*7	0-999999	
	Quantity onboard	OB			Quantity onboard the vessel by species in kilograms live weight rounded to the nearest 100 kg, in pairs as needed
	Species		Char*3	FAO Codes	
	Quantity		Num*7	6666666-0	

Category	Data Element	Field code	Type	Contents	Definitions
	Transferred species	KG			Information concerning the quantities transferred between vessels by species in kilograms live weight rounded to the nearest 100 kg. whilst operating in the R.A.
	Species		Char*3	FAO Codes in pairs	
	Quantity		Num*7	0-9999999	
	Transhipped From	TF	Char*7	IRCS Code	International Radio Call Sign of the donor vessel
	Transhipped To	ΤΤ	Char*7	IRCS Code	International Radio Call Sign of the receiving vessel
Reporting Details	Catch	CA			Aggregate catch, landed or transhipped, taken by fishing vessels of the Contracting Party, by species as listed in tonnes live weight, rounded to tonnes, in pairs as needed
	. Species		Char*3	FAO species code	
	Quantity		Num*7	6666666-0	
	Cumulative catch	CC			Cumulative aggregate catch, landed or transhipped, taken by fishing vessels of the Contracting Party, by species as listed in tonnes live weight, rounded to tonnes, in pairs as needed
	Species		Char*3	FAO species code	
	Quantity		Num*7	6666666-0	
	Relevant Arca	RA	Char*6	ICES/NAFO Codes	Code for the relevant fishing area
	Zone	ZO	Char*3	ISO-3166	The code for a Contracting Party's zone
	Year and month	ΜY	9*muN	ММҮҮҮҮ	The relevant year and month of reporting

Category	Data Element	Field code	Type	Contents	Definitions
Surveillanc	Latitude	ΥŢ	Char*5	S/NDDMM (WGS-84)	e.g. //LA/N6235 = 62°35' North
e/ Observer Details	Longitude	ГО	Char*6	E/WDDDMM (WGS-84)	E/WDDDMM (WGS-84) e.g. //LO/W02134 = 21°34' West
_	Speed	SP	Num*3	Knots * 10	e.g. //SP/105 = 10,5 knots
	Course	CO	Num*3	360° degree scale	e.g. //CO/270 = 270°
	Activity	AC	Char*3	Activity code	First 3 characters of the activity (see Annex XII.C.2)
	Means of Surveillance	IM	Char*3	NEAFC Code	"VES" = surface vessel, "AIR" = fixed wing aircraft, "HEL" helicopter
	Assigned Inspector CP ID	AI	Char*7	NEAFC Code	ISO-3160 code for the Contracting Party followed by 4 digit number repeated as needed
	Observation Serial. No.	OS	Num*3	666 - 0	Serial number of the observation during relevant patrol in the Regulatory Area
	Date of sighting	DA	Num*8	АТТАНИВВ	Date when the vessel is sighted
	Time of sighting	Ц	Num*4	MMHH	Time in UTC when the vessel is sighted
_	Object Identification	IO	Char*7	IRCS Code	International Radio Call Sign of the sighted vessel
	Photograph	Ηd	Char*1	ISO 8859.1	Was there a photograph taken, "Y" or "N"
	Free Text string	MS	Char*255 ISO 8859.1	ISO 8859.1	Free text area

E. Additional Codes to be included in "The North Atlantic Format"

1

Data Element	Field Type code	Type	Contents	Definitions
Directed Species	TS	TS Char*3	FAO Codes	Directed species (target species)
Masters Name	MA	MA Char*30		Name of master of fishing vessel
Type of Message	MT	TM Char*3		Indication of message code: MOV, ZON
DIVISION	Id	DI Char*2		Division into which the vessel is about to enter
FROM	FR	FR Char*3	ISO – 3	The code for a Contracting Party

F. Structure of reports and messages required by NEAFC

Where appropriate, each Contracting Party retransmits to the NEAFC Secretariat data received from its vessels, in accordance with Articles 4, 6 and 10; subject to the following amendments:

- the address (AD) shall be replaced by the address of the Secretariat (XNE)
- the data elements "record date" (RD), "record time" (RT), "record number" (RN) and "from" (FR) shall be inserted

Return messages

Return message format as defined by NEAFC is:

Data Element	Field Code	Mandatory / Optional	Remarks
Start Record	SR	M	System detail; indicates start of record
Address	AD	М	Message detail; destination Contracting Party sending the report
From	FR	М	Message detail; "XNE" for NEAFC
Type of message	ТМ	М	Message detail; message type " RET " for return message
Return Status	RS	М	Reporting detail; code showing whether the message is acknowledged or not (ACK or NAK)
Return error number	RE	0	Reporting detail; number showing the type of error: message unreadable (101), inconsistent data (102), sequence error (103)
Record number	RN	М	Reporting detail; record number of the message which is received
Date	DA	M	Message detail; date of transmission
Time	TI	M	Message detail; time of transmission
End of Record	ER	М	System detail; indicates end of the record

Annex 3. Proposal to amend the NAFO Conservation and Enforcement Measures Regarding Incidental Catch Limits

(STACTIC Working Paper 00/23 - Rev. 3)

Proposal:

Amend the NAFO Conservation and Enforcement Measures to add the following paragraph (f) to Part I A 5 Incidental Catch Limits

- (f) To avoid excessive incidental catch the following fishing strategy shall be implemented;
 - (i) If the amount of incidental catch of any one species listed in Schedule I for which no quota has been allocated in that division to that Contracting Party, in any one haul exceeds 10% of the total catch of the other species in that haul, the vessel shall immediately change fishing area to reduce the incidental catch. The vessel must move a minimum 5-nautical miles from any position of the previous haul.
 - (ii) In cases where a ban on fishing is in force for any particular species or an "Others" quota for any species has been fully utilized, and the amount of incidental catch of this species in any one haul exceeds 5% of the total catch of other species in that haul, the vessel shall immediately change fishing area to reduce the incidental catch. The vessel must move a minimum 5 nautical miles from any position of the previous haul.
 - (iii) If any future haul exceeds the permitted incidental catch limit outlined in (i) or (ii) above, whichever is applicable, the vessel shall again immediately change fishing area to reduce the incidental catch. The vessel must move a minimum 5 nautical miles from any position of the previous hauls and shall not return to the area for at least 48 hours.

Annex 4. Proposal to Amend the NAFO Conservation and Enforcement Measures regarding Part VII - Port Inspections (STACTIC W.P. 00/31 - Rev. 2)

Background

Part VII of the NAFO Conservation and Enforcement Measures requires Contracting Parties to ensure that port inspection take place on any occasion a fishing vessel having been fishing subject to NAFO Conservation and Enforcement Measures is discharging catch. According to the current measures, the results from port inspection shall be provided to the NAFO secretariat and shall be communicated to any other Contracting Party on request.

The content of port inspection should include verification of catches, of logbook records, mesh size and of inspection at sea. Sea inspection reports are sent to the Contracting Party without delay.

Communication of port inspection is sometimes delayed when vessels land in ports outside the Flag Contracting Party. In order to contribute to enhanced transparency and a better efficiency of the implementation of the NAFO Conservation and Enforcement Measures, it is proposed that the results of port inspection are communicated to the Flag Contracting Party without delay.

Furthermore, a standard report form would help to harmonise record of results of port inspection.

Proposal

1. Amend Part VII-1 of NAFO Conservation and Enforcement Measures to read :

Part VII-1

- "(v) Results of port inspection shall include at least the information listed in Part VII Schedule I -B.
- (vi) The authorities of the port State shall, on request, transmit the results of the port inspection to the flag State of the vessel, within 14 working days from the date on which the inspection has been completed.
- (vii) The copy of the results of the port inspection shall be transmitted to the NAFO Executive Secretary within 30 days as from the date on which the landing has been completed and shall be provided to other Contracting Party on request."
- (viii) Where possible, Contracting Parties should transmit the results of the port inspection as required in (v) to (vii) in the format defined in Part VII-Schedule I-Part A.
- 2. Insert <u>Part VII-Schedule I : "port inspection report"</u> (see annex)

Part VII-Schedule I

B. Information to be inserted in the report 1.

-- - ---

INSPECTION	REFERENCES

Data Element	M /O	Category ; Definition
Inspection authority	М	Inspection detail : Name of the inspection authority or of the alternate body nominated by the authority
Date	М	Inspection detail : Date the report is compiled
Port of inspection	M	Vessel activity detail : Place where the vessel is inspected : port followed by ISO -3 code of the country as "St Johns / CAN"
Vessel Name	М	Vessel registration detail; name of the vessel

TRIP INFORMATION 2.

Data Element	М /О	Category ; Definition
Date trip started	М	Vessel activity details : date started the current fishing trip
Vessel trip number	0	Vessel activity details : Number of the fishing trip in current year
Date Entry in the RA	М	Vessel activity details : Date the vessel entered the NRA for the current fishing trip
Date Exit from the RA	М	Vessel activity details : Date the vessel exited from the NRA for the current fishing trip
Other areas visited	0	Vessel activity detail : other area where vessel have been fishing during the current trip
Date trip Ended	М	Vessel activity details : date ended the current fishing trip

VESSEL IDENTIFICATION 3.

Data Element	M /O	Category ; Definition
External Identification Number	М	Vessel registration details : Side Number of the vessel
International Radio Call Sign	М	Vessel registration details : International Radio Call Sign of the vessel
Flag State	М	Vessel registration detail; State where the vessel is registered, 3-ISO country code

Data Element	M /O	Category ; Definition
NAFO Contracting Party	O (1)	Vessel registration detail :NAFO contracting party of the vessel, as ISO code of the country, EUR for European Community, NCP for Non Contracting Party
Home port	0	Vessel registration details : Port of registration of the vessel or homeport
Vessel owner	М	Vessel registration details : name and address of the vessel owner
Vessel operator	M (2)	Vessel registration details : responsible for using the vessel
Master name	0	Vessel activity details : name of the master

(1) if different from the flag state
(2) if different from the vessel owner

RESULT OF INSPECTION ON DISCHARGE 4.

4.1 General in	formation
----------------	-----------

Data Element	M /O	Category ; Definition
Start date of discharge	М	Discharge detail : date the vessel started discharge
End date of discharge	М	Discharge detail : date the vessel finished discharge
Has vessel landed all catches on board ?	М	Discharge detail : Has vessel landed all catches on board ?, answer Y if yes, N if not
Comments	0	Discharge detail : comments as necessary. If discharge as not been completed, please give an estimation on catch still on board

Quantity discharged 4.2

Data Element	M /O	Category ; Definition
Species	М	Discharge detail : FAO 3-alpha code (part V, schedule II, attachment II)
Presentation	M	Discharge detail : Product form
Live Weight	М	Quantities determined from the logbook.
Conversion factor	0	Product detail : Conversion factor as define by the master for the corresponding species, size and presentation, optional if already mention in table B

Process weight	М	Discharge detail : Quantities landed by species and presentation, in kilograms of product, rounded to the nearest 10 kg
Equivalent live weight	M	Discharge detail : Quantities landed in equivalent live weight, as "product weight x conversion factor", in kilograms, rounded to the nearest 10 kg
Comments	0	Discharge Details : free text area

4.3 Quantities staying on board the vessel

Data Element	M /O	Category ; Definition
Species	М	Discharge detail : FAO 3-alpha code (part V, schedule II, attachment II)
Presentation	М	Discharge detail : Product form
Conversion factor	0	Product detail : Conversion factor as define by the master for the corresponding species, size and presentation, optional if already mention in table B
Process weight	M .	Discharge detail : Quantities landed by species and presentation, in kilograms of product, rounded to the nearest 10 kg
Equivalent live weight	М	Discharge detail : Quantities landed in equivalent live weight, as "product weight x conversion factor", in kilograms, rounded to the nearest 10 kg
Comments	0	Discharge Details : free text area

5. **RESULT OF GEAR INSPECTION¹**

5.1 General information

Data Element	м /о	Category ; Definition
Date of inspection	М	Inspection detail : Date of current gear inspection
Inspected gear	М	Inspection detail : number of gear checked during port inspection

¹ Verification shall be done when non-compliance have been cited / observed during inspection at sea.

To be filled in when port inspection also concerns inspection of gears on board. A detail form shall be filled in for every gear having been subject to port inspection

Data Element	M /O	Category ; Definition
NAFO seal number	М	Inspection detail (if required) : Number of the NAFO seal attached to the gear after inspection at sea
Is Seal Undamaged ?	М	Whether NAFO inspection seal is intact. – "yes" or "no"
Gear type	М	International Standard Statistical Classification of the Fishing Gear, OTB for otter trawl
Attachments		Otter trawl detail : attachment to footrope
Grade bar spacing	М	Otter trawl detail : grade bar spacing in millimetres
Mesh type	М	Otter trawl detail : respectively mesh type: SQ for square mesh , DI for diamant mesh
Mesh size average	M	Otter trawl detail : average mesh size in the trawl part, by pair
Trawl part	М	Trawl part measured
Mesh size	М	Mesh size in millimetres

5.2 Otter trawl details

A. "Port inspection report" form

"Port inspection report"

Page n°	•] Of		
1. INSPECTION REFERENCE		 "				<u> </u>
Inspection authority						
Date of the report					· · · · · · · · · · · · · · · · · · ·	·
Port of inspection						
Vessel name			·			
2. TRIP INFORMATION ²						
Date trip started						
Trip number ³						
Activity in the NAFO RA :						
Date Entry in the RA]
Date Exit from the RA						
Other areas visited						
Date trip ended						

³ Where applicable

 $^{^{2}}$ To be filled in by the inspection authority or any alternate body nominated by the authorities as soon as the vessel land to port, based on logbook records.

3. VESSEL IDENTIFICATION⁴

External Identification	
International Radio Call Sign	
Flag State	·
NAFO Contracting Party	
Home port	
Vessel owner	
Vessel operator	
, Master name	

4. RESULT OF INSPECTION OF DISCHARGE⁵

4.1 General information

Starting of discharge :	Date	Time
Ending of discharge :	Date	Time
Has vessel discharged all catches on board ?	YES	If YES, fill in table
	NO	IF NO, fill table 4.3
Comments		

4.2 Quantity discharged

Species (FAO Code)	Presentation	Live Weight (Log Book, Kg)	Conversion factor	Landing Processed Wt (kg)	Equivalent live weight (kg)	Diff (Kg)	Diff (%)
							,
				l			L

⁴ To be filled in based on the license information.

⁵ To be filled in after completion of discharge

\$

Species (FAO Code)	Presentation	Live Weight (Log Book, Kg)	Conversion factor	Landing Processed Wt (kg)	Equivalent live weight (kg)	Diff (Kg)	Diff (%)
		•					

Comments		 ,	

4.3 Quantity staying on board the vessel

Species	Presentation	Conversion factor	Process weight (kg)	Equivalent live weight (kg)
		· · · · · · · · · · · · · · · · · · ·		
			· · · · · · · · · · · · · · · · · · ·	<u> </u>

To be filled where part of the catches stay on board after completion of discharge

Comments		

5. GEAR INSPECTION IN PORT⁶

5.1 General data

Number of gear inspected

Date gear inspection

Has the vessel been cited ?

If Yes, complete the full "verification of inspection in port" form.

If No, complete the form with the exception of the NAFO Seal Details.

			-
Yes		-	-
No			
· · ·			

5.2 Otter Trawl details

NAFO Seal number		· · · ·
Is seal undamaged ?	Yes	No
Gear Type:		
Attachments:		
Grate Bar Spacing (mm)		
Mesh Type:		

 $^{^{\}rm 6}$ Verification shall be done when non-compliance have been cited / observed during inspection at sea.

To be filled in when port inspection also concerns inspection of gears on board. A detail form shall be filled in for every gear having been subject to port inspection.

Average mesh sizes (mm)

İ

i

TRAWL PART	
Wings:	
Body:	
Lengthening. Piece:	
Codend:	

383

1

-

Annex 5. Paper on Chartering (STACTIC W.P. 00/33-Revised)

Proposal to Modify Part I.B. and I.G. of NAFO Conservation and Enforcement Measures (amendments underlined)

Amend Part I.B. as follows:

B. Chartering Arrangements

1. Replace the wording by:

"Each Contracting Party may <u>utilize</u> partly or wholly quota and shrimp fishing days allocated to that Party under Schedule I and Part I.G by way of charter arrangement with a fishing vessel flying the flag of another Contracting Party notified in accordance with Part III.D, subject to:

- the consent of the flag Contracting Party;
 - a favourable proposal adopted through a mail vote in accordance with Article XI.2 of the Convention.
- 2. Contracting Parties shall limit such <u>charter arrangements</u> to one fishing vessel per year and for a limited duration not exceeding 6 months.
- 3. Contracting Parties intending to have recourse to such charter arrangements shall [together with a request for a mail vote] notify the following information to the NAFO Executive Secretary:
 - the name and registration of the chartered vessel and the relevant flag Contracting Party
 - a copy of the charter
 - the fishing possibilities <u>concerned</u>
 - the date as from which the vessel is authorized to commence fishing on these fishing possibilities
 - the duration of the charter
- 4. The <u>relevant</u> flag Contracting Party shall notify in writing its consent to the NAFO Executive Secretary.
- 5. The NAFO Executive Secretary shall circulate the above information and the consent of the flag Contracting Party without delay to Contracting Parties.
- 6. The <u>relevant flag</u> Contracting Party is responsible for ensuring that the vessel complies with the requirements of the NAFO Conservation and Enforcement Measures. This does not nullify the obligations of the Contracting Party to which the quota and shrimp fishing days have been allocated under Part I of the Conservation and Enforcement Measures, as appropriate.
- 7. As a pilot project, these provisions shall apply only to the year 2001.

Amend Part I.G. as follows:

- Insert a new point I.G.4.j) which would read:
 - "j) Fishing days of a Contracting Party may only be utilized by a vessel flying the flag of another Contracting Party under the conditions provided in I.B (chartering arrangements)."
- Renumber point I.G.4.j) as point I.G.4.k) which would read:
 - k) "Fishing days are not transferable between Contracting Parties" (deletion of the last part of the sentence)

Annex 6. Proposal to Amend the NAFO Conservation and Enforcement Measures with a view to introducing new rules concerning obligatory inspection presence in the Regulatory Area

(STACTIC W.P. 00/29-Revised)

Background

Presence of inspection vessels in the Regulatory Area is of paramount importance for the effectiveness of the operation of the Scheme of Joint International Inspection and Surveillance. It appears, however, that the relevant rules in their present version do not make it an obligation for Contracting Parties to provide for adequate inspection presence. Under these circumstances, new such rules should be introduced in order to both make the existing rules more effective and share out the burdens connected with this means of inspection in a more equitable fashion and commensurate with the fishing activities of the different Contracting Parties.

Proposal

Amend Part IV, Section 3, as follows :

Sub-paragraph 2 shall read as follows:

"Where at any one time, more than 10 vessels of any one Contracting Party are engaged in fishing operations or in the processing or transferring of fish in the Regulatory Area, that Contracting Party shall, during that time:

- (a) have an inspection vessel in the Regulatory Area, or shall co-operate with another Contracting Party to jointly operate an inspection vessel; and
- (b) have an inspector or other designated authority present in the Regulatory Area, or other designated authority present in a country of a Contracting party adjacent to the Convention Area, to receive and respond, without delay, to notice of apparent infringements."